Discrete moments of the Riemann zeta function

Andrew Pearce-Crump

РнD

University of York Mathematics

September 2024

Abstract

An important problem in number theory is to calculate the moments of the Riemann zeta function $\zeta(s)$. Moments have a wide range of applications, for example in calculating proportions of non-trivial zeros $\rho = \beta + i\gamma$ of $\zeta(s)$ that satisfy the Riemann Hypothesis, or that are simple.

Shanks [280] noticed that $\zeta'(\rho)$ is real and positive on average, a strange result when we consider that this is a complex-valued function summed over complex points. Later, it was noticed that this peculiar behaviour continued to higher derivatives, where the sum remains real on average, but oscillates positive and negative depending on whether the order of the derivative is odd or even.

Generalisations of this observation are considered throughout this thesis. These involve sums of the form

$$\sum_{0 < t \le T} \zeta^{(n_1)} \left(\frac{1}{2} + it\right) \dots \zeta^{(n_k)} \left(\frac{1}{2} + it\right),$$

where for integers n_1, \ldots, n_k , t ranges over either the non-trivial zeros of zeta or the zeros of the derivative of the Hardy Z-function, and $\zeta^{(n)}$ denotes the n^{th} derivative of $\zeta(s)$.

After a comprehensive background on the zeta function, we begin with a simple heuristic for Shanks' observation and its generalisation, giving a clear reason for the oscillating behaviour before giving a rigorous proof of this fact with a full asymptotic expansion. A weighted first moment of this problem is then considered.

We build on the analogy between characteristic polynomials of random matrices and $\zeta(s)$, first noted by Keating and Snaith [211]. We present new conjectures for full asymptotic expansions of the above summation over the non-trivial zeros of $\zeta(s)$ after giving other supporting evidence for the leading order behaviour of these sums.

Finally we consider the above sum over the zeros λ of the derivative of the Hardy Z-function, and show that the behaviour of this sum oscillates in the opposite way to that over the non-trivial zeros, that is, $\zeta'(1/2 + i\lambda)$ is real and negative on average.

Acknowledgments

This thesis has only been possible due to the hard work and support of many people.

It goes without saying that a supervisor should be super in many ways, all of which Chris Hughes has been. Your support, encouragement, patience, guidance, kindness, and friendship has never wavered, and I hope to be just 1% as enthusiastic about maths as you are. The sheer amount of time and energy you have invested in me won't go to waste. If I ever need a fun anecdote, I know where to go. Chris considered me a peer from the get-go and shows us all exactly what a supervisor should be.

Several other academics have had a strong influence on my PhD. Hung Bui has always been available for a discussion on maths and has taught me a great deal, much of which has influenced what has appeared in this thesis. It has been a pleasure to already work with you and see how you approach tackling a problem. I am fortunate to have met Nathan Ng while in Canada on a conference. He has been a constant supporter of mine, and I appreciate every discussion we have had, all of which have been productive. A huge thanks also goes to him for being my external examiner and for his close reading of this thesis. I must also give thanks to Jon Keating, who was the first contact I had with academics in this field outside of Chris, and for being so welcoming and generous with his advice. Discussions with Jon on maths and careers have already helped shape my future in the field. I look forward to working with you all in the future.

The other group of people to mention are those involved with the PIMS CRG for *L*-functions in Analytic Number Theory, particularly the four organisers. I have already mentioned the impact Nathan has had on my PhD, but thanks also goes to Habiba Kadiri, Greg Martin, and Alia Hamieh. They have helped in many ways, from career advice from Habiba, collaborating and publishing a paper with Greg, and the all around support from Alia, particularly in inviting me to Canada in the first place. I have been lucky that the CRG has overlapped with my PhD, and I have benefited massively from interacting with all parts of the CRG.

Many other academics deserve a mention by name, but at the risk of missing people

off the list I have only mentioned a few. Thanks goes to everyone I have met and had any conversation or collaboration with, for your welcoming attitude, for your support, for the various invitations to give seminars and attend conferences, and for just being a good bunch of people in general.

The last academics who I want to mention are those based at the University of York, which has been my home for the last few years. Thanks to all those involved in the department, particularly Alet Roux for her encouragement with teaching, the Number Theory group who have been involved with all stages of my PhD, from interviewing me, inviting me to give talks, TAP support, and internal examining of this thesis, and to Sol Lugmayer for expanding my duo with Chris into a trio of people to talk maths with for the last couple of years.

A vast amount of support over these years has come from beyond the world of academia. To Mike, Esai, and Holly, thanks for your friendship, curry and pub visits, and for keeping me in touch with my old life as a teacher. Thanks also goes to Harry and the whole Hayes family, particularly Nick and Suz, for their love and support since I first met them. I look forward to putting a copy of this thesis in your hands and know I appreciate all you have done, treating me as part of the family over the years.

I finally come to the five people who deserve the biggest thanks of all. This thesis is dedicated entirely to you.

To Aza and Edward, thanks for being my siblings. That's all, really. Well, I suppose I should mention the non-stop dirty jokes and comments which would be weird to anyone else, and I look forward to all the others to come over the years ahead. Thanks especially to Aza for teaching me some of our best words in our second language, and to Edward for helping me with maths over the years.

There are no words to describe just how much I owe my parents, Donald and Amanda, for all they have done for me and my siblings. The endless support and love you pour into the three of us can never be repaid, and all I can say is thank you for all you have done and continue to do for us all. While you may not understand a word of my work, you always show an interest and provide the encouragement I need.

Lucy has been by my side since sending me a rose, and for some reason has agreed to move across the country with me for our next big adventure. I look forward to carrying on our evening routine of me summarising my day in under 30 seconds and in two words - I'm stuck! Somehow I spent just enough time being unstuck to put this thesis together over the years, and you played a huge role in this. Thank you for all of your love and support, and I can't wait to see where our future takes us.

Author's declaration

I declare that the work presented in this thesis, except where otherwise stated, is based on my own research carried out at the University of York and has not been submitted previously for any degree at this or any other university. Sources are acknowledged by explicit references.

Section 1.3.8.2 on the background of joint moments in Chapter 1 can be found in [183], written jointly with Christopher Hughes and Solomon Lugmayer, titled 'The second moment of the Riemann zeta function at its local extrema'.

Chapter 2 is based heavily on joint work with Christopher Hughes and Greg Martin [184], titled 'A heuristic for discrete mean values of the derivative of the Riemann zeta function', which has been published in the journal Integers.

Much of Chapter 3 can be found in a joint paper with Christopher Hughes [186], called 'A discrete mean-value theorem for the higher derivatives of the Riemann zeta function', and has appeared in the Journal of Number Theory.

Almost all of the content of Chapter 4 can be found in the paper [262], titled 'A further generalisation of sums of higher derivatives of the Riemann Zeta Function', which has been accepted for publication by the International Journal of Number Theory.

Chapter 5 is based heavily on joint work with Christopher Hughes [187], called 'Moments of derivatives of the Riemann zeta function: Characteristic polynomials and the hybrid formula', and has been submitted for publication.

Much of Chapter 6 will be in an upcoming paper [185], 'Moments of derivatives of the Riemann zeta function: the Ratios Conjecture', written jointly with Christopher Hughes.

Much of Chapter 7 will be found in [263] 'Moments of the Riemann zeta function at its local extrema'.

Versions of unpublished and upcoming papers have been shared privately with other mathematicians and will all soon be submitted for publication. I was involved in, and contributed to the work and the write up in all of the above papers in an equal manner to any co-authors. Permission has been obtained to include the above in this thesis.

Contents

A	bstra	ict		2
A	cknov	wledgn	nents	3
A	utho	r's dec	laration	5
С	onter	nts		7
$\mathbf{L}\mathbf{i}$	ist of	Figur	es	11
N	otati	on		13
P	refac	e		16
1	Intr	oducti	on	23
	1.1	The P	rime Number Theorem	23
		1.1.1	Historical Background	23
		1.1.2	The Riemann Zeta Function	24
		1.1.3	Zeta Zeros and the Riemann Hypothesis	27
			1.1.3.1 The Riemann Hypothesis	28
		1.1.4	A proof of the Prime Number Theorem	29
			1.1.4.1 The Prime Number Theorem in Arithmetic Progressions .	32
	1.2	Rando	m Matrix Theory	34
		1.2.1	The Classical Compact Groups	34
		1.2.2	Random Matrix Theory meets Number Theory: Act I	37
	1.3	Contin	nuous Moments of the Riemann zeta function	39
		1.3.1	The Lindelöf Hypothesis	40
		1.3.2	Continuous Moment Results	41
			1.3.2.1 Upper and Lower Bounds	43

	1.3.3	Random	Matrix Theory meets Number Theory: Act II	44
	1.3.4	The Hyl	brid Model	48
		1.3.4.1	Extreme Values of the Riemann zeta function	50
	1.3.5	Lower C	Order Terms for Moments	51
	1.3.6	Recent 1	Developments	52
	1.3.7	Applicat	tions of Moments	54
		1.3.7.1	Sub-convexity of $\zeta(s)$	54
		1.3.7.2	Twisted and Mollified Moments	54
		1.3.7.3	Zeros on the critical line	56
		1.3.7.4	Zero-density estimates	60
	1.3.8	Joint Me	oments of the Riemann zeta function and Joint Moments of	
		the Hare	dy Z-function \ldots \ldots \ldots \ldots \ldots \ldots	62
		1.3.8.1	The Hardy Z-function	63
		1.3.8.2	Joint Moments through RMT	64
		1.3.8.3	The Joint Moment of the Hardy Z -function and its derivative	65
1.4	Discre	te Mome	nts of the Riemann zeta function	67
	1.4.1	Discrete	Moment Results	67
		1.4.1.1	Upper and Lower Bounds	69
	1.4.2	Random	Matrix Theory meets Number Theory: Act III	71
	1.4.3	Applicat	tions of Discrete Moments	74
		1.4.3.1	Simple Zeros of $\zeta(s)$	74
		1.4.3.2	Multiple zeros of $\zeta(s)$	76
		1.4.3.3	Gaps between zeros of $\zeta(s)$	81
		1.4.3.4	Behaviour of $M(x)$	83
		1.4.3.5	The Error in the Prime Number Theorem	86
Sha	nks' C	onjectu	re	87
2.1	Overv	iew of Ch	apter	87
2.2	Histor	ical Back	ground of Shanks' Conjecture	88
2.3	Heuris	tic for th	e Generalised Shanks' Conjecture	95
	2.3.1	Prelimir	nary Lemmas	95
		2.3.1.1	Dirichlet Series Approximations	95
		2.3.1.2	The Landau–Gonek Theorem	97
	2.3.2	The Heu	rristic	99
		2.3.2.1	The Error Terms	00
2.4	Shank	s' Conjec	ture for the Reciprocal	01
	2.4.1	Negative	e Shanks' Conjecture via the Heuristic	03

 $\mathbf{2}$

3	The	e Gene	ralised Shanks' Conjecture	105
	3.1	The G	eneralised Shanks' Conjecture	106
		3.1.1	Overview of the Proof	107
		3.1.2	Preliminary Lemmas	108
		3.1.3	Beginning the proof	109
		3.1.4	Evaluating I	. 111
		3.1.5	Evaluating the sum S	. 113
		3.1.6	The Error Term	115
			3.1.6.1 The Unconditional Case	115
			3.1.6.2 The Conditional Case: No dependence on $n \ldots \ldots \ldots \ldots$. 116
			3.1.6.3 The Conditional Case: Dependence on n	. 116
		3.1.7	Finding the Leading Asymptotic Terms	120
		3.1.8	Examples of the first moment of specific n^{th} derivatives of the Rie-	
			mann zeta function $\zeta(s)$. 121
			3.1.8.1 The case $n = 1$ (Shanks' Conjecture)	. 123
			3.1.8.2 The case $n = 2 \dots \dots$. 124
	3.2	The G	eneralised Shanks' Conjecture from a shifted sum	126
	3.3	The G	eneralised Shanks' Conjecture in Short Intervals	128
4	A F	urther	Generalisation of Shanks' Conjecture	129
4	A F 4.1	`urther Histor	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and	129
4	A F 4.1	`urther Histor Mome	• Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ents of <i>L</i> -functions	129 129
4	A F 4.1	`urther Histor Mome Combi	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ents of <i>L</i> -functions	129 129 131
4	A F4.14.2	urther Histor Mome Combi 4.2.1	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ents of <i>L</i> -functions	129 129 131 132
4	A F 4.1 4.2	Yurther Histor Mome Combi 4.2.1 4.2.2	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ents of <i>L</i> -functions	129 131 132 133
4	A F 4.1 4.2	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ents of <i>L</i> -functions	129 131 132 133 134
4	A F 4.1 4.2	urther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions	129 131 132 133 134 135
4	A F 4.1 4.2	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions	129 131 132 133 133 134 135 135
4	A F 4.1 4.2	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjectureical Background of Combinations of the Landau–Gonek Theorem andints of L-functionsining the Generalised Shanks' Conjecture with Landau's TheoremOverview of the ProofPreliminary LemmasProof of Theorem 4.14.2.3.1Bounding I_B and I_T 4.2.3.2Evaluating I_R 4.2.3.3Evaluating I_L	129 131 132 133 134 135 135 137
4	A F 4.1 4.2	urther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjectureical Background of Combinations of the Landau–Gonek Theorem andints of L-functionsining the Generalised Shanks' Conjecture with Landau's TheoremOverview of the ProofPreliminary LemmasProof of Theorem 4.14.2.3.1Bounding I_B and I_T 4.2.3.2Evaluating I_R 4.2.3.3Evaluating I_L 4.2.3.4Finalising the Proof of Theorem 4.1	129 131 132 133 134 135 135 137 141
4	A F 4.1 4.2	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3	Generalisation of Shanks' Conjectureical Background of Combinations of the Landau–Gonek Theorem andints of L-functionsining the Generalised Shanks' Conjecture with Landau's TheoremOverview of the ProofPreliminary LemmasProof of Theorem 4.14.2.3.1Bounding I_B and I_T 4.2.3.2Evaluating I_R 4.2.3.3Evaluating I_L 4.2.3.4Finalising the Proof of Theorem 4.1Proof of Corollary 4.2	129 129 131 132 133 134 135 135 137 141 141
4	 A F 4.1 4.2 Hig 	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3 4.2.3	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions	129 129 131 132 133 134 135 135 135 137 141 141 141
4	 A F 4.1 4.2 Hig 5.1 	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3 4.2.3 4.2.4 Eher Ma Staten	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions ining the Generalised Shanks' Conjecture with Landau's Theorem Overview of the Proof Preliminary Lemmas Proof of Theorem 4.1 4.2.3.1 Bounding I_B and I_T 4.2.3.2 Evaluating I_L 4.2.3.3 Evaluating I_L 4.2.3.4 Finalising the Proof of Theorem 4.1 Proof of Corollary 4.2	129 129 131 132 133 134 135 135 137 141 141 144 145
4	 A F 4.1 4.2 Hig 5.1 5.2 	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3 4.2.3 4.2.4 ther Me Staten The ch	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions ining the Generalised Shanks' Conjecture with Landau's Theorem Overview of the Proof Preliminary Lemmas Proof of Theorem 4.1 4.2.3.1 Bounding I_B and I_T 4.2.3.2 Evaluating I_R 4.2.3.3 Evaluating I_L 4.2.3.4 Finalising the Proof of Theorem 4.1 Proof of Corollary 4.2 Another the Hybrid Model nent of Results	129 131 132 133 134 135 135 135 137 141 141 144 145 147
4 5	 A F 4.1 4.2 Hig 5.1 5.2 5.3 	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3 4.2.3 4.2.4 ther Me Staten The ch The hy	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions	129 129 131 132 133 134 135 135 135 137 141 141 141 144 145 147 154
4	 A F 4.1 4.2 Hig 5.1 5.2 5.3 	Yurther Histor Mome Combi 4.2.1 4.2.2 4.2.3 4.2.3 4.2.4 kler Ma Staten The ch The ch The hy 5.3.1	Generalisation of Shanks' Conjecture ical Background of Combinations of the Landau–Gonek Theorem and ints of L-functions	129 129 131 132 133 134 135 135 135 137 141 141 144 145 147 154 156

	$5.4 \\ 5.5$	Negat: A recu	ive Moments of the Derivatives of the characteristic polynomial nrrence relation hidden in the RMT calculation	166 168
6	Hig	her M	oments via the Ratios Conjecture	171
	6.1	The R	atios Conjecture	173
		6.1.1	Pair Correlation and <i>n</i> -level Correlations	173
		6.1.2	Moments of Logarithmic Derivatives of the Riemann zeta function .	174
		6.1.3	Mollified Moments of <i>L</i> -functions	174
		6.1.4	Non-vanishing results for various families of L -functions $\ldots \ldots \ldots$	174
		6.1.5	Discrete Moments of $\zeta(s)$ and Lower Order Terms $\ldots \ldots \ldots$	175
	6.2	Lower	Order Terms for Moments of the Derivatives of the Riemann Zeta	
		Functi	ion	175
		6.2.1	Preliminary Lemmas on Vandermonde Determinants	176
		6.2.2	Deriving Conjecture 6.3	180
		6.2.3	Recovering Conjectures 5.3 and 5.4	184
		6.2.4	Examples	185
			6.2.4.1 Recovering the Shanks' conjecture asymptotic $(k = 1)$	185
			6.2.4.2 The second moment of $\zeta'(1/2 + i\gamma)$ $(k = 2)$	187
7	Mo	ments	of the Riemann zeta function at its local extrema	192
	71			
	1.1	Mome	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the	
	1.1	Mome Riema	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the unn zeta function	193
	1.1	Mome Riema 7.1.1	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196
	1.1	Mome Riema 7.1.1 7.1.2	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the nn zeta function	193 196 200
	1.1	Mome Riema 7.1.1 7.1.2 7.1.3	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the num zeta function	193 196 200 201
	1.1	Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 205
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 207 207
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 207 207 207
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 207 207 207 208 218
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4 7.1.5 7.1.6	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 207 207 207 208 218 221
		Mome Riema 7.1.1 7.1.2 7.1.3 7.1.4 7.1.5 7.1.6 7.1.7	ents of $\zeta(s)$, its derivatives, and $\chi(s)$ at the local extrema of the ann zeta function	193 196 200 201 205 205 207 207 208 218 221 225

References

List of Figures

1.1	A comparison of the Circular Unitary Ensemble value distribution for $\Re(\log Z_N)$ with $N = 42$, Odlyzko's data for the distribution of $\Re(\log \zeta)$ near the $10^{20\text{th}}$ zero $(T \approx 10^{19})$ and the standard Gaussian, all scaled to have unit variance 47
1.2	A comparison of the Circular Unitary Ensemble value distribution for $\Im(\log Z_N)$ with $N = 42$, Odlyzko's data for the distribution of $\Im(\log \zeta)$ near the $10^{20\text{th}}$ zero $(T \approx 10^{19})$ and the standard Gaussian, all scaled to have unit variance 48
1.3	A plot of $Z(t)$ (dark blue) and $ \zeta(1/2 + it) $ (dashed) for $0 \le t \le 50$ 63
2.1	Plot of $t \mapsto \zeta(1/2 + it)$, with the original conjecture below the graph 88
2.2	Scatterplot of $\zeta'(\rho)$ for the first 100,000 zeros of the zeta function 90
2.3	Scatterplot of $\zeta''(\rho)$ for the first 100,000 zeros of the zeta function 93
2.4	Scatterplot of $\zeta'''(\rho)$ for the first 100,000 zeros of the zeta function
2.5	The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \le T} \frac{1}{\zeta'(\rho)} - \frac{T}{2\pi}$ plotted over the first 100,000 zeros
3.1	The real part of the sum $\sum_{0 < \gamma \leq T} \zeta'(\rho)$ with true value (blue), the true - leading (pink), true - leading - subleading (orange) and true - leading - subleading - subsubleading (green) terms subtracted from the true value, for T up to the height of the 100,000 th zero
3.2	Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta'(\rho)$ and the whole asymptotic result, for T up to the height of the 100,000 th zero, showing the real
	error at each point
3.3	The real part of the sum $\sum_{0 < \gamma \le T} \zeta''(\rho)$ with true value (blue), the true - leading (pink), true - leading - subleading (orange), true - leading - subleading - subleading - subsubleading (green), true - leading - subleading - subsubleading - subsubleading (purple) terms subtracted from the true value, for T up to the height of the 100 000 th zero.
	$100,000 2010. \dots \dots \dots \dots \dots \dots \dots \dots \dots $

3.4	Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta''(\rho)$ and the whole asymptotic result, for T up to the height of the 100,000 th zero, showing the real error at each point
5.1	The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^{-2}$ plotted over the first million zeros
6.1	The real part of the true value of $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$, for T up to the height of the 1,000,000 th zero
6.2	Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$ and the leading asymptotic result of the equation, for T up to the height of the
	1,000,000 th zero
6.3	Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$ and
	the whole asymptotic result of the equation, for T up to the height of the
	$1,000,000^{\text{th}}$ zero, showing the real error at each point. $\ldots \ldots \ldots \ldots \ldots 191$
7.1	The real part of the sum $\sum_{0 < \lambda \leq T} \zeta'(1/2 + i\lambda)$ with true value (blue), the true - leading (red), true - leading - subleading (green) and true - leading - subleading
	- subsubleading (purple) terms subtracted from the true value, for T up to the
	height of the 10,000 th zero λ of $Z'(t)$
7.2	The real part of the sum $\sum_{0 < \lambda \leq T} \zeta(1/2 + i\lambda)$ with true value (blue), the true -
	leading (red) and true - leading - subleading (green) terms subtracted from the
	true value, for T up to the height of the 1,000,000 th zero λ of $Z'(t)$
7.3	The real part of the sum $\sum_{0 < \lambda \leq T} \chi(1/2 + i\lambda)$ with true value (blue), the true -
	leading (red) and true - leading - subleading (green) terms subtracted from the
	true value, for T up to the height of the 1,000,000 th zero λ of $Z'(t)$ 199
7.4	The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \leq T} \chi(1/2 + i\gamma) + \frac{T}{2\pi}$ for T
	up to the height of the 1,000,000 th zero $1/2 + i\gamma$ of $\zeta(s)$

Notation

We remind the reader of the main notational conventions used in this thesis.

We write $\Re(z)$ and $\Im(z)$ for the real and imaginary parts of a complex number z.

The Riemann zeta function is denoted $\zeta(s)$ for a complex number $s = \sigma + it$ and the n^{th} derivative of $\zeta(s)$ is denoted $\zeta^{(n)}(s)$. A non-trivial zero of $\zeta(s)$ is written as $\rho = \beta + i\gamma$.

We also adopt the ε -notation: each occurrence of ε represents a sufficiently small positive real number which may vary from occurrence to occurrence, potentially even in the same line. We also let δ be some small real number.

We will use the standard asymptotic notations concerning limits at infinity:

• f(x) = O(g(x)) as $x \to \infty$ if there exists a positive constant A such that

$$\limsup_{x \to \infty} \left| \frac{f(x)}{g(x)} \right| \le A$$

• f(x) = o(g(x)) as $x \to \infty$ if

$$\lim_{x \to \infty} \left| \frac{f(x)}{g(x)} \right| = 0.$$

• $f(x) = \Omega(g(x))$ as $x \to \infty$ if there exists a positive constant B such that

$$\limsup_{x \to \infty} \left| \frac{f(x)}{g(x)} \right| \ge B$$

- $f(x) \asymp g(x)$ as $x \to \infty$ if f(x) = O(g(x)) and $f(x) = \Omega(g(x))$.
- $f(x) \sim g(x)$ as $x \to \infty$ if

$$\lim_{x \to \infty} \frac{f(x)}{g(x)} = 1.$$

• $f(x) \ll g(x)$ as $x \to \infty$ if f(x) = O(g(x)).

Any product or summation with a subscript p denotes a product or summation over the prime numbers p, for example \prod_p .

For a real number c we write

$$\frac{1}{2\pi i} \int_{(c)} f(z) \, dz = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} f(z) \, dz.$$

The Prime Number Theorem is discussed at various points throughout this thesis. We write:

- $\pi(x)$ for the number of primes less than or equal to some real number x.
- Li(x) is the logarithmic integral, given by Li(x) = $\int_2^x \frac{1}{\log t} dt$.
- $\pi(x; q, a)$ for the number of primes congruent to $a \mod q$ that are less than or equal to x.

The gamma function $\Gamma(z)$ is defined for all z by

$$\Gamma(z) = \int_0^\infty x^{z-1} e^{-x} \, dx$$

for $\Re(z) > 0$. The gamma function satisfies the functional equation

$$\Gamma(n+1) = n\Gamma(n)$$
, with $\Gamma(1) = 1$.

The Barnes G-function G(z) is defined for all z by

$$G(z+1) = (2\pi)^{z/2} \exp\left(-\frac{1}{2}(z^2 + \gamma_0 z^2 + z)\right) \prod_{n=1}^{\infty} \left(1 + \frac{z}{n}\right)^n e^{-z + z^2/2n}$$

where γ_0 is Euler's constant. The Barnes G-function satisfies the functional equation

$$G(z+1) = \Gamma(z)G(z)$$
, with $G(1) = 1$.

Arithmetic functions play a big part in this thesis. Let $n \in \mathbb{N}$. If $f, g : \mathbb{N} \to \mathbb{C}$ are two arithmetic functions, the Dirichlet convolution $f * g : \mathbb{N} \to \mathbb{C}$ is defined by

$$(f * g)(n) = \sum_{d|n} f(d)g\left(\frac{n}{d}\right) = \sum_{ab=n} f(a)g(b),$$

where the sum extends over all positive divisors d of n, or equivalently over all distinct pairs (a, b) of positive integers whose product is n. Dirichlet convolution appears naturally when we multiply Dirichlet series, as

$$\left(\sum_{n=1}^{\infty} \frac{f(n)}{n^s}\right) \left(\sum_{n=1}^{\infty} \frac{g(n)}{n^s}\right) = \left(\sum_{n=1}^{\infty} \frac{(f*g)(n)}{n^s}\right)$$

A wide range of arithmetic functions are used throughout this thesis. We have:

• The von Mangoldt function is given by

$$\Lambda(n) = \begin{cases} \log p & \text{if } n = p^k \text{ for } p \text{ prime and some integer } k \\ 0 & \text{otherwise.} \end{cases}$$

• The Euler totient function $\varphi(n)$ counts the positive integers up to a given integer n that are relatively prime to n.

• The Möbius function is defined by

$$\mu(n) = \begin{cases} 1 & \text{if } n = 1\\ (-1)^k & \text{if } n = p_1 \dots p_k \text{ with the } p_j \text{ distinct}\\ 0 & \text{if } n \text{ is not square-free.} \end{cases}$$

For k a positive integer,

- $\mu_k(n)$ denotes the n^{th} coefficient of the Dirichlet series for $\zeta(s)^{-k}$.
- $d_k(n)$ denotes the n^{th} coefficient of the Dirichlet series for $\zeta(s)^k$.
- $\Lambda_k(n)$ denotes the n^{th} coefficient of the Dirichlet series for $(-1)^k (\zeta'(s)/\zeta(s))^k$.

By our previous comments on Dirichlet convolution, we can describe each of μ_k, d_k and Λ_k via convolutions. For example, we can write

$$\Lambda_k = \underbrace{\Lambda * \Lambda * \Lambda * \cdots * \Lambda}_{k-1 \text{ convolutions}}$$

with the convention that $\Lambda_0(m)$ takes the value 1 if m = 1 and 0 otherwise, and where $\Lambda_1(m) = \Lambda(m)$ is the usual von Mangoldt function.

The classical compact groups U(N), SO(2N), Sp(2N) are used in this thesis with their definitions given in Section 1.2.1. The $N \times N$ identity matrix is written as I_N . The characteristic polynomial of an $N \times N$ matrix A from $\{U(N), SO(2N), Sp(2N)\}$ is

$$Z_{N,A}(\theta) = \det(I - Ae^{-i\theta}).$$

Preface

We give an overview of the work presented in this thesis. We consider the problem of moments of derivatives of the Riemann zeta function, and variations thereof. In particular, we will study moments of the form

$$\sum_{0 < \gamma \le T} \zeta^{(n)} \left(\rho\right)^k$$

as $T \to \infty$, for integer k, n, where $\zeta^{(n)}(s)$ denotes the n^{th} derivative of the Riemann zeta function $\zeta(s)$ and where $\rho = \beta + i\gamma$ denotes a non-trivial zero of the Riemann zeta function. More generally, we consider moments of mixed derivatives of the form

$$\sum_{0 < \gamma \le T} \zeta^{(n_1)}\left(\rho\right) \dots \zeta^{(n_k)}\left(\rho\right)$$

as $T \to \infty$, for integer $n_1, \ldots n_k$. We will also extend this problem to studying moments of derivatives of the Riemann zeta function over different points, as seen in Chapter 7.

Chapter 1

Chapter 1 serves as a comprehensive introduction to the field. Many topics are considered including, but not limited to, the Prime Number Theorem, the Riemann Hypothesis, continuous and discrete moments of the Riemann zeta function, random matrix theory (RMT), and applications of moments.

Chapter 2

In Chapter 2, we look at the history of the first moment of the derivatives of the Riemann zeta function, a problem known as (the Generalised) Shanks' Conjecture. This is an example of the moments problem above with k = 1 and n a positive integer. This conjecture, now a theorem, states that on average,

 $\zeta^{(n)}(\rho)$ is positive if n is odd, and negative if n is even in the mean.

We develop a heuristic which is used to non-rigorously prove the leading order behaviour of such moments. The heuristic relies on the Landau–Gonek Theorem, found in [144], which is an explicit formula (one that relates zeta zeros to the prime numbers). This heuristic, that is very basic and easy to use, derives the alternating plus/minus behaviour depending on whether the order of the derivative is odd/even, and even re-derives the correct order of magnitude to leading order. We also consider the question of the first negative moment of the first derivative of the Riemann zeta function.

Chapter 3

Chapter 3 involves the first original rigorous proof in this thesis. We consider the Generalised Shanks' Conjecture and derive a full asymptotic result with lower order terms, and (under the Riemann Hypothesis) a power-saving error term, although the Riemann Hypothesis isn't needed anywhere else in the proof. The result can be stated in the following form.

Theorem. Let $\rho = \beta + i\gamma$ denote a non-trivial zero of the Riemann zeta function. Then

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n+1} + \frac{T}{2\pi} \mathcal{P}_n\left(\log \frac{T}{2\pi} \right) + E_n(T)$$

where $\mathcal{P}_n(x)$ is an explicit polynomial in x of degree n, given in Theorem 3.1, and $E_n(T)$ is given by

$$E_n(T) = O\left(T e^{-C\sqrt{\log T}}\right)$$

with C is a positive constant. If we assume the Riemann hypothesis, then

$$E_n(T) = O\left(T^{1/2} (\log T)^{n+9/4}\right)$$

This conditional error term improves upon what was previously known due to Fujii in the first derivative case, and the one given in this thesis improves upon the published result that this chapter is based on.

Next, we employ a shifted sum over the non-trivial zeros of the Riemann zeta function of $\zeta(\rho + \alpha)$, where α is a small shift. This shifted sum follows from a result of Fujii's [126]. Using this result, we give an equivalent form of the main result in this chapter.

Theorem. For $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, and where $\zeta^{(n)}(s)$ is the nth derivative of $\zeta(s)$, we have for T sufficiently large,

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) = n! \sum_{\ell=0}^{n+1} \frac{(-1)^{n-\ell+1}}{(n-\ell+1)!} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-\ell+1} + n! \sum_{m=0}^{n} \sum_{\ell=0}^{n-m} \frac{(-1)^{n-m-\ell+1}\gamma_m}{(n-m-\ell)!m!} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-m-\ell} + n! A_n \frac{T}{2\pi} + O\left(Te^{-C\sqrt{\log T}}\right),$$

where the coefficients γ_j and A_n are from the Laurent expansions of $\zeta(s)$ and $\zeta'(s)/\zeta(s)$ about s = 1, respectively.

We end the chapter by briefly discussing the question of the Generalised Shanks' Conjecture over short intervals. That is, we previously sum over the imaginary part of the non-trivial zeros between 0 and T, but we could consider the analogous problem between T and T + H, where $T^{1/2+\varepsilon} \ll H \ll T$. This result to leading order is as expected, that is, we can show the following result in an analogous way to the proof of Theorem 3.1.

Theorem. For $T^{1/2+\varepsilon} \ll H \ll T$,

$$\sum_{T < \gamma \le T+H} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{H}{2\pi} \left(\log T\right)^{n+1} + O\left(H(\log T)^n\right) + O\left(T^{1/2+\varepsilon}\right)$$

Chapter 4

In Chapter 4, we combine the Landau's Theorem (a non-uniform version of the Landau– Gonek Theorem) with the Generalised Shanks' Conjecture. Specifically, we generalise a result of Fujii's to all derivatives.

Theorem. For a fixed positive real number X,

$$\begin{split} \sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} &= \\ & \left(-1\right)^n \left\{ \Delta(X) \frac{T}{2\pi} \left((\log X)^n \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4}\right) - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \right. \\ & \left. + X (\log X)^n \left(\frac{1}{2} \log X - \frac{\pi i}{4}\right) \sum_{m\leq \frac{T}{2\pi X}} e^{2\pi i m X} + \frac{X}{2} (\log X)^n \sum_{m\leq \frac{T}{2\pi X}} e^{2\pi i m X} \log m \right. \\ & \left. - X \sum_{mr\leq \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) (\log r X)^n \right\} + E_n(T), \end{split}$$

where $\Delta(X)$ equals 1 if X is a positive integer and 0 otherwise and $E_n(T)$ is the error term given in Theorem 3.1.

When X is a fixed positive integer, this result can be rewritten in a different way. This different way is of interest as the Landau–Gonek Theorem appears quite different (although to the same leading order) depending on whether the term X is a positive integer or a positive real number. This new asymptotic that we derive shows a clear distinction between the behaviour of such sums depending on whether the X is a generic positive real number, or specifically, a positive integer.

Corollary.

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} = (-1)^{n+1} \frac{T}{2\pi} (\log X)^n \left\{ \sum_{k=0}^n \sum_{u=0}^{k+1} \binom{n}{k} \binom{k+1}{u} (-1)^u \frac{1}{k+1} \left(\log \frac{T}{2\pi} \right)^{k+1-u} (\log X)^{u-k} + \sum_{k=0}^n \sum_{u=0}^k \sum_{u=0}^{k-l} \binom{n}{k} \binom{k}{l} \binom{k-l}{u} (-1)^l l! \left(-1 + \sum_{j=0}^l \frac{1}{j!} \gamma_j \right) \left(\log \frac{T}{2\pi} \right)^{k-l-u} (\log X)^{u-k} + \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} k! A_k (\log X)^{-k} - \left(\log \frac{T}{2\pi} - 1 - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \right\} + E_n(T)$$

with the error term $E_n(T)$ as given in Theorem 3.1 and where the coefficients γ_j and A_n are from the Laurent expansions of $\zeta(s)$ and $\zeta'(s)/\zeta(s)$ about s = 1, respectively.

Chapter 5

We consider higher moments of the derivatives of the Riemann zeta function in Chapter 5. We begin by considering an approach using characteristic polynomials of random unitary matrices, which involves proving rigorously a RMT result, and use it to conjecture a result on the Riemann zeta function. We also consider a hybrid approach to conjecture what the arithmetic term should be in these conjectures.

Conjecture. Assume the Riemann Hypothesis. For fixed, positive integers n_1, \ldots, n_k , the moments of mixed derivatives of $\zeta(s)$, evaluated at the non-trivial zeros of $\zeta(s)$, are given by

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta^{(n_1)} \left(\frac{1}{2} + i\gamma\right) \dots \zeta^{(n_k)} \left(\frac{1}{2} + i\gamma\right) \\ \sim (-1)^{n_1 + \dots + n_k + k} \frac{n_1! \dots n_k!}{(n_1 + \dots + n_k + 1)!} \left(\log \frac{T}{2\pi}\right)^{n_1 + \dots + n_k}$$

as $T \to \infty$, where N(T) is the number of non-trivial zeros of $\zeta(s)$ in the critical strip above the real axis up to height T.

A simple corollary is when all the derivatives are equal to each other.

Conjecture. Assume the Riemann Hypothesis. For $n, k \in \mathbb{N}$, the k^{th} moment of the n^{th} derivative of $\zeta(s)$ is given by

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right)^k \sim (-1)^{k(n+1)} \frac{(n!)^k}{(kn+1)!} \left(\log \frac{T}{2\pi}\right)^{kn}$$

as $T \to \infty$.

Using the hybrid model approach, we are able to extend this conjecture beyond integer k to a conjecture over complex k.

Conjecture. Assume the Riemann Hypothesis. For complex k with $\Re(k) > -3$,

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^k \sim \frac{1}{\Gamma(k+2)} \left(\log \frac{T}{2\pi}\right)^k$$

as $T \to \infty$.

This last conjecture suggests some further conjectures for the negative moments, and we find that in the case k = -1, they agree with what was previously known, specifically Theorem 2.3. Conjectures for the negative moments in the cases where k = -2, -3 are also considered. The chapter ends with a discussion on a certain recurrence relation.

Chapter 6

In Chapter 6 we consider the problem of finding the lower order terms of these asymptotic conjectures via the Ratios Conjecture. A comprehensive overview on the background of the Ratios Conjecture is given, with specific focus given to known applications of the Ratios Conjecture. We have found a new application of the Ratios Conjecture, and describe how to give a full asymptotic expansion for the moments of mixed derivatives of $\zeta(s)$, evaluated at the non-trivial zeros of $\zeta(s)$, first considered in Chapter 5.

Conjecture. Assume the Riemann Hypothesis. Let α_j be small complex numbers. Assume that $|\Re(\alpha_j)| < \frac{1}{4}$ and $|\Im(\alpha_j)| \ll_{\varepsilon} T^{1-\varepsilon}$ for every $\varepsilon > 0$. For $|\delta| < 1/4$, let

$$A_{\{\alpha_1,\dots,\alpha_k\}}(\delta) := \prod_p \frac{1 + F_1(p,\delta) + F_2(p,\delta) + \dots + F_k(p,\delta)}{(1 - p^{-(1 + \alpha_1)}) \dots (1 - p^{-(1 + \alpha_k)})}$$

where for $1 \leq m \leq k$

$$F_m(p,\delta) = (-1)^m \sum_{\substack{J \subset \{1,\dots,k\} \\ |J|=m}} p^{-(m+(m-1)\delta + \sum_{j \in J} \alpha_j)}.$$

The discrete moments of shifts of $\zeta(s)$, evaluated at the non-trivial zeros $\rho = 1/2 + i\gamma$ of the Riemann zeta function, are given by

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2} + i\gamma + \alpha_1\right) \dots \zeta\left(\frac{1}{2} + i\gamma + \alpha_k\right)$$
$$= \frac{1}{2\pi} \int_1^T \left(A'_{\{\alpha_1,\dots,\alpha_k\}}(0) + \sum_{j=1}^k W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t)\right) dt + O(T^{1/2+\varepsilon}),$$

where

$$W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t) = \frac{\zeta'(1+\alpha_j)}{\zeta(1+\alpha_j)} - \left(\frac{t}{2\pi}\right)^{-\alpha_j} \zeta(1-\alpha_j) A_{\{\alpha_1,\dots,\alpha_{j-1},0,\alpha_{j+1},\dots,\alpha_k\}}(-\alpha_j) \prod_{\substack{\ell=1\\\ell\neq j}}^k \frac{\zeta(1+\alpha_\ell-\alpha_j)}{\zeta(1+\alpha_\ell)}.$$

This approach allows us to re-derive the Generalised Shanks' Conjecture asymptotic in an integral form, showing a third way of writing this result in this thesis.

Theorem. Assume the Riemann Hypothesis. For n a positive integer, $L = \log \frac{t}{2\pi}$, we have

$$\begin{split} \sum_{0 < \gamma \le T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma \right) &= \\ \frac{n!}{2\pi} \int_{1}^{T} \left(A_n + \frac{(-1)^{n+1} L^{n+1}}{(n+1)!} + \sum_{m=0}^{n} \frac{(-1)^{m+1} L^m \gamma_{n-m}}{m!(n-m)!} \right) \ dt + O\left(T^{1/2+\varepsilon}\right), \end{split}$$

where the coefficients γ_j and A_n are from the Laurent expansions of $\zeta(s)$ and $\zeta'(s)/\zeta(s)$ about s = 1, respectively.

Additionally, we can now conjecture full asymptotic results for previously unknown moments, such as the second moment for the first derivative of the Riemann zeta function. We explain how to derive this conjecture, and give some numerical proof that we have both captured the full asymptotic, and that the error term is of a reasonable size.

Conjecture. Assume the Riemann Hypothesis. For $L = \log \frac{t}{2\pi}$ and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function,

$$\begin{split} \sum_{0 < \gamma \leq T} \zeta' \left(\frac{1}{2} + i\gamma\right)^2 &= \\ \frac{1}{2\pi} \int_1^T \left(\frac{1}{6}L^3 + \frac{1}{2}L^2 \left(2\gamma_0 + A^{(0,0,1)}\right) + \frac{1}{2}L \left(-8\gamma_1 + 4\gamma_0 A^{(0,0,1)} + A^{(0,0,2)} + 2A^{(0,1,1)}\right) \\ &+ \frac{1}{6} \left(-12\gamma_0^3 - 36\gamma_0\gamma_1 + 6\gamma_2 - 24\gamma_1 A^{(0,0,1)} + 6\gamma_0 A^{(0,0,2)} + A^{(0,0,3)} \right. \\ &\left. + 12\gamma_0 A^{(0,1,1)} + 3A^{(0,1,2)} - 3A^{(0,2,1)} + 6A^{(1,1,1)}\right) + O\left(T^{1/2+\varepsilon}\right) dt \end{split}$$

where the γ_m are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1 and the $A^{(i,j,k)}$ are arithmetic terms that are various products and sums over primes.

Chapter 7

Chapter 7 involves a different generalisation of the Generalised Shanks' Conjecture. Specifically, rather than summing the derivatives of the Riemann zeta function over the non-trivial zeros of the Riemann zeta function, we instead sum the derivatives of the Riemann zeta function over the local maxima of $|\zeta(s)|$. These local maxima can be written in terms of the local maxima of the Hardy Z-function Z(t). These maxima occur at the (under the Riemann Hypothesis) unique points λ such that $Z'(\lambda) = 0$. This has a natural analogy to a result of Conrey and Ghosh [66].

Theorem. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K, n \ge 1$ positive integers, we have

$$\sum_{0<\lambda\leq T} \zeta^{(n)}\left(\frac{1}{2}+i\lambda\right) = a_{n+1}\frac{T}{2\pi}L^{n+1} + \frac{T}{2\pi}\sum_{\ell=0}^{n}a_{n-\ell}L^{n-\ell} + \frac{T}{2\pi}\sum_{m=1}^{K}\frac{b_m}{L^m} + O\left(\frac{T}{L^{K+1}}\right),$$

as $T \to \infty$ where we give the coefficients a_m, b_m in the statement in Chapter 7.

Recall that for the Generalised Shanks' Conjecture, the average of $\zeta^{(n)}(s)$, summed over the non-trivial zeros of $\zeta(s)$, clearly alternates between positive and negative as nincreases. In the case of this theorem, the behaviour is reversed; the average of $\zeta^{(n)}(s)$, summed over the maxima of $|\zeta(s)|$, alternates between negative and positive as n increases. In particular, $\zeta'(1/2 + i\lambda)$ is negative and real in the mean. We also consider the sum of $\zeta(s)$ over these maxima (as it isn't trivially zero here, unlike the case when we sum over the non-trivial zeros of zeta).

Additionally, we sum the $\chi(s)$ factor from the functional equation for $\zeta(s)$ given in (1.1) over both the non-trivial zeros of the Riemann zeta function, and over the local maxima of $|\zeta(s)|$. As with the derivatives of the zeta function, the behaviour is reversed for the $\chi(s)$ factor, whether summed over the non-trivial zeros of zeta (in which case it is negative on average) or the zeros of the derivative of the Hardy Z-function (in which case it is positive on average).

Theorem. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K \ge 1$ a positive integer, we have

$$\sum_{0<\lambda\leq T} \chi\left(\frac{1}{2}+i\lambda\right) = (e^2-2)\frac{T}{2\pi} - 4e^2\gamma_0\frac{T}{2\pi}\frac{1}{L} + \frac{T}{2\pi}\sum_{m=2}^{K}\frac{e_m}{L^m} + O_K\left(\frac{T}{L^{K+1}}\right)$$

as $T \to \infty$ where we give the coefficients e_m in the statement in Chapter 7.

In the results where we are comparing the derivatives of the zeta function or the factor $\chi(s)$, the leading order term is of the same order of magnitude, whether we are summing over the non-trivial zeros of the Riemann zeta function, or over the maxima of $|\zeta(s)|$.

- 1 -

Introduction

1.1 THE PRIME NUMBER THEOREM

1.1.1 HISTORICAL BACKGROUND

Mankind has been interested in prime numbers since antiquity, dating back at least to Euclid and his Elements [111]. In Book VII, he defines prime numbers and in Book IX he proves both that there are infinitely many primes, and the Fundamental Theorem of Arithmetic, that every natural number can be factored uniquely as a product of primes.

Since then, mathematicians have wanted to understand everything there is to know about the primes. One question in particular that was of central interest to 18th and 19th century mathematicians was the distribution of the prime numbers, a result we now know as the Prime Number Theorem. This result states

$$\pi(x) \sim \frac{x}{\log x}$$

as $x \to \infty$, where $\pi(x)$ denotes the number of primes less than or equal to some real number x.

The Prime Number Theorem was first conjectured by Legendre [221] (up to some erroneous constants), and by Gauss [137], who at the age of 15 would calculate the number of primes in a chiliad (a group of one thousand numbers) whenever he had some spare time. He noted that the primes obey an inverse logarithm law. Gauss restated this conjecture in a different way, writing

$$\pi(x) \sim \operatorname{Li}(x)$$

as $x \to \infty$, where $\operatorname{Li}(x)$ is the logarithmic integral, given by $\operatorname{Li}(x) = \int_2^x \frac{1}{\log t} dt$.

This second form is numerically more accurate to the true value of $\pi(x)$. These two forms of the Prime Number Theorem are equivalent as it can be shown that for a positive integer N,

$$\operatorname{Li}(x) = \frac{x}{\log x} \sum_{k=0}^{N} \frac{k!}{(\log x)^k} + O\left(\frac{x}{(\log x)^{N+2}}\right).$$

Among the many celebrated mathematicians who attempted to prove the Prime Number Theorem, it is probably fair to say that Chebyshev [51] came closest before its eventual proof in 1896. He was able to show that if the limit

$$\lim_{x \to \infty} \pi(x) \Big/ \frac{x}{\log x}$$

exists, then the limit has to equal 1. Despite not obtaining the full proof of the Prime Number Theorem, he was able to prove Bertrand's Postulate [24], that there exists a prime number between n and 2n for any natural number $n \ge 2$.

Despite mathematician's best efforts, a new idea was needed. Enter Georg Friedrich Bernhard Riemann in 1859 and his eponymous function.

1.1.2 The Riemann Zeta Function

Upon his election as a corresponding member to the Berlin Academy in 1859, Riemann presented his seminal (and only) number theoretical paper, 'On the Number of Prime Numbers less than a Given Quantity' [271]. In this paper, Riemann introduces the zeta function that now bears his name.

The Riemann zeta function was originally studied by Euler in relation to solving the Basel problem concerning the infinite sum of inverse squares. Riemann's insight was to consider this function for a complex variable s rather than a real variable s. Riemann defines his zeta function as the Dirichlet series

$$\zeta(s) = \sum_{n=1}^{\infty} \frac{1}{n^s},$$

which converges for all complex s with $\Re(s) > 1$.

This Dirichlet series has an Euler product representation, that is, a product over prime numbers given by $(1)^{-1}$

$$\zeta(s) = \prod_{p} \left(1 - \frac{1}{p^s} \right)^-$$

which converges for all complex s with $\Re(s) > 1$, where we write \prod_p to mean a product over all prime numbers p.

Equality between these two ways of writing $\zeta(s)$ is equivalent to the Fundamental Theorem of Arithmetic, and provides a new way to prove that there are infinitely many prime numbers. This second fact follows by letting $s \to 1^+$, and noting that the Dirichlet series representation becomes the harmonic series which we know diverges, and the Euler

product representation could only tend to infinity if there were infinitely many terms in the product being multiplied together, which can only happen if there are infinitely many primes.

This connection between the natural numbers and prime numbers underlines the importance of the Riemann zeta function and motivates our study of it. Riemann wrote his paper with the aim to prove the Prime Number Theorem, and introduced many of the fundamental features and properties that we use today.

We will return to the Prime Number Theorem and a sketch of its proof in Section 1.1.4. We now look at the rest of the context of Riemann's paper and consider the results within as these will guide us, as they did Hadamard and de la Vallée Poussin, to the traditional proof of the Prime Number Theorem.

Firstly, Riemann established the meromorphic continuation of $\zeta(s)$ to the whole complex plane, except for a simple pole at s = 1 with residue 1. He did this by establishing a functional equation for $\zeta(s)$, an equation that relates the value of zeta at s to that at 1 - s. He gives two proofs of the functional equation, both of which we write in modern notation. We will sketch one way that Riemann proved the functional equation, and highlight some consequences of the second.

The sketch proof involves the modular form (a complex analytic function on the upper half-plane satisfying certain conditions) called the Jacobi theta function, defined by

$$\theta(x) = \sum_{n = -\infty}^{\infty} e^{-n^2 \pi x}.$$

First we note that

$$\zeta(s)\Gamma\left(\frac{s}{2}\right)\pi^{-s/2} = \int_0^\infty x^{s/2-1}\psi(x) \, ds$$

where

$$\psi(x) = \sum_{n=1}^{\infty} e^{-n^2 \pi x}$$

and where throughout this thesis $\Gamma(s)$ is the gamma function defined by

$$\Gamma(s) = \int_0^\infty x^{s-1} e^{-x} \ dx$$

for $\Re(s) > 0$. The gamma function satisfies the functional equation

$$\Gamma(n+1) = n\Gamma(n)$$
, with $\Gamma(1) = 1$.

Clearly

$$\psi(x) = \frac{\theta(x) - 1}{2}$$

and by Poisson summation we may show that $\theta(x)$ satisfies the functional equation

$$\theta(x) = \frac{1}{\sqrt{x}} \theta\left(\frac{1}{x}\right).$$

CHAPTER 1. INTRODUCTION

Using these facts we can show

$$\zeta(s)\Gamma\left(\frac{s}{2}\right)\pi^{-s/2} = \frac{1}{s(s-1)} + \int_1^\infty \left(x^{-s/2-1/2} + x^{s/2-1}\right)\psi(x) \ ds.$$

This integral converges for all s due to the rapid decay of $\psi(x)$, and so the meromorphic continuation follows. Since the integral is unchanged if s is replaced by 1 - s, we have established the formula

$$\zeta(s)\Gamma\left(\frac{s}{2}\right)\pi^{-s/2} = \zeta(1-s)\Gamma\left(\frac{1-s}{2}\right)\pi^{s/2-1/2}$$

which is the functional equation for $\zeta(s)$. This is often written as

$$\zeta(s) = \chi(s)\zeta(1-s) \tag{1.1}$$

where

$$\chi(s) = \pi^{s-1/2} \frac{\Gamma(1/2 - s/2)}{\Gamma(s/2)}$$

= $2^s \pi^{s-1} \sin\left(\frac{\pi s}{2}\right) \Gamma(1 - s).$ (1.2)

Riemann's second proof involves manipulating the integral

$$\zeta(s) = e^{-i\pi s} \Gamma(1-s) \frac{1}{2\pi i} \int_C \frac{x^{s-1}}{e^x - 1} \, dx,$$

where C is a certain keyhole contour. Using this representation, it can be shown that for all natural numbers

$$\zeta(-n) = (-1)^n \frac{B_{n+1}}{n+1} \tag{1.3}$$

where the B_n are the Bernoulli numbers, given by the coefficients in the Taylor series

$$\frac{z}{e^z - 1} = \sum_{j=0}^{\infty} B_j \frac{z^j}{j!}$$

for $|z| < 2\pi$.

Setting n = -1 + 2k for integer k in equation (1.3), we may show using the functional equation (1.1) that

$$\zeta(2k) = \frac{(-1)^{k+1}(2\pi)^{2k}B_{2k}}{2(2k)!}.$$

In particular, setting k = 1 recovers Euler's solution to the Basel problem, that is, that $\zeta(2) = \pi^2/6$. Many other proofs have followed Euler's original solution. Note that since integer powers of π are transcendental (and so irrational), so is $\zeta(2k)$ for all positive integers k. However, the problem of showing that $\zeta(2k + 1)$ is irrational for all positive integers k is still open in general. A special case is due to Apéry, who showed $\zeta(3)$ is irrational [4]. It is still unknown if it is transcendental. There is also no known closed form for $\zeta(3)$ in terms of π .

1.1.3 Zeta Zeros and the Riemann Hypothesis

We subscribe to the philosophy that if we want to understand a complex function, we need to understand its zeros. We now describe the locations of these zeros.

First note that there are no zeros in the plane $\Re(s) > 1$. To see this, note that we have the Euler product representation of $\zeta(s)$ in this region. Since a convergent infinite product of non-zero factors is not zero, we deduce that $\zeta(s)$ is non-zero here.

By the functional equation, $\zeta(s)$ is also non-zero for $\Re(s) < 0$ except where $\chi(s) = 0$. We see that this occurs at the negative even integers, either by (1.2) and noting that this is zero when the sine term is zero, or by (1.3) and noting that the odd Bernoulli numbers are zero for $n \ge 3$. We call these zeros the trivial zeros of $\zeta(s)$.

The only possible location of the zeros are in the region $0 \leq \Re(s) \leq 1$, the critical strip. As we will show in Section 1.1.4, there are no zeros with $\Re(s) = 1$, and so by the functional equation there are none with $\Re(s) = 0$, leaving the region $0 < \Re(s) < 1$. The zeros within the critical strip are called the non-trivial zeros which we will denote by $\rho = \beta + i\gamma$. If there is a non-trivial zero ρ then we automatically have that $1 - \rho$ is also a zero by the functional equation. Also, as $\zeta(s) = \overline{\zeta(\overline{s})}$, $\overline{\rho}$ is also a zero and then so is $1 - \overline{\rho}$, again by the functional equation. Therefore, if we have one zero in the critical strip, in reality we have up to four zeros (see the comments below on the Riemann Hypothesis to see why this number is actually more likely two).

While complex analysis hadn't reached an appropriate level of maturity when Riemann presented his paper, he predicted that the zeta function should have a third representation (beyond the infinite series and product over primes) as a product over its zeros. He was proved true by Hadamard in 1893 [154] who showed that $\zeta(s)$ satisfies a Hadamard product given by

$$\zeta(s) = \frac{e^{(\log(2\pi) - 1 - \gamma_0/2)s}}{2(s-1)\Gamma(1+s/2)} \prod_{\rho} \left(1 - \frac{s}{\rho}\right) e^{s/\rho},$$

where γ_0 is Euler's constant and the product is over all non-trivial zeros of $\zeta(s)$. In this way, the primes are related to the zeros of $\zeta(s)$ in a fundamental way which we see through the different representations of the zeta function. This is a theme that we come back to repeatedly through what are known as explicit formulae.

As with odd values of zeta, Euler's constant γ_0 has a host of open questions. For example, we do not know whether it is irrational or transcendental. One way of defining Euler's constant is as the limiting difference between the harmonic series and the natural logarithm, given by

$$\lim_{n \to \infty} \left\{ \sum_{k=1}^n \frac{1}{n} - \log n \right\} = \gamma_0.$$

Another way of defining Euler's constant, as we will see many times throughout this thesis, is from the Laurent expansion of the Riemann zeta function about s = 1, where

$$\lim_{s \to 1} \left\{ \zeta(s) - \frac{1}{s-1} \right\} = \gamma_0$$

The final representation that we mention, although don't use it in this thesis, is that we can write Euler's constant as

 $-\Gamma'(1) = \gamma_0.$

Finally, we can count the number of zeros within regions of the critical strip, typically above the real axis up to some height T. As before, Riemann stated the formula in his 1859 paper, and later von Mangoldt proved it [305]. Through the argument principle, they showed that this count of zeros,

$$N(T) = \#\{\rho = \beta + i\gamma : 0 < \beta < 1, \ 0 \le \gamma \le T\},$$
(1.4)

is given by the Riemann-von Mangoldt formula,

$$N(T) = \frac{T}{2\pi} \log \frac{T}{2\pi} - \frac{T}{2\pi} + \frac{7}{8} + S(T) + O\left(\frac{1}{T}\right),$$
(1.5)

with

$$S(T) = \frac{1}{\pi} \arg \zeta \left(\frac{1}{2} + it\right) \ll \log T,$$

where the argument is defined by continuous variation from $+\infty + iT$ to 1/2 + iT, starting with the value 0.

1.1.3.1 The Riemann Hypothesis

Why did we say that for each zero in the critical strip, there are 'up to' four zeros in total? This is because if the real part of the zero equals 1/2, then there are only two zeros, ρ and $\bar{\rho}$. If a zero has real part 1/2, we say it lies on the critical line. Indeed, we believe that all non-trivial zeros have this property, a conjecture first made by Riemann in his 1859 paper, which we now call the Riemann Hypothesis and one that has taken on a life of its own. Due to its significance, we state it separately below. We will often assume the truth of this conjecture in this thesis.

Conjecture 1.1 (Riemann Hypothesis). All non-trivial zeros of $\zeta(s)$ have real part 1/2.

The Riemann Hypothesis remains an open question to this day. It is considered one of, if not the most, important questions in mathematics. It appears on the list of the seven Millennium Problems set by the Clay Institute and comes with a prize of \$1 million for whoever can prove it. A full description can be found in [34] and in [58].

While we do not have a proof of this problem, some partial progress has been made. Numerically, all zeros that we have computed lie on the critical line - at time of writing, this is the first 10^{13} zeros [264], along with several billion other zeros higher up the critical line [259]. The Riemann zeta function is just one of a wider class of functions, called *L*-functions, each of which has its own representation as a Dirichlet series, Euler product, a meromorphic continuation, a functional equation, and a Riemann Hypothesis and each of which has all of its known zeros on the critical line.

In 1914, Hardy [163] showed that there are infinitely many zeros on the critical line. Selberg [276] improved this to show that a positive proportion lie on the line, and Min [238] gave an explicit constant for this proportion, showing that at least 1/14,074,731 of the zeros of $\zeta(s)$ lie on the line. Levinson [222] improved this to show that at least one-third of the zeros lie on the line through a method that now carries his name. Conrey [56] improved this to two-fifths, and the current record is due to Pratt, Robles, Zaharescu and Zeindler [265] who showed that just over five-twelfths lie on the line. A more detailed discussion of these results can be found in Section 1.3.7.3.

In addition to these partial results, in the theory of algebraic varieties over finite fields, the analogous Riemann Hypothesis (one of the Weil Conjectures) was proved true by Deligne [309, 99, 100] giving us further evidence for the truth of the Riemann Hypothesis in general.

Many equivalent forms of the Riemann Hypothesis are known, some of which we discuss later in this thesis.

While the Riemann Hypothesis answers the question of the horizontal distribution of the zeros, we can also ask questions on the vertical distribution of the zeros, a topic we come to in Section 1.2.2.

1.1.4 A proof of the Prime Number Theorem

We have given a summary of most of Riemann's paper in the previous section, but he went further still. We now sketch his arguments, and those of Hadamard [155] and de la Vallée Poussin [95] in proving the Prime Number Theorem (combined with some modern ideas that simply the proofs).

An explicit formula is one that relates zeros of an *L*-function to prime numbers. Riemann was able to establish an explicit formula for a function related to $\pi(x)$. We do this in a more modern way by establishing an explicit formula for a related, more mathematically natural function, the Chebyshev- ψ function, defined by

$$\psi(x) = \sum_{n \le x} \Lambda(n) = \sum_{\substack{p^k \le x \\ p \text{ prime}}} \log p \tag{1.6}$$

where

$$\Lambda(n) = \begin{cases} \log p & \text{if } n = p^k \text{ for some prime } p \text{ and integer } k \ge 1\\ 0 & \text{otherwise} \end{cases}$$

is the von Mangoldt function. While this new way of counting primes (and prime powers) may seem odd at first glance, it is actually mathematically natural. This is because the logarithmic derivative of $\zeta(s)$ is given by

$$\frac{\zeta'}{\zeta}(s) = -\sum_{n=2}^{\infty} \frac{\Lambda(n)}{n^s}$$

for $\Re(s) > 1$. Then we can show that the Prime Number Theorem is equivalent to

$$\psi(x) \sim x$$

as $x \to \infty$. By Perron's formula,

$$\psi(x) = -\frac{1}{2\pi i} \int_{(c)} \frac{\zeta'}{\zeta}(s) \frac{x^s}{s} \, ds$$

where c > 1, $\int_{(c)}$ denotes an integral up the vertical line $\Re(s) = c$, and where we are ignoring various technicalities, error terms, and whether x is an integer. Then move the line of integration to the left and compute residues. We have residues at s = 1 giving a contribution of x, at the trivial and non-zeros of $\zeta(s)$ giving a contribution of $-\sum_{z} x^{z}/z$, and at s = 0 giving a contribution of $-\log(2\pi)$. Overall, again ignoring technicalities and error terms, we have

$$\psi(x) = x - \sum_{\zeta(z)=0} \frac{x^z}{z} - \log(2\pi).$$

We can split the sum over zeros into trivial and non-trivial zeros. Recalling that the trivial zeros occur at z = -2n for n a natural number, we have

$$\sum_{n=1}^{\infty} \frac{1}{2nx^{2n}} = -\frac{1}{2} \log\left(1 - \frac{1}{x^2}\right)$$

which vanishes as $x \to \infty$. Therefore we have for large x,

$$\psi(x) \sim x - \sum_{\rho} \frac{x^{\rho}}{\rho},$$

as $x \to \infty$, where the sum is over the non-trivial zeros of $\zeta(s)$ and is conditionally convergent (the sum over zeros should be taken in increasing order of imaginary part in complex conjugate pairs).

1.1. The Prime Number Theorem

The Prime Number Theorem would then follow if we could show there are no non-trivial zeros with $\Re(\rho) = 1$. This is the step that Hadamard and de la Vallée Poussin completed in 1896. Taking logarithms of the Euler product for $\zeta(s)$, we have

$$\log \zeta(s) = \sum_{p \text{ prime}} \sum_{k=1}^{\infty} \frac{1}{k} p^{-ks}$$

so writing $s = \sigma + it$,

$$|\zeta(s)| = \exp\left(\sum_{p \text{ prime}} \sum_{k=1}^{\infty} \frac{\cos(kt \log p)}{kp^{k\sigma}}\right).$$

Then for $\sigma > 1$,

$$\zeta(\sigma)^3 |\zeta(\sigma+it)|^4 |\zeta(\sigma+2it)| = \exp\left(\sum_{p \text{ prime}} \sum_{k=1}^\infty \frac{3+4\cos(kt\log p) + \cos(2kt\log p)}{kp^{k\sigma}}\right) \ge 1$$

since $3 + 4\cos\theta + \cos 2\theta = 2(1 + \cos\theta)^2 \ge 0$.

Since we have a simple pole at s = 1, as $\sigma \to 1^+$, $\zeta(\sigma)^3 = O((\sigma - 1)^3)$. If we assume for a contradiction that there is some $t \neq 0$ with $\zeta(1+it) = 0$, then $|\zeta(\sigma+it)|^4 = O((\sigma - 1)^{-4})$. We make no assumptions about $|\zeta(\sigma + 2it)|$ but as we know it isn't a pole, it is just O(1). Combining this, as $\sigma \to 1^+$, we have

$$\zeta(\sigma)^3 |\zeta(\sigma+it)|^4 |\zeta(\sigma+2it)| = O(\sigma-1),$$

contradicting that this should be positive for $\sigma > 1$. Therefore $\zeta(s)$ does not vanish on $\Re(s) = 1$ and so the Prime Number Theorem follows.

We note that this is the only shrinking of the critical strip that has been proved since Riemann introduced his function. We cannot say unconditionally that there are no non-trivial zeros with $\Re(\rho) = 1 - \varepsilon$ for any $\varepsilon > 0$. However some zero-free regions are known, regions where $\zeta(s) \neq 0$ inside the critical strip. Note that for each region, as $T \to \infty$, each is equivalent to the fact that $\zeta(s)$ does not vanish on $\Re(s) = 1$. For sufficiently large t with c > 0 a positive constant, we have $\zeta(\sigma + it) \neq 0$ when

$$\sigma \geq 1 - \frac{c}{\log |t|},$$

which is due to de la Vallée Poussin [96], with improvements due to Littlewood [226], Chudakov [54], and Korobov [217] and Vinogradov [303]. Explicit versions of zero-free regions can be found in [117, 206].

We finish this section by mentioning two further points regarding the Prime Number Theorem. Firstly, if we want to write this as a full asymptotic rather than just to leading order, then de la Vallée Poussin showed in 1899 [96] that unconditionally for C > 0 that

$$\pi(x) = \operatorname{Li}(x) + O(xe^{-C\sqrt{\log x}}), \qquad (1.7)$$

with a further slight improvement by Korobov [217] and Vinogradov [303] who showed using their zero-free region for $\zeta(s)$ that for C > 0,

$$\pi(x) = \operatorname{Li}(x) + O\left(x \exp\left(-C(\log x)^{3/5} (\log\log x)^{-1/5}\right)\right).$$

These error terms in the Prime Number Theorem follow from the zero-free regions discussed above.

If we assume the Riemann Hypothesis then von Koch showed in 1901 [304] that

$$\pi(x) = \text{Li}(x) + O(x^{1/2} \log x).$$

This power of 1/2 on the x in the error term is exactly the 1/2 in the real part of the zeta zeros (to the extent that if we prove this is the true error term without assuming the Riemann Hypothesis, then we have proved Riemann was correct in making his hypothesis). If we have any zeros off the line, then von Koch [304] showed that the error term becomes $O(x^{\Theta} \log x)$, where

$$\Theta = \sup_{\rho} \Re(\rho)$$

which satisfies $1/2 \le \Theta \le 1$.

Finally, it appeared that it was always the case that $\text{Li}(x) > \pi(x)$, displaying what we call Chebyshev bias (see Section 1.4.3.4), with numerical evidence supporting this. However Littlewood [224] showed that $\text{Li}(x) - \pi(x)$ changes sign infinitely often. Skewes [283, 284] showed under the Riemann Hypothesis that the first time this happens is by $x \approx 10^{10^{10^{34}}}$. The best known upper bound for this first sign change is by $x \approx 1.397166161527 \times 10^{316}$, calculated in [286]. While a fascinating topic in its own right which we revisit later, it serves as a warning that just because we have what seems to be overwhelming numerical evidence, what we really need are rigorous proofs.

1.1.4.1 The Prime Number Theorem in Arithmetic Progressions

While we will mainly only consider the Riemann zeta function in this thesis, it belongs to a wider class of functions called *L*-functions. We mention one specific type, called Dirichlet *L*-functions. To understand what these are, we being with a Dirichlet character $\chi \mod q$ (where q is a positive integer) which is an arithmetic function $\chi : \mathbb{Z} \to \mathbb{C}$ that satisfies

- 1. $\chi(n+q) = \chi(n)$ for all $n \in \mathbb{Z}$
- 2. $\chi(n) = 0$ if $(n,q) \neq 1$
- 3. $\chi(mn) = \chi(m)\chi(n)$ for all $m, n \in \mathbb{Z}$
- 4. $\chi(1) = 1$.

There are exactly $\varphi(q)$ distinct Dirichlet characters mod q, where $\varphi(q)$ is the Euler totient function which counts the number of positive integers less than or equal to q that are coprime to q.

A Dirichlet *L*-function associated with a Dirichlet character $\chi \mod q$ has Dirichlet series representation

$$L(s,\chi) = \sum_{n=1}^{\infty} \frac{\chi(n)}{n^s}$$

which converges for all complex s with $\Re(s) > 1$. As with the Riemann zeta function, it has an Euler product representation given by

$$L(s,\chi) = \prod_{p} \left(1 - \frac{\chi(p)}{p^s}\right)^{-1}$$

which converges for all complex s with $\Re(s) > 1$.

The principal character χ_0 is given by

$$\chi_0(n) = \begin{cases} 1 & \text{if } (n,q) = 1 \\ 0 & \text{otherwise.} \end{cases}$$

The associated L-function is

$$L(s,\chi_0) = \zeta(s) \prod_{p|q} (1-p^{-s})$$

A Dirichlet L-function satisfies a function equation of the form

$$L(s,\chi) = F(s)L(1-s,\overline{\chi})$$

where F(s) is a certain function (similar to the term in the functional equation for the Riemann zeta function) and $\overline{\chi}$ is the complex conjugate of χ . This shows that $L(s,\chi)$ has an analytic continuation to the whole complex plane for $\chi \neq \chi_0$ and a meromorphic continuation with a simple pole if $\chi = \chi_0$.

Dirichlet L-functions have trivial zeros at $s = 0, -2, -4, \ldots$ if $\chi(-1) = 1$ and at $s = -1, -3, -5, \ldots$ if $\chi(-1) = -1$, and have their non-trivial zeros in the critical strip. They are believed to satisfy a Generalised Riemann Hypothesis, that is, it is believed that all non-trivial zeros of any Dirichlet L-function lie on the critical line.

We can ask many of the questions for Dirichlet L-functions that we do for the Riemann zeta function. Most are not considered in this thesis, but a general survey of such questions can be found in [2].

Part of the reason to mention Dirichlet L-functions at this point is to talk about their connection with prime numbers. Just as with using the Riemann zeta function to study prime numbers, Dirichlet L-functions can be used to study the prime numbers in arithmetic progressions. Questions of these forms are what led Dirichlet to first consider these functions. We mention two such results briefly, but many more questions about primes in arithmetic progressions can be considered, and we mention a few at other points in this introduction.

First, we start with Dirichlet's theorem on primes in arithmetic progressions [105]. It states that if (a,q) = 1, then there are infinitely many prime numbers congruent to $a \mod q$. He proved this by showing through the orthogonality of Dirichlet characters that

$$\sum_{\substack{p \le x \\ p \equiv a \bmod q}} \frac{1}{p} = \frac{1}{\varphi(q)} \sum_{p \le x} \frac{1}{p} + O(1)$$

where (a, q) = 1. Since the sum of the reciprocal of the primes diverges, his result follows. In particular, he was required to show

$$\sum_{p \le x} \frac{\chi(p)}{p} = O(1)$$

for all $\chi \neq \chi_0$. He noticed through the Euler product representation for Dirichlet *L*-functions that

$$\log L(1,\chi) = \sum_{p} \frac{\chi(p)}{p} + O(1).$$

Since $L(s, \chi)$ is analytic at s = 1 for $\chi \neq \chi_0$, it was sufficient to show that $L(1, \chi) \neq 0$.

Finally, we write $\pi(x; q, a)$ for the number of primes congruent to $a \mod q$ that are less than or equal to x. These primes equidistribute between equivalence classes in the way we expect from the Prime Number Theorem, that is, for (a, q) = 1, we expect

$$\pi(x;q,a) \sim \frac{1}{\varphi(q)} \frac{x}{\log x}$$

as $x \to \infty$ for fixed q, a fact that de la Vallée Poussin [95] proved in 1896.

1.2 RANDOM MATRIX THEORY

Random Matrix Theory (RMT) can be thought of as the probabilistic study of matrices whose entries are random variables. It has been used in a wide range of applications across mathematics and physics, including in particular the study of spectra of heavy atoms. Most surprisingly, it has found applications in number theory in a way that we describe in this section and throughout this thesis. There are many books on RMT, for example [231].

1.2.1 The Classical Compact Groups

The central objects of study in this section are randomly chosen elements of the classical compact matrix groups, namely the unitary, orthogonal, and symplectic groups. Writing I_N for the $N \times N$ identity matrix, we define these three groups together with some basic properties below.

1. Unitary Group U(N): The unitary group is the group of all $N \times N$ complex matrices U such that

$$UU^{\dagger} = U^{\dagger}U = I_N$$

where U^{\dagger} is the complex transpose of U.

The eigenvalues lie on the unit circle in the complex plane. We write them as $e^{i\theta_1}, ..., e^{i\theta_N}$ where $\theta_i \in [0, 2\pi)$.

2. Orthogonal Group O(N): The orthogonal group is the group of all $N \times N$ complex matrices O such that

$$OO^T = O^T O = I_N$$

where O^T is the transpose of O.

The special orthogonal group is the subgroup with $\det O = 1$.

Specialising to the group of even dimensional special orthogonal matrices SO(2N), the eigenvalues come in complex conjugate pairs, written as $e^{i\theta_1}, e^{-i\theta_1}, ..., e^{i\theta_N}, e^{-i\theta_N}$ where $\theta_j \in [0, \pi]$.

3. Symplectic Group Sp(2N): The symplectic group is the group of all $2N \times 2N$ complex matrices $S \in U(N)$ such that

$$SJS^T = J$$

where

$$J = \begin{pmatrix} 0 & I_N \\ -I_N & 0 \end{pmatrix}.$$

The eigenvalues come in complex conjugate pairs, denoted $e^{i\theta_1}, e^{-i\theta_1}, \ldots, e^{i\theta_N}, e^{-i\theta_N}$ where $\theta_j \in [0, \pi]$.

Each of the classical compact groups we have considered is a compact Lie group. This means that they have a unique invariant probability measure under the action of the group, called Haar measure. The unitary, orthogonal, and symplectic group ensemble is the group attached with its respective Haar measure. The unitary group with Haar measure is also known as the Circular Unitary Ensemble and is the key group ensemble that we focus on in this thesis. Note that the orthogonal/symplectic group with Haar measure is not the Circular Orthogonal/Symplectic Ensemble.

The calculation of the joint probability density of the eigenangles under Haar measure is due to Weyl [311]. We list the three joint probability densities here for completeness.

1. Unitary U(N):

$$\frac{1}{N!(2\pi)^N} \prod_{1 \le j < k \le N} \left| e^{i\theta_j} - e^{i\theta_k} \right| \prod_{n=1}^N d\theta_n$$

2. Even Special Orthogonal SO(2N):

$$\frac{2^{(N-1)^2}}{N!\pi^N} \prod_{1 \le j < k \le N} (\cos \theta_j - \cos \theta_k)^2 \prod_{n=1}^N d\theta_n$$

3. Symplectic Sp(2N):

$$\frac{2^{N^2}}{N!\pi^N} \prod_{1 \le j < k \le N} (\cos \theta_j - \cos \theta_k)^2 \prod_{n=1}^N \sin^2 \theta_n d\theta_n$$

These joint probability densities are used in the Weyl integration formula. Weyl's integration formula relates the average of a class function (a function which is symmetric in all of its variables) over a matrix ensemble to a multidimensional integral of the same function, over the eigenvalues of the matrices in that ensemble. In the unitary case, this is defined by

$$\mathbb{E}_{\mathbb{N}}\left\{f(\theta_{1},...,\theta_{N})\right\} = \frac{1}{N!(2\pi)^{N}} \int_{0}^{2\pi} ... \int_{0}^{2\pi} f(\theta_{1},...,\theta_{N}) \prod_{1 \le j < k \le N} \left|e^{i\theta_{j}} - e^{i\theta_{k}}\right|^{2} \prod_{n=1}^{N} d\theta_{n}$$
(1.8)

where $\mathbb{E}_{\mathbb{N}}$ denotes expectation with respect to the Circular Unitary Ensemble of an $N \times N$ matrix, the θ_n are the eigenangles of an $N \times N$ unitary matrix, and $f(\theta_1, ..., \theta_N)$ is a class function.

One of the major themes in this thesis are calculations relating to the characteristic polynomial of a matrix A from one of our groups above. For our purposes, we define the characteristic polynomial of such a matrix from one of these groups to be

$$Z_{N,A}(\theta) = \det(I - Ae^{-i\theta}) \tag{1.9}$$

where A is from one of U(N), SO(2N), Sp(2N). When it is clear which group we are working with we drop the subscript A. In the unitary case, the characteristic polynomial can be written as

$$Z_N(\theta) = \prod_{n=1}^N \left(1 - e^{i(\theta_n - \theta)} \right)$$

and in the even special orthogonal or symplectic cases, the characteristic polynomial can be written as

$$Z_N(\theta) = \prod_{n=1}^N \left(1 - e^{i(\theta_n - \theta)} \right) \left(1 - e^{-i(\theta_n + \theta)} \right).$$

Note that the characteristic polynomial, its derivative, and the logarithmic derivative are all class functions, and so Weyl's integration formula will be useful in our study of these functions.
1.2. RANDOM MATRIX THEORY

1.2.2 RANDOM MATRIX THEORY MEETS NUMBER THEORY: ACT I

The connection between RMT and number theory can be traced directly back to a single conversation at the Institute of Advanced Studies in Princeton in 1972, as described in [209]. Hugh Montgomery was visiting the IAS and was introduced to the physicist Freeman Dyson by Sarvadaman Chowla. When Dyson asked Montgomery what he was working on, Montgomery told him about his work on the vertical distributions of the zeros of $\zeta(s)$, in particular on a result on the pair correlation of the zeros. Dyson instantly recognised the result as being identical to that of the pair correlation of the eigenvalues of random unitary matrices. This chance meeting at a time when Montgomery had just completed this work together with Dyson's insight from his work on nuclei of heavy atoms was the birth of the future fruitful investigation into the links between RMT and the Riemann zeta function. While Riemann's original manuscript was the first epoch-making paper that kick started our interest in the zeta function, it is fair to say that Montgomery's paper on pair correlation was the second.

Montgomery [239] was one of the first to consider the vertical distribution of the zeros of $\zeta(s)$ under the assumption of the the Riemann Hypothesis. To motivate this problem, Montgomery considered the function

$$F(\alpha, T) = \left(\frac{T}{2\pi} \log T\right)^{-1} \sum_{0 < \gamma, \gamma' \le T} T^{i\alpha(\gamma - \gamma')} w(\gamma - \gamma')$$

where $w(u) = 4/(4+u^2)$ is a weight function, α is real and $T \ge 2$.

Montgomery observed that $F(\alpha, T)$ is a real-valued even function in α , and proved the following about $F(\alpha, T)$.

Theorem 1.2 (Montgomery). Assume the Riemann Hypothesis. Then

$$F(\alpha, T) = (1 + o(1))T^{-2\alpha}\log T + \alpha + o(1)$$
(1.10)

uniformly for $0 \le \alpha \le 1 - \varepsilon$ for any fixed $\varepsilon > 0$, as $T \to \infty$.

Later, Mueller and Heath-Brown noted that F is non-negative (see [52]) and Goldston and Montgomery [142] showed Montgomery's Theorem 1.2 holds uniformly for $0 \le \alpha \le 1$.

Using a quantitative form of the Hardy–Littlewood twin prime hypothesis, Montgomery conjectured that for $\alpha \geq 1$,

$$F(\alpha, T) = 1 + o(1) \tag{1.11}$$

as $T \to \infty$, uniformly in α .

Combining this conjecture with his Theorem 1.2, as $F(\alpha, T)$ is even in α ,

$$\lim_{T \to \infty} F(\alpha, T) = \begin{cases} |\alpha| + \delta(\alpha) & \text{if } |\alpha| \le 1\\ 1 & \text{if } |\alpha| \ge 1 \end{cases}$$
(1.12)

where $\delta(\alpha)$ is the Dirac delta function.

Now rescale the zeros so they have unit mean spacing, that is,

$$w_n = \frac{\gamma_n}{2\pi} \log \frac{\gamma_n}{2\pi}$$

where the γ_n are the ordinates of the non-trivial zeros of $\zeta(s)$, ordered by height above the real axis.

Assuming his conjecture and taking the Fourier transform of $F(\alpha, T)$, Montgomery [239] was lead to his conjecture known as the Pair Correlation Conjecture.

Conjecture 1.3 (Pair Correlation Conjecture). For fixed a < b,

$$\lim_{T \to \infty} \frac{1}{N(T)} \sum_{\substack{\gamma, \gamma' \in [0,T]\\a \le w_m - w_n \le b}} 1 = \int_a^b \left(1 - \left(\frac{\sin \pi x}{\pi x}\right)^2 \right) dx + \delta(a,b)$$

where

$$\delta(a,b) = \begin{cases} 1 & \text{if } 0 \in [a,b] \\ 0 & \text{otherwise.} \end{cases}$$

This term reflects the fact that if $a \leq 0 \leq b$, then the sum includes the terms $\gamma = \gamma'$.

There is considerable numerical support for this conjecture, initiated by Odlyzko [257, 258]. Higher correlations have since been studied. The three-point correlation of zeta zeros by Hejhal [178] matches the three-point correlation of U(N). Rudnick and Sarnak [275] then showed that the *n*-level correlations of zeros of principal *L*-functions match the analogous results from the RMT side. Both of these works are based on appropriate restrictions on the Fourier transform of the test function. Avoiding this issue, Bogomolny and Keating [30, 31] obtain the above conjectures heuristically, including all lower order terms that arise from arithmetical considerations. We will note later that the Ratios Conjecture [87] recovers these full asymptotics as discussed in Section 6.1. A key feature of these heuristics is that they do not restrict the support of the Fourier transform. Bogomolny and Leboeuf [32] have extended this heuristic to calculate the pair correlation for Dirichlet *L*-functions, which again agree with what we see from the U(N) result.

From the RMT side, to state Dyson's theorem [107, 108, 109] we let $U \in U(N)$ with eigenvalues $e^{i\theta_1}, ..., e^{i\theta_N}$. We rescale the eigenphases to have unit mean density to give

$$\phi_j = \theta_j \frac{N}{2\pi}$$

Theorem 1.4 (Dyson). For a test function f such that $f(x) \to 0$ as $|x| \to \infty$, the pair correlation of eigenvalues tends to a sine kernel as $N \to \infty$. That is,

$$\lim_{N \to \infty} \mathbb{E}_N \left\{ \frac{1}{N} \sum_{m,n \le N} f(\phi_m - \phi_n) \right\} = \int_{-\infty}^{\infty} f(x) \left(\delta(x) + 1 - \left(\frac{\sin \pi x}{\pi x}\right)^2 \right) dx$$

where \mathbb{E}_N denotes the expected value taken over U(N) with respect to Haar measure.

Taking the test function to be $f(x) \equiv 1$ for $x \in [a, b]$ and 0 otherwise gives

$$\mathbb{E}_{N}\left\{\frac{1}{N}\sum_{\substack{\phi_{m},\phi_{n}\leq N\\a<\phi_{m}-\phi_{n}< b}}1\right\} = \int_{a}^{b}\left(1-\left(\frac{\sin\pi x}{\pi x}\right)^{2}\right) dx + \delta(a,b)$$

as $N \to \infty$, with $\delta(a, b)$ defined as in Conjecture 1.3.

Comparing Montgomery's Theorem with the RMT result of Dyson makes it obvious why Dyson made his remarks at the IAS. If Montgomery's Pair Correlation Conjecture is true then this suggests there is a spectral interpretation for the Riemann zeta-function as an infinite dimensional operator, which is known as the Hilbert–Pólya conjecture. The idea of randomly choosing large matrices and averaging them should capture information about this one specific matrix related to the zeta function, even if we can't prove the Hilbert–Pólya conjecture. We have evidence supporting this conjecture beyond just the numerical. The strongest theoretical evidence comes from the function field setting as it is possible to provide a spectral interpretation of the zeros in terms of certain eigenvalues, as previously mentioned when discussing the Weil Conjectures.

1.3 CONTINUOUS MOMENTS OF THE RIEMANN ZETA FUNCTION

We saw in Section 1.1 how important it is to understand the Riemann zeta function for our understanding of the distribution of prime numbers. This interest in understanding the primes through the zeros of the Riemann zeta function has led us to ask questions about properties of the zeta function for its own sake.

One such question concerns the size of $\zeta(s)$, either for exceptional values, or on average. We will focus on the average in this thesis but will mention large values occasionally. Averages typically split into two main types that are traditionally considered, with a further one being the main topic of this thesis. The two types that have previously been studied deeply are those called continuous and discrete moments. The third kind is a variant of the typical discrete moments, with a general overview of these moments given in the Preface to this thesis, and with the full details starting in Chapter 2.

The continuous moments are integrals of the form

$$I_{2k}(T) = \int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} dt$$

and the discrete moments are sums of the form

$$J_{2k,n}(T) = \sum_{0 < \gamma \le T} \left| \zeta^{(n)} \left(\frac{1}{2} + i\gamma \right) \right|^{2k}$$

where the γ are the ordinates of the non-trivial zeros of $\zeta(s)$ and $\zeta^{(n)}(s)$ denotes the n^{th} derivative of $\zeta(s)$.

We can also consider continuous moments of derivatives of $\zeta(s)$, as mentioned in (1.16). For further examples of these moments, see [192, 55, 157].

Additionally, some authors use the convention of writing k rather than 2k in the subscripts for the labels of the moments moments, particularly for the discrete case. Some also divide the continuous moments by the length of the integral T and the discrete moments by the number of zeros in the sum N(T). For consistency throughout this thesis we will endeavour to use our convention, although in some places (notably after conjecturing a zeta result from an RMT result) it makes more sense to divide by this length/number of zeros.

1.3.1 The Lindelöf Hypothesis

Why might we want to know about the averages of $\zeta(s)$? The initial motivation came from attempts to prove the Lindelöf Hypothesis, which concerns the size of $\zeta(s)$ on the critical line. We state it as follows.

Conjecture 1.5 (Lindelöf Hypothesis). For t > 0, for every $\varepsilon > 0$,

$$\zeta\left(\frac{1}{2}+it\right) = O(t^{\varepsilon})$$

as $t \to \infty$.

We note that the Riemann Hypothesis implies the Lindelöf Hypothesis but the converse is not believed to be true. This is because the Lindelöf Hypothesis is equivalent to a less stringent condition on the zeta zeros than the Riemann Hypothesis. The reason that the Riemann Hypothesis implies the Lindelöf Hypothesis is because Littlewood [227] showed that the Riemann Hypothesis implies

$$\left|\zeta\left(\frac{1}{2}+it\right)\right| \ll \exp\left(\frac{c\log t}{\log\log t}\right) \tag{1.13}$$

for some c > 0.

It can be shown, for example in Section 13.2 of [298], that the Lindelöf Hypothesis is equivalent to the statement that

$$I_{2k}(T) = O\left(T^{1+\varepsilon}\right)$$

for every $\varepsilon > 0$ and all positive integers k, giving our first clear motivation for studying moments.

1.3.2 Continuous Moment Results

There are only two proven asymptotic formulae for continuous moments, beyond the trivial k = 0 case. The second moment is due to Hardy and Littlewood [164] in 1918 where they showed to leading order that

$$I_2(T) \sim T \log T \tag{1.14}$$

as $T \to \infty$. In 1926 Ingham [192] considered a shifted second moment of $\zeta(s)$ with $|\alpha|, |\beta| < 1/2$ and showed that for every $\varepsilon > 0$,

$$\int_0^T \zeta \left(\frac{1}{2} + it + \alpha\right) \zeta \left(\frac{1}{2} - it + \beta\right) dt = \int_0^T \left(\zeta (1 + \alpha + \beta) + \left(\frac{t}{2\pi}\right)^{-\alpha - \beta} \zeta (1 - \alpha - \beta)\right) \left(1 + O(t^{-1/2 + \varepsilon})\right) dt. \quad (1.15)$$

This idea of using small shifts has proved to be incredibly useful in recent years in obtaining full asymptotics, including lower order terms. In this case, Ingham was able to show that

$$I_2(T) = T \log \frac{T}{2\pi} + T(2\gamma - 1) + O\left(T^{1/2} \log T\right)$$

with improvements in the error term being made since then, for example in Heath-Brown– Huxley [175] and Watt [308]. The size of the shifts has been increased by Bettin [25], who showed we may take the shifts α, β with $\Re(\alpha), \Re(\beta) \ll 1/\log T$ and $\Im(\alpha), \Im(\beta) \ll T^{2-\varepsilon}$, for any $\varepsilon > 0$. He conjectures that the imaginary parts may be of size up to $T^{4-\varepsilon}$ for any $\varepsilon > 0$, but not larger due to Omega results for the error term in $I_2(T)$. (We say that $f(x) = \Omega(g(x))$ for some functions f, g if the lim sup of |f(x)/g(x)| is positive as $x \to \infty$.) For some Omega results, see [190, 191], where it is shown that the error term is $\Omega_{\pm} (T^{1/4})$. It is believed that the correct size of the error term is $T^{1/4} \log T$.

As well as allowing us to extract lower order terms, this shifted second moment allows us to write asymptotics for mixed moments of the derivatives of $\zeta(s)$. In particular Ingham's result shows for fixed $m, n \in \mathbb{N}$,

$$\int_0^T \zeta^{(m)} \left(\frac{1}{2} + it\right) \zeta^{(n)} \left(\frac{1}{2} - it\right) dt = \frac{(-1)^{m+n}}{m+n+1} T(\log T)^{m+n+1} + O\left(T(\log T)^{m+n}\right).$$
(1.16)

At the same time as proving this second moment result, Ingham [192] showed that the fourth moment satisfies to leading order

$$I_4(T) \sim \frac{1}{2\pi^2} T(\log T)^4$$
 (1.17)

as $T \to \infty$. Later Heath-Brown [172] showed that the lower order terms are given by

$$I_4(T) = \sum_{n=0}^4 a_n T \left(\log \frac{T}{2\pi} \right)^n + O\left(T^{7/8+\varepsilon}\right)$$

with a_4 as in Ingham's result, an explicit a_3 , and for some implicit computable constants a_n and for all $\varepsilon > 0$. An equivalent form is given as a residue by Conrey [57], and a shifted version of the fourth moment was studied by Motohashi [247]. Improvements to the error term for Heath-Brown's result have been made, for example by Zavorotnyi [317] and by Ivić-Motohashi [200]. Motohashi [246] also showed that error term is $\Omega_{\pm} \left(T^{1/2}\right)$.

Other than these two moments, a smoothed version has also been calculated for the first two moments. We write

$$\tilde{I}_{2k}(T) = \int_0^\infty \left| \zeta \left(\frac{1}{2} + it \right) \right| e^{-\delta t} dt$$

where δ is small.

For the second moment, Kober [216] showed that

$$\tilde{I}_2(T) \sim \frac{1}{\delta} \log \frac{1}{\delta}$$

as $\delta \to 0$. For the fourth moment, Titchmarsh [296] showed

$$\tilde{I}_4(T) \sim \frac{1}{2\pi^2\delta} \left(\log \frac{1}{\delta}\right)^4$$

as $\delta \to 0$ with Atkinson [10] giving lower order terms in 1941. A different method was used to obtain this full asymptotic in Motohasi [245]. It is possible to use a Tauberian theorem to switch between the smoothed and unsmoothed versions to leading order, but not to lower order.

Returning to our standard continuous moments $I_{2k}(T)$, almost 70 years went by before any new results were even conjectured. In 1992, Conrey and Ghosh [68] conjectured that

$$I_6(T) \sim \frac{42}{9!} \prod_p \left\{ \left(1 - \frac{1}{p}\right)^4 \left(1 + \frac{4}{p} + \frac{1}{p^2}\right) \right\} T(\log T)^9$$

as $T \to \infty$. In 1998 Conrey and Gonek [76] conjectured that

$$I_8(T) \sim \frac{24024}{16!} \prod_p \left\{ \left(1 - \frac{1}{p}\right)^9 \left(1 + \frac{9}{p} + \frac{9}{p^2} + \frac{1}{p^3}\right) \right\} T(\log T)^{16}$$

as $T \to \infty$, as well as rederiving the conjecture for the sixth moment in a different way. However, the methods used didn't generalise to higher moments. Indeed, for $k \ge 5$, the method predicts that $I_{2k}(T)$ is negative, which is clearly false from its definition. In Section 1.3.6 we will discuss recent developments that fix this method and predict exactly the moments that we have come to expect.

It became a folklore conjecture after a result of Conrey and Ghosh [65] that the moments should be of the form

$$I_{2k}(T) \sim a_k g_k T (\log T)^{k^2}$$
 (1.18)

as $T \to \infty$. The arithmetic factor a_k is given by

$$a_{k} = \prod_{p} \left(1 - \frac{1}{p} \right)^{k^{2}} \sum_{m=0}^{\infty} \left(\frac{\Gamma(m+k)}{m! \Gamma(k)} \right)^{2} p^{-m}$$
(1.19)

and $g_k(k^2)!$ is an integer whenever k is a positive integer. Note that while the arithmetic term is written as an infinite sum, it can be rewritten in a way to show it is actually finite. To do this, write this arithmetic term as

$$a_{k} = \prod_{p} \left(1 - \frac{1}{p} \right)^{k^{2}} \sum_{m=0}^{\infty} d_{k}^{2}(p^{m})p^{-m}$$

where $d_k(n)$ denotes the n^{th} coefficient of the Dirichlet series for $\zeta(s)^k$. As Conrey and Ghosh [65] showed $a_k = a_{1-k}$, it follows that

$$a_k = \prod_p \left(1 - \frac{1}{p}\right)^{(k-1)^2} \sum_{m=0}^{\infty} d_{1-k}^2(p^m) p^{-m}$$

Since

$$d_k(p^m) = \binom{k+m-1}{m} = (-1)^m \binom{-k}{m},$$

we have

$$a_k = \prod_p \left(1 - \frac{1}{p}\right)^{(k-1)^2} \sum_{m=0}^{k-1} \binom{k-1}{m}^2 p^{-m}.$$

It is then clear how the arithmetic terms in the sixth and eighth moments conjectures arise.

The issue however is that even without rigorous results for the sixth and eighth moments, there didn't seem to be a way to predict what the value of the geometric factor g_k should be. This all changed with the third epoch-making work on the Riemann zeta function after Riemann and Montgomery, that of Keating and Snaith, which we discuss in Section 1.3.3.

1.3.2.1 Upper and Lower Bounds

There is a huge amount of literature on the subject of upper and lower bounds of continuous moments. We give a brief summary below. We know that for k > 0,

$$T(\log T)^{k^2} \ll I_{2k}(T) = \int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} dt \ll T(\log T)^{k^2}.$$

Here the lower bound holds unconditionally for all k > 0 and the upper bound holds unconditionally for $0 < k \le 2$ and conditionally on the Riemann Hypothesis for all k > 2.

For the lower bounds, Ramachandra [268] established the bound for all even natural numbers and Heath-Brown [174] extended this to all positive rational rational numbers.

Under the Riemann Hypothesis, Conrey and Ghosh showed that the folklore conjecture (1.18) was of the right form, showing

$$I_{2k}(T) \ge \left(\frac{a_k}{\Gamma(k^2+1)} + o(1)\right) T(\log T)^{k^2}.$$

This result was improved by Balasubramanian and Ramachandra [16], and later by Soundararajan [288]. Radziwiłł and Soundararajan [267] obtained unconditional lower bounds of the correct form for $k \ge 1$ and Heap and Soundararajan [170] completed this by showing unconditional bounds of the correct form hold for $0 < k \le 1$.

For the unconditional upper bounds, Heath-Brown [174] showed that they hold for k = 1/n for positive integer n, while Bettin, Chandee and Radziwiłł [27] showed the same result but for k = 1 + 1/n. Heap, Radziwiłł, and Soundararajan [169] showed this is the correct upper bound for $0 < k \leq 2$.

For the conditional upper bounds on the Riemann Hypothesis, Soundararajan [291] showed that the upper bound is of the form $C_k T (\log T)^{k^2 + \varepsilon}$ for all $\varepsilon > 0$, while Harper [166] managed to remove the ε from this result, showing that the upper bound is of the expected form and believed to be sharp.

1.3.3 RANDOM MATRIX THEORY MEETS NUMBER THEORY: ACT II

On the centenary of the proof of the Prime Number Theorem, a conference was held in Seattle, Washington in 1996 celebrating the proof and gathering all the experts to discuss the Riemann Hypothesis. Among those in attendance was Jon Keating who, at the time, was primarily a physicist who worked in RMT. While some work had been done to investigate the connection between RMT and number theory as highlighted by Dyson, such as numerical evidence of Odlyzko [257, 258] and the *n*-level correlations of Rudnick and Sarnak [275], it hadn't been fully exploited or investigated and was treated more as a curiosity than a serious avenue of investigation. Shortly before this, the sixth moment had been conjectured by Conrey and Ghosh [68], and Keating was challenged by Peter Sarnak to use RMT to rederive the known and conjectured moments, as described in [209].

With this challenge in mind, after the conference Keating returned to Bristol and together with his new PhD student Nina Snaith, investigated the moments problem from a RMT perspective. They had one hint of a starting point, the one Dyson had highlighted — the eigenvalues of random unitary matrices behave statistically like the zeros of $\zeta(s)$. If the non-trivial zeros of the zeta function are by definition the zeros of $\zeta(s)$, then what is the function that the eigenvalues are the zeros of? This is the characteristic polynomial of the matrix. Perhaps it might then be possible to model moments of $\zeta(s)$ by the moments of characteristic polynomials.

Recall in (1.9) that the characteristic polynomial of $U \in U(N)$ is given by

$$Z_N(\theta) = \det(I - Ue^{-i\theta})$$
$$= \prod_{n=1}^N \left(1 - e^{i(\theta_n - \theta)}\right)$$

Then for a fixed $\theta \in [0, 2\pi)$, the $2k^{\text{th}}$ moments of Z_N are given by

$$M_N(2k) = \mathbb{E}_N \left\{ \prod_{n=1}^N \left| 1 - e^{i(\theta_n - \theta)} \right|^{2k} \right\} = \int_{U(N)} |Z_N(\theta)|^{2k} d \text{Haar}$$

where d Haar is Haar measure on U(N).

By the Weyl integration formula (1.8), we have

$$M_N(2k) = \frac{1}{N!(2\pi)^N} \int_0^{2\pi} \dots \int_0^{2\pi} \prod_{1 \le j < k \le N} \left| e^{i\theta_j} - e^{i\theta_k} \right|^2 \prod_{n=1}^N \left| 1 - e^{i(\theta_n - \theta)} \right|^{2k} d\theta_n.$$
(1.20)

This may be evaluated using Selberg's integral. This can be found in various places, for example [277], Chapter 17 of [231] and Chapter 8 of [3].

Theorem 1.6 (Selberg). Let $n \in \mathbb{N}$ and let $a, b, \alpha, \beta, \gamma \in \mathbb{C}$ with $\Re(a, b, \alpha, \beta) > 0$, $\Re(\alpha + \beta) > 1$, and

$$-\frac{1}{n} < \Re(\gamma) < \min\left(\frac{\Re(\alpha)}{n-1}, \frac{\Re(\beta)}{n-1}, \frac{\Re(\alpha+\beta+1)}{2(n-1)}\right)$$

Then

$$J(a,b,\alpha,\beta,\gamma,n) = \int_{-\infty}^{\infty} \dots \int_{-\infty}^{\infty} \prod_{1 \le j < k \le N} |x_j - x_k|^{2\gamma} \prod_{j=1}^n (a+ix_j)^{-\alpha} (b-ix_j)^{-\beta} dx_1 \dots dx_n$$
$$= \frac{(2\pi)^n}{(a+b)^{(\alpha+\beta)n-\gamma n(n-1)-n}} \prod_{j=0}^{n-1} \frac{\Gamma(1+\gamma+j\gamma)\Gamma(\alpha+\beta-(n-1+j)\gamma-1)}{\Gamma(1+\gamma)\Gamma(\alpha-j\gamma)\Gamma(\beta-j\gamma)}.$$

Then write (1.20) in terms of absolute values of sine-functions and set $\theta = 0$ which we can do thanks to the rotational invariance of Haar measure on U(N). After a change of variables, Keating and Snaith obtain for $\Re(2k) > -1$,

$$M_N(2k) = \frac{2^{N^2 + 2kN}}{N!(2\pi)^N} J(1, 1, N + k, N + k, 1, N)$$

= $\prod_{n=1}^N \frac{\Gamma(n)\Gamma(n+2k)}{\Gamma(n+k)^2}$ (1.21)

Note that (1.21) has a meromorphic continuation in k to the rest of the complex plane.

Since this is a theorem for finite N, we may take $N \to \infty$ after using the definition of the Barnes G-function in (1.21) and expanding its asymptotics to give

$$M_N(2k) = \frac{G^2(k+1)}{G(2k+1)} N^{k^2} + O\left(N^{k^2-1}\right)$$
(1.22)

where G(z) is the Barnes G-function defined by Barnes in [19] for all z by

$$G(z+1) = (2\pi)^{z/2} \exp\left(-\frac{1}{2}(z^2 + \gamma_0 z^2 + z)\right) \prod_{n=1}^{\infty} \left(1 + \frac{z}{n}\right)^n e^{-z + z^2/2n}$$
(1.23)

where γ_0 is Euler's constant. The Barnes G-function satisfies the functional equation

$$G(z+1) = \Gamma(z)G(z)$$
, with $G(1) = 1$.

When k is an integer, the leading order coefficient in (1.22) simplifies to

$$\frac{G^2(k+1)}{G(2k+1)} = \prod_{n=0}^{k-1} \frac{n!}{(n+k)!}$$

We also know for integer k that $M_N(2k)$ is a polynomial in N of degree k^2 . Finally, this ratio times $(k^2)!$ is an integer, a result first proved in [59].

Keating and Snaith wanted to establish the value distribution of the real and imaginary parts of the logarithm of $Z_N(\theta)$, which was part of the motivation for studying $M_N(2k)$. Note that this is precisely the generating function for the moments of the real part, and the imaginary parts may be computed in a similar manner. They showed [211] that as $N \to \infty$, that the real and imaginary parts tend independently to Gaussian random variables.

Theorem 1.7 (Keating–Snaith). Let $B \subset \mathbb{C}$. Then for fixed $\theta \in [0, 2\pi)$,

$$\lim_{N \to \infty} \text{Haar} \left\{ A \in U(N) : \frac{\log Z_N(\theta)}{\sqrt{\frac{1}{2} \log N}} \in B \right\} = \frac{1}{2\pi} \iint_B e^{-\frac{1}{2}(x^2 + y^2)} dx dy,$$

where the measure of the set is taken to be the usual Haar measure on U(N).

This should be compared with the Selberg Central Limit Theorem [278] for $\zeta(s)$, which shows that the real and imaginary parts of the logarithm of $\zeta(s)$ independently tend to a Gaussian random variable.

Theorem 1.8 (Selberg). Let $B \subset \mathbb{C}$,

$$\lim_{T \to \infty} \frac{1}{T} \max\left\{ T \le t \le 2T \ : \ \frac{\log \zeta(\frac{1}{2} + it)}{\sqrt{\frac{1}{2} \log \log \frac{t}{2\pi}}} \in B \right\} = \frac{1}{2\pi} \iint_B e^{-\frac{1}{2}(x^2 + y^2)} \ dx \ dy,$$

where the measure of the set is taken to be Lebesgue measure.

The similarity between these two theorems is impossible to miss. Compare the mean densities of the zeta zeros at height T, which is $\frac{1}{2\pi} \log \frac{T}{2\pi}$, to the mean density of eigenangles of an $N \times N$ unitary matrix, which is $\frac{N}{2\pi}$. Setting them equal to each other gives

$$N = \log \frac{T}{2\pi}.$$

With this connection the two central limit theorems are consistent. This scaling has really paved the way for us to understand $\zeta(s)$ through RMT.

To test the predictions of RMT against the Riemann zeta function results, we show plots the real and imaginary parts of the two Central Limit Theorems, with the graphs found in [287, 211]. These graphs have been standardised so as to be compared with the Gaussian of mean zero and variance one. While neither graph matches the asymptotic standard Gaussian limit, the RMT results agree extremely closely to the number theoretic results convincingly well.



Figure 1.1: A comparison of the Circular Unitary Ensemble value distribution for $\Re(\log Z_N)$ with N = 42, Odlyzko's data for the distribution of $\Re(\log \zeta)$ near the $10^{20\text{th}}$ zero $(T \approx 10^{19})$ and the standard Gaussian, all scaled to have unit variance.

Given this identification, it should be simple to transform the moments $M_N(2k)$ into conjectural moments of $\zeta(s)$. This works for k = 0 and for k = 1, but we have a problem when we consider k = 2. In this case, one might think that we would have $\frac{1}{12}T(\log T)^4$ for the fourth moment, but this disagrees with Ingham's result (1.17). What has gone wrong? As we consider higher moments from this approach and compare them with their conjectural results, we see that the arithmetic term is missing. We say colloquially that RMT doesn't know about primes (for example, there is no Euler product equivalent in RMT). Instead, treating our quotient of Barnes G-functions as g_k in the folklore conjecture



Figure 1.2: A comparison of the Circular Unitary Ensemble value distribution for $\Im(\log Z_N)$ with N = 42, Odlyzko's data for the distribution of $\Im(\log \zeta)$ near the $10^{20\text{th}}$ zero $(T \approx 10^{19})$ and the standard Gaussian, all scaled to have unit variance.

(1.18), and using the identification $N = \log \frac{T}{2\pi}$ in their moments calculation, Keating and Snaith were led to the following conjecture, first given in [211].

Conjecture 1.9 (Keating–Snaith). For k fixed with $\Re(k) > -1/2$,

$$I_{2k}(T) = \int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} dt \sim a_k \frac{G^2(k+1)}{G(2k+1)} T(\log T)^{k^2}$$

as $T \to \infty$, with a_k defined in (1.19) and G is the Barnes G-function defined in (1.23).

We note that this conjecture is consistent with all known and conjectured results. Indeed, this result and the eighth moment conjecture were announced at the same conference in the 1998 Vienna conference on the Riemann Hypothesis - fortunately, the two conjectures agreed with one another! We also note that Keating and Snaith went further, and also formulated moments conjectures for other *L*-functions from RMT results over the Orthogonal and Symplectic groups in [210].

Two questions remain: how can one make the arithmetic factor a_k appear naturally, and what are the lower order terms implied in the Keating–Snaith conjecture?

1.3.4 The Hybrid Model

Keating and Snaith were able to obtain the geometric term g_k solely through RMT methods, but as discussed, the arithmetic term a_k doesn't appear. Conversely, we can obtain the arithmetic term a_k solely through relatively simple number theoretical methods, but then the geometric term g_k disappears. We want a model for the moments of $\zeta(s)$ that treats both the arithmetic and geometric terms equally. Rather than considering $\zeta(s)$ as just an Euler product over primes or just a Hadamard product over zeros, Gonek, Hughes and Keating [149] consider a hybrid model where they write $\zeta(s)$ as a product over primes and over zeros. They write

$$\zeta(s) = P_X(s)Z_X(s)(1+\mathcal{E})$$

where $P_X(s)$ is a product over primes given by

$$P_X(s) = \exp\left(\sum_{n \le X} \frac{\Lambda(n)}{\log n} \frac{1}{n^s}\right),$$

where $\Lambda(n)$ is the von Mangoldt function, $Z_X(s)$ is a product over zeros given by

$$Z_X(s) = \exp\left(-\sum_{\rho_n} U((s-\rho_n)\log X)\right),\,$$

where U(z) is given in (5.12), X is a mediating variable, and \mathcal{E} is an error term. Letting $X \to \infty$ gives the Euler product over primes only, and letting $X \to 0$ gives the Hadamard product over zeros only. A balanced value of X gives the hybrid approach.

Next they assume that the moments of $\zeta(s)$ can be written as moments of the primes part multiplied by moments of the zeros part, a result they call the Splitting Conjecture. Specifically, they conjecture that the following holds.

Conjecture 1.10 (Splitting Conjecture). Let $X, T \to \infty$ with $X = O((\log T)^{2-\varepsilon})$. Then for k > -1/2, we have

$$\frac{1}{T} \int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} dt \sim \left(\frac{1}{T} \int_0^T \left| P_X \left(\frac{1}{2} + it \right) \right|^{2k} dt \right) \left(\frac{1}{T} \int_0^T \left| Z_X \left(\frac{1}{2} + it \right) \right|^{2k} dt \right).$$

The conjecture was shown to hold for k = 1 and k = 2 to leading order. The behaviour of the moments of the prime part is calculated rigorously, with

$$\frac{1}{T} \int_0^T \left| P_X \left(\frac{1}{2} + it \right) \right|^{2k} dt \sim a_k (e^{\gamma_0} \log X)^{k^2}$$

as $X, T \to \infty$ where γ_0 is Euler's constant and a_k is given in (1.19). They also conjecture an asymptotic for the zeros part using RMT, showing

$$\frac{1}{T} \int_0^T \left| Z_X \left(\frac{1}{2} + it \right) \right|^{2k} dt \sim \frac{G^2(k+1)}{G(2k+1)} \left(\frac{\log T}{e^{\gamma_0} \log X} \right)^{k^2}$$

as $X, T \to \infty$.

Combining these terms and using the Splitting Conjecture recovers the Conjecture 1.9, with the arithmetic term now appearing naturally. Note that the hybrid model has been extended in various settings to other *L*-functions, including the derivative of the Riemann zeta function (see Chapter 5 and [187]). For a collection of the literature on the hybrid model, see [40].

1.3.4.1 Extreme Values of the Riemann zeta function

Recall the Lindelöf Hypothesis asserts that for t > 0, for every $\varepsilon > 0$, that $\zeta(1/2 + it) \ll t^{\varepsilon}$. With the assumption of the Riemann Hypothesis, it has been shown [49] that

$$\zeta\left(\frac{1}{2} + it\right) = O\left(\exp\left(\left(\frac{\log 2}{2} + o(1)\right)\frac{\log t}{\log\log t}\right)\right)$$

as $t \to \infty$.

The best known unconditional lower bound can be found in [94] and is given by

$$\max_{t \in [0,T]} \left| \zeta \left(\frac{1}{2} + it \right) \right| \ge \exp\left((\sqrt{2} + o(1)) \sqrt{\frac{\log T \log \log \log T}{\log \log T}} \right),$$

as $T \to \infty$.

The true order of $|\zeta(1/2 + it)|$ remains elusive. Using the hybrid model, Farmer, Gonek and Hughes [113] have conjectured the following result concerning the extreme values achieved by the Riemann zeta function on the critical line.

Conjecture 1.11 (Farmer–Gonek–Hughes).

$$\max_{t \in [0,T]} \left| \zeta \left(\frac{1}{2} + it \right) \right| = \exp\left(\left(\frac{1}{\sqrt{2}} + o(1) \right) \sqrt{\log T \log \log T} \right)$$

where the o(1) term tends to zero as $T \to \infty$.

Motivated by the goal of understanding the global maximum order of $|\zeta(1/2 + it)|$, Fyodorov, Hiary, and Keating [130, 131] considered the problem of the local maximum size of the zeta function in randomly chosen intervals of constant length on the critical line. They obtained a precise conjecture for the distribution of this maximum over short intervals. This conjecture has been supported by numerical data [129, 1], something that the conjectures for zeta over large intervals cannot claim due to the length of the interval being calculated over.

Conjecture 1.12 (Fyodorov–Hiary–Keating). If t is chosen uniformly from [T, 2T], then

$$\max_{|h| \le 1} \log \left| \zeta \left(\frac{1}{2} + it + ih \right) \right| = \log \log T - \frac{3}{4} \log \log \log T + X_T,$$

where the random variable X_T is O(1) in size and converges weakly, as $T \to \infty$, to a sum of two independent Gumbel random variables.

As is often the case with understanding the Riemann zeta function, the Fyodorov–Hiary– Keating conjecture was motivated by a connection with random matrices and log-correlated processes, with a summary found in [13]. Much progress has been made on the conjecture in recent years, with sharp upper and lower bounds from the zeta function side found in [6, 7], while the RMT side can be found in [261], confirming the conjecture for both number theory and RMT. These papers also form a good summary of the field, a discussion of which takes us too far from the main aim of this thesis.

1.3.5 Lower Order Terms for Moments

By being guided through analogous results in RMT for the classical compact groups [60], the paper of Conrey, Farmer, Keating, Rubinstein and Snaith [61] developed what is now known as the recipe, a method for determining full asymptotics for moments of $\zeta(s)$ and other *L*-functions for integer moments. They showed that

$$I_{2k}(T) = \int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} dt = \int_0^T P_k \left(\log \frac{t}{2\pi} \right) dt + O\left(T^{1/2 + \varepsilon} \right)$$

where P_k is a polynomial of degree k^2 given by a certain residue. We note that they conjecture the error term here to be the correct size for general k (see Section 1.3.2 for a discussion for k = 2, 4). However, for other *L*-functions it can be shown that the recipe misses main terms of the size $O(T^{1-\delta})$ for $0 < \delta < 1/2$, as shown in [318, 102, 104].

Conrey, Farmer, Keating, Rubinstein and Snaith [61] begin with integrals of the form

$$\int_0^\infty \left(\prod_{\alpha \in A} \zeta\left(\frac{1}{2} + it + \alpha\right)\right) \left(\prod_{\beta \in B} \zeta\left(\frac{1}{2} - it + \beta\right)\right) \psi\left(\frac{t}{T}\right) dt$$
(1.24)

where A, B are sets of size k of 'shifts' and ψ is a smooth function of compact support. Letting the shifts tend to zero recovers the $2k^{\text{th}}$ moments.

The recipe is exactly what it sounds like – a non-rigorous method to follow to calculate moments. The rough idea is as follows: we ignore error terms throughout, and write each $\zeta(s)$ term using the approximate functional equation,

$$\zeta(s) \sim \sum_{n \le \sqrt{t/2\pi}} \frac{1}{n^s} + \chi(s) \sum_{m \le \sqrt{t/2\pi}} \frac{1}{m^{1-s}}.$$
(1.25)

Now keep only the non-oscillatory terms, which is when we can pair off a $\chi(s)$ term with a $\chi(1-s)$ term. Note that

$$\chi(s+\alpha)\chi(1-s+\beta) \sim \left(\frac{t}{2\pi}\right)^{-\alpha-\beta}.$$
(1.26)

Then we keep only the diagonal terms, which are those where the imaginary parts disappear, ignoring all off-diagonal terms. In this way we recover the polynomial and can integrate it to give the full asymptotic result.

At the same time and independently, Diaconu, Goldfeld and Hoffstein [103] used a multiple Dirichlet series approach to obtain the same conjectures. We note that this approach does predict the missing terms from the recipe of size $O(T^{1-\delta})$ for $0 < \delta < 1/2$. All these approaches are consistent for the main terms which are of size T and larger.

1.3.6 RECENT DEVELOPMENTS

The development of these conjectures has come from a fruitful link between RMT and number theory. However, as we saw earlier for the first four moments, we can obtain the same results and conjectures from number theoretical approaches which fail for $k \geq 5$ as they predict negative values. The approach taken is to write $\zeta(s)$ as a Dirichlet polynomial and calculate moments of these.

In a series of five papers, Conrey and Keating [78, 79, 80, 81, 84] showed how to fix this approach by taking long Dirichlet polynomials, much longer than were previously considered to avoid missing any terms. They showed that (1.24) can be calculated either by multiple contour integrals as in the recipe, or by considering Dirichlet series

$$\prod_{\alpha \in A} \zeta\left(\frac{1}{2} + it + \alpha\right) = \sum_{n=1}^{\infty} \frac{\tau_A(n)}{n^s}$$

and considering divisor sum correlations, where the arithmetic function $\tau_A(n)$ is defined by the Euler product expansion of the left-hand side of the equation. For the first time, they united the RMT and number theoretic approaches and showed that either approach gives exactly the same result. They also showed that in a further two papers [82, 83], the Ratios Conjecture (as seen later in Section 6.1) can also be derived in this way.

While this approach is still conjectural, some progress has been made. For example, in their series of papers, Conrey and Keating showed that what we call the 1-swap terms for moments of $\zeta(s)$ are a consequence of formulas for correlations of divisor sums. Hamieh and Ng [161] have made this rigorous, under the assumption of the asymptotic formulae for correlations of divisor sums, which come from applying the delta method of Duke, Friedlander and Iwaniec [106] to the sum

$$\sum_{\substack{m,n=1\\m-n=h}}^{\infty} \tau_A(m) \tau_B(n) f(m,n)$$

for suitable f and ignoring the error terms.

Knowing the results and methods from this approach has proved to be incredibly fruitful. Ng [255], assuming a certain ternary divisor conjecture, has recovered the full asymptotic formula predicted by the recipe for the sixth moment. Specifically, he recovered the analogous shifted form for the sixth moment that Ingham did for the second moment in (1.15). Also, Ng, Shen and Wong [256], assuming a quaternary ternary divisor conjecture, have recovered the leading order term for the eighth moment. Their approach is yet to be extended to higher order moments. Additionally, the problem of negative continuous moments has come to the fore recently. For k > 0 and $\alpha > 0$, let

$$I_{-2k}(\alpha,T) = \int_0^T \left| \zeta \left(\frac{1}{2} + \alpha + it \right) \right|^{-2k} dt.$$

Gonek [145] made the following conjecture for various ranges of k and α .

Conjecture 1.13 (Gonek). Let k > 0. Uniformly for $\frac{1}{\log T} \leq \alpha \leq 1$, we have

$$I_{-2k}(\alpha, T) \asymp T\left(\frac{1}{\alpha}\right)^k$$

and uniformly for $0 < \alpha \leq \frac{1}{\log T}$, we have

$$I_{-2k}(\alpha, T) \asymp \begin{cases} T(\log T)^{k^2} & \text{if } k < 1/2\\ T\log \frac{e}{\alpha \log T} (\log T)^{k^2} & \text{if } k = 1/2\\ T(\alpha \log T)^{1-2k} (\log T)^{k^2} & \text{if } k > 1/2 \end{cases}$$

More recent RMT computations due to Berry and Keating [23], and due to Forrester and Keating [120] suggest that certain corrections to Gonek's conjecture is due in some ranges. Namely, when $\alpha \leq 1/\log T$, RMT computations contradict Gonek's prediction for the negative moments when $k \geq 3/2$. In particular, the RMT predictions suggest certain transition regimes when k is half an odd positive integer. Gonek's conjecture already captures the first transition at k = 1/2 featuring a logarithmic correction, and in this case the two conjectures do agree. Interpreted correctly, the following conjectures have been made on the number theory side, based on the aforementioned RMT calculations.

Conjecture 1.14 (Berry–Keating, Forrester–Keating, Bui–Florea). Let k > 0. Uniformly for $\frac{1}{\log T} \leq \alpha \leq 1$, we have

$$I_{-2k}(\alpha,T) \sim a(k)T\left(\frac{1}{\alpha}\right)^{k^2}$$

and uniformly for $0 < \alpha \leq \frac{1}{\log T}$, we have

$$I_{-2k}(\alpha, T) \sim \begin{cases} a(k)T(\log T)^{k^2}(\alpha \log T)^{-j(2k-j)} & \text{if } j - 1/2 < k < j + 1/2 \\ a(k)T\log \frac{e}{\alpha \log T}(\log T)^{k^2}(\alpha \log T)^{-j(j-1)} & \text{if } k = j - 1/2 \text{ for } j \ge 1 \end{cases}$$

and where

$$a(k) = \prod_{p} \left(1 - \frac{1}{p^{1+2\alpha}} \right)^{k^2} \left(1 + \sum_{j=1}^{\infty} \frac{\mu_k(p^j)^2}{p^{(1+2\alpha)j}} \right),$$

where $\mu_k(n)$ denotes the n^{th} coefficient of the Dirichlet series for $\zeta(s)^{-k}$.

Various progress on the zeta function side has been made by Bui and Florea [43]. Additionally, results on negative moments of quadratic Dirichlet L-functions have been proved in the function field setting in [44, 116]. Problems relating to negative continuous moments are of interest due to their relation to the Ratios Conjecture, described in detail in Chapter 6.

1.3.7 Applications of Moments

1.3.7.1 Sub-convexity of $\zeta(s)$

We have already discussed the initial motivation for studying moments, that of proving the Lindelöf Hypothesis. As always, $\varepsilon > 0$ isn't necessarily the same throughout. From the approximate functional equation (1.25), we obtain the convexity bound

$$\left|\zeta\left(\frac{1}{2}+it\right)\right|\ll(1+|t|)^{1/4+\varepsilon}.$$

Going beyond the convexity bound involves showing cancellation in the exponential sums in the approximate functional equation (1.25). This is an active area of research, starting with Weyl [310], and Hardy and Littlewood (unpublished) who showed that

$$\left|\zeta\left(\frac{1}{2}+it\right)\right| \ll (1+|t|)^{1/6+\varepsilon}.$$

The exponent 1/6 is a consequence of Weyl's differencing method for estimating exponential sums, first introduced in 1916. The best known result is due to Bourgain [37] who showed that

$$\left|\zeta\left(\frac{1}{2}+it\right)\right| \ll (1+|t|)^{13/84+\varepsilon}$$

Sharp moment estimates encode Lindelöf on average, and in some cases, can also yield pointwise sub-convexity estimates. Moments encode this information. For example Ingham's fourth moment (1.17) yields the convexity bound above, while Heath-Brown's result [171]

$$\int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{12} dt \ll T^{2+\varepsilon}$$

implies (and is stronger than) the sub-convexity bound of Weyl, Hardy and Littlewood.

1.3.7.2 Twisted and Mollified Moments

Another type of moment question is that of twisted, or mollified moments, which are integrals of the form

$$\int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2k} \left| M_N \left(\frac{1}{2} + it \right) \right|^2 dt$$

where

$$M_N(s) = \sum_{n=1}^N \frac{a(n)}{n^s}$$

and the coefficients satisfy $a(n) \ll n^{\varepsilon}$ for all $\varepsilon > 0$.

Bohr and Landau [33] were the first to consider such questions, with the aim to calculate a zero-density estimate for zeta, described in Section 1.3.7.4. They used a mollified moment, one where the coefficients in $M_N(s)$ are given by $a(n) = \mu(n)$ where $\mu(n)$ is the Möbius function, defined by

$$\mu(n) = \begin{cases} 1 & \text{if } n = 1\\ (-1)^k & \text{if } n = p_1 \dots p_k \text{ with the } p_j \text{ distinct primes} \\ 0 & \text{if } n \text{ is not square-free.} \end{cases}$$
(1.27)

This sum $M_N(s)$ is essentially an approximation to $1/\zeta(s)$ for $\sigma > 1$. If this approximation is also valid in the critical strip, then $M_N(s)$ should dampen large values of $\zeta(s)$.

The twisted second moment was calculated by Balasubramanian, Conrey and Heath-Brown [15] for mollifiers of length $N = T^{1/2-\varepsilon}$ and where $a(n) \ll n^{\varepsilon}$.

Bettin, Chandee and Radziwiłł [28] calculated the twisted second moment for mollifiers of length $N = T^{1/2+\delta}$, where $\delta = 0.01515...$ and where $a(n) \ll n^{\varepsilon}$.

Setting k = 1 and $a(n) = \mu(n)P(\log(n/y)/\log y)$, where P(x) is a polynomial satisfying P(0) = 0, P(1) = 1 that is optimised through the calculus of variations, we have a mollified second moment which was considered by Levinson [222] when he showed that one-third of the zeros of $\zeta(s)$ lie on the critical line. Conrey [56] increased this length to $N = T^{4/7-\varepsilon}$ and proved that 40% of the zeros of $\zeta(s)$ lie on the critical line, a fact we discuss further in Section 1.3.7.3.

Conrey, Ghosh and Gonek [74] considered the problem of twisted second moments in a more general setting. To describe their result, we set up some notation. Let T be a large positive real number and let

$$B(s,P) = \sum_{n \le y} \frac{\mu(n) P\left(\frac{\log y/n}{\log y}\right)}{n^s}$$

where $y = T^{\theta}$ for $\theta < 1/2$, P(x) is an entire function with P(0) = 0, and $\mu(n)$ is the Möbius function (1.27). Let a, b be complex numbers and $Q_1(x)$ and $Q_2(x)$ be polynomials.

They claim that for fixed c with $1/2 \le c < 3/2 - \theta$, and $a, b \ll 1$, then as $T \to \infty$,

$$\frac{1}{i} \int_{c+i}^{c+iT} Q_1\left(-\frac{d}{da}\right) \zeta\left(s + \frac{a}{\log T}\right) Q_2\left(-\frac{d}{db}\right) \zeta\left(1 - s + \frac{b}{\log T}\right) B(s, P_1) B(1 - s, P_2) ds$$
$$\sim T \left\{ Q_1(0)Q_2(0)P_1(1)P_2(1) + \frac{\partial}{\partial u} \frac{\partial}{\partial v} \frac{1}{\theta} \left(\int_0^1 T_a Q_1 T_b Q_2 dx\right) \left(\int_0^1 P_1 P_2 dx\right) \Big|_{u=v=0} \right\},$$

where $P_1 = P_1(x+u), P_2 = P_2(x+v),$

$$T_aQ_1 = e^{-a(x+\theta u)}Q_1(x+\theta u)$$
 and $T_bQ_2 = e^{-b(x+\theta v)}Q_2(x+\theta v)$.

Then several results follow easily from here. For example, it is easy to see that this reduces down to the twisted second moment, and that

$$\int_{1}^{T} \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2} \left| B \left(\frac{1}{2} + it, P \right) \right|^{2} dt \sim T \left(P(1)^{2} + \frac{1}{\theta} \int_{0}^{1} P'(x)^{2} dx \right).$$

Also, if we let $P_1(x) = P_2(x) = x$, $Q_1(x) = Q_2(x) = -1 - x$, c = 1/2, a = -b = -1.3 and let $\theta \to 1/2^-$, then Levinson's [222] result that at least one-third of the zeros of the zeta function lie on the critical line. Various zero density results follow, for example those found in [276, 278, 204].

For the twisted fourth moment, Deshouillers and Iwaniec [101] were able to obtain upper bounds when the Dirichlet polynomial has length $N = T^{1/5-\varepsilon}$. This was later improved by Watt [307] for length $N = T^{1/4-\varepsilon}$. Gaggero Jara [203] was able to obtain a leading order result for the mollified fourth moment for length $N = T^{4/589-\varepsilon}$ but wasn't able to obtain lower order terms. Independently, Motohashi [248], and Hughes and Young [188] were able to obtain full asymptotics, with the later doing so for length $N = T^{1/1-\varepsilon}$. Finally, Bettin, Bui, Li and Radziwiłł [26] have extended this for length $N = T^{1/4-\varepsilon}$.

As hinted to above, these mollified moments have applications to proportions of zeros on the critical line, which we discuss further in Section 1.3.7.3, as well as to gaps between zeta zeros discussed in Section 1.4.3.3, and for lower bounds of moments discussed in Section 1.3.2.1.

1.3.7.3 Zeros on the critical line

As alluded to in Section 1.3.7.2, one use of these mollified moments is to calculate proportions of zeros on the critical line. We already gave a brief history of calculating proportions of zeros on the critical line when we discussed the Riemann Hypothesis in Section 1.1.3.1, but give a more detailed history here.

We write $N_0(T)$ for the number of zeros of $\zeta(s)$ on the critical line up to height T above the real axis, and let N(T) be as in (1.4). We let

$$\kappa = \liminf_{T \to \infty} \frac{N_0(T)}{N(T)}$$

denote the proportion of non-trivial zeros of $\zeta(s)$ that lie on the critical line.

Selberg [276] first implicitly showed that a positive proportion of the non-trivial zeros of the zeta function are on the critical line. He did this by introducing a mollifier to a method of Hardy and Littlewood [165] for detecting zeros of $\zeta(s)$. Min [238] later made this proportion explicit.

Since first introducing the mollifier method, a great deal of progress has been made on the proportion of zeros on the critical line. They are based on a different approach to the one taken by Selberg, and started when Levinson [222] introduced the mollifier method to an approach of Siegel's [282] in detecting zeros of $\zeta(s)$. Levinson showed that $\kappa \geq 0.3474$. while Conrey [56] improved this to $\kappa \geq 0.4088$. Further improvements have since been made, for example by Bui, Conrey, and Young [42] who showed $\kappa \geq 0.4105$. Feng [115] gave a further improvement, although it has been suggested that there is an error in this work. The value that is claimed to follow from this work is $\kappa \geq 0.4107$. Bui [41] pushed this further to show $\kappa \ge 0.4109$. The current record is due to Pratt, Robles, Zaharescu and Zeindler [265], who showed that $\kappa \geq 0.4172$, or rather, more than over five-twelfths of the non-trivial zeros of the Riemann zeta function lie on the critical line. Wu [315] independently showed this same result for Dirichlet L-functions as a side calculation when calculating a twisted second moment of a Dirichlet L-function, and showed agreement with the result of Pratt, Robles, Zaharescu and Zeindler. Note that these methods give as a free corollary various improvements in the proportions of simple zeros, as discussed in Section 1.4.3.1.

A third method has been explored by Baluyot [17], where he combines the mollifier method with a zero detecting method of Atkinson's [11] to show in a new way that the proportion κ is positive. This relies on the twisted fourth moment of Hughes and Young [188], discussed in the previous section. He uses this method to show that $\kappa > 0.0001049$. While this is obviously a smaller proportion than what was previously known, it does open up a new way to find these proportions.

The method we focus on here is Levinson's method, as he was the first to exploit it to show that one-third of the zeros of $\zeta(s)$ lie on the critical line. Levinson's original proof runs for over fifty pages, with the main difficulty being a certain moments result. Knowing how to calculate mollifiers using the Ratios Conjecture, discussed in Section 6.1.3, Young [316] was able to shorten this argument down to just eight pages. We give a rough outline of the steps used by Young here.

The first result to note is known as Littlewood's Lemma [227], which is an analogous result to Jenson's formula, a result that relates the average size of a function to its zeros. Littlewood's Lemma is the result for rectangles which is generally more useful when working with Dirichlet series. We state a version that ignores some technicalities as we aren't giving full proofs when describing these applications.

Lemma 1.15 (Littlewood). Let f be analytic and non-zero on the rectangular positively oriented contour C with vertices $\sigma_1, \sigma_1 + iT, \sigma_0 + iT, \sigma_0$, where $\sigma_0 < \sigma_1$. Then

$$\sum_{\rho \in C} \text{Dist}(\rho) = \frac{1}{2\pi} \int_0^T \log |f(\sigma_0 + it)| \, dt + \mathcal{E}$$

where $\text{Dist}(\rho)$ is the distance of the zero ρ to the line $\Re(s) = \sigma_0$ and \mathcal{E} is an error term that can be ignored and might be different on different occasions.

The second result to note is Speiser's Theorem [292], which provides an equivalent form of the Riemann Hypothesis.

Theorem 1.16 (Speiser). The Riemann Hypothesis is equivalent to the assertion that the derivative of $\zeta(s)$ has no zeros to the left of the critical line inside the critical strip.

We define $N_L(T)$ to be the number of zeros of $\zeta(s)$ to the left of the critical line inside the critical strip up to height T above the real axis and $N'_L(T)$ to be the number of zeros of $\zeta'(s)$ to the left of the critical line inside the critical strip up to height T above the real axis. Then Levinson and Montgomery [223] made Speiser's Theorem quantitative by showing that

$$N_L(T) = N'_L(T) + O(\log T).$$

Recall that $N_0(T)$ is the number of zeros of $\zeta(s)$ on the critical line up to height T above the real axis. Then we know from previous discussions that the number of zeros to the right of the critical line inside the critical strip is equal to the number of zeros to the left of the critical line inside the critical strip. Therefore,

$$N(T) = N_0(T) + 2N_L(T)$$

= $N_0(T) + 2N'_L(T) + O(\log T).$

Rearranging,

$$N_0(T) = N(T) - 2N'_L(T) + O(\log T)$$

so if we can find a small upper bound for $N'_L(T)$, then we will have a lower bound for $N_0(T)$ as we know what N(T) equals already by (1.5).

Due to technical reasons, we don't count zeros of $\zeta'(1-s)$, and instead count zeros of the function V(s), where

$$V(s) = Q\left(-\frac{1}{L}\frac{d}{ds}\right)\zeta(s)$$

and Q(x) is a real polynomial satisfying Q(0) = 1, with $L = \log T$. Levinson [222] originally takes Q(x) = 1 - x, but Conrey [56] showed that more general choices of Q(x) can be used to improve results.

 Set

$$\sigma_0 = \frac{1}{2} - \frac{R}{L}$$

where R > 0 is real number chosen later, and let $N = T^{\theta}$ where $0 < \theta < 1/2$.

Suppose $\psi(s)$ is a mollifier of the form

$$\psi(s) = \sum_{n \le N} \frac{\mu(n)}{n^{s+1/2-\sigma_0}} P\left(\frac{\log N/n}{\log N}\right)$$

where P(x) is a real polynomial with P(0) = 0 and P(1) = 1. Levinson originally takes P(x) = x but again this can be optimised in more general cases.

Then one can show

$$N_L'(T) \le \frac{T}{4\pi} \log\left(\frac{1}{T} \int_0^T |V\psi(\sigma_0 + it)|^2 dt\right) + \mathcal{E}$$

and evaluating this integral is where the main difficulty lies.

Young showed a simpler way to evaluate this moment, which we summarise as the following theorem.

Theorem 1.17 (Young). We have

$$\frac{1}{T} \int_0^T |V\psi(\sigma_0 + it)|^2 dt = c(P, Q, R, \theta) + \mathcal{E}$$

as $T \to \infty$, where

$$c(P,Q,R,\theta) = 1 + \frac{1}{\theta} \int_0^1 \int_0^1 e^{2Rv} \left(\frac{d}{dx} e^{R\theta x} P(x+u)Q(v+\theta x) \Big|_{x=0} \right)^2 \ dudv.$$

With Levinson's original choices of P(x) = x, Q(x) = 1 - x, R = 1.3, $\theta = 1/2$, we have, $c(P, Q, R, \theta) \approx 2.35$. Then we have

$$\kappa \ge 1 - \frac{1}{R} \log\left(\frac{1}{T} \int_0^T |V\psi(\sigma_0 + it)|^2 dt\right) + o(1).$$

and using Levinson's original inputs gives $\kappa \geq 0.34$.

Farmer [112] conjectures that we can set $\theta = \infty$ for mollifiers. This means that the mollifier can be made arbitrarily long, and so turns the mollifier into $1/\zeta(s)$. This prompted him to first consider what has become known as the Ratio's Conjecture, described in Chapter 6. This together with Theorem 1.17 implies that we have 100% of the zeros lie on the critical line. This conjecture is of particular interest as Bettin and Gonek [29] have shown that the $\theta = \infty$ conjecture implies the Riemann Hypothesis which is stronger than showing 100% of the zeros lie on the critical line. Farmer [112] also argues that a strong form of this conjecture implies Montgomery's Pair Correlation conjecture. Additionally, Radziwiłł [266] showed that as $\theta \to \infty$, $\psi(s)$ is essentially the best possible mollifier of length T^{θ} for $\zeta(s)$. In particular his work implies that Levinson's method can give $\kappa = 1$ only if it is used with mollifiers of length T^{θ} when θ is arbitrarily large.

1.3.7.4 Zero-density estimates

One can show that there are relatively few zeros in the region $1/2 < \sigma_0 < 1$ (and so few in the region $0 < \sigma_0 < 1/2$ by the functional equation).

These results are typically of the form

$$N(\sigma, T) \ll T^{A(\sigma)(1-\sigma)+\varepsilon}$$

as $T \to \infty$, where

$$N(\sigma, T) = \#\{\rho = \beta + i\gamma : \sigma \le \beta \le 1, 0 < \gamma < T\}$$

counts the number of zeros of $\zeta(s)$ with real part at least σ and imaginary part at most T. Note that the $\varepsilon > 0$ in the above equation for $N(\sigma, T)$ can be made (in theory) explicit in this section, replacing it with powers of logarithms.

The Density Hypothesis asserts that

$$N(\sigma, T) \ll T^{2(1-\sigma)+\varepsilon}$$

for all $\varepsilon > 0$, that is, $A(\sigma) = 2$.

Clearly the Riemann Hypothesis is equivalent to $N(\sigma, T) = 0$ for all $\sigma > 1/2$ and all T > 0. It is known [193] that the Lindelöf Hypothesis implies the Density Hypothesis. We also know thanks to the proof of the Prime Number Theorem and the fact there are no zeros on the line $\Re(s) = 1$ (see Section 1.1.4) that N(1,T) = 0, so A(1) = 0.

Ingham [194] proved the uniform bound for $N(\sigma, T)$ with

$$A(\sigma) = \frac{3}{2 - \sigma}$$

in the range $1/2 \le \sigma \le 3/4$, using the Montgomery–Vaughan Mean-Value Theorem [243]. Huxley [189] proved the uniform bound $N(\sigma, T)$ with

$$A(\sigma) = \frac{3}{3\sigma - 1}$$

in the range $3/4 \le \sigma \le 1$ using the Montgomery–Halász large value estimate [242] and [156].

Combined, these values of $A(\sigma)$ show

$$N(\sigma, T) \ll T^{\frac{12}{5}(1-\sigma)+\varepsilon}.$$

Together, these bounds show results on prime numbers, in particular,

$$p_{n+1} - p_n \ll p_n^{7/12 + \varepsilon}$$

where p_n denotes the n^{th} prime number. Note that the Density Hypothesis implies

$$p_{n+1} - p_n \ll p_n^{1/2 + \varepsilon}$$

while the Riemann Hypothesis implies

$$p_{n+1} - p_n \ll p_n^{1/2} \log p_n,$$

neither of which are strong enough to prove the conjecture that between any two square numbers there is always a prime number.

Guth and Maynaard [153] improved the Huxley bound above to show

$$N(\sigma, T) \ll T^{\frac{30}{13}(1-\sigma)+\varepsilon}$$

in the full range, showing

$$p_{n+1} - p_n \ll p_n^{17/30 + \varepsilon}.$$

For the summary of the intervening history between the combined result of Montgomery and Huxley, and of Montgomery and Guth–Maynaard, see Table 2 in Trudgian and Yang [300].

To see the link between these results and continuous moments, we sketch the argument of the first zero-density estimate of Bohr and Landau [33], following the sketch given in Gonek [148].

Let C be a positively oriented rectangular contour enclosing the region $1/2 \leq \Re(s) < 1$ with left side $\Re(s) = \sigma_0$ up to a height T. By Littlewood's Lemma 1.15 we have

$$\sum_{\rho \in C} \text{Dist}(\rho) = \frac{1}{2\pi} \int_0^T \log |f(\sigma_0 + it)| \, dt + \mathcal{E}$$

where $\text{Dist}(\rho)$ is the distance of the zero ρ to the line $\Re(s) = \sigma_0$.

Then for σ a fixed real number with $\sigma_0 < \sigma < 1$, for non-trivial zeros $\rho = \beta + i\gamma$ of $\zeta(s)$, we have

$$\sum_{\rho \in C} \operatorname{Dist}(\rho) \ge \sum_{\substack{\rho \in C \\ \sigma < \beta}} \operatorname{Dist}(\rho) \ge (\sigma - \sigma_0) N(\sigma, T).$$

Typically it is difficult to calculate the integral of the logarithm directly, so instead use the arithmetic-geometric mean to write

$$\frac{1}{2\pi} \int_0^T \log|f(\sigma_0 + it)| \ dt = \frac{1}{4\pi} \int_0^T \log(|f(\sigma_0 + it)|^2) \ dt \le \frac{T}{4\pi} \log\left(\frac{1}{T} \int_0^T |f(\sigma_0 + it)|^2 \ dt\right)$$

It is known (Theorem 7.2 of [298]) that

$$\int_0^T |\zeta(\sigma_0 + it)|^2 dt \sim \zeta(2\sigma_0)T$$

so this last inequality is O(T). Therefore,

$$N(\sigma, T) \ll \frac{T}{\sigma - \sigma_0}$$

and since $N(T) \sim T \log T$, we have

$$\frac{N(\sigma, T)}{N(T)} = O\left(\frac{1}{\log T}\right)$$

for any fixed $\sigma > 1/2$. This means that the proportion of zeros to the right of the critical line is infinitesimally small.

Note that if we had included mollifiers in this argument, then we would have obtained better zero-density estimates. As seen above, we now have much better zero-density estimates.

1.3.8 JOINT MOMENTS OF THE RIEMANN ZETA FUNCTION AND JOINT MOMENTS OF THE HARDY Z-FUNCTION

Hall [158] conjectured that the joint moments of the Riemann zeta function are of the form

$$\int_0^T \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2s - 2h} \left| \zeta' \left(\frac{1}{2} + it \right) \right|^{2h} dt \sim C(s, h) T(\log T)^{s^2 + 2h}$$

These joint moments can also be written in terms of the Hardy Z-function (see Section 1.3.8.1) as

$$\int_0^T |Z(t)|^{2s-2h} |Z'(t)|^{2h} dt \sim \tilde{C}(s,h) T(\log T)^{s^2+2h},$$

where the coefficients C(s, h) can be written in terms of $\tilde{C}(s, h)$ and vice versa, as shown in [181].

For the joint moments of the Riemann zeta function, we already know due to Hardy and Littlewood [164] that C(1,0) = 1, and to Ingham [192] that C(1,1) = 1/3. Also, Ingham [192] showed $C(2,0) = 1/2\pi^2$ and Conrey [55] (and independently Hall [157]) showed that $C(2,1) = 2/15\pi^2$ and $C(2,2) = 61/1680\pi^2$. The analogous results for the Hardy Z-function joint moments can be deduced from these, with $\tilde{C}(1,1) = 1/12$, $\tilde{C}(2,1) = 1/120\pi^2$ and $\tilde{C}(2,2) = 1/1120\pi^2$. Conrey and Ghosh [66, 67] have shown that $\tilde{C}(1,1/2) = (e^2 - 5)/4\pi$ but C(1,1/2) is currently unproven. We come back to this result of Conrey and Ghosh in Section 1.3.8.3 and in Chapter 7. As with the usual continuous moments, RMT has since been used to predict higher moments, which we describe to in Section 1.3.8.2.

We now take this opportunity to discuss the Hardy Z-function, a function that acts like the Riemann zeta function on the critical line, as well as some of its properties and various related points of interest, before giving some background on the RMT approach to joint moments.

1.3.8.1 The Hardy Z-function

The Riemann—Siegel theta function is defined as

$$e^{-2i\theta(t)} = \chi\left(\frac{1}{2} + it\right),\tag{1.28}$$

where $\chi(s)$ is the factor from the functional equation of $\zeta(s)$ stated in (1.1). The Hardy Z-function can be defined in terms of the Riemann—Siegel theta function, where

$$Z(t) = e^{i\theta(t)}\zeta\left(\frac{1}{2} + it\right).$$

This means we can write

$$Z(t) = \chi \left(\frac{1}{2} + it\right)^{-1/2} \zeta \left(\frac{1}{2} + it\right).$$

Note that Z(t) is real for real t, Z(-t) = Z(t), and satisfies $|Z(t)| = |\zeta(1/2 + it)|$.



Figure 1.3: A plot of Z(t) (dark blue) and $|\zeta(1/2 + it)|$ (dashed) for $0 \le t \le 50$.

From these observations, we know that the zeros of $\zeta(s)$ on the critical line correspond to the real zeros of Z(t) which makes Z(t) an invaluable tool in the study of the zeros of the zeta function on the critical line. For example, Hardy used his Z-function to give the first proof that there are infinitely many zeros of the Riemann zeta function on the critical line [199].

The Riemann–Siegel formula allows us to write

$$Z(t) = 2\sum_{n=1}^{N} n^{-1/2} \cos(\theta(t) - t \log n) + R$$

where $N = \lfloor (t/2\pi)^{1/2} \rfloor$ and R is a certain error term. This is a highly efficient way of calculating values of the Riemann zeta function, with increasing accuracy as t increases.

The Riemann–Siegel formula was derived and used by Riemann, but never published by him. It was not until 1932 that Carl Siegel uncovered the formula following a study of Riemann's unpublished notes. The method of choice before the rediscovery of the Riemann–Siegel formula to calculate values of zeta was Euler–Maclaurin summation, which requires O(t) steps to evaluate zeta on the half-line. The advantage to the Riemann–Siegel formula is that it requires only $O(t^{1/2})$ steps. This method has been used to verify the correctness of the Riemann Hypothesis up to very large values of t, with modern variations giving improvements to the calculations.

Turing [301] developed a method which allows one to calculate the number of zeros in the critical strip up to a height T. Once all zeros on the critical line up to height Thave been counted using the Riemann–Siegel formula, Turing's method is then used to verify that the zeros found are indeed the only zeros in this part of the critical strip. In fact, Turing was trying to use this method to show that the Riemann Hypothesis is false.

We return to the observation that the zeros of $\zeta(s)$ on the critical line correspond to the real zeros of Z(t). We note that for $\gamma \leq \gamma^+$ consecutive positive ordinates of non-trivial zeros of $\zeta(s)$, there is exactly one zero of Z'(t) (under the Riemann Hypothesis), call it λ , with $\gamma \leq \lambda \leq \gamma^+$. A proof of this result can be found in [199]. These points λ will be essential in our results given in Chapter 7, where we sum over these points, and this notation provides an unambiguous way of writing the results discussed there. Note that there are two zeros of Z'(t) between t = 0 and the first positive zero of Z(t). These additional zeros typically only introduce small errors and are generally ignored.

1.3.8.2 Joint Moments through RMT

We note that the background in this section has appeared in a paper [183]. Notation and cross-references have been updated for this thesis.

Following the groundbreaking work of Keating and Snaith [211, 210], the problem of calculating joint moments of the Riemann zeta function and its derivative through RMT was taken up by Hughes [181] in his thesis. The approaches taken involved calculating analogous moments of characteristic polynomials of random unitary matrices.

Hughes [181] considered the joint moment for the Riemann zeta function and conjectured the coefficient $F_{\zeta}(s,h)$ for integer s, h in terms of combinatorial sums, where $F_{\zeta}(s,h)$ is given in

$$\int_{0}^{T} \left| \zeta \left(\frac{1}{2} + it \right) \right|^{2s-2h} \left| \zeta' \left(\frac{1}{2} + it \right) \right|^{2h} dt \sim a_s F_{\zeta}(s,h) T(\log T)^{s^2 + 2h}$$

as $T \to \infty$, where a_s is defined in (1.19).

Similarly, Hughes predicted that

$$\int_0^T |Z(t)|^{2s-2h} |Z'(t)|^{2h} dt \sim a_s F_Z(s,h) T(\log T)^{s^2+2h}$$

as $T \to \infty$, where a_s is defined in (1.19), together with a way of translating between $F_{\zeta}(s,h)$ and $F_Z(s,h)$.

The conjectures agreed with all values known at the time as described in Section 1.3.8. The expressions for $F_{\zeta}(s,h)$ and $F_Z(s,h)$ made sense for real s as well, but not general real h. We focus on the Hardy Z-function joint moment in this thesis, but note that many of the following results also hold for $F_{\zeta}(s,h)$.

Alternative approaches yielding different representations of the conjectured $F_Z(s, h)$ with $s, h \in \mathbb{N}$ were given by Conrey, Rubinstein and Snaith [85] (in the special case where s = h), by Dehaye [97, 98], by Basor et al. [20], and by Bailey et al. [12]. Links between F(s, h) and Painlevé equations have been noted by Forrester and Witte [121], and Basor et al. [20].

However, the problem of non-integer s and h is much more difficult. Winn [312] was able to establish results for $s \in \mathbb{N}$ and $h \in \frac{1}{2}\mathbb{N}$ by finding connections with hypergeometric functions, giving an expression for $F_Z(s, h)$ in terms of a combinatorial sum. In particular, for s = 1 and h = 1/2 he obtained the $(e^2 - 5)/4\pi$ coefficient for the analogous RMT result to the Hardy Z-function joint moment considered by Conrey and Ghosh [66, 67].

Assiotis, Keating and Warren [9] proved results on the random matrix side for arbitrary real values of s > -1/2 and positive real values of h in the full range 0 < h < s + 1/2. They also give a full history of these problems in their paper.

In addition to the work on the standard joint moment, Keating and Wei [212, 213] have considered the RMT analogue of joint moments for higher derivatives. The case of the joint moments of arbitrary numbers of higher order derivatives with arbitrary positive real exponents has been settled by Assiotis, Gunes, Keating and Wei [8].

Curran [92] unconditionally obtained upper bounds of the conjectured order for the joint moments in the range $1 \le s \le 2$ and $0 \le h \le 1$. Curran and Heycock [93] extend these upper bounds to all $0 \le h \le s \le 2$, and obtain lower bounds for all $0 \le h \le s + 1/2$. By assuming the Riemann Hypothesis, they give sharp bounds for all $0 \le h \le s$. They also prove upper bounds of the conjectured order for joint moments of the Riemann zeta function with its higher derivatives.

1.3.8.3 The Joint Moment of the Hardy Z-function and its derivative

For a long time, there was a problem that number theory could solve that RMT could not. We mentioned previously that Conrey and Ghosh [66, 67] showed $\tilde{C}(1/2, 1) = (e^2 - 5)/4\pi$, which means that they showed

$$\int_0^T |Z(t)| |Z'(t)| \ dt \sim \frac{e^2 - 5}{4\pi} T(\log T)^2.$$

However, while Hughes [181] predicted the general joint moments for even integer powers, his method didn't generalise to odd integer powers, nor did the work that followed on this problem. It would be over a decade before Winn [312] would develop the RMT tools to predict these results. We note that the corresponding problem with the Hardy Z-function replaced by the Riemann zeta function is still unproven from the number theory side.

To prove the joint moment result, Conrey and Ghosh [66] showed that under the Riemann Hypothesis,

$$\sum_{0 < \gamma \le T} \max_{\gamma \le t \le \gamma^+} \left| \zeta \left(\frac{1}{2} + it \right) \right|^2 \sim \frac{e^2 - 5}{4\pi} T (\log T)^2$$

as $T \to \infty$ where for each non-trivial zero $\rho = 1/2 + i\gamma$, let γ^+ denote the ordinate of the successive non-trivial zero of $\zeta(s)$.

The relation between this sum and the joint moment follows by recalling that under the Riemann Hypothesis, there is a unique point $\gamma \leq \lambda \leq \gamma^+$ between successive zeros $\gamma \leq \gamma^+$ that satisfies $Z'(\lambda) = 0$. In [67], they split the integral as

$$\int_{0}^{T} |Z(t)| |Z'(t)| dt = \int_{0}^{\gamma_{0}} |Z(t)| |Z'(t)| dt + \sum_{0 < \lambda \le T} \left(\int_{\gamma}^{\lambda} + \int_{\lambda}^{\gamma+} \right) |Z(t)| |Z'(t)| dt + \int_{\gamma_{T}}^{T} |Z(t)| |Z'(t)| dt,$$

where γ_0 denotes the least positive zero of Z(t) and γ_T denotes the least zero of Z(t) that is larger than T. Note that if $\gamma \leq t \leq \lambda$, $Z(t)Z'(t) \geq 0$ and if $\lambda \leq t \leq \gamma^+$, $Z(t)Z'(t) \leq 0$. Then the integral above becomes

$$\sum_{0<\lambda\leq T} \left(\frac{Z(t)^2}{2} \Big|_{\gamma}^{\lambda} - \frac{Z(t)^2}{2} \Big|_{\lambda}^{\gamma^+} \right) + O\left(T^{\varepsilon}\right) = \sum_{0<\lambda\leq T} |Z(\lambda)|^2 + O\left(T^{\varepsilon}\right)$$

since $Z(t)Z'(t) \ll t^{\varepsilon}$ and $|\gamma_T - T| \ll 1$.

For higher moments, Conrey [55] established under the Riemann Hypothesis that

$$\frac{\sqrt{21}}{90\pi^2}T(\log T)^5 \lesssim \sum_{0<\gamma \le T} \max_{\gamma \le t \le \gamma^+} \left| \zeta\left(\frac{1}{2} + it\right) \right|^4 \lesssim \frac{1}{2\sqrt{15}\pi^2}T(\log T)^5,$$

where $A \leq B$ means $A \leq B(1+o(1))$, with some improvements on the upper bound due to Hall [159, 160]. By a similar argument to the above relating sums to integrals, this implies

$$\frac{\sqrt{21}}{180\pi^2} T(\log T)^5 \lesssim \int_0^T |Z(t)|^3 |Z'(t)| \ dt \lesssim \frac{1}{4\sqrt{15}\pi^2} T(\log T)^5.$$

Although no other asymptotics for higher moments are known, Milinovich [235] established sharp (up to an epsilon) upper and lower bounds. He showed that for $k \in \mathbb{N}$ and for $\varepsilon > 0$,

$$T(\log T)^{k^2+1-\varepsilon} \ll \sum_{0 < \gamma \le T} \max_{\gamma < t \le \gamma^+} |Z(t)|^{2k} \ll T(\log T)^{k^2+1+\varepsilon}$$

for sufficiently large T. By a similar argument to that done by Conrey and Ghosh [67], Milinovich [235] shows

$$\sum_{0 < \gamma \le T} \max_{\gamma < t \le \gamma^+} |Z(t)|^{2k} = k \int_0^T |Z(t)|^{2k-1} |Z'(t)| \ dt + O_{k,\varepsilon}(T^{\varepsilon}).$$

Finally, he conjectures for $k \in \mathbb{N}$ that

$$\sum_{0 < \gamma \le T} \max_{\gamma < t \le \gamma^+} |Z(t)|^{2k} \sim C_k T (\log T)^{k^2 + 1}$$

as $T \to \infty$, but did not conjecture the value of the implied constant. As previously mentioned, this constant has since been conjectured from the RMT side by [312].

1.4 DISCRETE MOMENTS OF THE RIEMANN ZETA FUNCTION

Having seen all about continuous moments in Section 1.3, we now describe the history relating to discrete moments, which we recall are sums of the form

$$J_{2k,n}(T) = \sum_{0 < \gamma \le T} \left| \zeta^{(n)}(\rho) \right|^{2\ell}$$

where the γ are the ordinates of the non-trivial zeros $\rho = \beta + i\gamma$ of $\zeta(s)$ and $\zeta^{(n)}(s)$ denotes the n^{th} derivative of $\zeta(s)$. This is defined for all $k \ge 0$ and if we assume all the zeros of $\zeta(s)$ are simple, for k < 0. (When we just discuss the moments of the first derivative we will write $J_{2k}(T)$ for simplicity.)

We will assume the Riemann Hypothesis in this section for simplicity unless otherwise stated, although some results are unconditional. The reason we assume this is that we often consider sums of the form

$$\sum_{0 < \gamma \le T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right)^k \zeta^{(n)} \left(\frac{1}{2} - i\gamma\right)^k$$

which is exactly $J_{2k,n}(T)$ under the Riemann Hypothesis.

1.4.1 DISCRETE MOMENT RESULTS

We have already seen a result on moments of derivatives of $\zeta(s)$, in Ingham's result (1.16) when proving the lower order terms for the second continuous moment. In the case n = mhe showed that

$$\int_{0}^{T} \left| \zeta^{(n)} \left(\frac{1}{2} + it \right) \right|^{2} dt \sim \frac{1}{2n+1} T(\log T)^{2n+1}$$
(1.29)

as $T \to \infty$.

Gonek initiated the study of the discrete moments in [143] by considering the analogous result for sums of the derivatives of $\zeta(s)$. In particular, he showed unconditionally for any real $|\alpha| \leq L/2$, where $L = \frac{1}{2\pi} \log \frac{T}{2\pi}$, that

$$\sum_{0 < \gamma \le T} \zeta^{(n)} \left(\rho + i \frac{\alpha}{L} \right) \zeta^{(m)} \left(1 - \rho + i \frac{\alpha}{L} \right)$$

= $(-1)^{n+m} \left(\frac{1}{n+m+1} - H(n,m,2\pi\alpha) - H(m,n,-2\pi\alpha) \right) \frac{T}{2\pi} (\log T)^{n+m+2}$
+ $O\left(T(\log T)^{n+m+1} \right)$ (1.30)

where

$$H(n, m, 2\pi\alpha) = n! \sum_{\ell=0}^{\infty} \frac{(2\pi\alpha i)^{\ell}}{(\ell + n + 1)!(\ell + n + m + 2)}.$$

Under the Riemann Hypothesis in the case where n = m = 0 and for T sufficient large with α as above, Gonek proved

$$\sum_{0 < \gamma \le T} \left| \zeta \left(\frac{1}{2} + i(\gamma + \alpha L^{-1}) \right) \right|^2 = \left(1 - \left(\frac{\sin \pi \alpha}{\pi \alpha} \right)^2 \right) \frac{T}{2\pi} (\log T)^2 + O(T \log T).$$
(1.31)

Setting $\alpha = 0$ and $n = m \ge 1$ gives the second moment of $\zeta(s)$ for all derivatives,

$$J_{2,n}(T) = \sum_{0 < \gamma \le T} \left| \zeta^{(n)} \left(\frac{1}{2} + i\gamma \right) \right|^2 = \frac{n^2}{(2n+1)(n+1)^2} \frac{T}{2\pi} (\log T)^{2n+2} + O\left(T (\log T)^{2n+1} \right).$$
(1.32)

In addition, Milinovich [233] has obtained a full asymptotic for the second moment including all lower order terms and a power saving error term, showing

$$J_2(T) = \sum_{0 < \gamma \le T} |\zeta'(\rho)|^2 = \sum_{n=0}^4 a_n T\left(\log\frac{T}{2\pi}\right)^n + O\left(T^{1/2+\varepsilon}\right)$$
(1.33)

for some explicit constants a_n and for all $\varepsilon > 0$.

While no asymptotic was known for any moment other than the second, Gonek [145] and Hejhal [177] independently conjectured that

$$J_{2k}(T) \sim b_k T (\log T)^{(k+1)^2}$$
(1.34)

as $T \to \infty$ for any real k but with no suggestion for what the coefficient b_k should be. Just as Keating and Snaith [211] used RMT to calculate the unknown constant in the continuous moments case, Hughes, Keating and O'Connell [182] used similar methods to calculate the unknown constant for this conjecture. We will discuss this work in detail in Section 1.4.2. The best known partial result in proving any moments higher than the second is due to Ng [253], who showed that there exists positive constants C_1, C_2 such that, for sufficiently large T,

$$C_1 T (\log T)^9 \le J_4(T) \le C_2 T (\log T)^9$$

(and who states that the method extends easily to higher derivatives), obtaining the conjectured order for the fourth moment but not the coefficient for the leading order term. Under some additional assumptions, Garunkštis and Steuding [136] show that the upper bound in Ng's fourth moment result can be improved slightly.

There is also a partial result for the first moment, due to Garaev [135] where he restricts the sum to simple zeros of $\zeta(s)$ with real part 1/2 (see also [220, 285]. He showed for a positive constant C that

$$J_1(T) = \sum_{0 < \gamma \le T} |\zeta'(\rho)| \le CT (\log T)^{9/4},$$

a result that relies on a mean-value estimate for Dirichlet series due to Ramachandra [269]. Note that this result is of the conjectured order of magnitude given in (1.34).

Gonek [145] conjectured that

$$J_{-2}(T) \sim \frac{3}{\pi^3} T \tag{1.35}$$

as $T \to \infty$. Note that this result implies that all the zeros of $\zeta(s)$ are simple otherwise the sum would diverge for sufficiently large T. Gonek was able to prove that $J_{-2}(T) \ll T$ under the assumption that all the zeros are simple.

Milinovich and Ng [236] showed that for all $\varepsilon > 0$,

$$J_{-2}(T) \ge \left(\frac{3}{2\pi^3} - \varepsilon\right)T \tag{1.36}$$

as $T \to \infty$, which is the correct lower bound up to a factor of a half.

1.4.1.1 Upper and Lower Bounds

It can be deduced from (1.13) that for $\sigma \ge 1/2$ and $t \ge 10$

$$\left|\zeta'\left(\frac{1}{2}+it\right)\right| \ll \exp\left(\frac{c\log t}{\log\log t}\right)$$

for some c > 0. Then for k > 0 it follows that

$$J_{2k}(T) \ll \exp\left(\frac{2kc\log t}{\log\log t}\right)$$

for some c > 0.

However we can do much better. Milinovich and Ng [237] showed under the Generalised Riemann Hypothesis that the lower bound in the Gonek–Hejhal (1.34) conjecture is correct for all natural numbers k. Gao [132] improved this lower bound to all real k > 0. As a corollary, Ng [254] showed that for each A > 0, the inequality

$$|\zeta'(\rho)| \ge (\log|\gamma|)^A$$

is satisfied infinitely often when $\rho = \beta + i\gamma$ is a non-trivial zero of $\zeta(s)$.

For the upper bounds, Milinovich [234] showed that the upper bound it almost sharp for all natural numbers k, with an extra ε in the power, as in the upper bound for continuous moments by Soundararajan [291]. Additionally he showed that the upper bound for derivatives holds, that is, he showed for integer k, n and $\varepsilon > 0$, that

$$J_{2k,n}(T) \ll T(\log T)^{k^2 + 2kn + 1 + \varepsilon}$$

Kirila [214] removed the ε in the upper bound and showed that it holds for all k > 0 and all derivatives.

In terms of the first derivative, we know that for k > 0,

$$T(\log T)^{(k+1)^2} \ll J_{2k}(T) = \sum_{0 < \gamma \le T} \left| \zeta' \left(\frac{1}{2} + i\gamma \right) \right|^{2k} \ll T(\log T)^{(k+1)^2}$$

Benli, Elma and Ng [22] have shown that the expected lower bound holds for all derivatives, and so under the Riemann Hypothesis, it is known for k, n integers that

$$T(\log T)^{k^2 + 2kn + 1} \ll J_{2k,n}(T) = \sum_{0 < \gamma \le T} \left| \zeta^{(n)} \left(\frac{1}{2} + i\gamma \right) \right|^{2k} \ll T(\log T)^{k^2 + 2kn + 1}$$

It is reasonable to assume that the lower bound holds for all real k > 0 but this has yet to be shown for general n > 1.

For negative moments, under the assumption that all zeros are simple, Heap, Li and Zhao [168] proved that the lower bound of the Gonek–Hejhal conjecture is consistent for rational k < 0 and Gao and Zhao [133] show this lower bound is consistent for all k < 0. However we do not expect the lower bound to be sharp for k < -3/2. To see this, Gonek [147] defines

$$\Theta = \inf\{\theta : |\zeta'(\rho)|^{-1} \ll \gamma^{\theta}\}.$$

Then under the Riemann Hypothesis, we expect $|\zeta'(\rho)| \ll |\rho|^{\varepsilon}$ and so we expect $\Theta \ge 0$. As we have discussed, we expect to be able to model statistical behaviour of zeta zeros through statistics of eigenvalues of random unitary matrices. This suggests $\Theta = 1/3$ (see [182], for example), and so it is possible that

$$|\zeta'(\rho)| \ll |\gamma|^{-1/3+\varepsilon}$$

occurs infinitely often.

Additionally, we do not expect to get a good upper bound for negative moments. This is explained by Hiary and Odlyzko [179] who note that extreme values of negative moments are caused by very few zeros. Small values of $|\zeta(1/2 + i\gamma)|$ typically occur at pairs of consecutive zeros that are close to each other, called Lehmer pairs, discussed further in Chapter 2. For example, they show that for 15 sets of $\approx 10^9$ zeros near the 10^{23} -rd zero, 87% of the value of the moment for these zeros comes from just 4 zeta zeros. Understanding this very rare behaviour with a good bound is tough in part due to the rarity of these zeros. There are very few upper bounds in the negative moment case. Bui, Florea, and Milinovich [45] obtained an upper bound for a related sum but the sum is not over all zeros. Zeros that are close in some sense to other zeros are removed from this sum, so while they include what is expected to be arbitrarily close to the full density inside the set of all zeros, they are still excluding zeros that are expected to contribute a large amount to the total sum.

1.4.2 RANDOM MATRIX THEORY MEETS NUMBER THEORY: ACT III

In a similar manner to the approach taken by Keating and Snaith in modelling continuous moments of $\zeta(s)$, Hughes, Keating and O'Connell [182] calculated the analogous result for discrete moments of $\zeta'(\rho)$. Recall that to model continuous moments of $\zeta(s)$ we calculated moments of characteristic polynomials of random unitary matrices. In this instance, they calculated moments of derivatives of characteristic polynomials of random unitary matrices. Differentiating the characteristic polynomial with respect to θ gives

$$Z'_N(\theta) = i \sum_{j=1}^N e^{i(\theta_j - \theta)} \prod_{\substack{m=1\\m \neq j}}^N (1 - e^{i(\theta_m - \theta)}),$$

 \mathbf{SO}

$$|Z'_N(\theta_j)| = \prod_{\substack{m=1\\m\neq j}}^N \left| 1 - e^{i(\theta_m - \theta_j)} \right|$$

To model moments of $J_{2k}(T)$, they calculated the expectation of

$$\frac{1}{N}\sum_{j=1}^{N}|Z_N'(\theta_j)|^{2k}.$$

By the rotational invariance of Haar measure,

$$\mathbb{E}_{N}\left\{\frac{1}{N}\sum_{j=1}^{N}|Z_{N}'(\theta_{j})|^{2k}\right\} = \mathbb{E}_{N}\left\{|Z_{N}'(\theta_{N})|^{2k}\right\} = \mathbb{E}_{N}\left\{\prod_{n=1}^{N-1}\left|1-e^{i(\theta_{n}-\theta_{N})}\right|^{2k}\right\}.$$
 (1.37)

The following lemma transforms expectation with respect to Haar measure over U(N) to expectation with respect to Haar measure over U(N-1). A proof can be found in [181].

Lemma 1.18 (Hughes). Writing \mathbb{E}_N to denote expectation with respect to Haar measure over U(N), if f is a 2π -periodic function, then

$$\mathbb{E}_N\left\{\prod_{n=1}^{N-1} f(\theta_n - \theta_N)\right\} = \frac{1}{N} \mathbb{E}_{N-1}\left\{\prod_{n=1}^{N-1} \left|1 - e^{i\theta_n}\right|^2 f(\theta_n)\right\}.$$

Applying Lemma 1.18 to (1.37) gives

$$\frac{1}{N}\mathbb{E}_{N-1}\left\{|Z_{N-1,\hat{U}}(0)|^{2k+2}\right\} = \frac{1}{N}\frac{G^2(k+2)}{G(2k+3)}\frac{G(N+2k+2)G(N)}{G^2(N+k+1)}$$

where $\hat{U} \in U(N-1)$. The evaluation of this expectation is essentially (1.21). This result holds for $\Re(k) > -3/2$. Clearly it can be continued meromorphically to the whole complex plane.

For k bounded and as $N \to \infty$,

$$\mathbb{E}_N\left\{|Z'_N(\theta_N)|^{2k}\right\} = \frac{G^2(k+2)}{G(2k+3)}N^{k(k+2)} + O\left(N^{k(k+2)-1}\right).$$

This calculation has modelled $J_{2k}(T)/N(T)$. Hughes, Keating and O'Connell then made the following conjecture.

Conjecture 1.19 (Hughes–Keating–O'Connell). For k fixed with $\Re(k) > -3/2$,

$$J_{2k}(T) = \sum_{0 < \gamma \le T} \left| \zeta' \left(\frac{1}{2} + i\gamma \right) \right|^{2k} \sim a_k \frac{G^2(k+2)}{G(2k+3)} \frac{T}{2\pi} (\log T)^{(k+1)^2}$$

as $T \to \infty$, with a_k defined in (1.19) and G is the Barnes G-function defined in (1.23).

Note that this conjecture is consistent with all known results and conjectures. Indeed, it also predicts what the true coefficient should be in Ng's fourth moment result, that is, it predicts

$$J_4(T) \sim \frac{1}{2880\pi^3} T(\log T)^9$$

as $T \to \infty$. This coefficient was conjectured from a purely number theoretical argument by Hughes [181], pushing a result of Conrey, Ghosh and Gonek [70] beyond its range of applicability. It also agrees with the Gonek–Hejhal conjecture (1.34), conjecturing the unknown coefficient b_k for k = 2.

As in the continuous moment case, part of the motivation for studying the discrete moments is to establish the value distribution of the real part of the logarithm of $Z'_N(\theta)$. Hughes, Keating and O'Connell [182] showed that as $N \to \infty$, that the real part tends to a Gaussian random variable.
1.4. DISCRETE MOMENTS OF THE RIEMANN ZETA FUNCTION

Theorem 1.20 (Hughes–Keating–O'Connell). For a < b,

$$\lim_{N \to \infty} \text{Haar} \left\{ \frac{\log \left| \frac{Z'_N(\theta_N)}{Ne^{\gamma_0 - 1}} \right|}{\sqrt{\frac{1}{2} (\log N + 3 + \gamma_0 - \frac{\pi^2}{2})}} \right\} = \frac{1}{\sqrt{2\pi}} \int_a^b e^{-\frac{x^2}{2}} dx$$

where the measure of the set is taken to be the usual Haar measure on U(N) and where γ_0 is Euler's constant.

This should be compared with Hejhal's Central Limit Theorem [177] for $\zeta'(s)$, which shows that the real part of the logarithm of $\zeta'(s)$ tends to a Gaussian random variable.

Theorem 1.21 (Hejhal). If we assume the Riemann Hypothesis and the existence of an α such that

$$\limsup_{T \to \infty} \frac{1}{N(2T) - N(T)} \left| \left\{ n : T \le \gamma_n \le 2T, \ 0 \le \gamma_{n+1} - \gamma_n \le \frac{c}{\log T} \right\} \right| \le Mc^{c}$$

holds uniformly for 0 < c < 1, with M a suitable constant, then for a < b,

$$\lim_{T \to \infty} \frac{1}{N(2T) - N(T)} \left| \left\{ n : T \le \gamma_n \le 2T, \frac{\log \left| \frac{\zeta'(\frac{1}{2} + i\gamma_n)}{\frac{1}{2\pi} \log \frac{\gamma_n}{2\pi}} \right|}{\sqrt{\frac{1}{2} \log \log T}} \in (a, b) \right\} \right| = \frac{1}{\sqrt{2\pi}} \int_a^b e^{-\frac{x^2}{2}} dx.$$

Recall that $N = \log T/2\pi$. Comparing the two Central Limit Theorems above, we see that the two results agree (up to errors of O(1) which are subdominant in the large N, Tlimit). Çiçek [48] gives an alternative proof of Hejhal's result and improves it by providing a rate of convergence to the distribution. A similar result in this paper is proved which shows the imaginary part of the logarithm of $\zeta'(s)$ tends to a Gaussian random variable.

As in the continuous moments case, two obvious questions arise from Conjecture 1.19. The first, that of lower order terms, is addressed in Section 6.1.5. The second question, a more subtle one, concerns the arithmetic factor. As in the Keating–Snaith conjecture, Hughes, Keating and O'Connell were forced to insert the arithmetic factor after obtaining the RMT moments result. Why should the same arithmetic factor a_k occur in both the continuous and discrete moments conjectures?

This question was again settled by the hybrid model, this time for discrete moments, of Bui, Gonek and Milinovich [47]. Recall that we write

$$\zeta(s) = P_X(s)Z_X(s)(1+\mathcal{E}).$$

with an error term \mathcal{E} which is not the same throughout and that we ignore here. Differentiating with respect to s we obtain

$$\zeta'(s) = \left(P'_X(s)Z_X(s) + P_X(s)Z'_X(s)\right)\left(1 + \mathcal{E}\right)$$

and on setting $s = \rho$ a non-trivial zero of $\zeta(s)$, we have

$$\zeta'(\rho) = P_X(\rho) Z'_X(\rho) (1 + \mathcal{E}).$$

Then they have the same prime part, $P_X(s)$, as in the continuous hybrid model of Gonek, Hughes and Keating, explaining why we should expect the same arithmetic factor in both moment conjectures. By assuming the splitting conjecture for discrete moments, Bui, Gonek and Milinovich re-established the Hughes, Keating and O'Connell conjecture, with the arithmetic term appearing naturally this time. They also showed that splitting holds to leading order for k = 1 in this case, corresponding to Gonek's second discrete moment result (1.32).

Returning to our questions, we now consider that of lower order terms for the discrete moments conjecture. In the continuous moments case, we have the recipe of Conrey, Farmer, Keating, Rubinstein and Snaith to predict the lower order terms for integer k. To calculate the lower order terms for the discrete moments we need a generalisation of the recipe, known as the Ratios Conjecture of Conrey, Farmer and Zirnbauer [63], discussed in detail in Section 6.1.

1.4.3 Applications of Discrete Moments

1.4.3.1 Simple Zeros of $\zeta(s)$

When calculating the number of zeros N(T) of $\zeta(s)$, as given in (1.4), we count each zero with multiplicity, denoted $m(\rho)$. It is conjectured that $m(\rho) = 1$ for all zeros, that is, that all the non-trivial zeros of $\zeta(s)$ are simple. We review the evidence for this conjecture below, before discussing (potentially hypothetical) multiple zeros in the next section.

Let $N^*(T)$ be the number of simple non-trivial zeros of $\zeta(s)$ with $0 < \gamma \leq T$. Then

$$\kappa^* = \liminf_{T \to \infty} \frac{N^*(T)}{N(T)}$$

is the proportion of simple zeros of $\zeta(s)$.

All zeros that have been found numerically are known to be simple. Unconditionally Heath-Brown [173] (and independently, Selberg) showed that Levinson's [222] result that one-third of the non-trivial zeros lie on the critical line implies

$$\kappa^* \ge 0.3474.$$

Through Conrey's [56] result that two-fifths of the non-trivial zeros lie on the critical line, he improved the above proportion to

$$\kappa^* \ge 0.4058.$$

Various small improvements were made following this method of Levinson's, and they were due to Bui, Conrey, and Young [42], Feng [115], and Bui [41], with the latest improvement using this method due to Pratt, Robles, Zaharescu and Zeindler [265]. They showed that the proportion of zeros on the critical line is $\kappa^* \geq 0.407$. In proving an unconditional form of Montgomery's Pair Correlation Conjecture 1.3, Baluyot, Goldston, Suriajaya and Turnage-Butterbaugh [18] showed that $\kappa^* \geq 0.617$.

Montgomery observed that

$$N^*(T) \ge \sum_{0 < \gamma \le T} (2 - m(\rho)).$$

Assuming the Riemann Hypothesis, Montgomery's Theorem 1.2 implies

$$\sum_{0 < \gamma \leq T} m(\rho) \leq \left(\frac{4}{3} + o(1)\right) N(T)$$

where the zeros are repeated according to multiplicity. Combining these two inequalities imply that

$$\kappa^* \ge 0.666...,$$

with incremental progress using this method due to Montgomery and Taylor [240] who showed $\kappa^* \ge 0.6725$, by Cheer and Goldston [52] who showed $\kappa^* \ge 0.6727$, and by Chirre, Gonçalves and de Laat [53] who showed $\kappa^* \ge 0.679$. Montgomery's Pair Correlation Conjecture 1.3 implies that 100% are simple [239].

By a mollifier method described below, Conrey, Ghosh, and Gonek [75] showed assuming the Riemann Hypothesis and the Generalised Lindelöf Hypothesis that

$$\kappa^* \geq 0.70307...$$

with Bui and Heath-Brown [39] removing the assumption of the Generalised Lindelöf Hypothesis.

To obtain a lower bound for κ^* using this method, we need to know the mollified first and second moments of $\zeta'(s)$. Note that ρ is a simple zero of $\zeta(s)$ if and only if $\zeta'(\rho) \neq 0$. Hence by the Cauchy–Schwarz inequality,

$$\left|\sum_{0<\gamma\leq T} B\zeta'(\rho)\right|^2 \leq N^*(T) \sum_{0<\gamma\leq T} \left|B\zeta'(\rho)\right|^2$$

for a mollifier B(s). We note that mollifiers are necessary to get a proportion of simple zeros, a theme we have already come across in this thesis in Sections 1.3.7.3 and 1.3.7.4. Conrey, Ghosh and Gonek [73] initially used this method without mollifiers to give a more straightforward proof that there are infinitely many simple zeros of $\zeta(s)$, as described in Chapter 2, equation (2.2). We mention one further result in the theme of the above work. In 1986, Conrey, Ghosh and Gonek [71] considered the problem of showing that there are infinitely many zeros of the Dedekind zeta function $\zeta_K(s)$ of a quadratic number field $K = \mathbb{Q}(\sqrt{d})$. Unlike in the case of the Riemann zeta function where this was already known (albeit with complicated proofs), it was not known before this work that there were infinitely many simple zeros of $\zeta_K(s)$. Their proof is based on the fact that for a quadratic number field, we have the factorisation

$$\zeta_K(s) = \zeta(s)L(s,\chi),$$

where $\chi \mod |d|$ is a real primitive character. In particular, a zero of the Riemann zeta function is also a zero of $\zeta_K(s)$.

Clearly a zero of $\zeta(s)$ is a simple zero of $\zeta_K(s)$ if and only if $\zeta'(\rho)L(1-\rho,\chi) \neq 0$. As before, they used the Cauchy–Schwarz inequality to write

$$\left| \sum_{0 < \gamma \le T} \zeta'(\rho) L(1-\rho, \chi) \right|^2 \le \sum_{0 < \gamma \le T}^* 1 \sum_{0 < \gamma \le T} |\zeta'(\rho) L(1-\rho, \chi)|^2$$

where the star on the middle sum denotes counting only the simple zeros of $\zeta_K(s)$. The main difficulty lies in calculating the sum on the left-hand side. They showed

$$\sum_{0 < \gamma \le T} \zeta'(\rho) L(1-\rho, \chi) = L(1, \chi) \frac{T}{4\pi} (\log T)^2 + O(T \log T)$$

and as they also showed that

$$\sum_{0 < \gamma \le T} \left| \zeta'(\rho) L(1-\rho,\chi) \right|^2 \ll T^{2-\delta}$$

where $\delta > 0$, this implies that there are infinitely many simple zeros of $\zeta_K(s)$. Improving on this in [72], they obtained a positive proportion of simple zeros, showing that under the Riemann Hypothesis, the proportion of simple zeros of $\zeta_K(s)$ is at least 1/54. By assuming the Riemann Hypothesis for $\zeta_K(s)$, this is improved to at least 1/27. This relatively low (albeit positive) proportion is due to the fact that $\zeta(s)$ and $L(s,\chi)$ can share zeros. They note that the approach that they take in [75] does not work in this instance to improve the proportion, and doing so in any way would pose an interesting problem.

1.4.3.2 Multiple zeros of $\zeta(s)$

While we have only focused on simple zeros of the Riemann zeta function, the possibility remains that there could be multiple zeros. We consider this possibility in this section and discuss the results that are known for these zeros.

Before rigorously proving that at least 70% of the zeros are simple (see [75]), Conrey, Ghosh and Gonek [74] provided a sketch of some more general results, including how to calculate proportions of multiple zeros through some discrete moments. To describe their results we set up some notation.

Let T be a large positive real number and let

$$B(s, P) = \sum_{n \le y} \frac{\mu(n) P\left(\frac{\log y/n}{\log y}\right)}{n^s}$$

where $y = T^{\theta}$ for $\theta < 1/2$, P(x) is an entire function with P(0) = 0, and $\mu(n)$ is the Möbius function given in (1.27). Let a, b be complex numbers and $Q_1(x)$ and $Q_2(x)$ be polynomials.

We consider a twisted discrete second moment, defined by

$$I_1(a, b, P_1, P_2, Q_1, Q_2) = \sum_{0 < \gamma \le T} Q_1\left(-\frac{d}{da}\right) \zeta\left(\rho + \frac{a}{\log T}\right) Q_2\left(-\frac{d}{db}\right) \zeta\left(1 - \rho + \frac{b}{\log T}\right) B(\rho, P_1) B(1 - \rho, P_2).$$

Theorem 1.22 (Conrey–Ghosh–Gonek). If $0 < \theta < 1/2$ and $a, b \ll 1$, then as $T \to \infty$,

$$\begin{split} I_{1}(a, b, P_{1}, P_{2}, Q_{1}, Q_{2}) &\sim \\ & \frac{T}{2\pi} \log T \frac{\partial}{\partial u} \frac{\partial}{\partial v} \left\{ \left(\frac{1}{\theta} \int_{0}^{1} P_{1} P_{2} \, dx + \int_{0}^{1} P_{1} \, dx \int_{0}^{1} P_{2} \, dx \right) \\ & \qquad \times \left(\int_{0}^{1} T_{a} Q_{1} T_{b} Q_{2} \, dx - \int_{0}^{1} T_{a} Q_{1} \, dx \int_{0}^{1} T_{b} Q_{2} \, dx \right) \\ & \qquad + \int_{0}^{1} P_{1} \, dx \int_{0}^{1} P_{2} \, dx \left(Q_{1}(0) - \int_{0}^{1} T_{a} Q_{1} \, dx \right) \left(Q_{2}(0) - \int_{0}^{1} T_{b} Q_{2} \, dx \right) \right\} \Big|_{u=v=0}, \end{split}$$

where $P_1 = P_1(x+u), P_2 = P_2(x+v),$

$$T_a Q_1 = e^{-a(x+\theta u)} Q_1(x+\theta u)$$
 and $T_b Q_2 = e^{-b(x+\theta v)} Q_2(x+\theta v).$

It should be possible to deduce this result from the more general result of Heap, Li, and Zhao [168] (as claimed in [46] but details should be written down), which builds on the work in [75] and [39].

Similarly, we consider a twisted discrete first moment, defined by

$$I_2(a, P, Q) = \sum_{0 < \gamma \le T} Q\left(-\frac{d}{da}\right) \zeta\left(\rho + \frac{a}{\log T}\right) B(\rho, P).$$

Theorem 1.23 (Conrey–Ghosh–Gonek). If $0 < \theta < 1/2$ and $a \ll 1$, then as $T \to \infty$,

$$I_2(a, P, Q) \sim -\frac{T}{2\pi} \log T \frac{d}{du} \left\{ \left(Q(0) - \int_0^1 T_a Q \ dx \right) \int_0^1 P \ dx \right\} \Big|_{u=0},$$

where P = P(x + u) and $T_a Q = e^{-a(x+\theta u)}Q(x + \theta u)$.

Likewise, it should be possible to deduce this result from the more general result of Benli, Elma and Ng [22] (but details should be written down).

To begin, Conrey, Ghosh and Gonek consider the proportion of simple zeros which we have already discussed in Section 1.4.3.1, but repeat here for completeness. Recall that N(T) is the number of non-trivial zeros with $0 < \gamma \leq T$, given in (1.4), and $N^*(T)$ is the number of simple non-trivial zeros of $\zeta(s)$ with $0 < \gamma \leq T$. Then we set

$$\kappa^* = \liminf_{T \to \infty} \frac{N^*(T)}{N(T)}$$

to be the proportion of simple zeros of $\zeta(s)$. By the Cauchy–Schwarz inequality,

$$N^*(T) \ge \frac{\left|\sum_{0 < \gamma \le T} \zeta'(\rho) B(\rho, P)\right|^2}{\sum_{0 < \gamma \le T} |\zeta'(\rho) B(\rho, P)|^2}.$$

Clearly the right-hand side can be evaluated using Theorems 1.22 and 1.23, with the denominator requiring the assumption of the Riemann Hypothesis. The optimal choice of P(x) can be calculated via the calculus of variations, giving

$$P(x) = -\theta x^2 + (1+\theta)x$$

 \mathbf{SO}

$$\kappa^* \ge 1 - \frac{1}{(1+\theta)^3}.$$

As $\theta \to 1/2^-$, we obtain

as before.

If we were to push Theorems 1.22 and 1.23 beyond their range of applicability in θ , we conjecture (assuming these two theorems still hold) that

 $\kappa^* \ge \frac{19}{27}$

$$\kappa^* \ge \begin{cases} \frac{988}{1331} \approx 0.7423... & \text{as } \theta \to 4/7^- \\ \frac{7}{8} = 0.875 & \text{as } \theta \to 1^- \\ 1 & \text{as } \theta \to \infty. \end{cases}$$

We choose the values of 4/7 to compare with the result of Conrey [56] on zeros on the critical line, 1 as we expect mollifiers to hold true to length T, and ∞ for Farmer's [112] $\theta = \infty$ conjecture.

The purpose of this section is to consider multiple zeros, however. We let $N^{(2)}(T)$ denote the number of simple and double non-trivial zeros with $0 < \gamma \leq T$ and let

$$\kappa^{(2)} = \liminf_{T \to \infty} \frac{N^{(2)}(T)}{N(T)}$$

be the proportion of simple and double zeros of $\zeta(s)$. Again by the Cauchy—Schwarz inequality,

$$N^{(2)}(T) \ge \frac{\left|\sum_{0 < \gamma \le T} \zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2)\right|^2}{\sum_{0 < \gamma \le T} \left|\zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2)\right|^2}$$

Again, the right-hand side can be evaluated using Theorems 1.22 and 1.23, with the denominator requiring the assumption of the Riemann Hypothesis. Then optimising the polynomials (with a slightly more optimal choice than that given in [74]), we have

$$P_{1}(x) = -\left(\frac{7\theta + 3\theta^{2} + 3\theta^{3}}{7}\right)x^{3} + \left(\frac{-77 - 35\theta + 65\theta^{2} + 30\theta^{3} + 45\theta^{4}}{70\theta}\right)x^{2} + \left(\frac{42 + 182\theta + 158\theta^{2} + 85\theta^{3} + 45\theta^{4}}{70\theta^{2}}\right)x$$
$$P_{2}(x) = x^{3} - 3\left(\frac{7 + 14\theta + 3\theta^{2} + 3\theta^{3}}{14\theta}\right)x^{2} + \left(\frac{7 + 35\theta + 31\theta^{2} + 15\theta^{3} + 9\theta^{4}}{14\theta^{2}}\right)x$$

and so

$$\kappa^{(2)} \ge 1 - \frac{7 + 3\theta^2}{7 + 56\theta + 199\theta^2 + 416\theta^3 + 434\theta^4 + 280\theta^5 + 150\theta^6 + 45\theta^7}$$

Then similarly to the simple zeros case above, if we were to push Theorems 1.22 and 1.23 beyond their range of applicability in θ , we conjecture (assuming these two theorems still hold) that

$$\kappa^{(2)} \ge \begin{cases} \frac{173}{181} \approx 0.9556\dots & \text{as } \theta \to 1/2^{-1} \\ \frac{512096}{528903} \approx 0.9682\dots & \text{as } \theta \to 4/7^{-1} \\ \frac{200180384}{206751921} \approx 0.9682\dots & \text{as } \theta \to 1^{-1} \\ 1 & \text{as } \theta \to \infty. \end{cases}$$

This means that under the Riemann Hypothesis, fewer than 4.44% of the non-trivial zeros of $\zeta(s)$ have multiplicity 3 or greater, with this percentage decreasing as the length of the mollifier increases.

In an entirely analogous way, we let $N^{(3)}(T)$ denote the number of simple, double, and triple non-trivial zeros with $0 < \gamma \leq T$ and let $N^{(4)}(T)$ denote the number of simple, double, triple, and quadruple non-trivial zeros with $0 < \gamma \leq T$. Similarly, let

$$\kappa^{(3)} = \liminf_{T \to \infty} \frac{N^{(3)}(T)}{N(T)} \text{ and } \kappa^{(4)} = \liminf_{T \to \infty} \frac{N^{(4)}(T)}{N(T)}$$

be the proportion of simple, double, and triple (and quadruple) zeros of $\zeta(s)$. Again by the Cauchy—Schwarz inequality,

$$N^{(3)}(T) \ge \frac{\left|\sum_{0 < \gamma \le T} \zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2) + \zeta'''(\rho) B(\rho, P_3)\right|^2}{\sum_{0 < \gamma \le T} |\zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2) + \zeta'''(\rho) B(\rho, P_3)|^2}$$

and similarly

$$N^{(4)}(T) \geq \frac{\left|\sum_{0 < \gamma \leq T} \zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2) + \zeta'''(\rho) B(\rho, P_3) + \zeta^{(IV)}(\rho) B(\rho, P_4)\right|^2}{\sum_{0 < \gamma \leq T} \left|\zeta'(\rho) B(\rho, P_1) + \zeta''(\rho) B(\rho, P_2) + \zeta'''(\rho) B(\rho, P_3) + \zeta^{(IV)}(\rho) B(\rho, P_4)\right|^2}$$

Again, we can evaluate the right-hand side of both of these inequalities using Theorems 1.22 and 1.23, with the denominators requiring the assumption of the Riemann Hypothesis. Next we optimise the polynomials P_1, P_2, P_3 for $N^{(3)}(T)$ and P_1, P_2, P_3, P_4 for $N^{(4)}(T)$. We don't state any of these polynomials for brevity as the coefficients are fairly complicated. In the optimisation that we have found for the $N^{(3)}(T)$ case, P_1 is of degree 6, P_2 is of degree 5 and P_3 is of degree 4, while for the $N^{(4)}(T)$ case, P_1 is of degree 8, P_2 is of degree 7, P_3 is of degree 6, and P_4 is of degree 5. We then have

$$\kappa^{(3)} \ge 1 - \frac{1}{1 + 15\theta + 105\theta^2 + 455\theta^3 + 1065\theta^4 + 1353\theta^5 + 875\theta^6 + 255\theta^7}$$

and

$$\frac{\kappa^{(4)} \ge 1 - 1}{1 + 24\theta + 276\theta^2 + 2024\theta^3 + 8526\theta^4 + 21504\theta^5 + 32984\theta^6 + 30096\theta^7 + 14994\theta^8 + 3136\theta^9}$$

Then similarly to the cases above, if we were to push Theorems 1.22 and 1.23 beyond their range of applicability in θ , we conjecture (assuming these two theorems still hold) that

$$\kappa^{(3)} \geq \begin{cases} \frac{27507}{27635} \approx 0.9954\dots & \text{as } \theta \to 1/2^{-1}\\ \frac{295389868}{296213411} \approx 0.9972\dots & \text{as } \theta \to 4/7^{-1}\\ \frac{4093}{4094} \approx 0.9998\dots & \text{as } \theta \to 1^{-1}\\ 1 & \text{as } \theta \to \infty \end{cases}$$

and

$$\kappa^{(4)} \geq \begin{cases} \frac{301321}{301449} \approx 0.9996 \dots & \text{as } \theta \to 1/2^{-1} \\ \frac{3820131104}{3820954647} \approx 0.9998 \dots & \text{as } \theta \to 4/7^{-1} \\ \frac{113564}{113565} \approx 0.999991 \dots & \text{as } \theta \to 1^{-1} \\ 1 & \text{as } \theta \to \infty. \end{cases}$$

Clearly the proportion of multiple zeros is decreasing quickly as n increases. It would be a good exercise to extend this method to show that the proportion of simple, double, ..., n-tuple zeros tends to zero at $n \to \infty$. It would also be interesting to see if Theorems 1.22 and 1.23 could be proved with longer mollifiers as this also increases each of the proportions of simple and multiple zeros. Deducing full proofs of these two theorems from [168] and [22] should also be written formally. Finally, it would be interesting to see if the $\theta = \infty$ conjecture from [112] implies all the zeros of the Riemann zeta function are simple, not just 100%, in the same way that Bettin and Gonek [29] showed that the $\theta = \infty$ conjecture implies all the zeros of the Riemann zeta function are on the critical line, not just 100%.

1.4.3.3 Gaps between zeros of $\zeta(s)$

While questions have already been considered regarding the horizontal and vertical distribution of the zeros of $\zeta(s)$, further questions can be asked. Consider the sequence $0 < \gamma_1 \leq \gamma_2 \leq \dots$ of consecutive ordinates of the non-trivial zeros of $\zeta(s)$ and note that the average size of $\gamma_{n+1} - \gamma_n$ is $2\pi/\log \gamma_n$. Normalising, we set

$$\lambda = \limsup_{n \to \infty} \frac{(\gamma_{n+1} - \gamma_n) \log \gamma_n}{2\pi}$$

and

$$\mu = \liminf_{n \to \infty} \frac{(\gamma_{n+1} - \gamma_n) \log \gamma_n}{2\pi}$$

By definition, $\mu \leq 1 \leq \lambda$ but it is conjectured that $\mu = 0$ and $\lambda = \infty$. This means that there are infinitely many pairs of zeros arbitrarily close, and arbitrarily far apart relative to the average spacing. The idea of looking at gaps between zeros started with Montgomery and his Pair Correlation Conjecture 1.3. This conjecture implies that $\mu = 0$, as for any c > 0, where the sum is over normalised zeros w_n ,

$$0 < \lim_{T \to \infty} \frac{1}{N(T)} \sum_{\substack{\gamma, \gamma' \in [0, T] \\ 0 \le w_m - w_n \le c}} 1 = \int_0^c \left(1 - \left(\frac{\sin \pi x}{\pi x}\right)^2 \right) \, dx < c$$

so take c arbitrarily small and this quantity will still be positive, giving us a pair of zeros arbitrarily close. For large gaps, recall that we believe that we can model zeta zeros through random unitary matrices thanks to Montgomery's Pair Correlation Conjecture 1.3 and the link noted by Dyson. Together with the conjecture that n-level correlations of zeros of $\zeta(s)$ also follow these random matrix statistics due to Rudnick and Sarnak [275], this implies that $\lambda = \infty$.

Selberg (unpublished but announced in [279]) and Fujii [122, 123] showed that $\mu < 1 < \lambda$. For references to these results and corrections to some misprints that have occurred, see [90]. There is a vast literature of results, both unconditional and conditional, about the sizes of μ and λ . For a selection and summary of this literature, see Page 15 of [302].

There are relations between the class number problem, gaps between zeros, and exceptional zeros of *L*-functions. We let h(K) denote the class number of $K = \mathbb{Q}(\sqrt{d})$, where *d* is a fundamental discriminant. In Article 303 of *Disquisitiones Arithmeticae* [138], Gauss conjectured that $k(K) \to \infty$ as *d* runs through negative discriminants. For a full history of this problem and its solutions, see Section 6 in Chapter 20 of Ireland and Rosen 82

[196]. This implies that there are only finitely many imaginary quadratic fields K with h(K) = n for any fixed positive integer n. The Class Number Problem for imaginary quadratic fields is to give a complete list of fundamental discriminants for each fixed positive integer n.

For n = 1, this is a resolved conjecture of Gauss, who conjectured the only d are

$$d = -3, -4, -7, -8, -11, -19, -43, -67, -163$$

which was proved independently by Heegner [176], Stark [293] and Baker [14]. What happens for $n \ge 2$?

The Class Number Formula (for d < -4) states

$$h(K) = \frac{1}{\pi} \sqrt{|d|} L(1, \chi_d)$$

where $L(s, \chi_d)$ is the Dirichlet *L*-function with coefficients $\chi_d = \left(\frac{d}{n}\right)$ where $\left(\frac{1}{n}\right)$ denotes the Kronecker symbol.

We want a bound on h(K). Under the Generalised Riemann Hypothesis, we have $(\log \log |d|)^{-1} \ll L(1, \chi_d) \ll (\log \log |d|)$, so

$$\sqrt{|d|}(\log\log|d|)^{-1} \ll h(K) \ll \sqrt{|d|}(\log\log|d|).$$

The problem with finding a stronger, unconditional bound on $L(1, \chi_d)$ is that we may have an Riemann–Siegel zero of $L(s, \chi_d)$, a real zero which is close to 1. Siegel [281] gave an unconditional bound, but the implied constant is not computable. Goldfeld [140] and Gross–Zagier [152] showed a better unconditional bound with computable constant, and Watkins [306] used this to give a complete list of fundamental discriminants with h(K) = 2, 3, ..., 100.

The connection of gaps to the class number problem comes from a result of Conrey and Iwaniec [77] which states as a corollary that if for all large T there are $\gg T(\log T)^{4/5}$ non-trivial zeros of $\zeta(s)$ up to height T where the average spacing between pairs of these consecutive zeros is less than 1/2, then

$$L(1, \chi_d) \gg \frac{1}{(\log |d|)^{90}}$$

with computable implied constant.

Therefore if we can find better values of μ , and in particular $\mu < 1/2$, then we get stronger lower bounds on $L(1, \chi_d)$ (and so better bounds on h(K)), and in turn will imply that there are no Riemann–Siegel zeros.

How can we compute small and large gaps? We begin by returning to Gonek's second discrete moment result (1.30). For any real $|\alpha| \leq L/2$, setting n = m = 0 and assuming

1.4. Discrete Moments of the Riemann zeta function

the Riemann Hypothesis gives

$$\sum_{0<\gamma\leq T} \left| \zeta \left(\frac{1}{2} + i \left(\gamma + \frac{\alpha}{L} \right) \right) \right|^2 = \left(1 - \left(\frac{\sin \pi \alpha}{\pi \alpha} \right)^2 \right) \frac{T}{2\pi} (\log T)^2 + O(T \log T), \quad (1.38)$$

Notice that the coefficient of the leading order term is exactly that from the integrand in Montgomery's Pair Correlation 1.3.

Under the Riemann Hypothesis, Mueller [249, 250] then exploited this connection to prove results for large gaps, and Conrey, Ghosh and Gonek [69] used this method to also calculate small gaps between zeros.

As an example, we sketch how to calculate large gaps. Integrate both sides of (1.38) over $[-\beta/2, \beta/2]$ for some $\beta > \lambda$, giving

$$\sum_{0 < \gamma \le T} \int_{-\beta/2}^{\beta/2} \left| \zeta \left(\frac{1}{2} + i\gamma + it \right) \right|^2 dt \sim F(\beta)T \log T,$$

using (1.14), where

$$F(\beta) = \int_{-\beta/2}^{\beta/2} \left(1 - \left(\frac{\sin \pi \alpha}{\pi \alpha}\right)^2 \right) \, d\alpha.$$

Now if $\beta > \lambda$ then $F(\beta) \ge 1$. Using a computer package we can show F(1.9) = 0.997 so

 $\lambda > 1.9.$

Hughes [181] showed that if we assume results on the fourth discrete moment, using Conjecture 1.19, then we have $\lambda > 2.7$. Steuding and Steuding [294] then assumed the whole discrete moments Conjecture 1.19 to show that $\lambda = \infty$.

We noted above that if we can show that $\mu < 1/2$ then we have no Riemann–Siegel zeros. However, in their paper on small gaps between zeros of $\zeta(s)$, Conrey, Ghosh and Gonek [69] showed that this method cannot break through this barrier, and so we would need a different approach to this problem.

1.4.3.4 Behaviour of M(x)

The summatory function of the Möbius function $\mu(n)$ is given by

$$M(x) = \sum_{n \le x} \mu(n).$$

Based on numerical evidence, some number theorists used to believe Merten's conjecture, that

$$M(x) = O(\sqrt{x})$$

holds for $x \ge 2$. This was an attractive belief as it implies the Riemann Hypothesis. However, Ingham [195] showed that Merten's conjecture implied that the ordinates of the zeros of $\zeta(s)$ satisfy some rational linear relations. This would contradict what we call the Linear Independence Hypothesis, which states that there are no rational linear relations between zeros of any *L*-functions. This hypothesis seemed to first appear in the work of Wintner [313] to study the error term in the Prime Number Theorem, which we come back to at the end of this section.

It was believed that it was more likely that the Linear Independence Hypothesis was true, and in 1985, Odlyzko and te Reile [260] showed that Merten's conjecture is false by showing

$$\liminf_{x \to \infty} \frac{M(x)}{\sqrt{x}} < -1.009 \text{ and } \limsup_{x \to \infty} \frac{M(x)}{\sqrt{x}} > 1.06.$$

For a full history of Merten's conjecture, see Odlyzko and te Reile's paper [260]. We note that their method doesn't give an x such that these inequalities hold, and we still do not know one such that $M(x) > \sqrt{x}$. In fact, Ingham went further and showed that the Linear Independence Hypothesis implies

$$\liminf_{x\to\infty} \frac{M(x)}{\sqrt{x}} = -\infty \text{ and } \limsup_{x\to\infty} \frac{M(x)}{\sqrt{x}} = +\infty$$

The Linear Independence Hypothesis, together with the Riemann Hypothesis, has some fascinating consequences. Chebyshev, in 1853, in a letter to Fuss [50], asked questions about whether there are more primes congruent to 1 mod 4 or 3 mod 4 to some real number x, denoted $\pi(x; 4, 1)$ and $\pi(x; 4, 3)$ respectively. Littlewood [225] proved that $\pi(x; 4, 3) - \pi(x; 4, 1)$ changes sign infinitely often, and unlike the $\pi(x) - \text{Li}(x)$ case (as discussed in Section 1.1.4), it is easier to find when these sign changes occur - the first is at x = 26861. The notion of comparing primes in different arithmetic progressions is called a prime number race, and many results are reliant on both the Linear Independence Hypothesis and the Riemann Hypothesis, as shown by Rubinstein and Sarnak [273]. We note that prime number races can be generalized to an arbitrary modulus and to more than two residue classes. For a summary of this area, see [151] for a friendly introduction in the topic of prime number races and see [229] for the annotated bibliography that lists and summarises every publication in the field of comparative prime number theory.

Returning to M(x), it is still possible that the weak Merten's conjecture is true. This states that

$$\int_{1}^{X} \left(\frac{M(x)}{x}\right) dx \ll \log X$$

This conjecture implies that the Riemann Hypothesis is true, that all the zeros are simple, and that

$$\sum_{\gamma>0}\frac{1}{|\rho\zeta'(\rho)|^2}$$

converges, showing a clear link between M(x) and discrete moments of $\zeta(s)$. We can obtain this last sum through partial summation of $J_{-2}(T)$. In fact, it can be shown that the Riemann Hypothesis and the Gonek–Hejhal conjecture (1.34) implies the weak Merten's conjecture. Moreover, it seems likely that

$$\int_1^X \left(\frac{M(x)}{x}\right) \ dx \sim \sum_{\gamma > 0} \frac{2}{|\rho \zeta'(\rho)|^2} \log X.$$

In showing this, Ng [252] argues that

$$M(x) = O\left(\sqrt{x}(\log\log x)^{3/2}\right)$$

except on a set of finite logarithmic measure.

While we know that Merten's conjecture is false, thankfully this does not disprove the Riemann Hypothesis. The Riemann Hypothesis is equivalent to

$$M(x) = O\left(x^{1/2 + \varepsilon}\right)$$

for every $\varepsilon > 0$.

The best known unconditional upper bound is

$$M(x) = O\left(x \exp\left(-c(\log x)^{3/5} (\log\log x)^{-1/5}\right)\right)$$

for some constant c > 0 while the Riemann Hypothesis implies

$$M(x) = O\left(\sqrt{x} \exp\left(A\frac{\log x}{\log\log x}\right)\right)$$

for some constant A > 0. For Omega results, the best unconditional result is

$$M(x) = \Omega\left(x^{1/2}\right)$$

while if the Riemann Hypothesis is false,

$$M(x) = \Omega\left(x^{\theta}\right)$$

for some $\theta > 1/2$. Note the similarities between the error terms here and those in the Prime Number Theorem. For a summary of these results, see [251].

The question still remains about the true size of M(x). We first note that Ng [252] showed that a limiting distribution exists for

$$\phi(y) = e^{-y/2} M(e^{-y/2}),$$

assuming the Riemann Hypothesis and that $J_{-2}(T) \ll T$. Assuming results for $J_{-1}(T)$ and $J_{-2}(T)$, Ng conjectured

$$M(x) = \Omega_{\pm} \left(\sqrt{x} (\log \log \log x)^{5/4} \right).$$

Note that this power of 5/4 comes from the Gonek–Hejhal conjecture for k = -1/2. This conjecture was proved under the same assumptions on $J_{-1}(T)$ and $J_{-2}(T)$ and an effective version of the Linear Independence Hypothesis by Lamzouri [218].

Gonek (unpublished, with no specified value of the constant) and Ng [251] conjectured that

$$\liminf_{x \to \infty} \frac{M(x)}{\sqrt{x} (\log \log \log x)^{5/4}} = -\frac{8a}{5}, \text{ and } \limsup_{x \to \infty} \frac{M(x)}{\sqrt{x} (\log \log \log x)^{5/4}} = \frac{8a}{5},$$

where Ng (unpublished) conjectures that

$$a = \frac{1}{\sqrt{\pi}} e^{3\zeta'(-1) - \frac{11}{12}\log 2} \prod_{p} \left(1 - \frac{1}{p}\right)^{\frac{1}{4}} \sum_{m=0}^{\infty} \left(\frac{\Gamma(m-1/2)}{m!\Gamma(-1/2)}\right)^{2} p^{-m} \approx 0.16712\dots,$$

based on that coefficient from the discrete moment Conjecture 1.19 of Hughes, Keating and O'Connell with k = -1/2. Additionally, assuming this effective version of the Linear Independent Hypothesis, together the bounds on negative discrete moment for $J_{-1}(T)$ and $J_{-2}(T)$, Ng (upcoming paper) proved

$$\liminf_{x \to \infty} \frac{M(x)}{\sqrt{x} (\log \log \log x)^{5/4}} \le -\frac{8a}{5}, \text{ and } \limsup_{x \to \infty} \frac{M(x)}{\sqrt{x} (\log \log \log x)^{5/4}} \ge \frac{8a}{5}.$$

1.4.3.5 The Error in the Prime Number Theorem

We now return to the Prime Number Theorem, specifically to the error term in the Prime Number Theorem and come full circle with the initial motivation for this chapter. Define

$$E(x) = \frac{\psi(x) - x}{\sqrt{x}}$$

where $\psi(x)$ is defined in (1.6). Assuming the Riemann Hypothesis and the Linear Independence Hypothesis, Wintner [314] showed that there exists an absolutely continuous probability measure ν such that

$$\lim_{X \to \infty} \int_2^X f(E(x)) \ \frac{dx}{x} = \int_{-\infty}^\infty f(x) \ d\nu(x)$$

for all bounded continuous functions f on \mathbb{R} . By studying the tail of the limiting distribution ν , Montgomery [241] conjectured in 1980 that

$$\liminf_{x \to \infty} \frac{\psi(x) - x}{\sqrt{x} (\log \log \log x)^2} = -\frac{1}{2\pi}, \text{ and } \limsup_{x \to \infty} \frac{\psi(x) - x}{\sqrt{x} (\log \log \log x)^2} = \frac{1}{2\pi},$$

Lamzouri [218], assuming an effective version of the Linear Independence Hypothesis, proved

$$\liminf_{x \to \infty} \frac{\psi(x) - x}{\sqrt{x} (\log \log \log x)^2} \le -\frac{1}{2\pi}, \text{ and } \limsup_{x \to \infty} \frac{\psi(x) - x}{\sqrt{x} (\log \log \log x)^2} \ge \frac{1}{2\pi}.$$

-2 - -

Shanks' Conjecture

2.1 OVERVIEW OF CHAPTER

Shanks' conjecture [280], now a theorem, dates from 1961 and states that

 $\zeta'(\rho)$ is real and positive in the mean

as $\rho = \beta + i\gamma$ ranges over non-trivial zeros of the Riemann zeta function. This simple assertion is the basis for the thesis, including various proofs and generalisations of this problem.

More recently this assertion, again now a theorem proved in Chapter 3 and in [207, 186], has been generalised to higher derivatives, which we call the Generalised Shanks' Conjecture and states that

on average, $\zeta^{(n)}(\rho)$ is positive if n is odd, and negative if n is even.

Note that Shanks' conjecture was first conjectured, and later proved, in the usual order of events, but that we still call the established assertion Shanks' "conjecture". This nomenclature leads to an unusual situation where the extension is reasonably labelled the "Generalised Shanks' Conjecture" even though its first appearance in the literature was as a proven result!

In this first viewing of Shanks' conjecture and its generalisation, we begin by presenting a historical overview of the topic. Furthermore we will also give a simple heuristic that provides the leading term, including the sign, of the asymptotic formula for the average value of $\zeta^{(n)}(\rho)$. This heuristic follows almost immediately from an explicit formula known as the Landau–Gonek Theorem which we also discuss further.

Additionally, we will also use this heuristic to find the behaviour of the averages of other L-functions, including $1/\zeta'(s)$ over the non-trivial zeros of the Riemann zeta function, and our method applies directly to derivatives of Dirichlet L-functions over the non-trivial

zeros of the associated Dirichlet L-function in an identical proof (up to those obvious changes between the Riemann zeta function and Dirichlet L-functions).

Much of the work presented in this chapter has already appeared in print in [184], written jointly with Chris Hughes and Greg Martin. Notation and cross-references have been updated for this thesis.

2.2 HISTORICAL BACKGROUND OF SHANKS' CONJECTURE

Shanks first made his conjecture [280] when he was reviewing Haselgrove's tables [167] of numerical values of the Riemann zeta function. He plotted the graph of $t \mapsto \zeta(1/2 + it)$, copied below, and noticed that the way this curve approaches the origin, mainly through the third and fourth quadrants. This suggests the phase of $\zeta'(1/2 + i\gamma)$ is close to zero in the mean and thus $\zeta'(1/2 + i\gamma)$ is positive and real in the mean. This observation stands in contrast to the general behaviour of the function $\zeta'(1/2 + it)$, whose mean value tends quickly to 0.



FIG. 1.— ζ $(\frac{1}{2} + it)$ in the complex plane, for $0 \leq t \leq 30$.

This suggests the conjecture* that $\lim_{N\to\infty} (1/N) \sum_{1}^{N} \phi_n = 0$, that is, that $\zeta'(\frac{1}{2} + i\gamma_n)$

is positive real in the mean. Since violations against the "Law" are associated with exceptionally close zeros or exceptionally large values of Z(t), some extremes discovered by D. H. Lehmer are of interest, and are listed in Table IV:

Figure 2.1: Plot of $t \mapsto \zeta(1/2 + it)$, with the original conjecture below the graph.

We note that in Shanks' comments below his graph in Figure 2.1, he refers to violations of some 'Law'. This refers to Gram's Law, which states that given two consecutive Gram points g_n and g_{n+1} , we say that Gram's Law holds true for $[g_n, g_{n+1})$ if this Gram interval contains exactly one zero of $\zeta(\frac{1}{2} + it)$. A Gram point is one on the critical line where where $\zeta(\frac{1}{2} + ig_n)$ takes real, non-zero values. The Riemann–Siegel function $\theta(t)$, defined in (1.28), allows us to give another, equivalent definition of Gram points. These are precisely the points g_n for $n \ge -1$ with t > 7 such that

$$\theta(g_n) = n\pi.$$

This technique was initially used by Danish mathematician Gram [150] in 1903 to find the first 15 zeros of zeta.

Gram's Law is a bit of a misnomer - Gram himself didn't believe that there would always be exactly one zero in a Gram interval. The first time Gram's Law fails is the interval $[g_{125}, g_{126})$, which doesn't contain any zeros, while the next interval $[g_{126}, g_{127})$ contains two zeros. Titchmarsh [297] proved that Gram's Law fails infinitely many times. In the case of the original Gram's Law, it is still unknown whether it is true a positive proportion of the time, let alone infinitely many times. While this is an interesting topic of discussion in its own right, it takes us too far away from the theme of this thesis. We refer the interested reader to [162] for further information, as well as a probabilistic model of Gram's Law.

The other tibit of information that Shanks alluded to in his comments below Figure 2.1 are about exceptionally close zeros of the Riemann zeta function, called Lehmer pairs. Lehmer first found a close pair of zeros with imaginary parts 7005.06266... and 7005.10056... (the 6709th and 6710th zeros). More precisely, Csordas, Smith and Varga [91] define a Lehmer pair as having the property that their ordinates γ_n and γ_{n+1} obey the inequality

$$\frac{1}{(\gamma_n - \gamma_{n+1})^2} \ge C \sum_{m \notin \{n, n+1\}} \left(\frac{1}{(\gamma_m - \gamma_n)^2} + \frac{1}{(\gamma_m - \gamma_{n+1})^2} \right)$$

for some constant C > 5/4.

For each $t \in \mathbb{R}$, define the entire function

$$H_t(z) := \int_0^\infty e^{tu^2} \Phi(u) \cos(uz) \ du$$

where $\Phi(u)$ is a certain super-exponentially decaying function. Newman [38] showed there is a finite constant Λ , the De Bruijn–Newman constant, such that the zeros of $H_t(z)$ are all real precisely when $t \geq \Lambda$. The Riemann Hypothesis is equivalent to the fact that $\Lambda \leq 0$ and Newman conjectured $\Lambda \geq 0$. If there are infinitely many Lehmer pairs, then it would follow that $\Lambda \geq 0$. This has been shown unconditionally by Rodgers and Tao [272]. Newman is known to have said "This new conjecture is a quantitative version of the dictum that the Riemann hypothesis, if true, is only barely so."

We now return to Shanks' conjecture and the generalisation to higher derivatives. We present a different graph to the one Shanks created: Figure 2.2 shows a scatter plot of $\zeta'(\rho)$ at the first 100,000 non-trivial zeros of $\zeta(s)$. This new graph clearly displays symmetry along the real axis and a bias towards the positive side of the complex plane. Interestingly the graph appears to be a cardioid in shape, although we have set aside the search for a proof of why this should be the case for now.



Figure 2.2: Scatterplot of $\zeta'(\rho)$ for the first 100,000 zeros of the zeta function.

As mentioned in Section 1.4.3.1, in 1985 Conrey, Ghosh and Gonek [73] were looking for a straightforward proof of the fact that there are infinitely many simple zeros of $\zeta(s)$. To this end, they used the Cauchy–Schwarz inequality to write

$$\left|\sum_{0<\gamma\leq T}\zeta'(\rho)\right|^2\leq\sum_{0<\gamma\leq T}^*1\sum_{0<\gamma\leq T}|\zeta'(\rho)|^2$$

where the star on the middle sum denotes counting only the simple zeros. We denoted this sum as $N^*(T)$ previously. This sum only counts simple zeros as if there were any zeros of higher degree, the sum on the left-hand side would be zero and so wouldn't contribute to the answer. To complete the proof they used the asymptotic for the discrete second moment on the right-hand side, given in (1.32) and so only needed to find the leading order behaviour of the sum on the left-hand side. They showed

$$\sum_{0 < \gamma \le T} \zeta'(\rho) = \frac{T}{4\pi} (\log T)^2 + O(T \log T)$$
(2.1)

which coincidently proved Shanks' conjecture. Combining these two asymptotics with the Cauchy–Schwarz inequality above shows

$$N^*(T) \gg T \tag{2.2}$$

as $T \to \infty$, which completes their proof that there are infinitely many simple zeros of the Riemann zeta function.

The downside of the method to prove that infinitely many of the zeros of the Riemann zeta function are simple is that it is not strong enough to show that a positive proportion of the non-trivial zeros of $\zeta(s)$ are simple, whereas other methods are, as we have already discussed in Section 1.4.3.1.

It is now clear what the third type of moment is that we alluded to earlier when we first discussed continuous and discrete moments. Moments of the form found in (2.1) are the key point of interest in this thesis. Specifically, we will study moments of the form

$$\sum_{0 < \gamma \le T} \zeta^{(n)} \left(\rho \right)^k$$

as $T \to \infty$, for integer k, n, or more generally,

$$\sum_{0<\gamma\leq T}\zeta^{\left(n_{1}\right)}\left(\rho\right)\ldots\zeta^{\left(n_{k}\right)}\left(\rho\right)$$

as $T \to \infty$, for integer $n_1, \ldots n_k$, where $\rho = \beta + i\gamma$ denotes a non-trivial zero of the Riemann zeta function. The specific questions we ask have been given in the Preface to this thesis, with full details given in each chapter from this one onwards.

Before we describe the generalisation of Shanks' conjecture to higher derivatives, it is worth describing the ongoing research progress on the distribution of $\zeta'(\rho)$. In 1994 Fujii [126] found explicit lower order terms for the asymptotic formula (2.1) (he later corrected a slight error in the lowest-order coefficient in [128], when writing a paper where he combined $\zeta'(\rho)$ with Landau's Theorem, as seen in Chapter 4). The full correct asymptotic is given by

$$\sum_{0 < \gamma \le T} \zeta'(\rho) = \frac{T}{4\pi} \left(\log \frac{T}{2\pi} \right)^2 + (-1 + \gamma_0) \frac{T}{2\pi} \log \frac{T}{2\pi} + (1 - \gamma_0 - \gamma_0^2 - 3\gamma_1) \frac{T}{2\pi} + E(T)$$
(2.3)

where γ_0 and γ_1 are coefficients in the Laurent expansion of $\zeta(s)$ about s = 1, given by

$$\zeta(s) = \frac{1}{s-1} + \sum_{j=0}^{\infty} \frac{(-1)^j}{j!} \gamma_j (s-1)^j,$$

where the γ_j are the Stieltjes constants and γ_0 is Euler's constant.

Unconditional bounds for E(T) were given by Fujii, where

$$E(T) = O\left(Te^{-C\sqrt{\log T}}\right),$$

with C a positive constant. Under the Riemann Hypothesis, he showed

$$E(T) = O\left(T^{1/2}(\log T)^{7/2}\right).$$

The best known conditional error term is shown in Section 3.1.6 and can be found in [186], to be

$$E(T) = O\left(T^{1/2}(\log T)^{13/4}\right)$$

In 2021 Kobayashi [215] gave a similar result for Dirichlet L-functions, with the same leading order asymptotic term.

In 2010 Trudgian [299] gave an alternative proof of Shanks' conjecture, based on Shanks' observation of the connection between $\arg \zeta'(\rho)$ and Gram's Law. Although care needs to be taken in the definition of the argument (*a priori* it is only defined up to a multiple of 2π), Trudgian was able to show that

$$\sum_{0 < \gamma \le T} \arg \zeta'(\rho) \ll_{\varepsilon} T^{\varepsilon}$$

for every $\varepsilon > 0$.

We are now ready to explore the generalisation of Shanks' Conjecture to $\zeta^{(n)}(s)$, the n^{th} derivative of $\zeta(s)$, for every positive integer n. This extension, which we call the Generalised Shanks' Conjecture, states that $\zeta^{(n)}(\rho)$ is real and positive in the mean if n is even, while $\zeta^{(n)}(\rho)$ is real and negative in the mean if n is odd.

In 2011, using ideas similar to those in Conrey, Ghosh and Gonek's proof of Shanks' original conjecture, Kaptan, Karabulut and Yıldırım [207] found the main term for the summatory function of $\zeta^{(n)}(\rho)$. Specifically, they showed that

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) = \frac{(-1)^{n+1}}{n+1} \frac{T}{2\pi} (\log T)^{n+1} + O(T(\log T)^n)$$
(2.4)

which clearly implies the Generalised Shanks' Conjecture as stated above. They have also proved a similar result for derivatives of Dirichlet L-functions, with the same leading order asymptotic term.

2.2. HISTORICAL BACKGROUND OF SHANKS' CONJECTURE



Figure 2.3: Scatterplot of $\zeta''(\rho)$ for the first 100,000 zeros of the zeta function.

In 2022, together with Hughes [186], we investigated the Generalised Shanks' Conjecture and found lower order terms for the asymptotic formula (2.4), the proof of which will be given in Chapter 3. The result can be stated in the form

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n+1}$$

$$+ (-1)^{n+1} \sum_{k=1}^{n} \binom{n}{k} (-1)^{k} k! \left(-1 + \sum_{j=1}^{k} \frac{1}{j!} \gamma_{j}\right) \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-k} + n! A_{n} \frac{T}{2\pi} + E_{n}(T)$$

$$(2.5)$$

$$+ (-1)^{k+1} \sum_{k=0}^{\infty} {\binom{k}{(-1)^{k}k!}} {\binom{-1}{j=0}} \frac{-1}{j!} \gamma_j \frac{1}{2\pi} \left(\log \frac{1}{2\pi} \right)^{k+1} + n! A_n \frac{1}{2\pi} + E_n(1)$$
(2.6)

where the γ_j are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1, and where the A_j are the coefficients in the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1.

The unconditional error term is the same as given by Fujii in the first derivative case. If we assume the Riemann hypothesis, then we can improve his error term in the first derivative case, and in general, we have

$$E_n(T) = O\left(T^{1/2} (\log T)^{n+9/4}\right).$$

In Chapter 3 we are able to give an equivalent version of this result derived from a



Figure 2.4: Scatterplot of $\zeta'''(\rho)$ for the first 100,000 zeros of the zeta function.

result of Fujii's [126], which can be written unconditionally, for T sufficiently large, as

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) = n! \sum_{\ell=0}^{n+1} \frac{(-1)^{n-\ell+1}}{(n-\ell+1)!} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n-\ell+1} + n! \sum_{m=0}^{n} \sum_{\ell=0}^{n-m} \frac{(-1)^{n-m-\ell+1} \gamma_m}{(n-m-\ell)!m!} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n-m-\ell} + n! A_n \frac{T}{2\pi} + O\left(T e^{-C\sqrt{\log T}} \right),$$

where the coefficients γ_j and A_n are from the Laurent expansions of $\zeta(s)$ and $\zeta'(s)/\zeta(s)$ about s = 1, respectively.

In Chapter 6 we are able to give an integral form of this result, using the Ratios Conjecture. This can be written (under the Riemann Hypothesis) as

$$\sum_{0<\gamma\leq T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right) = \frac{n!}{2\pi} \int_{1}^{T} \left(A_n + \frac{(-1)^{n+1}L^{n+1}}{(n+1)!} + \sum_{m=0}^{n} \frac{(-1)^{m+1}L^m \gamma_{n-m}}{m!(n-m)!}\right) dt + O\left(T^{1/2+\varepsilon}\right),$$

where $L = \log t/2\pi$ and again the γ_j are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1, and where the A_j are the coefficients in the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1. Furthermore, we have also extended Fujii's combination (mentioned earlier) of $\zeta'(s)$ and Landau's Theorem to all derivatives of $\zeta(s)$, as will be shown in Chapter 4 and has appeared in [262].

In a recent paper, Benli, Elma and Ng [22] have generalised (2.5), giving a full asymptotic for sums of the form

$$\sum_{0 < \gamma \le T} \zeta(\rho + \alpha) X(\rho) Y(1 - \rho)$$

where α is a small complex number and X(s), Y(s) are Dirichlet polynomials. As an application of this result, they set $X(s) \equiv Y(s) \equiv 1$ and differentiate with respect to α *n*-times and set $\alpha = 0$ to reobtain (2.5).

All of the results we have described have relatively complicated proofs, and it is unclear that any of them provide an intuitive explanation for the alternating signs in the Generalised Shanks' Conjecture, or indeed of the positive sign in Shanks' original conjecture. We now present a simple heuristic that explains these signs, and indeed recovers the leading order term in the asymptotic formula (2.1) and (2.4).

2.3 HEURISTIC FOR THE GENERALISED SHANKS' CONJECTURE

2.3.1 Preliminary Lemmas

2.3.1.1 Dirichlet Series Approximations

We begin with some preliminaries concerning the Riemann zeta function, which we recall for $\Re(s) > 1$ can be written as the convergent Dirichlet series

$$\zeta(s) = \sum_{m=1}^{\infty} \frac{1}{m^s}.$$

Since Dirichlet series converge locally uniformly we may differentiate term by term; and since the derivative of $\frac{1}{m^s} = e^{-s \log m}$ is $(-\log m)m^{-s}$, we see that

$$\zeta^{(n)}(s) = (-1)^n \sum_{m=1}^{\infty} \frac{(\log m)^n}{m^s}$$

for $\Re(s) > 1$. This can be extended beyond the range of absolute convergence in a myriad of ways. We sketch the proof of one such method in the following lemma.

Lemma 2.1. For $s = \sigma + it$,

$$\zeta^{(n)}(s) = (-1)^n \sum_{m \le x} \frac{(\log m)^n}{m^s} - \frac{d^n}{ds^n} \left[\frac{x^{1-s}}{1-s} \right] + O\left(\frac{(\log x)^n}{x^{\sigma}} \right)$$

uniformly for $\sigma \ge \sigma_0 > 0$ and |t| < Cx where C is a positive constant less than 2π . Restricting ourselves to t > 2 and setting x = t we have

$$\zeta^{(n)}(s) = (-1)^n \sum_{m \le t} \frac{(\log m)^n}{m^s} + O(t^{-\sigma} (\log t)^n).$$
(2.7)

Proof. We sketch the proof for (2.7), using an adaptation of the method found in Titchmarsh [298, pages 74–77]. Equation (4.11.2) from that book shows that for $\sigma > 0$ and $N \in \mathbb{N}$,

$$\zeta(s) = \sum_{m=1}^{N} \frac{1}{m^s} - \frac{N^{1-s}}{1-s} + s \int_N^\infty \frac{[u] - u + 1/2}{u^{s+1}} \, du - \frac{1}{2} N^{-s}.$$

If we differentiate this equality n times with respect to s, we get

$$\begin{aligned} \zeta^{(n)}(s) &= (-1)^n \sum_{m=1}^N \frac{(\log m)^n}{m^s} - \frac{d^n}{ds^n} \left[\frac{N^{1-s}}{1-s} \right] - \frac{1}{2} (-\log N)^n N^{-s} \\ &+ n \int_N^\infty \frac{([u] - u + 1/2)(-\log u)^{n-1}}{u^{s+1}} \, du + s \int_N^\infty \frac{([u] - u + 1/2)(-\log u)^n}{u^{s+1}} \, du. \end{aligned}$$
(2.8)

When $\sigma > \sigma_0$ for a fixed $\sigma_0 > 0$, the last three terms can be uniformly estimated as

$$\begin{split} & -\frac{1}{2}(-\log N)^n N^{-s} \ll \frac{(\log N)^n}{N^{\sigma_0}} \\ & \int_N^\infty \frac{([u]-u+1/2)(-\log u)^{n-1}}{u^{s+1}} du \ll \frac{(\log N)^{n-1}}{N^{\sigma_0}} \\ & s \int_N^\infty \frac{([u]-u+1/2)(-\log u)^n}{u^{s+1}} du \ll |s| \frac{(\log N)^n}{N^{\sigma_0}} \end{split}$$

Moreover, if we define $g(m) = \frac{(\log m)^n}{m^{\sigma}}$ and $f(m) = -\frac{t \log m}{2\pi}$, we may apply [298, Lemma 4.10] to get

$$(-1)^{n} \sum_{t < m \le N} \frac{(\log m)^{n}}{m^{s}} = (-1)^{n} \sum_{t < m \le N} g(m) e^{2\pi i f(m)}$$
$$= (-1)^{n} \int_{t}^{N} g(u) e^{2\pi i f(u)} du + O\left(\frac{(\log t)^{n}}{t^{\sigma}}\right)$$
$$= \int_{t}^{N} \frac{(-\log u)^{n}}{u^{s}} du + O\left(\frac{(\log t)^{n}}{t^{\sigma}}\right).$$

On the other hand,

$$\int_t^N \frac{(-\log u)^n}{u^s} du = \frac{d^n}{ds^n} \left[\frac{N^{1-s}}{1-s} \right] - \frac{d^n}{ds^n} \left[\frac{t^{1-s}}{1-s} \right]$$

(to see this, note that for n = 0 the integral is directly calculable, and for n > 1 simply differentiate both sides with respect to s the appropriate number of times). If we restrict ourselves to t > 2, we obtain

$$\frac{d^n}{ds^n} \left[\frac{t^{1-s}}{1-s} \right] \ll \frac{t^{1-\sigma} (\log t)^n}{|1-\sigma+it|} \ll t^{-\sigma} (\log t)^n.$$

Therefore, plugging all this back into (2.8), we have shown that when $\sigma > \sigma_0$ for a fixed $\sigma_0 > 0$,

$$\zeta^{(n)}(s) = (-1)^n \sum_{m \le t} \frac{(\log m)^n}{m^s} + O\left(|s| \frac{(\log N)^n}{N^{\sigma_0}}\right) + O\left(\frac{(\log t)^n}{t^{\sigma}}\right).$$

Letting $N \to \infty$ completes the derivation of (2.7).

2.3.1.2 The Landau–Gonek Theorem

We begin this section by stating the Landau–Gonek Theorem, found in [146] which concerns sums of the form X^{ρ} over the non-trivial zeros of the Riemann zeta function. This is in essence an exponential sum. It was proved by Landau [219] for fixed X and made uniform by Gonek [144, 146]. This formula has a great many applications. For example, Gonek [146] reproved the leading order behaviour for the second discrete moment of the Riemann zeta function given in (1.31) using the Landau–Gonek Theorem, as well as a result of Montgomery's on simple zeros. Other applications will be seen throughout this thesis.

Theorem (Landau–Gonek). For X, T > 1,

$$\sum_{0 < \gamma \le T} X^{\rho} = -\frac{T}{2\pi} \Lambda(X) + O(X \log(2XT) \log \log(3X)) + O\left(\log X \min\left(T, \frac{X}{\langle X \rangle}\right)\right) + O\left(\log T \min\left(T, \frac{1}{\log X}\right)\right)$$
(2.9)

where $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, $\Lambda(X)$ denotes the von Mangoldt function, and where $\langle X \rangle$ denotes the distance from X to the nearest prime power other than X itself.

The large number of error terms are explained by Gonek [144]. They are due to the different behaviour of the sum in different ranges of X. He also notes that when $X = m \in \mathbb{N}$ and $T \gg m$, the last two error terms of (2.9) are absorbed by the first error term and so (2.9) becomes

$$\sum_{0 < \gamma \le T} m^{\rho} = -\frac{T}{2\pi} \Lambda(m) + O(m \log(2mT) \log \log(3m)).$$
(2.10)

The difference in formula (2.9) and formula (2.10) highlights the differences in behaviour of sums of this form when X is no longer an arbitrary positive real number but is instead a positive integer.

Fujii has generalised Gonek's result (assuming the Riemann Hypothesis) in [124] and [125] to find the subleading and subsubleading behaviour. This is given by

$$\sum_{0 < \gamma \le T} X^{1/2 + i\gamma} = -\frac{T}{2\pi} \Lambda(X) + X^{1/2 + iT} \left(\frac{\log \frac{T}{2\pi}}{2\pi i \log X} + \frac{1}{\pi} \arg \zeta \left(\frac{1}{2} + iT \right) \right) + O\left(\frac{\log T}{(\log \log T)^2} \right)$$
(2.11)

We state a corollary of the Landau–Gonek Theorem, found in [146]. Note that under the Riemann Hypothesis, for a non-trivial zero ρ of $\zeta(s)$, we have $\overline{\rho} = 1 - \rho$. Then

$$\sum_{0 < \gamma \le T} m^{1-\rho} = \sum_{0 < \gamma \le T} \overline{m^{\rho}} = \overline{\sum_{0 < \gamma \le T} m^{\rho}} = -\frac{T}{2\pi} \Lambda(m) + O(m \log(2mT) \log \log(3m))$$

since the right-hand side of (2.10) is real. Dividing through by m gives the following result.

Corollary 2.2 (Gonek). Under the Riemann Hypothesis, for $T > 1, m \in \mathbb{N}$ with $m \ge 2$ and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function $\zeta(s)$,

$$\sum_{0 < \gamma \le T} m^{-\rho} = -\frac{T}{2\pi} \frac{\Lambda(m)}{m} + O(\log(2mT)\log\log(3m)),$$
(2.12)

where $\Lambda(m)$ is the von Mangoldt function.

We now sketch where the main term in the Landau–Gonek Theorem comes from. By Cauchy's Theorem, we can write

$$\sum_{0 < \gamma \le T} X^{\rho} = \frac{1}{2\pi i} \int_{\mathcal{C}} \frac{\zeta'}{\zeta}(s) X^s \, ds$$

where C is a certain contour that encloses the non-trivial zeros of $\zeta(s)$ that goes past the 1-line, say c + i to c + iT, where $c = 1 + 1/\log T$. The other sides of the contour aren't relevant to our sketch, as it can be shown that the integrals along them only contribute to the error term.

The integral along this line gives

$$\frac{1}{2\pi} \int_{c+i}^{c+iT} \frac{\zeta'}{\zeta}(s) X^s \, ds = \frac{1}{2\pi} \int_1^T \left(-\sum_{n=2}^\infty \frac{\Lambda(n)}{n^{c+it}} \right) X^{c+it} \, ds$$
$$= -\sum_{n=2}^\infty \Lambda(n) \left(\frac{X}{n} \right)^c \frac{1}{2\pi} \int_1^T \left(\frac{X}{n} \right)^{it} \, dt$$

since we can expand the logarithmic derivative of $\zeta(s)$ as a Dirichlet series as c > 1. The diagonal terms (those where X = n) contribute to the main term, again with the rest of the terms contributing to the error terms. The right-hand side contributes

$$-\sum_{n=2}^{\infty} \Lambda(n) \left(\frac{X}{n}\right)^c \frac{1}{2\pi} \int_1^T \left(\frac{X}{n}\right)^{it} dt = -\frac{T}{2\pi} \Lambda(X) + \mathcal{E},$$

where \mathcal{E} is some error term, and so we can see where the Landau–Gonek Theorem should reasonably come from. That is, we have sketched that

$$\sum_{0<\gamma\leq T}X^{\rho}=-\frac{T}{2\pi}\Lambda(X)+\mathcal{E}$$

where \mathcal{E} is some error term that Landau and Gonek considered and found explicitly.

Other generalisations of the Landau–Gonek Theorem can be found. For example, in the case of a Dirichlet L-function, we could follow the above method to show

$$\sum_{0 < \gamma_{\chi} \le T} X^{\rho_{\chi}} = -\frac{T}{2\pi} \Lambda(X) \chi(X) + \mathcal{E}$$

where the sum is over the non-trivial zeros $\rho_{\chi} = \beta_{\chi} + \gamma_{\chi}$ of a Dirichlet *L*-function, and where we can derive a similar error term to that in the Landau–Gonek Theorem, rather than leaving it implicit by just writing \mathcal{E} . Similarly, under the Generalised Riemann Hypothesis, we could show for an integer $m \geq 2$ that

$$\sum_{0 < \gamma_{\chi} \le T} m^{-\rho_{\chi}} = -\frac{T}{2\pi} \frac{\Lambda(m)\overline{\chi}(m)}{m} + \mathcal{E}.$$

2.3.2 The Heuristic

We now state the short heuristic argument which yields the Generalised Shanks' Conjecture as stated in Section 2.1. We ignore error terms on the whole so the equal signs are not 'true' equalities but show the general argument. (See Chapter 3 on how to derive the rigorous result.)

We note that while the heuristic requires the assumption of the Riemann Hypothesis (as we will use Corollary 2.2), the general results with a full, rigorous proof of (2.1) and (2.4) do not, except for improvements in the error term.

Using the approximate formula (2.7) for $\zeta(s)$ and Corollary 2.2, we may write for $n \ge 1$,

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) \approx (-1)^n \sum_{0 < \gamma \le T} \sum_{m \le \gamma} (\log m)^n m^{-\rho}$$

$$= (-1)^n \sum_{m \le T} (\log m)^n \sum_{m < \gamma \le T} m^{-\rho}$$

$$\approx (-1)^{n+1} \frac{T}{2\pi} \sum_{m \le T} \frac{(\log m)^n \Lambda(m)}{m} - (-1)^{n+1} \frac{1}{2\pi} \sum_{m \le T} (\log m)^n \Lambda(m).$$
(2.13)

To finish the heuristic, we need to sum the two series in the last line. By Chebyshev's Theorem [5],

$$C(x) = \sum_{m \le x} \frac{\Lambda(m)}{m} = \log x + O(1)$$

so by partial summation, we have

$$\sum_{m \le T} \frac{(\log m)^n \Lambda(m)}{m} = C(T)(\log T)^n - n \int_1^T \frac{C(x)(\log x)^{n-1}}{x} \, dx$$
$$= \frac{1}{n+1} (\log T)^{n+1} + O((\log T)^n).$$

and similarly

$$\sum_{m \le T} (\log m)^n \Lambda(m) = O(T(\log T)^n).$$

Combining this with our argument above, we have

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{T}{2\pi} \frac{1}{n+1} (\log T)^{n+1} + O(T(\log T)^n),$$

as $T \to \infty$, which is the leading order the asymptotic result for the Generalised Shanks' Conjecture, as given in (2.4).

Remark. Note that this result assumes $n \ge 1$. When n = 0 the left-hand side is trivially zero, whilst the right-hand side, as written, is not. This is explained by noticing that when $n \ge 1$ the term when m = 1 in the sum (2.13) is not present, since $(\log 1)^n = 0$. However, if n = 0 this term is present and is not accounted for in the Landau–Gonek Theorem, which holds for $m \ge 2$ only. The m = 1 term in that case clearly contributes N(T), which perfectly cancels the $-\frac{T}{2\pi} \log T + O(T)$ term coming from the calculation given above.

Remark. This argument can be applied *mutatis mutandis* for Dirichlet *L*-functions, yielding exactly the same leading order behaviour as for the Riemann zeta function. This result was already known [207, 215].

2.3.2.1 The Error Terms

In the previous section in the last part of equation (2.13), we ignored the error terms coming from Corollary 2.2. Including them one can see that they contribute

$$\ll \sum_{m \le T} (\log m)^n \log \log(3m) \log T \ll T (\log T)^{n+1} \log \log T$$

which dominates the main term! However, these are worst-case point-wise estimates and take no account of any potential cancellation when averaged, so it is likely that when summed over m the true error is smaller and the main term is correct. (Indeed that is what is proved, by different methods, in [73, 207] for n = 1 and $n \ge 2$ respectively.)

The point is that whilst this method must remain a heuristic, it is a much quicker approach to find the main term in the overage of $\zeta^{(n)}(\rho)$ and gives some sense of why the Generalised Shanks' Conjecture holds true, that is, why the mean of $\zeta^{(n)}(\rho)$ is real and positive/negative.

Remark. In Chapter 3 we prove that the true value of the mean of $\zeta^{(n)}(\rho)$ (that is, the main term and all the lower order terms up to a power saving) is given by

$$\sum_{\substack{m,r\\n,r\leq\frac{T}{2\pi}}}\Lambda(r)(\log r)^n$$

100

which shows that the heuristic in the previous section can only yield the main term since the last line of equation (2.13) differs from the true value by $O(T(\log T)^n)$. This result is found with full details in Chapter 3 and in [186].

2.4 SHANKS' CONJECTURE FOR THE RECIPROCAL

As we have seen in Sections 1.3.6 and 1.4.1, negative moments are also of interest. It is natural to ask if there is an analogue to Shanks' conjecture for negative moments of the derivatives of $\zeta(s)$. Garaev and Sankaranarayanan in [134] have proved a negative moment of $\zeta'(\rho)$ (under the assumption of simple zeros) which we can rederive with the heuristic. Note that if we could find a proof that didn't assume the simple zeros hypothesis, then proving the following asymptotic would imply that all zeros of the Riemann zeta function are simple (else the right-hand side of the asymptotic would blow up as $T \to \infty$). We also note that the following result can be found [195, 244, 236] in under the additional assumption of the Riemann Hypothesis.

Theorem 2.3 (Garaev–Sankaranarayanan). Assume all the non-trivial zeros of $\zeta(s)$ are simple. Then

$$\sum_{0 < \gamma \le T} \frac{1}{\zeta'(\rho)} \sim \frac{T}{2\pi}$$
(2.14)

as $T \to \infty$.



Figure 2.5: The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \leq T} \frac{1}{\zeta'(\rho)} - \frac{T}{2\pi}$ plotted over the first 100,000 zeros.

Consider Theorem 2.3 above, together with the Figure 2.5. Clearly if we were to state a Shanks' style conjecture (although again, it is already a theorem), we would claim

 $1/\zeta'(\rho)$ is real and positive in the mean.

However it is also clear that there is some constant, subleading real term from the graph. We make the following conjecture (although believe we could remove the assumptions of simple zeros and the Riemann Hypothesis). As before, a proof without assuming the simple zeros hypothesis would imply that all the non-trivial zeros of the Riemann zeta function are simple.

Conjecture. Assume all the non-trivial zeros of $\zeta(s)$ are simple and the Riemann Hypothesis is true. Then

$$\sum_{0 < \gamma \le T} \frac{1}{\zeta'(\rho)} = \frac{T}{2\pi} + C + O\left(T^{1/2+\varepsilon}\right)$$
(2.15)

as $T \to \infty$.

We have suggested an error term of $O(T^{1/2+\varepsilon})$ to be consistent with other zeta function conjectures, but the numerics suggest the true size of the error term should be far smaller. The constant C however, is a different story. Typically, the lower order terms of a moments problem involves some arithmetic information, so a safe bet would be to say that $C = f(\gamma_0)$. At a numerical value of roughly -1.6, we suggest that $C = -1 - \gamma_0$ but this guess is based solely on a numerical approximation. Clearly the constant should be contained within the error term, but we can see a clear systematic negative bias from Figure 2.5 and so we include the constant explicitly in this conjecture. Investigating this error term and the value of the missing constant would be of interest.

We can also compare this result with a conjecture of Gonek's [145], stated in (1.35). He conjectured the following result.

Conjecture 2.4 (Gonek). Assume all the non-trivial zeros of $\zeta(s)$ are simple and the Riemann Hypothesis is true. Then

$$J_{-2}(T) = \sum_{0 < \gamma \le T} \frac{1}{|\zeta'(\rho)|^2} \sim \frac{3}{\pi^3} T$$

as $T \to \infty$.

Again, we note that if we could find a proof that didn't assume the simple zeros hypothesis, then proving the following asymptotic would imply that all zeros of the Riemann zeta function are simple.

Recall also that Milinovich and Ng [236] have shown a lower bound for this result, correct up to a factor of a half, stated in (1.36).

Interestingly, as a final comment before we use the heuristic to derive Theorem 2.3, we note that both Theorem 2.3 and Conjecture 2.4 are of the same order of magnitude.

2.4.1 Negative Shanks' Conjecture via the Heuristic

Given

$$\zeta'(\rho) = \lim_{x \to 0} \frac{\zeta(\rho + x) - \zeta(\rho)}{x} = \lim_{x \to 0} \frac{\zeta(\rho + x)}{x},$$

it is plausible (but certainly not rigorously proven, since we have ignored all issues with rearranging the, at best, conditionally convergent sums) that

$$\sum_{0 < \gamma \le T} \frac{1}{\zeta'(\rho)} = \sum_{0 < \gamma \le T} \lim_{x \to 0} \frac{x}{\zeta(\rho + x)}$$
$$= \lim_{x \to 0+} x \sum_{0 < \gamma \le T} \sum_{m=1}^{\infty} \frac{\mu(m)}{m^{\rho + x}}$$
$$= \lim_{x \to 0^+} x \left(N(T) + \sum_{m=2}^{\infty} \frac{\mu(m)}{m^x} \sum_{0 < \gamma \le T} m^{-\rho} \right)$$

where the second line is conditionally convergent for $\Re(x) > 0$ if the Riemann Hypothesis holds, and the N(T) term comes from the m = 1 term in that sum. Then applying the Landau–Gonek Theorem (again ignoring the error terms) to the second sum in the previous equation, this equals

$$\lim_{x \to 0^+} x \left(\sum_{m=2}^{\infty} \frac{\mu(m)}{m^x} \sum_{0 < \gamma \le T} m^{-\rho} \right) \approx -\frac{T}{2\pi} \lim_{x \to 0^+} x \sum_{n=1}^{\infty} \frac{\mu(n)\Lambda(n)}{n^{1+x}}$$
$$= \frac{T}{2\pi} \lim_{x \to 0^+} x \sum_{p \text{ prime}} \frac{\log p}{p^{1+x}}$$

where last line follows since $\mu(n)\Lambda(n) = -\log(p)$ if n = p a prime, and zero otherwise.

By the Prime Number Theorem we have

$$\lim_{x \to 0^+} x \sum_{p} \frac{\log p}{p^{1+x}} = \lim_{x \to 0^+} x \int_2^\infty \frac{\log y}{y^{1+x}} \frac{dy}{\log y} = 1$$

giving us the heuristic derivation of Theorem 2.3.

Remark. This argument can be applied *mutatis mutandis* for Dirichlet *L*-functions, yielding exactly the same leading order behaviour as for the Riemann zeta function. This result

doesn't seem to already be known, but a similar proof to that of Garaev and Sankaranarayanan should follow (under the assumptions of simple zeros and the Generalised Riemann Hypothesis). We state this result in full for future interest.

Conjecture 2.5. Assume the Generalised Riemann Hypothesis, and that all the zeros of a Dirichlet L-function $L(s,\chi)$ are simple. Then for $\rho_{\chi} = \beta_{\chi} + i\gamma_{\chi}$ a non-trivial zero of $L(s,\chi)$, we have

$$\sum_{0 < \gamma_{\chi} \le T} \frac{1}{L'(\rho_{\chi}, \chi)} \sim \frac{T}{2\pi}$$

The Generalised Shanks' Conjecture

In Chapter 2 we introduced Shanks' conjecture, which states for $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, that

 $\zeta'(\rho)$ is real and positive in the mean.

While Conrey, Ghosh and Gonek [75] established the truth of this conjecture to leading order, Fujii [126, 128] extended their result by showing for sufficiently large T that

$$\sum_{0 < \gamma \le T} \zeta'(\rho) = \frac{T}{4\pi} \left(\log \frac{T}{2\pi} \right)^2 + (-1 + \gamma_0) \frac{T}{2\pi} \log \frac{T}{2\pi} + (1 - \gamma_0 - \gamma_0^2 - 3\gamma_1) \frac{T}{2\pi} + E(T)$$
(3.1)

where γ_0 and γ_1 are coefficients in the Laurent expansion of $\zeta(s)$ about s = 1 and E(T) is a particular error term, depending on whether we assume the Riemann hypothesis or not. The unconditional error term is the same as that in the Prime Number Theorem (1.7) and the conditional error term given when assuming the Riemann hypothesis is given by

$$E(T) = O\left(T^{1/2}(\log T)^{7/2}\right).$$

The aim of Section 3.1 is to do the same for the Kaptan, Karabulut and Yıldırım [207] result for sums of $\zeta^{(n)}(\rho)$, that is to find lower order terms for

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) = \frac{(-1)^{n+1}}{n+1} \frac{T}{2\pi} (\log T)^{n+1} + O\left(T(\log T)^n\right).$$

This comes in the form of a polynomial in log T of degree n + 1, with a similar error term to that of Fujii's. Their proof verifies the Generalised Shanks' Conjecture, which states for $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, that

 $\zeta^{(n)}(\rho)$ is real and positive/negative in the mean if n is odd/even.

In Section 3.1.6 we show how we can improve the error term in Fujii's result, as well as how to improve the initial error term that we gave Hughes and Pearce-Crump [186]. This improvement comes in the form of a saving of a fourth-root of a logarithm in T. In Section 3.2 we show how we can derive an equivalent asymptotic to the one given in Theorem 3.1 using a shifted asymptotic formula of Fujii's.

We end this chapter by briefly discussing the Generalised Shanks' Conjecture over short intervals. This means that we instead sum over the ordinates of zeros of size between T and T + H, where $T^{1/2+\varepsilon} \ll H \ll T$. This result to leading order is as expected, and shows for such H, for sufficiently large T,

$$\sum_{T < \gamma \le T+H} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{H}{2\pi} \left(\log T\right)^{n+1} + O\left(H(\log T)^n\right) + O\left(T^{1/2+\varepsilon}\right).$$

The work presented in Section 3.1 has already appeared in print in [186]. Notation and cross-references have been updated for this thesis, as well as an improved error term as previously mentioned. This improved error term written in this chapter takes the place of what was given in the original paper.

3.1 THE GENERALISED SHANKS' CONJECTURE

In this section we prove the following result.

Theorem 3.1. For $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, and where $\zeta^{(n)}(s)$ denotes the n^{th} derivative of $\zeta(s)$, we have for T sufficiently large,

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n+1} + (-1)^{n+1} \sum_{k=0}^{n} \binom{n}{k} (-1)^{k} k! \left(-1 + \sum_{j=0}^{k} \frac{1}{j!} \gamma_{j}\right) \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-k} + n! A_{n} \frac{T}{2\pi} + E_{n}(T)$$

$$(3.2)$$

where the γ_j are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1, given by

$$\zeta(s) = \frac{1}{s-1} + \sum_{j=0}^{\infty} \frac{(-1)^j}{j!} \gamma_j (s-1)^j,$$

and where the A_j are the coefficients in the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1, given by

$$A_{n} = \begin{cases} \gamma_{0} & \text{if } n = 0\\ \frac{(-1)^{n}(n+1)}{n!} \gamma_{n} - \sum_{j=0}^{n-1} A_{j} \frac{(-1)^{n-1-j}}{(n-1-j)!} \gamma_{n-1-j} & \text{otherwise.} \end{cases}$$
(3.3)

The error term is given by

$$E_n(T) = O\left(T e^{-C\sqrt{\log T}}\right)$$

with C is a positive constant. If we assume the Riemann hypothesis, then

$$E_n(T) = O\left(T^{1/2}(\log T)^{n+9/4}\right).$$

The expression for A_n given in (3.3) is found in [197] with equivalent forms in [230, 35]. *Remark.* Two simpler but equivalent representations of this theorem can be found in Theorem 3.17 and in Section 6.2.4.1, both of which are derived in different ways.

3.1.1 Overview of the Proof

In this section we give a brief outline of the proof, explaining the method we will follow to prove Theorem 3.1.

In Subsection 3.1.2, we state some preliminary lemmas that we will use throughout this section.

In Subsection 3.1.3, we begin by considering the integral S given by

$$S = \frac{1}{2\pi i} \int_R \frac{\zeta'}{\zeta}(s) \zeta^{(n)}(s) \, ds$$

where R is the positively oriented rectangular contour with vertices c + i, c + iT, 1 - c + iT, 1 - c + iT, 1 - c + i with $c = 1 + \frac{1}{\log T}$. The non-trivial zeros of $\zeta(s)$ up to a height T are contained within R and so by Cauchy's Theorem the integral represents the summation

$$S = \sum_{0 < \gamma \le T} \zeta^{(n)}(\rho)$$

in question. We then show that the contribution to the integral from the bottom, top and right-hand side of the contour is contained within the error term of the theorem, so the main contribution comes from the integral along the left-hand side of the integral.

In Subsection 3.1.4, we evaluate this part of the contour integral. Through this, by Lemma 3.11 we link the integral to a summation

$$(-1)^n \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^n$$

which is then the main object in question.

In Subsection 3.1.5, Lemma 3.12 we use Perron's formula to evaluate this sum as a complex integral up a vertical line just to the right of the critical strip.

In Subsection 3.1.6, we will evaluate that complex integral and show that

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + E_n(Y)$$

where $Y = \frac{T}{2\pi}$ and V is given explicitly to optimise the error term $E_n(Y)$, which we can describe both unconditionally and under the assumption of the Riemann hypothesis.

Finally, in Subsection 3.1.7, we will evaluate the residue at s = 1 to find the asymptotic expansion described in Theorem 3.1 by considering the Laurent expansions around the pole s = 1 of the terms in the above integrand.

Combining all of these steps together gives the result.

We finish the section by giving some examples in Subsection 3.1.8. Firstly, we can recover the asymptotic in (3.1) by specialising to the case n = 1, and in particular recover Shanks' Conjecture. We then give the first new case where n = 2, as well as providing some good numerical evidence for the strength of the result.

3.1.2 Preliminary Lemmas

We start by stating the strong form of the convexity bound for the zeta function and the n^{th} derivatives of the zeta function. The case n = 0 can be found in Ivić [198, Ch. I, Sect. 1.5], while the cases for all other n can be derived from this using Cauchy's theorem on derivatives of analytic functions on $\zeta(s)$ in a small disc of radius $\frac{1}{\log t}$ centred at $s = \sigma + it$.

Lemma 3.2. For $t \ge t_0 > 0$ uniformly in σ ,

$$\zeta^{(n)}(\sigma + it) \ll \begin{cases} t^{1/2 - \sigma} (\log t)^{n+1} & \text{if } \sigma \le 0\\ t^{(1-\sigma)/2} (\log t)^{n+1} & \text{if } 0 \le \sigma \le 1\\ (\log t)^{n+1} & \text{if } \sigma \ge 1. \end{cases}$$

The following result follows from Gonek [143, Sect. 2, p. 126].

Lemma 3.3. If T is such that $|T - \gamma| \gg \frac{1}{\log T}$ for any ordinate γ , uniformly for $-1 \leq \sigma \leq 2$ we have

$$\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \ll (\log T)^{n+2}.$$
(3.4)

To obtain the form of the functional equation for $\zeta^{(n)}(1-s)$ that we will need, we will use the following result from Gonek [143], which is his Lemma 6.

Lemma 3.4. For σ fixed, $n \ge 0$ and $t \ge 1$ we have

$$\chi^{(n)}(1-s) = (-1)^n \chi(1-s) \left(\log \frac{t}{2\pi}\right)^n + O(t^{\sigma-3/2}(\log t)^{n-1}).$$

Lemma 3.5. For $\sigma \ge 1$ and $t \ge 1$

$$\zeta^{(n)}(1-s) = (-1)^n \chi(1-s) \sum_{k=0}^n \binom{n}{k} \left(\log \frac{t}{2\pi}\right)^{n-k} \zeta^{(k)}(s) + O(t^{\sigma-3/2}(\log t)^n).$$

Proof. By the functional equation (1.1), and using the Leibniz product rule

$$(-1)^{n}\zeta^{(n)}(1-s) = \sum_{k=0}^{n} \binom{n}{k} (-1)^{n-k}\chi^{(n-k)}(1-s)\zeta^{(k)}(s).$$

108
3.1. The Generalised Shanks' Conjecture

Using Lemma 3.4 and Lemma 3.2 to bound $\zeta^{(k)}(s)$ for $\sigma \geq 1$, this equals

$$\begin{split} \sum_{k=0}^{n} \binom{n}{k} \chi(1-s) \left(\log \frac{t}{2\pi} \right)^{n-k} \zeta^{(k)}(s) + O\left(t^{\sigma-3/2} \sum_{k=0}^{n} (\log t)^{n-k-1} (\log t)^{k+1} \right) \\ &= \sum_{k=0}^{n} \binom{n}{k} \chi(1-s) \left(\log \frac{t}{2\pi} \right)^{n-k} \zeta^{(k)}(s) + O\left(t^{\sigma-3/2} (\log t)^{n} \right). \end{split}$$

Dividing through by the factor of $(-1)^n$ gives the result.

As another application of Lemma 6 from [143] we can rewrite the logarithmic derivative of the functional equation for $\zeta(s)$ as

$$\frac{\zeta'}{\zeta}(1-s) = -\log\frac{t}{2\pi} - \frac{\zeta'}{\zeta}(s) + O\left(t^{\sigma-3/2}\right) \tag{3.5}$$

valid for any fixed σ , which we will also need later.

3.1.3 Beginning the proof

Throughout we assume T is sufficiently large and satisfies $|T - \gamma| \gg \frac{1}{\log T}$, where γ is the imaginary part of any zero ρ . (This constraint simplifies the arguments, but has no effect on the resulting expressions or final theorem.) Consider the integral

$$S = \frac{1}{2\pi i} \int_R \frac{\zeta'}{\zeta}(s) \zeta^{(n)}(s) \ ds \tag{3.6}$$

where the contour R is the positively oriented rectangular contour with vertices c + i, c + iT, 1 - c + iT, 1 - c + i with $c = 1 + \frac{1}{\log T}$.

By Cauchy's Residue Theorem,

$$S = \sum_{0 < \gamma \le T} \zeta^{(n)}(\rho)$$

We need to evaluate S in another way to determine the behaviour of $\sum_{0 < \gamma \leq T} \zeta^{(n)}(\rho)$. To do this, we begin by splitting the integral S along each part of the contour, so

$$S = \frac{1}{2\pi i} \left(\int_{c+i}^{c+iT} + \int_{c+iT}^{1-c+iT} + \int_{1-c+iT}^{1-c+i} + \int_{1-c+i}^{c+i} \right) \frac{\zeta'}{\zeta}(s) \zeta^{(n)}(s) \ ds$$
$$= S^R + S^T + S^L + S^B,$$

say. We will first bound S^B, S^T, S^R trivially within error terms as follows. Our main aim will then be to evaluate S^L which we will do in the next section.

Lemma 3.6. The integral along the bottom of the contour is $S^B = O(1)$.

Proof. This follows as the integral is of finite length with a bounded integrand.

Lemma 3.7. The integral along the top of the contour is $S^T = O\left(T^{1/2}(\log T)^{n+2}\right)$.

Proof. By the convexity result from Lemma 3.2 and using Lemma 3.3 to bound the logarithmic derivative, we may write

$$\begin{split} S^T \ll & \int_{1-c}^c \left| \frac{\zeta'}{\zeta} (\sigma + iT) \zeta^{(n)} (\sigma + iT) \right| \, d\sigma \\ \ll & (\log T)^2 \left\{ \int_{1-c}^0 + \int_0^1 + \int_1^c \right\} \left| \zeta^{(n)} (\sigma + iT) \right| \, d\sigma \\ \ll & (\log T)^2 \left\{ \int_{1-c}^0 T^{1/2-\sigma} (\log T)^{n+1} \, d\sigma + \int_0^1 T^{(1-\sigma)/2} (\log T)^{n+1} \, d\sigma + \int_1^c (\log T)^{n+1} \, d\sigma \right\} \\ \ll & T^{1/2} (\log T)^{n+2} \end{split}$$

since $c = 1 + \frac{1}{\log T}$.

Lemma 3.8. The integral along the right-hand side of the contour is $S^R = O((\log T)^{n+3})$.

Proof. Along the right-hand side of the contour we need to evaluate

$$S^{R} = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{\zeta'}{\zeta}(s) \zeta^{(n)}(s) \, ds.$$

Writing $\frac{\zeta'}{\zeta}(s)$ and $\zeta^{(n)}(s)$ in terms of their Dirichlet series, we have

$$\frac{\zeta'}{\zeta}(s) = -\sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^s} \text{ and } \zeta^{(n)}(s) = (-1)^n \sum_{m=1}^{\infty} \frac{(\log m)^n}{m^s}$$
(3.7)

where $\Lambda(r)$ is the von Mangoldt function given by

$$\Lambda(r) = \begin{cases} \log p & \text{if } r = p^k \text{ for some prime } p \text{ and some integer } k \ge 1, \\ 0 & \text{otherwise.} \end{cases}$$

Substituting into S^R we have

$$\begin{split} S^{R} &= \frac{1}{2\pi i} \int_{1}^{T} \left(-\sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^{c+it}} \right) \left((-1)^{n} \sum_{m=1}^{\infty} \frac{(\log m)^{n}}{m^{c+it}} \right) \ idt \\ &\ll \sum_{r=1}^{\infty} \sum_{m=1}^{\infty} \frac{\Lambda(r)(\log m)^{n}}{r^{c}m^{c}} \int_{1}^{T} (rm)^{-it} \ dt \\ &\ll \sum_{r=1}^{\infty} \sum_{m=1}^{\infty} \frac{\Lambda(r)(\log m)^{n}}{r^{c}m^{c}\log rm} \\ &\ll \frac{\zeta'}{\zeta} (c) \zeta^{(n)}(c) \\ &\ll (\log T)^{n+3}. \end{split}$$

where the final line follows from Lemmas 3.2 and 3.3.

3.1. The Generalised Shanks' Conjecture

Since S^B, S^T, S^R are all within an error term of $O\left(T^{1/2}(\log T)^{n+2}\right)$, only the integral over the left-hand side of the contour will contribute in any meaningful way. Observe that

$$S^{L} = \frac{1}{2\pi i} \int_{1-c+iT}^{1-c+i} \frac{\zeta'}{\zeta}(s)\zeta^{(n)}(s) \, ds = -\frac{1}{2\pi i} \int_{c-iT}^{c-i} \frac{\zeta'}{\zeta}(1-s)\zeta^{(n)}(1-s) \, ds = -\overline{I},$$

where

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{\zeta'}{\zeta} (1-s)\zeta^{(n)}(1-s) \, ds.$$
(3.8)

3.1.4 Evaluating I

Our main aim then for this section is to evaluate I. To do this we want to write I in such a way that we can apply Gonek's Lemma 5 from [143] which we will state below when we use it later in this section.

Lemma 3.9. We can write the integral I given in (3.8) as

$$I = \frac{(-1)^{n+1}}{2\pi} \int_{1}^{T} \chi(1-c-it) \left[\sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k+1} \sum_{m=1}^{\infty} \frac{(-1)^{k} (\log m)^{k}}{m^{c+it}} + \sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^{\infty} \frac{(-1)^{k+1} (\log m)^{k}}{m^{c+it}} \sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^{c+it}} \right] dt + O\left(T^{1/2} (\log T)^{n+2} \right)$$

Proof. Substituting the results from Lemma 3.5 and (3.5) into I gives

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(-\log\frac{t}{2\pi} + \frac{\zeta'}{\zeta}(s) + O(t^{c-3/2}) \right) \\ \times \left((-1)^n \chi(1-s) \sum_{k=0}^n \binom{n}{k} \left(\log\frac{t}{2\pi} \right)^{n-k} \zeta^{(k)}(s) + O\left(t^{c-3/2} (\log t)^n\right) \right) \, ds.$$

Writing s = c + it, expanding the bracket in the integral and simplifying the error term gives

$$I = \frac{(-1)^{n+1}}{2\pi} \int_{1}^{T} \chi(1-c-it) \left[\sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k+1} \zeta^{(k)}(c+it) + \sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k} \zeta^{(k)}(c+it) \frac{\zeta'}{\zeta}(c+it) \right] dt + O\left(T^{1/2} (\log T)^{n+2} \right).$$
(3.9)

where we have used $\chi(1-c-it) \ll t^{1/2-c}$, and $\zeta^{(k)}(c+it) \ll (\log t)^{k+1}$, and $\frac{\zeta'}{\zeta}(c+it) \ll (\log t)^2$. Writing $\zeta^{(k)}(s)$ and $\frac{\zeta'}{\zeta}(s)$ as their Dirichlet series (3.7) and substituting into I gives the result.

We will now require the use of Lemma 5 from Gonek [143, Sect. 4, p. 131]. As it is such an important result for our proof, we state it here, with the phrasing adapted to suit our needs.

Lemma 3.10 (Gonek). Let $\{b_m\}_{m=1}^{\infty}$ be a sequence of complex numbers such that for any $\varepsilon > 0, b_m \ll m^{\varepsilon}$. Let c > 1 be as before and let k be a non-negative integer. Then for T sufficiently large,

$$\frac{1}{2\pi} \int_{1}^{T} \left(\sum_{m=1}^{\infty} b_m m^{-c-it} \right) \chi(1-c-it) \left(\log \frac{t}{2\pi} \right)^k \, dt = \sum_{m \le \frac{T}{2\pi}} b_m (\log m)^k + O\left(T^{c-1/2} (\log T)^k \right).$$

We can finally simplify I to get a single sum that we will work on evaluating in the next section.

Lemma 3.11. The integral I can be written as

$$I = (-1)^n \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^n + O(T^{1/2} (\log T)^{n+2})$$

where the sum is taken over all integers m and r such that $mr \leq T/2\pi$.

Proof. Before we apply Lemma 3.10, we will split the integral I given in (3.9) substituting the Dirichlet series to simplify the argument slightly. To do this, we write

$$\begin{split} I &= \frac{(-1)^{n+1}}{2\pi} \int_{1}^{T} \chi (1-c-it) \sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k+1} \sum_{m=1}^{\infty} \frac{(-1)^{k} (\log m)^{k}}{m^{c+it}} \, dt \\ &+ \frac{(-1)^{n+1}}{2\pi} \int_{1}^{T} \chi (1-c-it) \sum_{k=0}^{n} \binom{n}{k} \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^{\infty} \frac{(-1)^{k+1} (\log m)^{k}}{m^{c+it}} \sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^{c+it}} \, dt \\ &+ O \left(T^{1/2} (\log T)^{n+2} \right) \\ &= J_{1} + J_{2} + O \left(T^{1/2} (\log T)^{n+2} \right), \end{split}$$

say.

For J_1 , we have

$$\begin{aligned} J_1 &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^k \frac{1}{2\pi} \int_1^T \chi (1-c-it) \left(\log \frac{t}{2\pi} \right)^{n-k+1} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} \, dt \\ &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^k \left(\sum_{m \le \frac{T}{2\pi}} (\log m)^k (\log m)^{n-k+1} + O\left(T^{1/2} (\log T)^{n-k+1}\right) \right) \\ &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^k \sum_{m \le \frac{T}{2\pi}} (\log m)^{n+1} + O\left(T^{1/2} (\log T)^{n+1}\right) \\ &= 0 + O\left(T^{1/2} (\log T)^{n+1}\right) \end{aligned}$$

since by the Binomial Theorem,

$$\sum_{k=0}^{n} \binom{n}{k} (-1)^{k} = (1+(-1))^{n} = 0.$$

For J_2 , we have

$$\begin{split} J_2 &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} \frac{1}{2\pi} \int_1^T \chi(1-c-it) \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{\sigma+it}} \sum_{r=1}^\infty \frac{\Lambda(r)}{r^{c+it}} \, dt \\ &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} \frac{1}{2\pi} \int_1^T \chi(1-c-it) \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{mr=1}^\infty \frac{\Lambda(r)(\log m)^k}{(mr)^{c+it}} \, dt \\ &= (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} \left(\sum_{mr \le \frac{T}{2\pi}} \Lambda(r)(\log m)^k (\log mr)^{n-k} + O\left(T^{1/2}(\log T)^{n-k}\right) \right) \\ &= (-1)^n \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) \sum_{k=0}^n \binom{n}{k} (-\log m)^k (\log mr)^{n-k} + O\left(T^{1/2}(\log T)^n\right) \\ &= (-1)^n \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log mr - \log m)^n + O\left(T^{1/2}(\log T)^n\right) \\ &= (-1)^n \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^n + O\left(T^{1/2}(\log T)^n\right). \end{split}$$

where the last three sums are over all integers m and r such that $mr \leq T/2\pi$. Our results follow from combining J_1 and J_2 .

Therefore

$$S^{L} = -\overline{I}$$

= $(-1)^{n+1} \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^{n} + O\left(T^{1/2} (\log T)^{n+2}\right).$

Since we have shown that the terms from S^B, S^T, S^R are harmless within the error term of $O\left(T^{1/2}(\log T)^{n+2}\right)$, we have that the integral given in equation (3.6) is equal to the sum

$$S = (-1)^{n+1} \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^n + O\left(T^{1/2} (\log T)^{n+2}\right).$$
(3.10)

3.1.5 Evaluating the sum S

As we have discussed above, all that remains to do is to evaluate the sum in (3.10). To do this, we first note that by Perron's formula, we have

$$(-1)^{n+1} \sum_{mr \le \frac{T}{2\pi}} \Lambda(r) (\log r)^n = \frac{1}{2\pi i} \int_{c-i\infty}^{c+i\infty} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds$$

where we set $Y = \frac{T}{2\pi}$ and $c = 1 + \frac{1}{\log T}$ as before.

Since we want to be able to evaluate the integral on the right-hand side, we will modify this slightly and instead use a truncated Perron formula. **Lemma 3.12.** For $2 \leq V \leq Y$, as $Y \to \infty$,

$$(-1)^{n+1} \sum_{mr \le Y} \Lambda(r) (\log r)^n = \frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds + R(Y,V),$$

where

$$R(Y,V) \ll \frac{Y}{V} (\log Y)^{n+2}.$$

We will use a specific V later in this section. The choice of V will depend on whether we assume the Riemann hypothesis or not.

Proof. If we let a(k) denote the coefficients in Dirichlet series for $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}\zeta(s)$, namely

$$a(k) = (-1)^{n+1} \sum_{r|k} \Lambda(r) (\log r)^n$$

then by the truncated Perron formula we have that the integral in the lemma equals

$$\sum_{k \le Y} a(k) + R(Y, V)$$

with

$$R(Y,V) \ll \sum_{\frac{Y}{2} < k < 2Y} |a(k)| \min\left(1, \frac{Y}{V|Y-k|}\right) + \frac{4^c + Y^c}{V} \sum_{k=1}^{\infty} \frac{|a(k)|}{k^c}$$

= A + B.

Writing k = mr we see $\sum_{k \leq Y} a(k)$ equals the sum in the lemma. To evaluate the error, note that $|a(k)| \leq (\log k)^{n+1}$ (with equality only if k is prime). Therefore

$$A \ll (\log Y)^{n+1} \sum_{\frac{Y}{2} < k < 2Y} \min\left(1, \frac{Y}{V|Y-k|}\right)$$
$$\ll (\log Y)^{n+1} \sum_{\ell \le Y} \frac{Y}{V} \frac{1}{\ell}$$
$$\ll \frac{Y}{V} (\log Y)^{n+2}.$$

For B, since $c = 1 + \frac{1}{\log T}$, the Dirichlet series converges, and so

$$B = \frac{4^c + Y^c}{V} \left| \left(\frac{\zeta'}{\zeta}(c)\right)^{(n)} \zeta(c) \right| \ll \frac{Y}{V} (\log Y)^{n+2}$$

where we use the fact that $Y \simeq T$.

Combining A and B gives the required bound on R(Y, V).

114

3.1.6 The Error Term

We will show that the error term can be described explicitly as follows, depending on whether we assume the Riemann hypothesis or not.

Lemma 3.13. For $Y = \frac{T}{2\pi}$, as $Y \to \infty$, we have

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + E_n(Y)$$

where $E_n(Y)$ is an error term given by one of the following three cases:

1. Unconditionally, by setting $V = e^{C\sqrt{\log Y}}$, we obtain

$$E_n(Y) = O\left(Y e^{-C\sqrt{\log Y}}\right)$$

where C a positive constant that is not necessarily the same in each instance.

2. Assuming the Riemann hypothesis, setting $V = 2\pi Y$, we obtain

$$E_n(Y) = O\left(Y^{1/2+\varepsilon}\right)$$

for $\varepsilon > 0$.

3. Assuming the Riemann hypothesis, setting $V = Y/(\log Y)^{3/2}$, we obtain

$$E_n(Y) = O\left(Y^{1/2} (\log Y)^{n+9/4}\right).$$

We will calculate the residue in Section 3.1.7. First, we will show how we can obtain the different expressions for $E_n(Y)$ in the following subsections. We will also show in the final subsection that we are able to obtain a slightly better error term than that given in Fujii [126]. This argument first appeared in [186] for the n = 1 case and beats what is given in [186] for $n \ge 2$. We call the error term $E_n(T)$ here for consistency with our later comments on the error term as it grows with n, even though the first two error terms given in Lemma 3.13 don't have any explicit n-dependence.

3.1.6.1 The Unconditional Case

From Titchmarsh [298, Sect. 3.8, p. 54], we know there is a positive constant C such that for $c' = 1 - \frac{C}{\log V}$, all the zeros of $\zeta(s)$ are $\gg \frac{1}{\log T}$ away from the line running from c' - iVto c' + iV. By Cauchy's Residue Theorem, the integral is

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \\ + \frac{1}{2\pi i} \left(\int_{c'+iV}^{c+iV} + \int_{c'-iV}^{c'+iV} - \int_{c'-iV}^{c-iV}\right) \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds.$$

By Lemmas 3.2 and 3.3, if $V \ll T$, the integral on the horizontal lines can be estimated

as

$$\frac{1}{2\pi i} \int_{c'\pm iV}^{c\pm iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll (\log V)^{n+3} \frac{Y^c}{V} (c-c') \ll \frac{Y}{V} (\log V)^{n+2}$$

where we use the fact that $c - c' \ll \frac{1}{\log V}$ and $Y^c \ll Y$ since $c = 1 + \frac{1}{\log(2\pi Y)}$.

For the integral on the vertical line, we have

$$\frac{1}{2\pi i} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll Y^{c'} (\log V)^{n+3} \int_{-V}^{V} \frac{1}{1+|t|} \, dt \ll Y^{c'} (\log V)^{n+4} \, dt \ll Y^{c'} \, dt \ll Y^{c'}$$

Since $c' = 1 - \frac{C}{\log V}$, balancing the two errors comes from taking $V = \exp(C\sqrt{\log Y})$ for some positive constant C, and so we have

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + O\left(Y \mathrm{e}^{-C\sqrt{\log Y}}\right).$$

3.1.6.2 The Conditional Case: No dependence on n

We assume the Riemann Hypothesis. One approach would be to choose $c' = \frac{1}{2} + \frac{1}{\log T}$ to guarantee a zero-free region, and set $V = T = 2\pi Y$. In that case, an application of Cauchy's Residue Theorem yields

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + O\left(T^{1/2+\varepsilon}\right)$$

for $\varepsilon > 0$, where the horizontal pieces of the contour are estimated in a manner similar to that in the previous section, and the vertical piece of the contour uses the bound $\zeta(s) \ll t^{\varepsilon}$ for $\sigma \ge 1/2$.

As we have previously noted, we can get a better bound, one that depends explicitly upon n as we show in the next section. This error term originally appeared in our paper [186] for the case n = 1 but we give a general proof in this thesis for all n, and so we have combined the proofs and omitted the special case.

3.1.6.3 The Conditional Case: Dependence on n

We again assume the Riemann Hypothesis in this section. We can get a better bound, one that depends explicitly upon n, by choosing $c' = 1 - c = -\frac{1}{\log T}$ (that is, just to the left of the critical strip) and $V \ll T$. By Cauchy's Residue Theorem,

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + \sum_{|\gamma| < V} \operatorname{Res}_{s=\rho} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + \frac{1}{2\pi i} \left(\int_{c'+iV}^{c+iV} + \int_{c'-iV}^{c'+iV} - \int_{c'-iV}^{c-iV}\right) \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds,$$

3.1. The Generalised Shanks' Conjecture

where the sum runs through the non-trivial zeros $\rho = \beta + i\gamma$ of $\zeta(s)$, lying inside the contour.

To estimate the integral on the horizontal lines, choose V so that all the zeros of zeta are bounded by $\gg \frac{1}{\log V}$ away from the horizontal line. Therefore we may use Lemma 3.3 to bound the logarithmic derivative by $(\log V)^{n+2}$ along the line. Using the convexity result from Lemma 3.2, we obtain, unconditionally,

$$\begin{split} \frac{1}{2\pi i} \int_{c'\pm iV}^{c\pm iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} ds \ll (\log V)^{n+2} \int_{-\frac{1}{\log V}}^{0} V^{1/2-\sigma} \log V \frac{Y^{\sigma}}{V} \ d\sigma \\ &+ (\log V)^{n+2} \int_{0}^{1} V^{(1-\sigma)/2} \log V \frac{Y^{\sigma}}{V} \ d\sigma \\ &+ (\log V)^{n+2} \int_{1}^{1+\frac{1}{\log V}} \log V \frac{Y^{\sigma}}{V} \ d\sigma \\ \ll \frac{Y}{V} (\log V)^{n+2}. \end{split}$$

For the integral on the vertical line, since $V \leq T$ and $c' = -\frac{1}{\log T}$ we may again use Lemma 3.3 as the vertical line is bounded away from any zeros of zeta. Using Lemma 3.2 to bound zeta just to the left of the critical strip, we obtain (again unconditionally),

$$\frac{1}{2\pi i} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^V (\log t)^{n+2} \times t^{1/2} \times \log t \times \frac{1}{t} \, dt \ll V^{1/2} (\log V)^{n+3} + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^{c'+iV} \left(\frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} ds + \frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds \ll \int_1^{c'+iV} \left(\frac{1}{2} \int_{c'-iV}^{c'+iV} \left(\frac{1}{2} \int_$$

We now consider the poles at $s = \rho$ for each ρ with $|\gamma| < V$, where ρ is a non-trivial zero of $\zeta(s)$.

First we note that for $-1 \leq \sigma \leq 2$ and $0 < t_0 \leq t \leq V$, we have

$$\frac{\zeta'}{\zeta}(\sigma+it) = \sum_{|\gamma-t|<1} \frac{1}{s-\rho} + O(\log V).$$

Being careful with the error term, we may differentiate this n times to give

$$\left(\frac{\zeta'}{\zeta}(\sigma+it)\right)^{(n)} = \sum_{|\gamma-t|<1} \frac{(-1)^n n!}{(s-\rho)^{n+1}} + O(\log V).$$

For each ρ , we need to consider the coefficient of $(s - \rho)^n$ in the expansion of $\zeta(s)\frac{Y^s}{s}$ to find the residue at $s = \rho$. For this, note that by the triple product rule, we may write

$$\left(\zeta(s)\frac{Y^s}{s}\right)^{(n)} = \sum_{\substack{k_1+k_2+k_3=n\\k_1,k_2,k_3 \ge 0}} \binom{n}{k_1,k_2,k_3} \zeta^{(k_1)}(s)(Y^s)^{(k_2)} \left(\frac{1}{s}\right)^{(k_3)}$$

where $\binom{n}{k_1,k_2,k_3}$ is the multinomial coefficient given by

$$\binom{n}{k_1, k_2, k_3} = \frac{n!}{k_1! k_2! k_3!}.$$

Therefore, at each zero ρ ,

$$\left(\zeta(s)\frac{Y^s}{s}\right)^{(n)}\bigg|_{s=\rho} = \sum_{\substack{k_1+k_2+k_3=n\\k_1,k_2,k_3 \ge 0}} (-1)^{k_3}k_3! \binom{n}{k_1,k_2,k_3} \frac{\zeta^{(k_1)}(\rho)Y^{\rho}(\log Y)^{k_2}}{\rho^{k_3+1}}.$$

As we are just bounding these terms, we do not worry about the coefficient such as the n!, $(-1)^n$ and the multinomial coefficients. Assuming RH, so $\rho = \frac{1}{2} + i\gamma$ and summing over all zeros with $|\gamma| < V$, we have

$$\sum_{|\gamma| < V} \operatorname{Res}_{s=\rho} \left(\frac{\zeta'}{\zeta}(s) \right)^{(n)} \zeta(s) \frac{Y^s}{s} \ll Y^{1/2} \sum_{\substack{k_1 + k_2 + k_3 = n \\ k_1, k_2, k_3 \ge 0}} (\log Y)^{k_2} \sum_{\gamma < V} \frac{\left| \zeta^{(k_1)}(\frac{1}{2} + i\gamma) \right|}{\left| \frac{1}{2} + i\gamma \right|^{k_3 + 1}}.$$
 (3.11)

Under the Riemann Hypothesis, Garaev [135] showed that

$$\sum_{0 < \gamma \le T} \left| \zeta' \left(\frac{1}{2} + i\gamma \right) \right| \ll T (\log T)^{9/4}.$$

To prove this result, he uses two key facts. The first is the following lemma.

Lemma 3.14. Let f(t) be a real-valued function that is continuously differentiable on the interval $[t_0, t_k]$. Let $t_0 < t_1 < ... < t_k$, and assume that f(t) has at least one zero in each interval $[t_j, t_{j+1}], 0 \le j \le j - 1$. Then

$$\sum_{1 \le j \le k} |f(t_j)| \le \int_{t_0}^{t_k} |f'(t)| \, dt.$$

The other result Garaev uses is the following inequality from Ramachandra [270].

Lemma 3.15 (Ramachandra). The inequality

$$\frac{1}{T} \int_{1}^{T} \zeta^{(n)} \left(\frac{1}{2} + it\right) dt \ll (\log T)^{n+1/4}$$

holds.

To obtain a bound for the n^{th} derivative, we begin by proving the following corollary of Lemma 3.14.

Corollary 3.16. Let S(t) be a complex-valued function that is (n + 1)-times differentiable on the interval $[t_0, t_k]$ with $t_0 < t_1 < ... < t_k$ and $S(t_j) = 0$ for all $0 \le j \le k$. Then

$$\sum_{1 \le j \le k} |S^{(n)}(t_j)| \le \sqrt{2} \int_{t_0}^{t_k} |S^{(n+1)}(t)| \, dt.$$

Proof. We write $S(t) = A_1(t) + iA_2(t)$ where the $A_m(t)$ for m = 1, 2 are (n+1)-differentiable functions on the interval $[t_0, t_k]$. Then $A_m(t_j) = 0$ for $0 \le j \le k$. By the generalised form of Rolle's theorem, the $A_m^{(n)}(t)$ have zeros in each interval (t_j, t_{j+n}) for $0 \le j \le k - n$. Applying Lemma 3.14 to $f(t) = A_m^{(n)}(t)$ gives

$$\sum_{1 \le j \le k} |A_m^{(n)}(t_j)| \le \int_{t_0}^{t_k} |A_m^{(n+1)}(t)| \ dt.$$

Combining these,

$$\sum_{1 \le j \le k} |S^{(n)}(t_j)| \le \sqrt{2} \int_{t_0}^{t_k} |A_1^{(n+1)}(t)| + |A_2^{(n+1)}(t)| \, dt \le \sqrt{2} \int_{t_0}^{t_k} |S^{(n+1)}(t)| \, dt.$$

Applying Lemma 3.15 to Corollary 3.16 with $S(t) = \zeta(\frac{1}{2} + it)$ and $t_j = \gamma_j$ gives the following bound for all zeros with $1 \leq \gamma_k \leq T$. Under the Riemann Hypothesis, we have shown that

$$\sum_{0 < \gamma \le T} \left| \zeta^{(n)} \left(\frac{1}{2} + i\gamma \right) \right| \ll T (\log T)^{n+5/4}.$$

Now by partial summation,

$$(\log Y)^{k_2} \sum_{\gamma < V} \frac{\left| \zeta^{(k_1)}(\frac{1}{2} + i\gamma) \right|}{\left| \frac{1}{2} + i\gamma \right|^{k_3 + 1}} \ll \int_1^V \frac{(\log t)^{k_1 + 5/4}}{t^{k_3 + 1}} dt$$
$$\ll \begin{cases} \log^{k_2} Y (\log V)^{k_1 + 9/4} & \text{if } k_3 = 0\\ \log^{k_2} Y & \text{if } k_3 \ge 1 \end{cases}$$

Clearly the dominant error term is when $k_3 = 0$ (which forces $k_2 = n - k_1$), and so the sum over the residues at the zeros in (3.11) is bounded by

$$\sum_{|\gamma| < V} \operatorname{Res}_{s=\rho} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \ll \sum_{k_1=0}^n Y^{1/2} (\log Y)^{n-k_1} (\log V)^{k_1+9/4} \\ \ll Y^{1/2} (\log Y)^{n+9/4}.$$

Balancing this error term with the error term from the vertical line comes from taking $V = Y/(\log Y)^{3/2}$, and so we have

$$\frac{1}{2\pi i} \int_{c-iV}^{c+iV} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} \, ds = \operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{Y^s}{s} + O\left(Y^{1/2} (\log Y)^{n+9/4}\right).$$

3.1.7 Finding the Leading Asymptotic Terms

We now evaluate

$$\operatorname{Res}_{s=1}\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}\zeta(s)\frac{Y^s}{s}$$

from Lemma 3.13. We expand each of the terms in this residue calculation in their Laurent expansions about s = 1. Since

$$\frac{\zeta'(s)}{\zeta(s)} = -\frac{1}{s-1} + A_0 + A_1(s-1) + \dots + A_k(s-1)^k + \dots + A_n(s-1)^n + O\left((s-1)^{n+1}\right),$$

where the A_n are defined in (3.3), we have

1. The Laurent expansion for $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}$ about s = 1:

$$\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} = \frac{(-1)^{n+1}n!}{(s-1)^{n+1}} + n!A_n + O((s-1)).$$

2. The Laurent expansion for $\zeta(s)$ about s = 1:

$$\zeta(s) = \frac{1}{s-1} + \gamma_0 - \gamma_1(s-1) + \dots + \frac{(-1)^k}{k!}(s-1)^k + \dots + \frac{(-1)^n}{n!}(s-1)^n + O\left((s-1)^{n+1}\right)$$

3. The Laurent expansion for Y^s about s = 1:

$$\begin{split} Y^s = & Y\left(1 + (s-1)\log Y + \ldots + \frac{(s-1)^k}{k!}(\log Y)^k + \ldots + \frac{(s-1)^n}{n!}(\log Y)^n \\ & + \frac{(s-1)^{n+1}}{(n+1)!}(\log Y)^{n+1} + O\left((s-1)^{n+2}\right)\right). \end{split}$$

4. The Laurent expansion for $\frac{1}{s}$ about s = 1:

$$\frac{1}{s} = 1 - (s-1) + (s-1)^2 + \dots + (-1)^k (s-1)^k + \dots + (-1)^n (s-1)^n + O\left((s-1)^{n+1}\right)$$

We now work out the residue at s = 1 by considering the terms in powers of $(\log Y)^k$, as k runs from k = n + 1 down to k = 0. We consider a combinatorial style argument to make sure we consider all possible terms. The following table shows this for various powers of $(\log Y)^k$.

To calculate the resulting contribution to the residue for the leading order behaviour, that is, the coefficient of $Y(\log Y)^{n+1}$, there is only one way to obtain a factor of $\frac{1}{s-1}$ for the residue calculation. We have to have the $\frac{(-1)^{n+1}n!}{(s-1)^{n+1}}$ term from $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}$, the $\frac{1}{s-1}$ term from $\zeta(s)$, the $\frac{(s-1)^{n+1}}{(n+1)!}$ term from Y^s and the 1 term from $\frac{1}{s}$. Combined, the coefficient for $Y(\log Y)^{n+1}$ is $\frac{(-1)^{n+1}}{n+1}$.

To calculate the subleading term there are two distinct possibilities.

3.1. The Generalised Shanks' Conjecture

- 1. The first possibility is to have the $\frac{(-1)^{n+1}n!}{(s-1)^{n+1}}$ term from $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}$, the γ_0 term from $\zeta(s)$, the $\frac{(s-1)^n}{n!}$ term from Y^s and the 1 term from $\frac{1}{s}$. The resulting contribution to the residue for this term is $(-1)^{n+1}\gamma_0$.
- 2. The other possibility is to have the $\frac{(-1)^{n+1}n!}{(s-1)^{n+1}}$ term from $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}$, the $\frac{1}{s-1}$ term from $\zeta(s)$, the $\frac{(s-1)^n}{n!}$ term from Y^s and the -(s-1) term from $\frac{1}{s}$. The resulting contribution to the residue for this term is $(-1)^{n+2}$.

Combined, the coefficient for the $Y(\log Y)^n$ term is $(-1)^{n+1}(\gamma_0 - 1)$.

The other terms in all lower order cases follow in a similar manner - Table 3.1 shows all possibilities, with one extra caveat for the lowest order term which has an additional term.

In the lowest-leading order case, the coefficient of the Y term, there are all the usual terms for the subleading behaviour, with one additional term. The additional term comes from the $n!A_n$ term from $\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}$, the $\frac{1}{s-1}$ term from $\zeta(s)$, the 1 term from Y^s , and the 1 term from $\frac{1}{s}$. Combined, the extra term for Y has coefficient $n!A_n$.

Combining all of the terms in the above table, and putting $Y = \frac{T}{2\pi}$ we have shown that

$$\operatorname{Res}_{s=1} \left(\frac{\zeta'}{\zeta}(s)\right)^{(n)} \zeta(s) \frac{\left(\frac{T}{2\pi}\right)^s}{s} = (-1)^{n+1} \frac{1}{n+1} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n+1} + (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (-1)^k k! \left(-1 + \sum_{j=0}^k \frac{1}{j!} \gamma_j\right) \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-k} + n! A_n \frac{T}{2\pi}$$

Combining this with the results from Lemma 3.13 and using the appropriate choices for V in Lemma 3.12 we see that

$$(-1)^{n+1}\sum_{mr\leq\frac{T}{2\pi}}\Lambda(r)(\log r)^n = \operatorname{Res}_{s=1}\left(\frac{\zeta'}{\zeta}(s)\right)^{(n)}\zeta(s)\frac{\left(\frac{T}{2\pi}\right)^s}{s} + E_n(T).$$

Finally, (3.10) shows that sum is equal to $S = \sum_{0 < \gamma \leq T} \zeta^{(n)}(\rho)$ (plus an error that is smaller than $E_n(T)$), and the result follows.

3.1.8 Examples of the first moment of specific n^{th} derivatives of the Riemann zeta function $\zeta(s)$

The first test of a good theorem is if it can recover any previously known results. We begin this section by doing exactly that in the n = 1 case where we recover Shanks' Conjecture and Fujii's asymptotics, given in (3.1). After that we give another example, as well as showing some numerical data showing that our result is in excellent agreement with the true values. We have chosen n = 2 to show the most simple new case in full detail. The calculations for higher level derivatives become unwieldy to do by hand so we could employ a computer package to manage this for us to obtain any specific n^{th} order derivative that we choose.

(s)	(<u>)</u>	(-)	 (<u>)</u>	() ()	$\left(\frac{\zeta}{\zeta'}\right)$
$\frac{1)^{n+1}n!}{-1)^{n+1}}$	$\frac{1)^{n+1}n!}{-1)^{n+1}}$	$\frac{1)^{n+1}n!}{-1)^{n+1}}$	 $\frac{1)^{n+1}n!}{-1)^{n+1}}$	$\frac{1)^{n+1}n!}{-1)^{n+1}}$	$(s)\Big)^{(n)}$
$\frac{1}{s-1}$	γ_0	$-\gamma_1(s-1)$	 $\frac{(-1)^{k-1}}{(k-1)!}\gamma_{k-1}(s-1)^{k-1}$	$rac{(-1)^k}{k!}\gamma_k(s-1)^k$	$\zeta(s)$
$\frac{(s-1)^{n-k}Y(\log Y)^{n-k}}{(n-k)!}$	$\frac{(s-1)^{n-k}Y(\log Y)^{n-k}}{(n-k)!}$	$\frac{(s-1)^{n-k}Y(\log Y)^{n-k}}{(n-k)!}$	 $\frac{(s-1)^{n-k}Y(\log Y)^{n-k}}{(n-k)!}$	$\frac{(s-1)^{n-k}Y(\log Y)^{n-k}}{(n-k)!}$	Y^s
$(-1)^{k+1}(s-1)^{k+1}$	$(-1)^k(s-1)^k$	$(-1)^{k-1}(s-1)^{k-1}$	 -(s-1)	1	<u>سا</u> «
$(-1)^{n+k} \frac{n!}{(n-k)!} Y(\log Y)^{n-k}$	$(-1)^{n+k+1} \frac{n!}{(n-k)!} \gamma_0 Y (\log Y)^{n-k}$	$(-1)^{n+k} \frac{n!}{(n-k)!} (-\gamma_1) Y (\log Y)^{n-k}$	 $(-1)^n \frac{n!}{(n-k)!} \frac{(-1)^{k-1}}{(k-1)!} \gamma_{k-1} Y(\log Y)$	$(-1)^{n+1} \frac{n!}{(n-k)!} \frac{(-1)^k}{k!} \gamma_k Y(\log Y)^n$	Contribution to the residue

term in the residue (k = 0, ..., n). Table 3.1: Table to show which ter ms in the Laurent expansion of each piece of L combine to form a contribution to the $Y(\log Y)^{n-k}$

3.1. The Generalised Shanks' Conjecture

3.1.8.1 The case
$$n = 1$$
 (Shanks' Conjecture)

We can use Theorem 3.1 to recover Fujii's asymptotic formula (2.3) (with an improved error term) and hence Shanks' Conjecture as first stated at the start of Chapter 2. To do this, set n = 1 in the theorem to obtain

$$\sum_{0<\gamma\leq T} \zeta'(\rho) = \frac{1}{2} \frac{T}{2\pi} \left(\log \frac{T}{2\pi}\right)^2 + \sum_{k=0}^1 \binom{1}{k} (-1)^k k! \left(-1 + \sum_{j=0}^k \frac{1}{j!} \gamma_j\right) \frac{T}{2\pi} \left(\log \frac{T}{2\pi}\right)^{1-k} + A_1 \frac{T}{2\pi} + E_1(T).$$

Note that $A_1 = -2\gamma_1 - \gamma_0^2$. Substituting this into the expression above and simplifying gives

$$\sum_{0 < \gamma \le T} \zeta'(\rho) = \frac{T}{4\pi} \left(\log \frac{T}{2\pi} \right)^2 + (-1 + \gamma_0) \frac{T}{2\pi} \log \frac{T}{2\pi} + (1 - \gamma_0 - \gamma_0^2 - 3\gamma_1) \frac{T}{2\pi} + E_1(T)$$

in perfect agreement with (3.1). In Figure 3.1 we show the accuracy of this asymptotic by considering the true value of the real part of the sum, and start subtracting the main terms, for T up to the height of the 100,000th zero.



Figure 3.1: The real part of the sum $\sum_{0 < \gamma \leq T} \zeta'(\rho)$ with true value (blue), the true - leading (pink), true - leading - subleading (orange) and true - leading - subleading - subleading (green) terms subtracted from the true value, for T up to the height of the 100,000th zero.

The error term $E_1(T)$ is $O\left(Te^{-C\sqrt{\log T}}\right)$ unconditionally. Note that under the Riemann Hypothesis, we have

$$E_1(T) = O\left(T^{1/2}(\log T)^{13/4}\right)$$

as previously claimed (and proved in [186]). This error term is then plotted in Figure 3.2, for T up to the height of the 100,000th zero.



Figure 3.2: Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta'(\rho)$ and the whole asymptotic result, for T up to the height of the 100,000th zero, showing the real error at each point.

3.1.8.2 The case n = 2

We set n = 2 in Theorem 3.1 to obtain

$$\sum_{0<\gamma\leq T} \zeta''(\rho) = -\frac{1}{3} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^3 - \sum_{k=0}^2 \binom{2}{k} (-1)^k k! \left(-1 + \sum_{j=0}^k \frac{1}{j!} \gamma_j\right) \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{2-k} + 2A_2 \frac{T}{2\pi} + E_2(T).$$

Note that $A_2 = \frac{3}{2}\gamma_2 + 3\gamma_0\gamma_1 + \gamma_0^3$. Substituting this into the expression above and simplifying gives

$$\sum_{0<\gamma\leq T} \zeta''(\rho) = -\frac{T}{6\pi} \left(\log\frac{T}{2\pi}\right)^3 - (-1+\gamma_0)\frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^2 + (-2+2\gamma_0+2\gamma_1)\frac{T}{2\pi}\log\frac{T}{2\pi} + (2-2\gamma_0-2\gamma_1+2\gamma_2+6\gamma_0\gamma_1+2\gamma_0^3)\frac{T}{2\pi} + E_2(T).$$

In Figure 3.3 we show the accuracy of this asymptotic by considering the true value of the real part of the sum, and start subtracting the main terms, for T up to the height of the 100,000th zero.



Figure 3.3: The real part of the sum $\sum_{0 < \gamma \leq T} \zeta''(\rho)$ with true value (blue), the true - leading (pink), true - leading - subleading (orange), true - leading - subleading - subsubleading (green), true - leading - subleading - subsubleading - subsubleading (purple) terms subtracted from the true value, for T up to the height of the 100,000th zero.

The error term $E_2(T)$ is $O\left(Te^{-C\sqrt{\log T}}\right)$ unconditionally. Note that under the Riemann Hypothesis, we have

$$O\left(T^{1/2}(\log T)^{17/4}\right).$$

This error term is then plotted in Figure 3.4, for T up to the height of the $100,000^{\text{th}}$ zero.



Figure 3.4: Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta''(\rho)$ and the whole asymptotic result, for T up to the height of the 100,000th zero, showing the real error at each point.

3.2 THE GENERALISED SHANKS' CONJECTURE FROM A SHIFTED SUM

Fujii [126] also gave a shifted version of the Riemann zeta function at the non-trivial zeros of zeta, written in [22] in the case $\alpha = i\delta$, $0 \neq \delta \ll 1$ as

$$\sum_{0 < \gamma \le T} \zeta(\rho + \alpha) = \frac{T}{2\pi} \frac{\zeta'}{\zeta} (1 + \alpha) - \left(\frac{T}{2\pi}\right)^{1 - \alpha} \frac{\zeta(1 - \alpha)}{1 - \alpha} + \frac{T}{2\pi} \log \frac{T}{2\pi e} + O\left(T e^{-C\sqrt{\log T}}\right).$$
(3.12)

Fujii [126] shows the error term can be replaced by $O\left(T^{1/2}(\log T)^{5/2}\right)$ under the Riemann Hypothesis. By a similar argument to that given in Section 3.1.6, this can be improved to $O\left(T^{1/2}(\log T)^{9/4}\right)$.

In this section we use the above result (3.12) to recover the asymptotic in Theorem 3.1. As a theme that we see later in this thesis when using shifts, mainly in Chapters 5 and 6, we will Taylor expand the $\zeta(\rho + \alpha)$ term as

$$\zeta(\rho + \alpha) = \zeta'(\rho)\frac{\alpha}{1!} + \zeta''(\rho)\frac{\alpha^2}{2!} + \dots + \zeta^{(n)}(\rho)\frac{\alpha^n}{n!} + \dots,$$
(3.13)

recalling $\zeta(\rho) = 0$ by definition.

To do this, we begin by writing the appropriate functions in their Laurent expansions about s = 1. Specifically, with $L = \log \frac{T}{2\pi}$, we write

$$\begin{aligned} \frac{\zeta'}{\zeta}(1+\alpha) &= -\frac{1}{\alpha} + \sum_{n=0}^{\infty} A_n \alpha^n & \left(\frac{T}{2\pi}\right)^{1-\alpha} &= \frac{T}{2\pi} \sum_{n=0}^{\infty} \frac{(-1)^n L^n \alpha^n}{n!} \\ \zeta(1-\alpha) &= -\frac{1}{\alpha} + \sum_{m=0}^{\infty} \frac{\gamma_m \alpha^m}{m!} & \frac{1}{1-\alpha} &= \sum_{\ell=0}^{\infty} \alpha^\ell \end{aligned}$$

where A_n is given explicitly in (3.3).

Next, substitute these into (3.12) to give

$$\begin{split} \sum_{0<\gamma\leq T} \zeta(\rho+\alpha) &= \frac{T}{2\pi} \left(-\frac{1}{\alpha} + \sum_{n=0}^{\infty} A_n \alpha^n \right) \\ &- \frac{T}{2\pi} \left(\sum_{n=0}^{\infty} \frac{(-1)^n L^n \alpha^n}{n!} \right) \left(-\frac{1}{\alpha} + \sum_{m=0}^{\infty} \frac{\gamma_m \alpha^m}{m!} \right) \left(\sum_{\ell=0}^{\infty} \alpha^\ell \right) \\ &+ \frac{T}{2\pi} \log \frac{T}{2\pi e} + O\left(T e^{-C\sqrt{\log T}} \right). \end{split}$$

After some simplifying, we have

$$\begin{split} \sum_{0<\gamma\leq T}\zeta(\rho+\alpha) &= -\frac{T}{2\pi}\frac{1}{\alpha} + \frac{T}{2\pi}\left(\sum_{n=0}^{\infty}A_n\alpha^n\right) \\ &+ \frac{T}{2\pi}\frac{1}{\alpha}\sum_{n=0}^{\infty}\left(\sum_{\ell=0}^{n}\frac{(-1)^{n-\ell}L^{n-\ell}}{(n-\ell)!}\right)\alpha^n \\ &+ \frac{T}{2\pi}\sum_{n=0}^{\infty}\left(\sum_{m=0}^{n}\sum_{\ell=0}^{n-m}\frac{(-1)^{n-m-\ell+1}\gamma_mL^{n-m-\ell}}{(n-m-\ell)!m!}\right)\alpha^n \\ &+ \frac{T}{2\pi}\log\frac{T}{2\pi e} + O\left(Te^{-C\sqrt{\log T}}\right). \end{split}$$

Now extract the pole from the second line and re-index to obtain

$$\begin{split} \sum_{0 < \gamma \le T} \zeta(\rho + \alpha) &= \frac{T}{2\pi} \left(\sum_{n=0}^{\infty} A_n \alpha^n \right) \\ &+ \frac{T}{2\pi} \sum_{n=0}^{\infty} \left(\sum_{\ell=0}^{n+1} \frac{(-1)^{n-\ell+1} L^{n-\ell+1}}{(n-\ell+1)!} \right) \alpha^n \\ &+ \frac{T}{2\pi} \sum_{n=0}^{\infty} \left(\sum_{m=0}^n \sum_{\ell=0}^{n-m} \frac{(-1)^{n-m-\ell+1} \gamma_m L^{n-m-\ell}}{(n-m-\ell)!m!} \right) \alpha^n \\ &+ \frac{T}{2\pi} \log \frac{T}{2\pi e} + O\left(T e^{-C\sqrt{\log T}} \right). \end{split}$$

For the general n^{th} derivative of $\zeta(s)$, where $n \in \mathbb{N}$, compare the coefficient of α^n on both sides of the previous equation. Recall that we have a Taylor expansion for $\zeta(\rho + \alpha)$ about $\alpha = 0$, so

$$\sum_{0<\gamma\leq T} \frac{1}{n!} \zeta^{(n)}(\rho) = \frac{T}{2\pi} A_n + \frac{T}{2\pi} \sum_{\ell=0}^{n+1} \frac{(-1)^{n-\ell+1} L^{n-\ell+1}}{(n-\ell+1)!} + \frac{T}{2\pi} \sum_{m=0}^n \sum_{\ell=0}^{n-m} \frac{(-1)^{n-m-\ell+1} \gamma_m L^{n-m-\ell}}{(n-m-\ell)!m!} + O\left(Te^{-C\sqrt{\log T}}\right).$$

Multiplying through by n! gives us an equivalent way of writing Theorem 3.1.

Theorem 3.17. For $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, and where $\zeta^{(n)}(s)$ is the nth derivative of $\zeta(s)$, we have for T sufficiently large,

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) = n! \sum_{\ell=0}^{n+1} \frac{(-1)^{n-\ell+1}}{(n-\ell+1)!} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-\ell+1} + n! \sum_{m=0}^{n} \sum_{\ell=0}^{n-m} \frac{(-1)^{n-m-\ell+1}\gamma_m}{(n-m-\ell)!m!} \frac{T}{2\pi} \left(\log\frac{T}{2\pi}\right)^{n-m-\ell} + n! A_n \frac{T}{2\pi} + O\left(Te^{-C\sqrt{\log T}}\right),$$

where the coefficients γ_j are from the Laurent expansions of $\zeta(s)$ about s = 1 and the coefficients A_n are from the Laurent expansions of $\zeta'(s)/\zeta(s)$ about s = 1, given in (3.3).

It is clear then that the leading-order behaviour comes from the first sum with $\ell = 0$, and returns the expected leading-order coefficient.

We remark that care should be taken with regards to the error term; however it is fair that we should expect the error term from Theorem 3.1 to carry through, both unconditionally and under the Riemann Hypothesis.

3.3 THE GENERALISED SHANKS' CONJECTURE IN SHORT INTERVALS

We end this chapter by briefly considering sums of the derivative of the Riemann zeta function over the non-trivial zeros of the Riemann zeta function but instead over short intervals of length $T^{1/2+\varepsilon} \ll H \ll T$, rather than one of full length T. In an analogous method to that given in Section 3.1, we can show to leading order the following result.

Theorem. Let $T^{1/2+\varepsilon} \ll H \ll T$. Then for $\rho = \beta + i\gamma$ a non-trivial zero of $\zeta(s)$, and where $\zeta^{(n)}(s)$ is the nth derivative of $\zeta(s)$, we have

$$\sum_{T < \gamma \le T+H} \zeta^{(n)}(\rho) = (-1)^{n+1} \frac{1}{n+1} \frac{H}{2\pi} \left(\log T\right)^{n+1} + O\left(H(\log T)^n\right) + O\left(T^{1/2+\varepsilon}\right).$$

— 4 —

A Further Generalisation of Shanks' Conjecture

In this chapter we obtain a full asymptotic formula for the sum

$$\sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) X^{\rho}$$

where $\zeta^{(n)}(s)$ denotes the *n*th derivative of the Riemann zeta function, X is a positive real number, and $\rho = \beta + i\gamma$ denotes a non-trivial zero of the Riemann zeta function.

We also specify what the asymptotic formula becomes when X is a positive integer, highlighting the differences in the asymptotic expansions as X changes its arithmetic nature. This is interesting as the Landau–Gonek formula, as stated in (2.9) has numerous error terms but when X becomes a positive integer, as shown in (2.10), the behaviour changes. We see this phenomenon happening in the results in this chapter.

We have already seen the strength of the Landau–Gonek theorem in developing the heuristic presented in Chapter 2. The application of the Landau–Gonek theorem given here when combining with the Generalised Shanks' Conjecture takes a different form.

The work presented in this chapter has already appeared in print in [262]. Notation and cross-references have been updated for this thesis, as well as incorporating the improved error term given in Section 3.1.6.

4.1 HISTORICAL BACKGROUND OF COMBINATIONS OF THE LANDAU– GONEK THEOREM AND MOMENTS OF *L*-FUNCTIONS

Recall that the Landau–Gonek Theorem (2.9) is an explicit formula, a formula that relates the zeros of the Riemann zeta function to the prime numbers (and the further generalisations relating primes to zeros of other *L*-functions). By combining an explicit formula with an asymptotic for an *L*-function gives another type of explicit formula. Note that these combined formulae still link the zeros of an *L*-function with the prime numbers as will be highlighted in this historical section and in the new results we present. Fujii started the idea of combining the asymptotic (2.9) from the Landau–Gonek Theorem and the derivatives of the Riemann zeta function in [128]. More specifically, he considers X to be a fixed positive real number, so he actually combines Landau's Theorem [219] with the derivatives of the Riemann zeta function (that is, the non-uniform version of the Landau–Gonek Theorem). For T sufficiently large, he gives the following explicit formula for a fixed positive real number X,

$$\sum_{0<\gamma\leq T} \zeta'(\rho) X^{\rho} = -\Delta(X) \frac{T}{2\pi} \left(\log X \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) - \sum_{mr=X} \Lambda(r) \log m \right) + X \sum_{m\leq \frac{T}{2\pi X}} e^{2\pi i m X} (\log m)^2 + \frac{1}{2} X \log X \sum_{m\leq \frac{T}{2\pi X}} e^{2\pi i m X} \log m - \left(\frac{1}{2} X (\log X)^2 - \frac{\pi i}{4} X \log X \right) \sum_{m\leq \frac{T}{2\pi X}} e^{2\pi i m X} - X \sum_{mr\leq \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) \log m + O\left(T^{1/2} (\log T)^3 \right)$$
(4.1)

where $\Delta(X)$ is defined by

$$\Delta(X) = \begin{cases} 1 & \text{if } X \in \mathbb{Z} \\ 0 & \text{otherwise.} \end{cases}$$
(4.2)

throughout this chapter.

Fujii then narrowed down X to be a positive integer and gave an asymptotic formula for this, again exhibiting the behaviour that Gonek noted in [128], about how the arithmetic nature of X can affect the asymptotic expansion. Specifically, he finds for X a positive integer,

$$\sum_{0 < \gamma \le T} \zeta'(\rho) X^{\rho} = \frac{T}{4\pi} \left(\log \frac{T}{2\pi} \right)^2 + (-1 + \gamma_0 - \log X) \frac{T}{2\pi} \log \frac{T}{2\pi} + \frac{T}{2\pi} \left(1 - \gamma_0 - \gamma_0^2 - 3\gamma_1 + \sum_{X=mn} \Lambda(n) \log m - \left(\gamma_0 - 1 + \frac{1}{2} \log X \right) \log X \right) + O\left(T e^{-C\sqrt{\log T}} \right)$$
(4.3)

where C is a positive constant. By assuming the Riemann Hypothesis, Fujii gives an improved error term of $O\left(T^{1/2}(\log T)^{7/2}\right)$. Clearly setting X = 1 then recovers (2.3).

A generalisation of Fujii's result has been given in Jakhlouti and Mazhouda [202] to give an analogue of (4.1) for Dirichlet *L*-functions. This states for *T* sufficiently large and

X a positive real number,

$$\begin{split} \sum_{0<\gamma_{\chi}\leq T} L'(\rho_{\chi},\chi)X^{\rho_{\chi}} &= \\ &-\Delta(X)\chi[\Delta(X)X]\frac{T}{2\pi}\bigg(\log X\left(\frac{1}{2}\log\frac{qT}{2\pi}-\frac{1}{2}+\frac{\pi i}{4}\right)-\sum_{mr=X}\Lambda(r)\log m\bigg) \\ &+\frac{X}{\sqrt{q}}\sum_{m\leq\frac{qt}{2\pi X}}e^{2\pi i m\frac{X}{q}}\overline{\chi}(m)(\log m)^2+\frac{1}{2\sqrt{q}}X\log X\sum_{m\leq\frac{qt}{2\pi X}}e^{2\pi i m\frac{X}{q}}\overline{\chi}(m)\log m \\ &-\bigg(\frac{1}{2\sqrt{q}}X(\log X)^2-\frac{\pi i}{4\sqrt{q}}X\log X\bigg)\sum_{m\leq\frac{qt}{2\pi X}}e^{2\pi i m\frac{X}{q}}\overline{\chi}(m) \\ &-\frac{X}{\sqrt{q}}\sum_{mr\leq\frac{qt}{2\pi X}}e^{2\pi i mr\frac{X}{q}}\Lambda(r)\overline{\chi}(r)\overline{\chi}(m)\log m+O\left(T^{1/2}(\log T)^3\right) \end{split}$$

as $T \to \infty$ for a fixed positive number X and where χ is a primitive character mod q. Fixing q = 1 gives (4.1) and setting q = 1 and X = 1 gives (2.3).

Other ideas have grown from (2.9). For example, Ford and Zaharescu [119] start (2.9) and investigate the distribution of the fractional parts of $\alpha\gamma$, where α is a fixed non-zero real number. This idea is then expanded upon by Ford, Soundararajan and Zaharescu in [118]. A further example of a result that begins with the Landau–Gonek Theorem can be found in [205], where Kaczorowski, Languasco, and Perelli consider a different weighted version of the Landau–Gonek Theorem.

4.2 COMBINING THE GENERALISED SHANKS' CONJECTURE WITH LAN-DAU'S THEOREM

We generalise Fujii's result to higher derivatives.

Theorem 4.1. For X a fixed positive real number and T sufficiently large, we have

$$\begin{split} \sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) X^{\rho} &= \\ & \left(-1 \right)^{n} \left\{ \Delta(X) \frac{T}{2\pi} \left((\log X)^{n} \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) - \sum_{mr = X} \Lambda(r) (\log m)^{n} \right) \\ & + X (\log X)^{n} \left(\frac{1}{2} \log X - \frac{\pi i}{4} \right) \sum_{m \le \frac{T}{2\pi X}} e^{2\pi i m X} + \frac{X}{2} (\log X)^{n} \sum_{m \le \frac{T}{2\pi X}} e^{2\pi i m X} \log m \\ & - X \sum_{mr \le \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) (\log r X)^{n} \right\} + O\left(T^{1/2} (\log T)^{n+2} \right) . \end{split}$$

where $\Delta(X)$ is given in (4.2).

Remark. Setting n = 1 in the theorem recovers Fujii's result in (4.1).

When we restrict X to being a positive integer we obtain a special case of the above results. It is evident from the statement of the following corollary that we see the changing behaviour of our asymptotic expansions depending on whether X > 0 is any real number or when $X \ge 1$ is an integer.

Corollary 4.2. If X is a positive integer and T is sufficiently large, we have

$$\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} = (-1)^{n+1} \frac{T}{2\pi} (\log X)^n \left\{ \sum_{k=0}^n \sum_{u=0}^{k+1} \binom{n}{k} \binom{k+1}{u} (-1)^u \frac{1}{k+1} \left(\log \frac{T}{2\pi} \right)^{k+1-u} (\log X)^{u-k} + \sum_{k=0}^n \sum_{u=0}^k \sum_{u=0}^{k-l} \binom{n}{k} \binom{k}{l} \binom{k-l}{u} (-1)^l l! \left(-1 + \sum_{j=0}^l \frac{1}{j!} \gamma_j \right) \left(\log \frac{T}{2\pi} \right)^{k-l-u} (\log X)^{u-k} + \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} k! A_k (\log X)^{-k} - \left(\log \frac{T}{2\pi} - 1 - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \right\} + E_n(T).$$

where $E_n(T)$ is given in Theorem 3.1. The γ_j are the coefficients in the Laurent expansion for $\zeta(s)$ about s = 1, and the A_j are the coefficients in the Laurent expansion for the logarithmic derivative of $\zeta(s)$ about s = 1, given in (3.3).

Remark. Setting n = 1 in the corollary recovers Fujii's result in (4.3). Setting n = 1 and X = 1 recovers Fujii's result in (2.3). Setting X = 1 recovers our result from [186], stated in Theorem 3.1.

4.2.1 Overview of the Proof

So far we have described the motivation for studying asymptotic expansions of the sums given in Section 4.1. We have stated the main results in Section 4.2 that we will prove in the following sections.

In Section 4.2.2 we will recall some basic facts about the Riemann zeta function $\zeta(s)$ and the Riemann xi function $\xi(s)$, which is a 'completed' version of the Riemann zeta function.

In Section 4.2.3 we will use the tools from Section 4.2.2 to prove Theorem 4.1. The integral we use to prove this theorem is given by

$$I = \frac{1}{2\pi i} \int_R \frac{\xi'}{\xi}(s) \zeta^{(n)}(s) X^s \, ds$$

where $\xi(s)$ is the Riemann xi function, $\zeta^{(n)}(s)$ denotes the n^{th} derivative of $\zeta(s)$ and R denotes the rectangular positively oriented contour with the vertices are c + i, c + iT, 1 - iT

c + it, 1 - c + i connected in this order and $c = 1 + \frac{1}{\log T}$. The non-trivial zeros of $\zeta(s)$ up to a height T are contained within R and so by Cauchy's Theorem the integral represents the summation

$$I = \sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) X^{\rho}.$$

We split this section up into several subsections, each corresponding to different parts of the contour that we will be integrating over, to show that most of the contribution comes from the left-hand side of the contour, while the other sides mostly only contribute to the error term (apart from one term from the right-hand side).

Finally in Section 4.2.4 we will prove Corollary 4.2 which will highlight the differences between the general case proved in Section 4.2.3 for any positive number X and the case when X is a positive integer. As described above we will use a result from [186] to do most of the work here. This will again highlight the observation of Gonek's in [144] that the asymptotic formulae tend to change quite dramatically depending on the arithmetic nature of X.

4.2.2 Preliminary Lemmas

In this section we recall some basic information about $\zeta(s)$ and $\xi(s)$, as well as recalling some results from other papers that will be useful in our proof. Any facts that are not explicitly referenced in this section can be found in any good text about the Riemann zeta function, for example Titchmarsh [298].

We state a more general functional equation for $\zeta^{(n)}(s)$ that is proved using the functional equation (1.1) for $\zeta(s)$ and the Leibniz product rule.

Lemma 4.3. The general functional equation for $\zeta^{(n)}(s)$ is given by the following formula

$$\zeta^{(n)}(s) = \frac{1}{\chi(1-s)} \sum_{k=0}^{n} \binom{n}{k} (-1)^k \frac{\chi^{(n-k)}(s)}{\chi(s)} \zeta^{(k)}(1-s)$$

We will also need the following result that we proved in Lemma 3.7 in Section 3.1.3 for the integral along the top and the bottom of our contour.

Lemma 4.4. For $c = 1 + \frac{1}{\log T}$, we have

$$\int_{1-c}^{c} |\zeta^{(n)}(\sigma + iT)| \, d\sigma \ll T^{1/2} (\log T)^n.$$

For the Riemann xi function we write

$$\xi(s) = \frac{1}{2}s(s-1)\pi^{-s/2}\Gamma\left(\frac{s}{2}\right)\zeta(s)$$

so the functional equation for $\xi(s)$ is given by

$$\xi(s) = \xi(1-s).$$

We now observe that

$$\frac{\xi'}{\xi}(s) = \frac{2s-1}{s(s-1)} - \frac{1}{2}\log\pi + \frac{1}{2}\psi\left(\frac{s}{2}\right) + \frac{\zeta'}{\zeta}(s) \tag{4.4}$$

where we have written

$$\psi(s) = \frac{\Gamma'}{\Gamma}(s)$$

and for $|\arg s| < \pi - \delta$ with arbitrarily fixed positive δ and for $|s| \ge \frac{1}{2}$ we have

$$\psi(s) = \log s + O\left(\frac{1}{|s|}\right) = \log t + \frac{\pi i}{2} + O\left(\frac{1}{t}\right)$$

as $t \to \infty$.

Combining these two observations, we have the following lemma.

Lemma 4.5. With the conditions written above, we have

$$\frac{\xi'}{\xi}(s) = \frac{1}{2}\log\frac{t}{2\pi} + \frac{\pi i}{4} + \frac{\zeta'}{\zeta}(s) + O\left(\frac{1}{t}\right).$$

A key component to upcoming the proof is the method of stationary phase. Applications of this method to these types of problems can be found in Gonek [143, Sect.4, p.131], in Levinson [222] and in Jakhlouti and Mazhouda [202, Sect.2, p.13], amongst other places, including throughout this thesis. In an entirely analogous way to the proof that Gonek writes in [143], we are able to prove the following result.

Lemma 4.6. Let X be a fixed positive real. Let $\{b_m\}_{m=1}^{\infty}$ be a sequence of complex numbers such that for any $\epsilon > 0, b_m \ll m^{\epsilon}$. Let c > 1 and let $k \ge 0$ be an integer. Then for T sufficiently large, we have

$$\frac{1}{2\pi} \int_{1}^{T} \chi(1 - c - it) \left(\sum_{m=1}^{\infty} b_m m^{-c - it} \right) \left(\log \frac{t}{2\pi} \right)^k e^{-it \log X} dt$$
$$= X^c \sum_{1 \le m \le \frac{T}{2\pi X}} b_m (\log mX)^k e^{-2\pi i mX} + O\left(T^{c - 1/2} (\log T)^k \right)$$

4.2.3 Proof of Theorem 4.1

Let X be a fixed positive real. We write $s = \sigma + it$ with $\sigma, t \in \mathbb{R}$ and $\rho = \beta + i\gamma$ for a non-trivial zero of the Riemann zeta function $\zeta(s)$. Suppose $T > T_0$ and T is not the imaginary part of the zeros of $\zeta(s)$ and further that $|T - \gamma| \gg \frac{1}{\log T}$ where γ is the imaginary part of any non-trivial zero ρ . This restriction on T is harmless within our remainder terms.

Set $c = 1 + \frac{1}{\log T}$ and consider the integral

$$I = \frac{1}{2\pi i} \int_{R} \frac{\xi'}{\xi}(s) \zeta^{(n)}(s) X^{s} \, ds \tag{4.5}$$

where $\xi(s)$ is the Riemann xi function, $\zeta^{(n)}(s)$ denotes the n^{th} derivative of $\zeta(s)$, and where R denotes the rectangular positively oriented contour with vertices given by c + i, c + iT, 1 - c + iT, 1 - c + i, connected in this order.

By Cauchy's Theorem,

$$I = \sum_{0 < \gamma \le T} \zeta^{(n)}(\rho) X^{\rho}.$$

$$\tag{4.6}$$

We now need to evaluate I in another way to obtain our asymptotic expansion. We decompose the integral (4.5) along the sides of the contour as

$$I = \frac{1}{2\pi i} \left(\int_{1-c+i}^{c+i} + \int_{c+i}^{c+iT} + \int_{c+iT}^{1-c+iT} + \int_{1-c+iT}^{1-c+i} \right) \frac{\xi'}{\xi}(s) \zeta^{(n)}(s) X^s \ ds$$

= $I_B + I_R + I_T + I_L.$

4.2.3.1 Bounding I_B and I_T

Notice that by our choice of T we may bound I_B and I_T trivially within our error. To do this, recall that for $-1 \le \sigma \le 2$ and since we are assuming $|T - \gamma| \gg \frac{1}{\log T}$, we have from [143, Sect.2, p.127] that

$$\frac{\xi'}{\xi}(s) \ll (\log T)^2.$$

Combining this with Lemma 4.4, since X is fixed we have

$$I_B + I_T \ll (\log T)^2 \int_{1-c}^{c} |\zeta^{(n)}(\sigma + it) X^{\sigma + it}| \, d\sigma \ll T^{1/2} (\log T)^{n+2}.$$
(4.7)

4.2.3.2 Evaluating I_R

Writing s = c + it and using (4.4), we may write

$$I_R = \frac{1}{2\pi} \int_1^T \frac{\xi'}{\xi} (c+it) \zeta^{(n)}(c+it) X^{c+it} dt$$

= $\frac{1}{2\pi} \int_1^T \left(\frac{2(c+it)-1}{(c+it)(c+it-1)} - \frac{1}{2} \log \pi + \psi \left(\frac{c+it}{2} \right) + \frac{\zeta'}{\zeta} (c+it) \right) \zeta^{(n)}(c+it) X^{c+it} dt.$

Using the Dirichlet series given in (3.7), and Lemma 4.5, we may rewrite this as

$$\begin{split} I_R &= \frac{(-1)^n}{2\pi} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4} - \sum_{r=1}^\infty \frac{\Lambda(r)}{r^{c+it}} + O\left(\frac{1}{t}\right) \right) \sum_{m=1}^\infty \frac{(\log m)^n}{m^{c+it}} X^{c+it} \ dt \\ &= \frac{(-1)^n}{2\pi} X^c \sum_{m=1}^\infty \frac{(\log m)^n}{m^c} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4} + O\left(\frac{1}{t}\right) \right) \left(\frac{X}{m} \right)^{it} \ dt \\ &+ \frac{(-1)^{n+1}}{2\pi} X^c \sum_{m=1}^\infty \frac{(\log m)^n}{m^c} \sum_{r=1}^\infty \frac{\Lambda(r)}{r^c} \int_1^T \left(\frac{X}{mr} \right)^{it} \ dt \\ &= I_{R,1} + I_{R,2}. \end{split}$$

Consider $I_{R,1}$ first. We have (with $\Delta(X)$ defined as in (4.2)),

$$\begin{split} I_{R,1} &= \frac{(-1)^n}{2\pi} X^c \sum_{m=1}^{\infty} \frac{(\log m)^n}{m^c} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4} + O\left(\frac{1}{t}\right)\right) \left(\frac{X}{m}\right)^{it} \, dt \\ &= \frac{(-1)^n}{2\pi} \Delta(X) (\log X)^n \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4}\right) \, dt \\ &+ \frac{(-1)^n}{2\pi} X^c \sum_{\substack{m=1\\m \neq X}}^{\infty} \frac{(\log m)^n}{m^c} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4}\right) \left(\frac{X}{m}\right)^{it} \, dt \\ &+ O\left((\log T)^{n+2}\right) \\ &= I_{R,1,1} + I_{R,1,2} + O\left((\log T)^{n+2}\right). \end{split}$$

Firstly,

$$I_{R,1,1} = (-1)^n \Delta(X) (\log X)^n \frac{T}{2\pi} \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4}\right) + O(1).$$

Next, integrating by parts and summing we obtain

$$I_{R,1,2} \ll X^c \sum_{\substack{m=1\\m \neq X}}^{\infty} \frac{(\log m)^n}{m^c} \frac{\log T}{\left|\log \frac{X}{m}\right|} \ll (\log T)^{n+2}.$$

If 0 < X < 1 we can do slightly better than this error term. Recombining, we have

$$I_{R,1} = (-1)^n \Delta(X) (\log X)^n \frac{T}{2\pi} \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) + O\left((\log T)^{n+2} \right).$$
(4.8)

Now consider $I_{R,2}$. We have

$$\begin{split} I_{R,2} &= \frac{(-1)^{n+1}}{2\pi} X^c \sum_{m=1}^{\infty} \frac{(\log m)^n}{m^c} \sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^c} \int_1^T \left(\frac{X}{mr}\right)^{it} dt \\ &= \frac{(-1)^{n+1}}{2\pi} X^c \sum_{k=1}^{\infty} \frac{1}{k^c} \sum_{k=mr} (\log m)^n \Lambda(r) \int_1^T \left(\frac{X}{k}\right)^{it} dt \\ &= \frac{(-1)^{n+1}}{2\pi} \Delta(X) \sum_{\substack{X=mr}} (\log m)^n \Lambda(r) \int_1^T 1 dt \\ &+ \frac{(-1)^{n+1}}{2\pi} X^c \sum_{\substack{k=1\\k\neq X}}^{\infty} \frac{1}{k^c} \sum_{k=mr} (\log m)^n \Lambda(r) \int_1^T \left(\frac{X}{k}\right)^{it} dt \\ &= I_{R,2,1} + I_{R,2,2}. \end{split}$$

Then

$$I_{R,2,1} = (-1)^{n+1} \Delta(X) \frac{T}{2\pi} \sum_{X=mr} (\log m)^n \Lambda(r) + O(1)$$

and as with the case for $I_{R,1,2}$ above, we have

$$I_{R,2,2} \ll X^c \sum_{\substack{k=1\\k \neq X}}^{\infty} \frac{1}{k^c} \sum_{k=mr} (\log m)^n \Lambda(r) \frac{1}{\log |X/k|} \ll (\log T)^{n+2}$$

where again the error can be improved slightly for 0 < X < 1. Combining, we have

$$I_{R,2} = (-1)^{n+1} \Delta(X) \frac{T}{2\pi} \sum_{X=mr} (\log m)^n \Lambda(r) + O\left((\log T)^{n+2}\right).$$
(4.9)

Finally, we may combine (4.8) and (4.9) to obtain I_R , given by

$$I_R = (-1)^n \Delta(X) \frac{T}{2\pi} \left((\log X)^n \left\{ \frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right\} - \sum_{X=mr} (\log m)^n \Lambda(r) \right) + O\left((\log T)^{n+2} \right).$$
(4.10)

4.2.3.3 Evaluating I_L

Finally we evaluate I_L , which is where most of the contribution to the asymptotic expansion comes from. Using Lemma 4.3 we have

By complex conjugation we obtain

$$\begin{split} \overline{I_L} &= \frac{X^{1-c}}{2\pi} \int_1^T \frac{\xi'}{\xi} (c+it) \left(\frac{1}{\chi(c+it)} \sum_{k=0}^n \binom{n}{k} (-1)^k \frac{\chi^{(n-k)} (1-c-it)}{\chi(1-c-it)} \zeta^{(k)} (c+it) \right) X^{-it} dt \\ &= \frac{X^{1-c}}{2\pi} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4} - \sum_{r=1}^\infty \frac{\Lambda(r)}{r^{c+it}} + O\left(\frac{1}{t}\right) \right) \chi(1-c-it) \\ &\quad \times \left(\sum_{k=0}^n \binom{n}{k} (-1)^k \left((-1)^{n-k} \left(\log \frac{t}{2\pi} \right)^{n-k} + O\left(\frac{1}{t}\right) \right) (-1)^k \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} \right) X^{-it} dt \\ &= \frac{X^{1-c}}{2\pi} \int_1^T \left(\frac{1}{2} \log \frac{t}{2\pi} + \frac{\pi i}{4} - \sum_{r=1}^\infty \frac{\Lambda(r)}{r^{c+it}} \right) \chi(1-c-it) \\ &\quad \times \left(\sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} \right) e^{-it\log X} dt + O\left(T^{1/2} (\log T)^{n+2}\right) \end{split}$$

where the second line follows from Lemma 4.5.

We now split this integral and evaluate each part separately.

$$\begin{split} \overline{I_L} &= \frac{X^{1-c}}{2\pi} \frac{1}{2} \int_1^T \chi (1-c-it) \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \left(\log \frac{t}{2\pi} \right)^{n-k+1} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} e^{-it\log X} dt \\ &+ \frac{X^{1-c}}{2\pi} \frac{\pi i}{4} \int_1^T \chi (1-c-it) \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} e^{-it\log X} dt \\ &+ \frac{X^{1-c}}{2\pi} \int_1^T \chi (1-c-it) \sum_{r=1}^\infty \frac{\Lambda(r)}{r^{c+it}} \sum_{k=0}^n \binom{n}{k} (-1)^{n+k+1} \\ &\times \left(\log \frac{t}{2\pi} \right)^{n-k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^{c+it}} e^{-it\log X} dt + O\left(T^{1/2} (\log T)^{n+2} \right) \\ &= J_1 + J_2 + J_3 + O\left(T^{1/2} (\log T)^{n+2} \right). \end{split}$$

4.2. Combining the Generalised Shanks' Conjecture with Landau's Theorem 139

Now apply Lemma 4.6 to each of the summations J_k , k = 1, 2, 3. For J_1 , we have

$$\begin{split} J_1 &= \frac{X^{1-c}}{2\pi} \frac{1}{2} \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^c} \\ &\qquad \times \int_1^T \chi (1-c-it) \left(\log \frac{t}{2\pi} \right)^{n-k+1} e^{-it\log(mX)} \, dt \\ &= \frac{X}{2} \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \sum_{m \leq \frac{T}{2\pi X}} (\log m)^k (\log mX)^{n-k+1} e^{-2\pi imX} + O\left(T^{1/2}(\log T)^{n+1}\right) \\ &= \frac{X}{2} \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi imX} \left[\sum_{k=0}^n \binom{n}{k} (-1)^{n-k} (\log m)^k (\log mX)^{n-k} \right] (\log(mX)) \\ &\quad + O\left(T^{1/2}(\log T)^{n+1}\right) \\ &= \frac{X}{2} \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi imX} (\log mX) \left[-(\log mX) + \log m \right]^n + O\left(T^{1/2}(\log T)^{n+1}\right) \\ &= (-1)^n \frac{X}{2} \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi imX} (\log mX) (\log X)^n + O\left(T^{1/2}(\log T)^{n+1}\right) \\ &= (-1)^n \frac{X}{2} (\log X)^{n+1} \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi imX} \\ &\quad + (-1)^n \frac{X}{2} (\log X)^n \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi imX} \log m + O\left(T^{1/2}(\log T)^{n+1}\right). \end{split}$$

Similarly,

$$\begin{split} J_2 &= \frac{X^{1-c}}{2\pi} \frac{\pi i}{4} \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \sum_{m=1}^\infty \frac{(\log m)^k}{m^c} \int_1^T \chi (1-c-it) \left(\log \frac{t}{2\pi}\right)^{n-k} e^{-it\log(mX)} dt \\ &= \frac{\pi i}{4} X \sum_{k=0}^n \binom{n}{k} (-1)^{n+k} \sum_{m \leq \frac{T}{2\pi X}} (\log m)^k (\log mX)^{n-k} e^{-2\pi i mX} + O\left(T^{1/2} (\log T)^n\right) \\ &= \frac{\pi i}{4} X \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi i mX} \left[\sum_{k=0}^n \binom{n}{k} (-1)^{n+k} (\log m)^k (\log mX)^{n-k} \right] + O\left(T^{1/2} (\log T)^n\right) \\ &= \frac{\pi i}{4} X \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi i mX} \left[\log m - (\log mX) \right]^n + O\left(T^{1/2} (\log T)^n\right) \\ &= (-1)^n \frac{\pi i}{4} X (\log X)^n \sum_{m \leq \frac{T}{2\pi X}} e^{-2\pi i mX} + O\left(T^{1/2} (\log T)^n\right). \end{split}$$

Finally,

$$\begin{split} J_{3} &= \frac{X^{1-c}}{2\pi} \sum_{k=0}^{n} \binom{n}{k} (-1)^{n+k+1} \sum_{r=1}^{\infty} \frac{\Lambda(r)}{r^{c}} \sum_{m=1}^{\infty} \frac{(\log m)^{k}}{m^{c}} \\ &\times \int_{1}^{T} \chi (1-c-it) \left(\log \frac{t}{2\pi} \right)^{n-k} e^{-it \log(mrX)} dt \\ &= X \sum_{k=0}^{n} \binom{n}{k} (-1)^{n+k+1} \sum_{mr \leq \frac{T}{2\pi X}} \Lambda(r) (\log m)^{k} (\log mrX)^{n-k} e^{-2\pi i mrX} \\ &+ O \left(T^{1/2} (\log T)^{n} \right) \\ &= -X \sum_{mr \leq \frac{T}{2\pi X}} e^{-2\pi i mrX} \Lambda(r) \left(1 \sum_{k=0}^{n} \binom{n}{k} (-1)^{n+k} (\log m)^{k} (\log mrX)^{n-k} \right] \\ &+ O \left(T^{1/2} (\log T)^{n} \right) \\ &= -X \sum_{mr \leq \frac{T}{2\pi X}} e^{-2\pi i mrX} \Lambda(r) \left[-(\log mrX) + \log m \right]^{n} + O \left(T^{1/2} (\log T)^{n} \right) \\ &= (-1)^{n+1} X \sum_{mr \leq \frac{T}{2\pi X}} e^{-2\pi i mrX} \Lambda(r) (\log rX)^{n} + O \left(T^{1/2} (\log T)^{n} \right). \end{split}$$

Recombining these, we have

$$\overline{I_L} = (-1)^n \left\{ X (\log X)^n \left(\frac{1}{2} \log X + \frac{\pi i}{4} \right) \sum_{m \le \frac{T}{2\pi X}} e^{-2\pi i m X} + \frac{1}{2} X (\log X)^n \sum_{m \le \frac{T}{2\pi X}} e^{-2\pi i m X} \log m - X \sum_{mr \le \frac{T}{2\pi X}} e^{-2\pi i m r X} \Lambda(r) (\log r X)^n \right\} + O\left(T^{1/2} (\log T)^{n+2} \right).$$

Taking complex conjugates,

$$I_{L} = (-1)^{n} \left\{ X (\log X)^{n} \left(\frac{1}{2} \log X - \frac{\pi i}{4} \right) \sum_{m \le \frac{T}{2\pi X}} e^{2\pi i m X} + \frac{X}{2} (\log X)^{n} \sum_{m \le \frac{T}{2\pi X}} e^{2\pi i m X} \log m - X \sum_{mr \le \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) (\log r X)^{n} \right\} + O\left(T^{1/2} (\log T)^{n+2} \right).$$
(4.11)

4.2. Combining the Generalised Shanks' Conjecture with Landau's Theorem 141

4.2.3.4 Finalising the Proof of Theorem 4.1

Combining (4.7), (4.10), and (4.11) gives I in the second way that we are looking for. Specifically, we have

$$\begin{split} I &= (-1)^n \left\{ \Delta(X) \frac{T}{2\pi} \left((\log X)^n \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \\ &+ X (\log X)^n \left(\frac{1}{2} \log X - \frac{\pi i}{4} \right) \sum_{m \leq \frac{T}{2\pi X}} e^{2\pi i m X} \\ &+ \frac{X}{2} (\log X)^n \sum_{m \leq \frac{T}{2\pi X}} e^{2\pi i m X} \log m - X \sum_{mr \leq \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) (\log r X)^n \right\} \\ &+ O\left(T^{1/2} (\log T)^{n+2} \right), \end{split}$$

Combining this asymptotic expansion with our observation that $I = \sum_{0 < \gamma \leq T} \zeta^{(n)}(\rho) X^{\rho}$ in (4.6) completes the proof of Theorem 4.1.

4.2.4 Proof of Corollary 4.2

We rewrite Theorem 4.1 in a slightly different way as follows.

Corollary 4.7. For X a fixed positive real number, we have

$$\begin{split} \sum_{0 < \gamma \leq T} \zeta^{(n)}(\rho) X^{\rho} &= \\ & \left(-1 \right)^{n} \left\{ \Delta(X) \frac{T}{2\pi} \left((\log X)^{n} \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) - \sum_{mr=X} \Lambda(r) (\log m)^{n} \right) \right. \\ & + X (\log X)^{n} \left(\frac{1}{2} \log X - \frac{\pi i}{4} \right) \sum_{m \leq \frac{T}{2\pi X}} e^{2\pi i m X} + \frac{X}{2} (\log X)^{n} \sum_{m \leq \frac{T}{2\pi X}} e^{2\pi i m X} \log m \\ & - X \sum_{k=0}^{n} \binom{n}{k} (\log X)^{n-k} \sum_{mr \leq \frac{T}{2\pi X}} e^{2\pi i m r X} \Lambda(r) (\log r)^{k} \right\} + O\left(T^{1/2} (\log T)^{n+2} \right), \end{split}$$

where $\Delta(X)$ is given in Theorem 4.1.

Proof. This follows from Theorem 4.1 by using the binomial expansion on $(\log rX)^n$ in the last summation in the braces in Theorem 4.1.

Remark. The advantage to rewriting Theorem 4.1 in the form of Corollary 4.7 is that none of the summations involving exponentials have any reliance on powers of $\log X$.

When $X \ge 1$ and $X \in \mathbb{Z}$, notice that

$$\sum_{\substack{m \le \frac{T}{2\pi X}}} e^{2\pi i m X} = \frac{T}{2\pi X} + O(1),$$

$$\sum_{\substack{m \leq \frac{T}{2\pi X}}} e^{2\pi i m X} \log m = \frac{T}{2\pi X} \log \frac{T}{2\pi X} - \frac{T}{2\pi X} + O(\log T),$$
$$\sum_{\substack{mr \leq \frac{T}{2\pi X}}} e^{2\pi i m r X} \Lambda(r) (\log r)^k = \sum_{\substack{mr \leq \frac{T}{2\pi X}}} \Lambda(r) (\log r)^k.$$

Define

$$S = (-1)^{k+1} \sum_{mr \le \frac{T}{2\pi X}} \Lambda(r) (\log r)^k$$
(4.12)

Then in an entirely analogous way to that done in Chapter 3 and in Hughes and Pearce-Crump [186], we have

$$S = (-1)^{k+1} \frac{1}{k+1} \frac{T}{2\pi X} \left(\log \frac{T}{2\pi X} \right)^{k+1}$$

+ $(-1)^{k+1} \sum_{l=0}^{k} \binom{k}{l} (-1)^{l} l! \left(-1 + \sum_{j=0}^{l} \frac{1}{j!} \gamma_{j} \right) \frac{T}{2\pi X} \left(\log \frac{T}{2\pi X} \right)^{k-l}$
+ $k! A_{k} \frac{T}{2\pi X} + E_{k}(T)$

where

$$E_k(T) = \begin{cases} O\left(Te^{-C\sqrt{\log T}}\right) & \text{unconditionally} \\ O\left(T(\log T)^{k+9/4}\right) & \text{under the Riemann Hypothesis} \end{cases}$$

and C > 0 is a constant. Multiplying S by $(-1)^{k+1}$ gives the summation that we were originally looking for.

Substituting these expressions into the asymptotic expansion from Corollary 4.7 gives

$$\begin{split} \sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} &= (-1)^n \left\{ \frac{T}{2\pi} \left((\log X)^n \left(\frac{1}{2} \log \frac{T}{2\pi} - \frac{1}{2} + \frac{\pi i}{4} \right) - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \\ &+ (\log X)^n \left(\frac{1}{2} \log X - \frac{\pi i}{4} \right) \frac{T}{2\pi} + \frac{1}{2} (\log X)^n \left(\frac{T}{2\pi} \log \frac{T}{2\pi X} - \frac{T}{2\pi} \right) \\ &- \sum_{k=0}^n \binom{n}{k} (\log X)^{n-k} \left(\frac{1}{k+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi X} \right)^{k+1} \right. \\ &+ \sum_{l=0}^k \binom{k}{l} (-1)^l l! \left(-1 + \sum_{j=0}^l \frac{1}{j!} \gamma_j \right) \frac{T}{2\pi} \left(\log \frac{T}{2\pi X} \right)^{k-l} \\ &+ (-1)^{k+1} k! A_k \frac{T}{2\pi} \right) \right\} + E_n(T), \end{split}$$

where the γ_j and A_k are defined in the statement of Corollary 4.2 and $E_n(T)$ is defined in Theorem 3.1.

142

4.2. Combining the Generalised Shanks' Conjecture with Landau's Theorem 143

We want to simplify this so that the $\log \frac{T}{2\pi X}$ terms can be written separately in terms of $\log \frac{T}{2\pi}$ and $\log X$. To do this, we begin by writing this previous expansion in a slightly different form and do some simplifying.

$$\begin{split} \sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} &= (-1)^n \frac{T}{2\pi} (\log X)^n \left(\log \frac{T}{2\pi} - 1 - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \\ &+ (-1)^{n+1} \sum_{k=0}^n \binom{n}{k} (\log X)^{n-k} \bigg\{ \frac{1}{k+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} - \log X \right)^{k+1} \\ &+ \sum_{l=0}^k \binom{k}{l} (-1)^l l! \left(-1 + \sum_{j=0}^l \frac{1}{j!} \gamma_j \right) \frac{T}{2\pi} \left(\log \frac{T}{2\pi} - \log X \right)^{k-l} \\ &+ (-1)^{k+1} k! A_k \frac{T}{2\pi} \bigg\} + E_n(T). \end{split}$$

Applying the binomial theorem, we have

$$\begin{split} \sum_{0 < \gamma \leq T} \zeta^{(n)}(\rho) X^{\rho} &= (-1)^{n} \frac{T}{2\pi} (\log X)^{n} \left(\log \frac{T}{2\pi} - 1 - \sum_{mr = X} \Lambda(r) (\log m)^{n} \right) \\ &+ (-1)^{n+1} \frac{T}{2\pi} \left\{ \sum_{k=0}^{n} \sum_{u=0}^{k+1} \binom{n}{k} \binom{k+1}{u} (-1)^{u} \frac{1}{k+1} \left(\log \frac{T}{2\pi} \right)^{k+1-u} (\log X)^{n-k+u} \right. \\ &+ \sum_{k=0}^{n} \sum_{l=0}^{k} \sum_{u=0}^{k-l} \binom{n}{k} \binom{k}{l} \binom{k-l}{u} (-1)^{l} l! \\ &\times \left(-1 + \sum_{j=0}^{l} \frac{1}{j!} \gamma_{j} \right) \left(\log \frac{T}{2\pi} \right)^{k-l-u} (\log X)^{n-k+u} \\ &+ \sum_{k=0}^{n} \binom{n}{k} (-1)^{k+1} k! A_{k} (\log X)^{n-k} \right\} + E_{n}(T). \end{split}$$

Finally, we may rearrange slightly to complete the proof of Corollary 4.2 to give

$$\begin{split} &\sum_{0<\gamma\leq T} \zeta^{(n)}(\rho) X^{\rho} = (-1)^{n+1} \frac{T}{2\pi} (\log X)^n \\ &\left\{ \sum_{k=0}^n \sum_{u=0}^{k+1} \binom{n}{k} \binom{k+1}{u} (-1)^u \frac{1}{k+1} \left(\log \frac{T}{2\pi} \right)^{k+1-u} (\log X)^{u-k} \\ &+ \sum_{k=0}^n \sum_{l=0}^k \sum_{u=0}^{k-l} \binom{n}{k} \binom{k}{l} \binom{k-l}{u} (-1)^l l! \left(-1 + \sum_{j=0}^l \frac{1}{j!} \gamma_j \right) \left(\log \frac{T}{2\pi} \right)^{k-l-u} (\log X)^{u-k} \\ &+ \sum_{k=0}^n \binom{n}{k} (-1)^{k+1} k! A_k \log^{-k} X - \left(\log \frac{T}{2\pi} - 1 - \sum_{mr=X} \Lambda(r) (\log m)^n \right) \right\} + E_n(T). \end{split}$$

— 5 —

Higher Moments of the Derivatives of the Riemann Zeta Function via RMT and the Hybrid Model

In this chapter we assume the Riemann Hypothesis, unless otherwise stated. Note that in this chapter we have divided through by N(T), while in most of the rest of the thesis we do not make this normalisation. This is because in this chapter we develop the heuristic RMT connection to model higher moments of zeta, and that model works at the level of averages.

We will develop the philosophy of Keating and Snaith [211] and Hughes, Keating and O'Connell [182] from RMT to study the discrete moments of the n^{th} derivative of $\zeta(s)$, denoted $\zeta^{(n)}(s)$. That is, we study

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right)^k,\tag{5.1}$$

where N(T) is given in (1.4). In fact, we will study the more general problem of moments of mixed derivatives,

$$\frac{1}{N(T)}\sum_{0<\gamma\leq T}\zeta^{(n_1)}\left(\frac{1}{2}+i\gamma\right)\ldots\zeta^{(n_k)}\left(\frac{1}{2}+i\gamma\right).$$

Clearly if $n_1 = \cdots = n_k = n$, then we recover (5.1).

Recall that the model proposed by Keating and Snaith [211] is that the characteristic polynomial of a large random unitary matrix can be used to model the value distribution of the Riemann zeta function near a large height T. They used the characteristic polynomial

$$Z_N(\theta) = \det(I - Ue^{-i\theta}) = \prod_{m=1}^N (1 - e^{i(\theta_m - \theta)}),$$
(5.2)

where the θ_m are the eigenangles of an $N \times N$ random unitary matrix U, to model $\zeta(s)$. The motivation for studying this is that it has been conjectured that the limiting distribution of the non-trivial zeros of $\zeta(s)$, on the scale of their mean spacing, is asymptotically the
same as that of the eigenangles θ_m of matrices chosen according to Haar measure, in the limit as $N \to \infty$, as previously discussed in Section 1.3.3. Recall that equating the mean densities yields the connection between matrix size and height up the critical line

$$N = \log \frac{T}{2\pi}.$$

Averaging over all Haar-distributed $N \times N$ unitary matrices, Keating and Snaith [211] calculated the moments of $|Z(\theta)|$, leading them to form Conjecture 1.9. Hughes, Keating and O'Connell [182] followed a similar approach and calculated the discrete moments of $|Z'(\theta)|$ and formed Conjecture 1.19.

In both of these conjectures, the arithmetic factor was inserted artificially after studying the analogous problem in RMT. The hybrid model was formed to fix this problem, treating the zeros and primes part of the Riemann zeta function equally, recreating Conjectures 1.9 and 1.19, as previously discussed in Section 1.3.4.

Both of the problems discussed considered moments of a real function, where an absolute value of the characteristic polynomial (and so zeta) was taken. For our problem (5.1) and its generalisation, we consider the moments of the complex function. In this chapter we attack our problem in two ways. First, by direct calculation of the characteristic polynomial, and then by modelling zeta using the hybrid approach. We also consider some problems relating to negative moments, and to a 'hidden' recurrence relation in our characteristic polynomial approach. The problem of considering lower order terms is taken up in Chapter 6.

Other than the work presented in Section 5.4 and in Section 5.5, the work presented in this chapter has already appeared in print in [187]. Notation and cross-references have been updated for this thesis.

5.1 STATEMENT OF RESULTS

In Section 5.2 we will calculate

$$\mathbb{E}_N\left[\frac{1}{N}\sum_{m=1}^N Z_N(\theta_m + \alpha_1)\dots Z_N(\theta_m + \alpha_k)\right]$$

where recall from Section 1.2.1 that \mathbb{E}_N denotes expectation with respect to Haar measure. By Taylor expanding each characteristic polynomial about their respective shift, we will be able to prove the following theorem. **Theorem 5.1.** For fixed positive integers n_1, \ldots, n_k , we have

$$\mathbb{E}_{N}\left[\frac{1}{N}\sum_{m=1}^{N}Z_{N}^{(n_{1})}(\theta_{m})\dots Z_{N}^{(n_{k})}(\theta_{m})\right]$$

~ $(-1)^{n_{1}+\dots+n_{k}+k} i^{n_{1}+\dots+n_{k}}\frac{n_{1}!\dots n_{k}!}{(n_{1}+\dots+n_{k}+1)!}N^{n_{1}+\dots+n_{k}},$

as $N \to \infty$.

Setting $n_1 = \cdots = n_k = n$ gives the following Corollary.

Corollary 5.2. For fixed positive integers n, k, we have

$$\mathbb{E}_{N}\left[\frac{1}{N}\sum_{m=1}^{N}Z_{N}^{(n)}(\theta_{m})^{k}\right] \sim (-1)^{k(n+1)} i^{kn}\frac{(n!)^{k}}{(kn+1)!}N^{kn}$$

as $N \to \infty$.

In order to use this to create a conjecture for zeta, we need to know what the arithmetic term equals. To do that, we take the hybrid approach, discussed in Section 5.3. In Section 5.3.1 (Theorems 5.8 and 5.9) we calculate the discrete moments of $P_X^{(n_1)} \dots P_X^{(n_k)}$ and show that they asymptotically vanish unless all the n_j are 0 (that is, no derivatives are taken), in which case the mean is 1. Furthermore, we predict that the discrete moments of Z_X are accurately modelled by the characteristic polynomial, and we will verify this in the case of the first derivative in Theorem 5.12 in Section 5.3.2. That is, we conjecture that there is no arithmetic term at the leading order for these moments of the complex Riemann zeta function.

Note that the derivative of $Z_N(\theta)$ is with respect to θ , and this acts like the derivative of $\zeta(1/2 + it)$ with respect to t. This is obviously equal to $i\zeta'(1/2 + it)$, and so we expect there to be no factors of i for the Riemann zeta function. To be precise, we make the following prediction.

Conjecture 5.3. For fixed positive integers n_1, \ldots, n_k , the moments of mixed derivatives of $\zeta(s)$, evaluated at the non-trivial zeros of $\zeta(s)$, are given by

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta^{(n_1)} \left(\frac{1}{2} + i\gamma\right) \dots \zeta^{(n_k)} \left(\frac{1}{2} + i\gamma\right) \\ \sim (-1)^{n_1 + \dots + n_k + k} \frac{n_1! \dots n_k!}{(n_1 + \dots + n_k + 1)!} \left(\log \frac{T}{2\pi}\right)^{n_1 + \dots + n_k}$$

as $T \to \infty$.

Setting $n_1 = \cdots = n_k = n$ gives the following conjecture.

Conjecture 5.4. For fixed positive integers n, k, the k^{th} moment of the n^{th} derivative of $\zeta(s)$ is given by

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right)^k \sim (-1)^{k(n+1)} \frac{(n!)^k}{(kn+1)!} \left(\log \frac{T}{2\pi}\right)^{kn}$$

as $T \to \infty$.

Very little is rigorously known about these types of results. For k = 0 the result is trivial, as the sum is just N(T), defined in equation (1.5). For k = 1, they have been previously studied in relation to Shanks' Conjecture, a full history of which can be found in Chapter 2, and in [184], and the theoretical results as given in Chapter 3 agree with Conjecture 5.4 in this case.

These results are for leading order only. In Chapter 6 and in a forthcoming paper [185], we use a different method to conjecture the full asymptotic for zeta.

These results and conjectures are stated for k a positive integer. However, in the case of the first derivative, when n = 1, Section 5.3.2 applies for $k \in \mathbb{C}$ with $\Re(k) > -3$, and some of our results in Section 5.3.1 (namely Theorem 5.9) also apply for non-integer k. Therefore it is plausible that in the case of the first derivative, Conjecture 5.4 also holds for complex k.

Conjecture 5.5. For $\Re(k) > -3$ as $T \to \infty$

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^k \sim \frac{1}{\Gamma(k+2)} \left(\log \frac{T}{2\pi}\right)^k.$$

In the case k = -1, this result is already known under the additional assumption of simple zeros (but without the Riemann Hypothesis), and can be found as equation (3.9) in a paper of Garaev and Sankaranarayanan, [134], and in (2.14).

When k = -2, the right hand side of Conjecture 5.5 vanishes. A separate calculation that will be shown explicitly in Section 5.4 shows that

$$\mathbb{E}_N\left[\frac{1}{N}\sum_{m=1}^N Z'_N(\theta_m)^{-2}\right] = 0$$

exactly. For the case of zeta, what we mean is the left hand side is bounded by some error terms with no leading-order asymptotic. An educated guess suggests the error terms will be $O(T^{-1/2+\varepsilon})$. We discuss this further in Section 5.4.

5.2 THE CHARACTERISTIC POLYNOMIAL APPROACH

In this section we prove Theorem 5.1 concerning moments of characteristic polynomials. We start with the product representation of the characteristic polynomial of an $N \times N$ random unitary matrix $Z_N(\theta)$ in (5.2), given by

$$Z_N(\theta) = \prod_{m=1}^N (1 - e^{i(\theta_m - \theta)}).$$

The Haar probability density of U(N) equals [311]

$$d \operatorname{Haar} = \frac{1}{N! (2\pi)^N} \prod_{1 \le j < \ell \le N} |e^{i\theta_j} - e^{i\theta_\ell}|^2 \ d\theta_1 \dots d\theta_N.$$
(5.3)

To extract the moments that we wish to calculate, we use shifts and calculate

$$\mathbb{E}_N\left[\frac{1}{N}\sum_{m=1}^N Z_N(\theta_m + \alpha_1)\dots Z_N(\theta_m + \alpha_k)\right].$$

By invariance of Haar measure there are no distinguished eigenvalues, so this expectation equals

$$\mathbb{E}_N \left[Z_N(\theta_N + \alpha_1) \dots Z_N(\theta_N + \alpha_k) \right].$$

Since we can Taylor expand assuming the α_j are very small, we obtain

$$\mathbb{E}_{N}\left[\left(Z_{N}'(\theta_{N})\frac{\alpha_{1}}{1!}+Z_{N}''(\theta_{N})\frac{\alpha_{1}^{2}}{2!}+\ldots+Z_{N}^{(n_{1})}(\theta_{N})\frac{\alpha_{1}^{n_{1}}}{n_{1}!}+\ldots\right)\right.\\ \left.\times\ldots\times\left(Z_{N}'(\theta_{N})\frac{\alpha_{k}}{1!}+Z_{N}''(\theta_{N})\frac{\alpha_{k}^{2}}{2!}+\ldots+Z_{N}^{(n_{k})}(\theta_{N})\frac{\alpha_{k}^{n_{k}}}{n_{k}!}+\ldots\right)\right],$$

where obviously $Z_N(\theta_N) = 0$. By calculating the $\alpha_1^{n_1} \dots \alpha_k^{n_k}$ -th coefficient, we can find moments of the mixed derivatives of the characteristic polynomial.

Writing out the expectation in terms of the Weyl density (5.3), we have

$$\mathbb{E}_N\left[\prod_{r=1}^k Z_N(\theta_N + \alpha_r)\right] = \frac{1}{N!(2\pi)^N} \int \cdots \int_0^{2\pi} \prod_{1 \le j < \ell \le N} |e^{i\theta_j} - e^{i\theta_\ell}|^2 \prod_{m=1}^N \prod_{r=1}^k (1 - e^{i\theta_m} e^{-i\theta_N} e^{-i\alpha_r}) \ d\theta_m.$$

We split off the terms in the first product with $\ell = N$, and all the terms in the second product with m = N (which can then be pulled out of the integral) to give

$$\frac{1}{N!(2\pi)^N} \prod_{r=1}^k (1 - e^{-i\alpha_r}) \int \cdots \int_0^{2\pi} \prod_{1 \le j < \ell \le N-1} |e^{i\theta_j} - e^{i\theta_\ell}|^2 \times \prod_{m=1}^{N-1} |e^{i\theta_m} - e^{i\theta_N}|^2 \prod_{r=1}^k (1 - e^{i\theta_m} e^{-i\theta_N} e^{-i\alpha_r}) \ d\theta_m.$$

We may make a change of variables $\phi_m = \theta_m - \theta_N$ for m = 1, ..., N - 1 and trivially perform the θ_N integral to obtain

$$\frac{1}{N!(2\pi)^{N-1}} \prod_{r=1}^{k} (1-e^{-i\alpha_r}) \int \cdots \int_0^{2\pi} \prod_{1 \le j < \ell \le N-1} \left| e^{i\phi_j} - e^{i\phi_\ell} \right|^2 \\ \times \prod_{m=1}^{N-1} (1-e^{i\phi_m})(1-e^{-i\phi_m}) \prod_{r=1}^k (1-e^{i\phi_m}e^{-i\alpha_r}) \ d\phi_m.$$

Notice then that we can rewrite the integrals back in terms of Haar measure for $(N-1) \times (N-1)$ unitary matrices, obtaining

$$\frac{1}{N} \prod_{r=1}^{k} (1 - e^{-i\alpha_r}) \times \mathbb{E}_{N-1} \left[\prod_{m=1}^{N-1} (1 - e^{i\phi_m})(1 - e^{-i\phi_m}) \prod_{r=1}^{k} (1 - e^{i\phi_m} e^{-i\alpha_r}) \right] \\ = \frac{(-1)^{(k+1)(N-1)}}{N} \prod_{r=1}^{k} A_r^{-N} (A_r - 1) \times \mathbb{E}_{N-1} \left[\prod_{m=1}^{N-1} e^{-i\phi_m} (e^{i\phi_m} - 1)^2 \prod_{r=1}^{k} (e^{i\phi_m} - A_r) \right]$$
(5.4)

where we have done some simple algebraic manipulations to put it into a suitable form for later, and written $A_r = e^{i\alpha_r}$. We may then use Heine's Lemma [295] to write the expectation as a Toeplitz determinant, and thus we have

$$\mathbb{E}_{N-1}\left[\prod_{m=1}^{N-1} e^{-i\phi_m} (e^{i\phi_m} - 1)^2 \prod_{r=1}^k (e^{i\phi_m} - A_r)\right] = D_{N-1}[f]$$

where

$$D_{N-1}[f] := \det \left| \hat{f}_{j-\ell} \right|_{1 \le j,\ell \le N-1}$$

with

$$\hat{f}_{j-\ell} = \frac{1}{2\pi} \int_0^{2\pi} f(\vartheta) e^{-i(j-\ell)\vartheta} \, d\vartheta$$

for the symbol

$$f(z) = z^{-1}(z-1)^2 \prod_{r=1}^{k} (z-A_r).$$
(5.5)

As a historical aside, though the identity was first written down in 1939 by Szegő [295], he gave the credit to Heine (who lived from 1821 to 1881).

Putting this together, we have

$$\mathbb{E}_{N}\left[\prod_{r=1}^{k} Z_{N}(\theta_{N} + \alpha_{r})\right] = \frac{(-1)^{(k+1)(N-1)}}{N} \prod_{r=1}^{k} A_{r}^{-N} \left(A_{r} - 1\right) \times D_{N-1}[f].$$
(5.6)

Basor and Forrester [21] evaluate $D_{N-1}[f]$ with symbols of the form (5.5) as a $(k + 2) \times (k + 2)$ determinant,

$$\begin{split} D_{N-1}[f] &= \frac{(-1)^{(k+1)(N-1)} \prod_{r=1}^{k+2} (-A_r)}{\prod_{1 \le j < \ell \le k+2} (A_\ell - A_j)} \\ & \times \det \begin{bmatrix} A_1^{N-1} & A_1^N & A_1^{N+1} & \dots & A_1^{N-1+k} & (-A_1)^{-1} \\ A_2^{N-1} & A_2^N & A_2^{N+1} & \dots & A_2^{N-1+k} & (-A_2)^{-1} \\ A_3^{N-1} & A_3^N & A_3^{N+1} & \dots & A_3^{N-1+k} & (-A_3)^{-1} \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ A_k^{N-1} & A_k^N & A_k^{N+1} & \dots & A_k^{N-1+k} & (-A_k)^{-1} \\ A_{k+2}^{N-1} & A_{k+2}^N & A_{k+2}^{N+1} & \dots & A_{k+2}^{N-1+k} & (-A_{k+2})^{-1} \end{bmatrix}, \end{split}$$

where we write A_{k+1} and A_{k+2} for now, but will let them both tend to 1 later.

We can manipulate this determinant through elementary row operations. We begin by factoring A_r^{N-1} from each row, and factor the -1 from the last column. This determinant becomes

$$D_{N-1}[f] = \frac{(-1)^{N(k+1)} \prod_{r=1}^{k+2} A_r^N}{\prod_{1 \le j < \ell \le k+2} (A_\ell - A_j)} \det \begin{bmatrix} 1 & A_1 & A_1^2 & \dots & A_1^k & A_1^{-N} \\ 1 & A_2 & A_2^2 & \dots & A_2^k & A_2^{-N} \\ 1 & A_3 & A_3^2 & \dots & A_3^k & A_3^{-N} \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 1 & A_k & A_k^2 & \dots & A_k^{N-1+k} & A_k^{-N} \\ 1 & A_{k+1} & A_{k+1}^2 & \dots & A_{k+1}^k & A_{k+1}^{-N} \\ 1 & A_{k+2} & A_{k+2}^2 & \dots & A_{k+2}^k & A_{k+2}^{-N} \end{bmatrix}$$

Subtract the first row from each other row, and bring the factor of $1/(A_r - A_1)$ inside

the determinant for each row $r = 2, \ldots k + 2$ to show that $D_{N-1}[f]$ equals

$$\frac{(-1)^{N(k+1)}\prod_{r=1}^{k+2}A_r^N}{\prod_{2\leq j<\ell\leq k+2}(A_\ell - A_j)} \det \begin{bmatrix} 1 & A_1 & A_1^2 & \dots & A_1^k & A_1^{-N} \\ 0 & 1 & A_1 + A_2 & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_2^{k-1-j_1} & C_{2,2} \\ 0 & 1 & A_1 + A_3 & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_3^{k-1-j_1} & C_{3,2} \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & 1 & A_1 + A_k & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_k^{k-1-j_1} & C_{k,2} \\ 0 & 1 & A_1 + A_{k+1} & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_{k+1}^{k-1-j_1} & C_{k+1,2} \\ 0 & 1 & A_1 + A_{k+2} & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_{k+2}^{k-1-j_1} & C_{k+2,2} \end{bmatrix}$$

where in the r^{th} row and m^{th} column we make use of the following identity

$$\frac{A_r^{m-1} - A_1^{m-1}}{A_r - A_1} = \sum_{j_1=0}^{m-2} A_1^{j_1} A_r^{m-2-j_1}$$
(5.7)

for m from 2 to k + 1, and for the $(k + 2)^{nd}$ column we use the identity

$$C_{r,2} := \frac{A_r^{-N} - A_1^{-N}}{A_r - A_1} = -\sum_{j_1=1}^N A_1^{-j_1} A_r^{-(N+1-j_1)}.$$
(5.8)

Now subtract the second row from each other row below it and bring the factor of $1/(A_r - A_2)$ inside the determinant for each row $r = 3, \ldots k + 2$ to give $D_{N-1}[f]$ as

$$\frac{(-1)^{N(k+1)}\prod_{r=1}^{k+2}A_r^N}{\prod_{3\leq j<\ell\leq k+2}(A_\ell-A_j)}\det \begin{bmatrix} 1 & A_1 & \dots & A_1^k & A_1^{-N} \\ 0 & 1 & \dots & \sum_{j_1=0}^{k-1}A_1^{j_1}A_2^{j_2}A_3^{k-1-j_1} & C_{2,2} \\ 0 & 1 & \dots & \sum_{j_1=0}^{k-2}\sum_{j_2=0}^{k-2-j_1}A_1^{j_1}A_2^{j_2}A_3^{k-2-j_1-j_2} & C_{3,3} \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & 1 & \dots & \sum_{j_1=0}^{k-2}\sum_{j_2=0}^{k-2-j_1}A_1^{j_1}A_2^{j_2}A_k^{k-2-j_1-j_2} & C_{k,3} \\ 0 & 1 & \dots & \sum_{j_1=0}^{k-2}\sum_{j_2=0}^{k-2-j_1}A_1^{j_1}A_2^{j_2}A_{k+1}^{k-2-j_1-j_2} & C_{k+1,3} \\ 0 & 1 & \dots & \sum_{j_1=0}^{k-2}\sum_{j_2=0}^{k-2-j_1}A_1^{j_1}A_2^{j_2}A_{k+2}^{k-2-j_1-j_2} & C_{k+2,3} \end{bmatrix}$$

where we have used (5.7) for the r^{th} row and m^{th} column for m = 3, ..., k + 1 and where the last column uses the identity (5.8) with A_2 instead of A_1 with

$$C_{r,3} = -\sum_{j_1=1}^{N} A_1^{-j_1} \left(\frac{A_r^{-(N+1-j_1)} - A_2^{-(N+1-j_1)}}{A_r - A_2} \right)$$
$$= (-1)^2 \sum_{j_1=1}^{N} \sum_{j_2=1}^{N+1-j_1} A_1^{-j_1} A_2^{-j_2} A_r^{-(N+2-j_1-j_2)}.$$

Continuing in this manner, it is clear that $D_{N-1}[f]$ becomes

$$D_{N-1}[f] = (-1)^{N(k+1)} \prod_{r=1}^{k+2} A_r^N$$

$$\times \det \begin{bmatrix} 1 & A_1 & A_1^2 & \dots & A_1^k & A_1^{-N} \\ 0 & 1 & A_1 + A_2 & \dots & \sum_{j_1=0}^{k-1} A_1^{j_1} A_2^{k-1-j_1} & C_{2,2} \\ 0 & 0 & 1 & \dots & \sum_{j_1=0}^{k-2} \sum_{j_2=0}^{k-2-j_1} A_1^{j_1} A_2^{j_2} A_3^{k-2-j_1-j_2} & C_{3,3} \\ \vdots & \vdots & \vdots & \vdots & \vdots & \vdots & \vdots \\ 0 & 0 & 0 & \dots & A_1 + A_2 + \dots + A_k & C_{k,k} \\ 0 & 0 & 0 & \dots & 0 & C_{k+1,k+1} \\ 0 & 0 & 0 & \dots & 0 & C_{k+2,k+2} \end{bmatrix},$$

that is, we have an upper-triangular matrix with 1's on the diagonal and zeros underneath. This means that the determinant equals the bottom right-most element, which is

$$C_{k+2,k+2} = (-1)^{(k+1)} \sum_{j_1=1}^{N} \sum_{j_2=1}^{N+1-j_1} \cdots \sum_{j_k=1}^{N+k-1-j_1-\cdots-j_{k-1}} A_1^{-j_1} \cdots A_k^{-j_k} \times \sum_{j_{k+1}=1}^{N+k-j_1-\cdots-j_k} A_{k+1}^{j_{k+1}} A_{k+2}^{-(N+k+2-j_1-\cdots-j_{k+1})}.$$

We now combine $D_{N-1}[f]$ with the other terms in (5.6), and let $A_{k+1}, A_{k+2} \to 1$. This gives

$$\mathbb{E}_{N}\left[\prod_{r=1}^{k} Z(\theta_{N} + \alpha_{r})\right]$$

$$= \frac{1}{N} \prod_{r=1}^{k} (A_{r} - 1) \sum_{j_{1}=1}^{N} \sum_{j_{2}=1}^{N+1-j_{1}} \cdots \sum_{j_{k}=1}^{N+k-1-j_{1}-\dots-j_{k-1}} A_{1}^{-j_{1}} \dots A_{k}^{-j_{k}} \sum_{j_{k+1}=1}^{N+k-j_{1}-\dots-j_{k}} 1.$$

Re-indexing the summations to start from 0 instead of 1 and evaluating the last sum, we have

$$\mathbb{E}_{N}\left[\prod_{r=1}^{k} Z(\theta_{N} + \alpha_{r})\right]$$

$$= \frac{1}{N} \prod_{r=1}^{k} A_{r}^{-1} \left(A_{r} - 1\right) \sum_{j_{1}=0}^{N-1} \sum_{j_{2}=0}^{N-1-j_{1}} \cdots \sum_{j_{k}=0}^{N-1-j_{1}-\dots-j_{k-1}} A_{1}^{-j_{1}} \dots A_{k}^{-j_{k}} \left(N - j_{1} - \dots - j_{k}\right).$$
(5.9)

Recall that the coefficient of $\alpha_1^{n_1} \dots \alpha_k^{n_k}$, which we denote by $[\alpha_1^{n_1} \dots \alpha_k^{n_k}]$, from (5.9) will yield the desired mixed derivative, and we wish to find this to leading order as $N \to \infty$. It is clear that to do this, we want to obtain the highest possible power of N from the summations as they depend upon N, and the smallest possible contribution from the product $\prod_{r=1}^k A_r^{-1} (A_r - 1)$ as there is no N dependence in that product.

Recalling that $A_r = e^{i\alpha_r}$ and Taylor expanding gives

$$\prod_{r=1}^{k} A_r^{-1} \left(A_r - 1 \right) = i^k \alpha_1 \dots \alpha_k + \text{ higher order terms.}$$
(5.10)

Therefore we need to find the large N behaviour of $[\alpha_1^{n_1-1} \dots \alpha_k^{n_k-1}]$ from the summations in (5.9), that is from

$$\sum_{j_1=0}^{N-1} \sum_{j_2=0}^{N-1-j_1} \cdots \sum_{j_k=0}^{N-1-j_1-\cdots-j_{k-1}} e^{-ij_1\alpha_1} \cdots e^{-ij_k\alpha_k} (N-j_1-\cdots-j_k)$$

The coefficient of $\alpha_1^{n_1-1}$ in the Taylor expansion of $e^{-ij_1\alpha_1}$ equals $(-ij_1)^{n_1-1}/(n_1-1)!$, so the coefficient $[\alpha_1^{n_1-1}...\alpha_k^{n_k-1}]$ equals

$$\sum_{j_1=0}^{N-1} \cdots \sum_{j_k=0}^{N-1-j_1-\dots-j_{k-1}} \frac{(-ij_1)^{n_1-1}}{(n_1-1)!} \cdots \frac{(-ij_k)^{n_k-1}}{(n_k-1)!} (N-j_1-\dots-j_k)$$
$$= \frac{(-i)^{n_1+\dots+n_k-k}}{(n_1-1)!\dots(n_k-1)!} \sum_{j_1=0}^{N-1} \cdots \sum_{j_k=0}^{N-1-j_1-\dots-j_{k-1}} j_1^{n_1-1} \dots j_k^{n_k-1} (N-j_1-\dots-j_k).$$

To evaluate the leading order coefficient of these sums, we will turn them into Riemann sums and hence into an integral. The sums equal

$$N^{n_1+\dots+n_k+1} \sum_{j_1=0}^{N-1} \sum_{\substack{j_2=0\\N-1-j_1-\dots-j_{k-1}\\\sum_{j_k=0}}}^{N-1-j_1} \frac{1}{N^k} \left(\frac{j_1}{N}\right)^{n_1-1} \dots \left(\frac{j_k}{N}\right)^{n_k-1} \left(1-\frac{j_1}{N}-\dots-\frac{j_k}{N}\right)$$

and now each summation can be considered as a Riemann sum and so we can rewrite the summation in the above line as the k-fold integral

$$\int_0^1 \int_0^{1-x_1} \cdots \int_0^{1-x_1-\dots-x_{k-1}} x_1^{n_1-1} \dots x_k^{n_k-1} (1-x_1-\dots-x_k) \, dx_1 dx_2 \dots dx_k.$$

By Theorem 1.8.6 in [3], this integral equals

$$\frac{(n_1-1)!\dots(n_k-1)!}{(n_1+\dots+n_k+1)!}$$

and so we have

$$[\alpha_1^{n_1-1}...\alpha_k^{n_k-1}] \sim \frac{(-i)^{n_1+\dots+n_k-k}}{(n_1-1)!\dots(n_k-1)!} \frac{(n_1-1)!\dots(n_k-1)!}{(n_1+\dots+n_k+1)!} N^{n_1+\dots+n_k+1}.$$
 (5.11)

Combining (5.10) with (5.11) and the term 1/N in (5.9), we obtain the coefficient $[\alpha_1^{n_1}...\alpha_k^{n_k}]$ of $\mathbb{E}_N[Z(\theta_N + \alpha_1)...Z(\theta_N + \alpha_k)]$ is asymptotic to

$$\frac{(-1)^{n_1+\dots+n_k+k} i^{n_1+\dots+n_k}}{(n_1+\dots+n_k+1)!} N^{n_1+\dots+n_k}$$

Recalling that the coefficient of $\alpha_r^{n_r}$ in the Taylor expansion of $Z(\theta_N + \alpha_r)$ equals $Z^{(n_r)}(\theta_N)/n_r!$, we see that

$$\mathbb{E}_{N}\left[Z^{(n_{1})}(\theta_{N})\dots Z^{(n_{k})}(\theta_{N})\right] \sim (-1)^{n_{1}+\dots+n_{k}+k} i^{n_{1}+\dots+n_{k}} \frac{n_{1}!\dots n_{k}!}{(n_{1}+\dots+n_{k}+1)!} N^{n_{1}+\dots+n_{k}},$$

and this completes the proof of Theorem 5.1.

5.3 THE HYBRID MODEL APPROACH

As discussed in Section 1.3.4, in 2007, Gonek, Hughes and Keating [149] proved a hybrid formula for the zeta function, expressing it as a partial product over primes times a rapidly decaying product over non-trivial zeros. We state a result with slightly more control over the smoothing (in [149] and other later papers, X = Y is taken).

Theorem 5.6 (Gonek–Hughes–Keating). Let $s = \sigma + it$ with $\sigma \ge 0$ and $|t| \ge 2$, let $X, Y \ge 2$ be real parameters, and let K be any fixed positive integer. Let f(x) be a non-negative C^{∞} function of mass 1 supported on [0,1] and set $u(x) = Yf(Y\log \frac{x}{e} + 1)/x$. Thus, u(x) is a function of mass 1 supported on $[e^{1-1/Y}, e]$. Set

$$U(z) = \int_{1}^{e} u(y) E_{1}(z \log y) \, dy \,, \tag{5.12}$$

where $\mathbb{E}_1(z)$ is the exponential integral $\int_z^{\infty} e^{-w}/w \, dw$. Then

$$\zeta(s) = P_X(s)Z_X(s)\left(1 + O\left(\frac{Y^{K+1}X^{\max(1-\sigma,0)}}{(|s|\log X)^K}\right) + O\left(\frac{X^{1-\sigma}\log X}{Y}\right)\right) + O\left(\frac{X^{1-\sigma}\log X}{Y}\right)\right)$$

where

$$P_X(s) = \exp\left(\sum_{n \le X} \frac{\Lambda(n)}{\log n} \frac{1}{n^s}\right),\tag{5.13}$$

 $\Lambda(n)$ is von Mangoldt's function, and

$$Z_X(s) = \exp\left(-\sum_{\rho_n} U((s-\rho_n)\log X)\right).$$
(5.14)

The constants implied by the O terms depend only on f and K.

Remark. The proof of this result follows precisely the method in [149], with obvious changes for the different smoothing.

Note that U(z) is not an analytic function; it has a logarithmic singularity at z = 0. However, recall the formula

$$E_1(z) = -\log z - \gamma_0 - \sum_{m=1}^{\infty} \frac{(-1)^m z^m}{m! m}, \qquad (5.15)$$

where $|\arg z| < \pi$, $\log z$ denotes the principal branch of the logarithm, and γ_0 is Euler's constant. (It is clear that the sum is an entire function of z since it is absolutely convergence for all $z \in \mathbb{C}$.) From this and (5.12) we observe that we may interpret $\exp(-U(z))$ to be an analytic function, asymptotic to Cz as $z \to 0$, for some constant C which depends upon the smoothing function u.

Just as in work of Bui, Gonek and Milinovich [47], we can differentiate whilst still obtaining the asymptotic, obtaining under the Riemann Hypothesis

$$\zeta'(\rho) = P_X(\rho) Z'_X(\rho) \left(1 + O\left(\frac{Y^{K+1} X^{1/2}}{(|\rho| \log X)^K}\right) + O\left(\frac{X^{1/2} \log X}{Y}\right) \right).$$

It is believed that P_X and Z_X operate pseudo-independently. Hence the moments of zeta are products of moments of P_X and Z_X . This is known as the Splitting Conjecture, given in Conjecture 1.10 for the analogous continuous moment case. We conjecture it here in our case, where it will become clear later in the section where the technical assumptions come from.

Conjecture 5.7. Let $X, T \to \infty$ with $X = O(\log T)$. Then for k > -3, we have

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta'(\rho)^k \sim \left(\frac{1}{N(T)} \sum_{0 < \gamma \le T} P_X(\rho)^k\right) \left(\frac{1}{N(T)} \sum_{0 < \gamma \le T} Z'_X(\rho)^k\right).$$

In the next section we will evaluate the moments of mixed derivatives of P_X , when averaged over zeros of zeta. We will show that the moments only really contribute when no derivatives are taken. In the last section we will model the moments of the first derivative of Z_X when averaged over zeros of zeta, using a method that allows non-integer moments to be taken. Taken collectively, these results give weight to Conjectures 5.4 and 5.5.

5.3.1 The moments of P_X

In this section we will show that the mixed moments of P_X vanish if any derivatives are taken, whilst the mean of P_X^k is asymptotically 1, when averaged over zeros of the zeta function.

Theorem 5.8. Under the Riemann Hypothesis, for X > 2 with $X = O(\log T)$, for any non-negative integers n_1, \ldots, n_k we have,

$$\sum_{0 < \gamma \le T} P_X^{(n_1)}(\rho) \dots P_X^{(n_k)}(\rho) = \begin{cases} N(T) + O(T \log \log T) & \text{if } n_1 = \dots = n_k = 0\\ O\left(T(\log \log T)^{1+n_1}\right) & \text{if } n_1 > 0 \text{ and } n_2 = \dots = n_k = 0\\ O\left(T\right) & \text{if more than one } n_i > 0 \end{cases}$$

as $T \to \infty$, where $\rho = \frac{1}{2} + i\gamma$ is a non-trivial zero of the Riemann zeta function, $P_X(s)$ is given in (5.13) and N(T) is the number of such zeros up to height T, given by (1.5).

In the case that no derivatives are taken, one can permit k to be real and positive.

Theorem 5.9. Under the Riemann Hypothesis, for X > 2 with $X = O(\log T)$, for k a fixed positive real number,

$$\sum_{0 < \gamma \leq T} P_X(\rho)^k = N(T) + O(T \log \log T)$$

as $T \to \infty$.

This should be compared to a result of Bui, Gonek and Milinovich (Theorem 2.2 of [47]) which stated that, under the Riemann Hypothesis, for $\varepsilon > 0$ and $X, T \to \infty$ with $X = O\left((\log T)^{2-\varepsilon}\right)$ then for any $k \in \mathbb{R}$ one has

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} |P_X(\rho)|^{2k} = a_k \left(e^{\gamma_0} \log X \right)^{k^2} \left(1 + O_k \left((\log X)^{-1} \right) \right)$$

where a_k is an explicit product over primes given by (1.19).

Before we prove Theorem 5.8, we need to state and prove some preliminary lemmas.

Lemma 5.10. For k a fixed positive real number, the function $P_X(s)^k$ can be written as a non-vanishing absolutely convergent Dirichlet series

$$P_X(s)^k = \sum_{m=1}^{\infty} \frac{a_k(m)}{m^s}$$

with $0 \leq a_k(m) \leq d_k(m)$, where $d_k(m)$ is the k^{th} divisor function.

Proof. Since $P_X(s)$ given in (5.13) is the exponential of a finite Dirichlet polynomial, so its k^{th} power is an entire non-vanishing function, so can be written as a Dirichlet series

$$P_X(s)^k = \exp\left(k\sum_{n\leq X} \frac{\Lambda(n)}{\log n} \frac{1}{n^s}\right)$$
$$= \sum_{m=1}^{\infty} \frac{a_k(m)}{m^s}$$

where the sum converges absolutely for any s, and $a_k(m) = 0$ if m has any prime divisor greater than X. If $p \leq X$ is a prime, then $a_k(p) = k$. Since these coefficients come from the exponential of positive terms, the $a_k(m)$ are positive. In the case when $m \leq X$, these coefficients will equal $d_k(m)$ and in the case when m > X they will be less than $d_k(m)$ due to missing contributions from terms with n > X in the exponent.

In the proofs of the two theorems, it will be important to truncate the infinite sum.

Lemma 5.11 (Gonek-Hughes-Keating). For $2 \leq X \ll (\log T)^{2-\varepsilon}$ with $\varepsilon > 0$, and k a fixed positive real number, and θ a small positive number, we have

$$P_X\left(\frac{1}{2}+it\right)^k = \sum_{m \le T^\theta} \frac{a_k(m)}{m^{1/2+it}} + O_k\left(T^{-\varepsilon\theta/2}\right).$$

Proof. This follows immediately from the proof of Lemma 2 in [149].

As previously discussed, Gonek [146] proved unconditionally a uniform version of Landau's formula concerning sums over the non-trivial zeros of the Riemann zeta function, including the corollary given in Lemma 2.2 under the assumption of the Riemann Hypothesis. We restate it below for ease of reference.

Lemma (Gonek). Under the Riemann Hypothesis, for $T > 1, m \in \mathbb{N}$ with $m \ge 2$ and ρ a non-trivial zero of the Riemann zeta function $\zeta(s)$,

$$\sum_{0 < \gamma \le T} m^{-\rho} = -\frac{T}{2\pi} \frac{\Lambda(m)}{m} + O(\log(2mT)\log\log(3m)),$$

where $\Lambda(m)$ is the von Mangoldt function.

Remark. Recall that this result does not apply in the case when m = 1, when the sum over zeros trivially equals N(T).

Proof of Theorem 5.8. To differentiate $P_X(\sigma)$ *j*-times, we return to the exponential form (5.13). The Faà di Bruno formula on multiple derivatives of e^f states

$$\frac{d^j}{d\sigma^j}e^{f(\sigma)} = \left(f^{(j)} + \dots + (f')^j\right)e^f$$

where the sum ranges over all possible combinations of derivatives of f so that total derivative is j, with certain multinomial coefficients that don't matter for us.

In our case, one can check that the longest sum comes from the first derivative raised to the j^{th} power, essentially giving a sum of length X^{j} . That is

$$P_X^{(j)}(s) = \sum_{n \le X^j} \frac{\tilde{\beta}_j(n)}{n^s} \exp\left(\sum_{n \le X} \frac{\Lambda(n)}{\log n} \frac{1}{n^s}\right)$$

for coefficients $\tilde{\beta}_j(n)$ that come from appropriate combinations of derivatives of the sum in the exponential, whose precise nature does not concern us, other than noting that

$$\tilde{\beta}_j(1) = \begin{cases} 0 & \text{if } j \neq 0 \\ 1 & \text{if } j = 0 \end{cases}$$

and for prime $p \leq X$,

$$\tilde{\beta}_j(p) = \begin{cases} (-\log p)^j & \text{ if } j \neq 0\\ 0 & \text{ if } j = 0 \end{cases}$$

(this comes from differentiating the sum in the exponential j times; other terms in the Faà di Bruno formula contribute to coefficients of higher powers of p). If n is divisible by any prime greater than X, $\tilde{\beta}_j(n) = 0$.

Combining k of these expressions together (replacing j with derivatives n_1, n_2, \ldots, n_k , we have

$$P_X^{(n_1)}(s)\dots P_X^{(n_k)}(s) = \sum_{n \le X^{n_1+n_2+\dots+n_k}} \frac{\beta(n)}{n^s} P_X(s)^k$$
(5.16)

where

$$\beta(n) = (\tilde{\beta}_{n_1} * \tilde{\beta}_{n_2} * \dots * \tilde{\beta}_{n_k})(n)$$

is the convolution of the appropriate $\tilde{\beta}_i$.

In the case when $\Re(s) = 1/2$ and $X = (\log T)^{2-\varepsilon}$, using Lemma 5.11 we have

$$P_X^{(n_1)}(s) \dots P_X^{(n_k)}(s) = \sum_{\substack{n \le X^{n_1+n_2+\dots+n_k}}} \frac{\beta(n)}{n^s} \left(\sum_{\substack{m \le T^{\theta}}} \frac{a_k(m)}{m^s} + O_k\left(T^{-\varepsilon\theta/2}\right) \right)$$
$$= \sum_{\substack{m \le T^{\theta'}}} \frac{b(m)}{m^s} + O\left(T^{-\varepsilon'}\right)$$
(5.17)

for some $\varepsilon' > 0$ and $\theta' > \theta$, which both depend upon ε , k, θ , and n_1, \ldots, n_k . Essentially, since the first sums length is only a power of $\log T$, the two sums together are truncated at $T^{\theta'}$ for some $\theta' > \theta$, as T gets large.

Shortly we will use some facts about the b(m), which we record here:

- Only if $n_1 = n_2 = \cdots = n_k = 0$ does b(1) = 1, otherwise b(1) = 0 (essentially because $\tilde{\beta}_j(1) = 0$ for any $j \neq 0$ since differentiation kills constants).
- If $n_1 = n_2 = \cdots = n_k = 0$, then $b(p) = a_k(p)$ and for prime $p \le X$, $a_k(p) = k$.
- If exactly all bar one of the n_i vanish (say $n_1 \neq 0$, and $n_2 = n_3 = \cdots = n_k = 0$) then $b(p) = \beta(p) = \tilde{\beta}_{n_1}(p) = (-\log p)^{n_1}$. To see this, note that $\tilde{\beta}_{n_1}(1) = 0$, which kills all other contributions in the convolution other than $\tilde{\beta}_{n_1}(p)$.
- If more than one $n_i \neq 0$, then b(p) = 0 since there will be at least two pieces in the convolution that have $\tilde{\beta}_{n_1}(1) = 0$ and $\tilde{\beta}_{n_2}(1) = 0$, so every term vanishes.

Assuming the Riemann Hypothesis, we can sum (5.16) over the zeros of zeta, obtaining

$$\sum_{0 < \gamma \le T} P_X^{(n_1)}(\rho) \dots P_X^{(n_k)}(\rho) = b(1)N(T) + \sum_{0 < \gamma \le T} \sum_{2 \le m \le T^{\theta'}} \frac{b(m)}{m^{\rho}} + O\left(N(T)T^{-\varepsilon'}\right)$$
(5.18)

and applying Corollary 2.2 to the middle sum yields

$$\sum_{0<\gamma\leq T}\sum_{2\leq m\leq T^{\theta}}\frac{b(m)}{m^{\rho}} = -\frac{T}{2\pi}\sum_{2\leq m\leq T^{\theta'}}\frac{b(m)\Lambda(m)}{m} + \sum_{2\leq m\leq T^{\theta}}b(m)O(\log(2mT)\log\log(3m)).$$
(5.19)

To bound the first term on the right-hand side of (5.19), note that the $\Lambda(m)$ forces the sum to be over primes and prime powers only, and the b(m) ensures we only need primes $p \leq X$. Since $b(m) \ll m^{\varepsilon}$, the square and higher powers of primes form a convergent sum, so

$$-\frac{T}{2\pi}\sum_{2\leq m\leq T^{\theta'}}\frac{b(m)\Lambda(m)}{m}=O\left(T\sum_{\substack{p\leq X\\p \text{ prime}}}\frac{b(p)\log p}{p}\right)+O(T).$$

The value of b(p) depends on how many $n_i = 0$

$$b(p) = \begin{cases} k & \text{if } n_1 = \dots = n_k = 0\\ (-\log p)^{n_1} & \text{if } n_1 > 0 \text{ and } n_i = 0 \text{ otherwise}\\ 0 & \text{if more than one } n_i > 0. \end{cases}$$

Hence, summing over all primes up to X and using the Prime Number Theorem,

$$\sum_{p \le x} \frac{b(p)\log p}{p} \ll \begin{cases} \log X & \text{if } n_1 = \dots = n_k = 0\\ (\log X)^{1+n_1} & \text{if } n_1 > 0 \text{ and } n_i = 0 \text{ otherwise}\\ 0 & \text{if more than one } n_i > 0. \end{cases}$$

For the second error term in (5.19), since the b(m) all have the same sign (for any given fixed n_1, \ldots, n_k), we have that this equals

$$O\left(\log T \log \log T \sum_{m=1}^{\infty} b(m)\right).$$

By (5.17) and (5.16) the sum is simply

$$\sum_{m=1}^{\infty} b(m) = \sum_{n \le X^{n_1 + n_2 + \dots + n_k}} \beta(n) P_X(0)^k.$$

To bound the first sum on the right-hand side, recall that $\beta(n) \ll n^{\varepsilon}$, and to bound the $P_X(0)^k$ note that since

$$\log P_X(0) = \sum_{n \le X} \frac{\Lambda(n)}{\log n}$$
$$= \frac{X}{\log X} + O\left(\frac{X}{(\log X)^2}\right)$$

by the Prime Number Theorem, we have

$$\sum_{m=1}^{\infty} b(m) = O\left(\exp\left(c'\frac{X}{\log X}\right)\right)$$

for any fixed c' > k.

Plugging these two error terms back into (5.19), we have shown

$$\sum_{0 < \gamma \le T} \sum_{2 \le m \le T^{\theta}} \frac{b(m)}{m^{\rho}} = O\left(T(\log X)^{1+n_1}\right) + O(T) + O\left(\log T \log \log T \exp\left(c'\frac{X}{\log X}\right)\right)$$

where the first error term is only present if $n_2 = \cdots = n_k = 0$.

To complete the proof we now choose X to balance the various error terms that appear when we substitute this into (5.18). If we insist $X = O(\log T)$ then we have

$$\sum_{0 < \gamma < T} P_X^{(n_1)}(\rho) \dots P_X^{(n_k)}(\rho) = \frac{T}{2\pi} \log T + O\left(T(\log \log T)\right)$$

in the case when $n_1 = \cdots = n_k = 0$,

$$\sum_{0 < \gamma < T} P_X^{(n_1)}(\rho) \dots P_X^{(n_k)}(\rho) = O\left(T(\log \log T)^{1+n_1}\right)$$

if $n_1 > 0$ and $n_2 = \cdots = n_k = 0$, and

$$\sum_{0 < \gamma < T} P_X^{(n_1)}(\rho) \dots P_X^{(n_k)}(\rho) = O(T)$$

if more than one $n_i > 0$.

This completes the proof of Theorem 5.8.

Proof of Theorem 5.9. In the case when no derivatives are taken, we can exploit the multiplicative nature of P_X to allow k to be a fixed positive real number, not just an integer. Much of the proof goes through unchanged.

160

5.3. The hybrid model approach

By Lemma 5.11 and Lemma 2.2 we have

$$\sum_{0<\gamma\leq T} P_X(\rho)^k = N(T) + \sum_{0<\gamma\leq T} \sum_{2\leq m\leq T^{\theta}} \frac{a_k(m)}{m^{\rho}} + O\left(N(T)T^{-\varepsilon'}\right)$$
$$= N(T) - \frac{T}{2\pi} \sum_{2\leq m\leq T^{\theta}} \frac{a_k(m)\Lambda(m)}{m} + O\left(\log T \log \log T \sum_{2\leq m\leq T^{\theta}} a_k(m)\right)$$

since $a_k(1) = 1$.

As in the case of Theorem 5.8, assuming $X < T^{\theta}$ we can bound

$$\sum_{2 \le m \le T^{\theta}} \frac{a_k(m)\Lambda(m)}{m} = \sum_{p < X} \frac{k \log p}{p} + O(1)$$
$$\ll \log X$$

and, exactly as before,

$$\sum_{m=1}^{\infty} a_k(m) = P_X(0)^k \ll \exp\left(c'\frac{X}{\log X}\right)$$

for any fixed c' > k.

Therefore

$$\sum_{0 < \gamma \le T} P_X(\rho)^k = N(T) + O\left(T\log X\right) + O\left(\log T\log\log T\exp\left(c'\frac{X}{\log X}\right)\right).$$

As before, the error terms can be made to approximately match when $X = O(\log T)$. \Box

Remark. In the previous proofs we can take X to be anything up to $o(\log T \log \log T)$ before the second error term dominates the main term of $\frac{T}{2\pi} \log T$. Since the error term as we have written it comes from a subsidiary main term, we cannot reduce the error down to O(T).

5.3.2 The moments of Z'_X

In this section we will concentrate on the random matrix equivalent for Z_X , given in (5.14), and for simplicity just look at its first derivative. Again in this section, we will not require k to be an integer. Taking an $N \times N$ unitary matrix with eigenvalues $e^{i\theta_1}, \ldots, e^{i\theta_N}$, we wish to calculate

$$\mathbb{E}_N\left[Z'_{N,X}(\theta_N)^k\right]$$

where

$$Z_{N,X}(\theta) = \prod_{m=1}^{N} \exp\left(-\sum_{j=-\infty}^{\infty} U(i(\theta - \theta_m + 2\pi j)\log X)\right)$$

plays the role of $Z_X(s)$ in Theorem 5.6, and where U is given in (5.12).

Remark. The sum over j makes the arguments 2π -periodic, thus permitting the move from $e^{i\theta}$ to θ without any branch-cut ambiguities. The decay of U for large inputs means that the infinite sum over j is absolutely summable.

Theorem 5.12. If $k \in \mathbb{C}$ is such that $k \notin \{-3, -4, -5, \dots\}$, then

$$\mathbb{E}_N\left[\frac{1}{N}\sum_{m=1}^N Z'_{N,X}(\theta_m)^k\right] \sim \frac{e^{ik\pi/2}}{\Gamma(k+2)}N^k$$

as $N \to \infty$.

Remark. The proof deals with the analytic continuation of the objects under consideration. However, for our purposes as a model for zeta, we must stop before the first singularity; that is, in Conjecture 5.5 we take $\Re(k) > -3$.

Remark. Restricting to $k \in \mathbb{N}$, we see this Theorem has the same right hand side as Corollary 5.2 in the case n = 1. That is, $Z'_{N,X}$ and the first derivative of the characteristic polynomial asymptotically have the same moments.

Proof. It is convenient to factor out the behaviour around the eigenvalues (where $Z_{N,X}$ vanishes) as follows

$$Z_{N,X}(\theta) = \prod_{m=1}^{N} \left(1 - e^{-i(\theta - \theta_m)} \right) e^{F_X(\theta - \theta_m)}$$

where

$$F_X(\vartheta) = -\log\left(1 - e^{-i\vartheta}\right) - \sum_{j=-\infty}^{\infty} U(i(\vartheta + 2\pi j)\log X)$$
(5.20)

is a 2π -periodic function which is continuously differentiable for all real ϑ - the logarithmic singularities cancel out. The first statement is obvious. The second statement can be seen from the definition of U(z) given in (5.12) and the series expansion of $E_1(z)$ around z = 0given in (5.15) (and indeed, this fact will be made clear when we look at the decay of its Fourier coefficients in Lemma 5.13).

Differentiating,

$$\begin{aligned} Z_{N,X}'(\theta) &= \sum_{n=1}^{N} i e^{-i(\theta-\theta_n)} e^{F_X(\theta-\theta_n)} \prod_{\substack{m=1\\m\neq n}}^{N} \left(1 - e^{-i(\theta-\theta_m)}\right) e^{F_X(\theta-\theta_m)} \\ &+ \sum_{n=1}^{N} \left(1 - e^{-i(\theta-\theta_n)}\right) F_X'(\theta-\theta_n) e^{F_X(\theta-\theta_n)} \prod_{\substack{m=1\\m\neq n}}^{N} \left(1 - e^{-i(\theta-\theta_m)}\right) e^{F_X(\theta-\theta_m)} \end{aligned}$$

and substituting $\theta = \theta_N$, we see that for every $n \neq N$ there is a term in the product that vanishes (so only the n = N terms survives), and every term in the second term vanishes,

5.3. The hybrid model approach

including the n = N term. That is,

$$Z'_{N,X}(\theta_N) = i e^{F_X(0)} \prod_{m=1}^{N-1} \left(1 - e^{-i(\theta_N - \theta_m)} \right) e^{F_X(\theta_N - \theta_m)}.$$

Remark. There is nothing special about picking θ_N over any of the other eigenvalues. This is made for notational convenience. The true calculation would be to evaluate

$$\frac{1}{N}\sum_{m=1}^{N}Z'_{N,X}(\theta_m)$$

similar to what we do in Section 5.2, but by rotation invariance the result will be the same.

We now calculate the expected value of $Z'_{N,X}(\theta_N)^k$ for $k \in \mathbb{C}$ when averaged over all $N \times N$ unitary matrices distributed according to Haar measure, employing a trick found in [182] and written in this thesis as Lemma 1.18.

$$\mathbb{E}_{N}\left[Z_{N,X}'(\theta_{N})^{k}\right] = e^{ik\pi/2}e^{kF_{X}(0)}\mathbb{E}_{N}\left[\prod_{m=1}^{N-1}\left(1 - e^{-i(\theta_{N} - \theta_{m})}\right)^{k}e^{kF_{X}(\theta_{N} - \theta_{m})}\right]$$
$$= e^{ik\pi/2}e^{kF_{X}(0)}\frac{1}{N}\mathbb{E}_{N-1}\left[\prod_{m=1}^{N-1}\left|1 - e^{i\vartheta_{m}}\right|^{2}\left(1 - e^{i\vartheta_{m}}\right)^{k}e^{kF_{X}(-\vartheta_{m})}\right]$$
(5.21)

where we interpret $e^{i\vartheta_m}$ as the eigenvalues of an $(N-1) \times (N-1)$ unitary matrix. (Briefly, what we have done is write the first expectation out as a N-dimensional Weyl integral, then change variables to $\vartheta_m = \theta_m - \theta_N$ for m = 1, ..., N - 1. Making use of the fact the integrand is 2π -periodic, we turn it back into a (N-1)-dimension Weyl integral, plus an extra trivial integral over θ_N .)

Because k is complex, we need to be precise about the choice of branch taken. We define $(1 - e^{i\vartheta})^k$ to be the value obtained from continuous variation of $(1 - e^{i\vartheta}e^{-\varepsilon})^k$ for $\varepsilon > 0$, converging to the value 1 as $\varepsilon \to \infty$.

As in Section 5.2, the inner expectation can be calculated by Heine's identity to be equal to the Toeplitz determinant

$$D_{N-1}[f] := \det \left| \hat{f}_{j-\ell} \right|_{1 \le j, \ell \le N-1}$$

with symbol

$$f(\vartheta) = \left|1 - e^{i\vartheta}\right|^2 \left(1 - e^{i\vartheta}\right)^k e^{kF_X(-\vartheta)}.$$

Note that $f(\vartheta)$ has only one singularity at $\vartheta = 0$. Toeplitz determinants with such symbols have been calculated asymptotically by Ehrhardt and Silbermann in [110] (building on extensive previous work by several other authors).

They show (in their Theorem 2.5) that if $b(\vartheta)$ is a suitably smooth (defined in their paper) 2π -periodic function with winding number zero and whose logarithm has the Fourier

series expansion

$$\log b(\vartheta) = \sum_{m=-\infty}^{\infty} s_m e^{im\vartheta},$$

then for symbols of the form

$$f(\vartheta) = b(\vartheta) \left(1 - e^{i\vartheta}\right)^{\gamma} \left(1 - e^{-i\vartheta}\right)^{\delta}$$

subject to $\gamma + \delta \notin \{-1, -2, -3, \dots\}$, then for any $\varepsilon > 0$

$$D_{N-1}[f] = C_1 N^{\gamma \delta} C_2^{N-1} \left(1 + O\left(\frac{1}{N^{1-\varepsilon}}\right) \right)$$

where

$$C_1 = \frac{G(1+\gamma)G(1+\delta)}{G(1+\gamma+\delta)} \exp\left(\sum_{m=1}^{\infty} m s_m s_{-m}\right) \exp\left(-\delta \sum_{m=1}^{\infty} s_m\right) \exp\left(-\gamma \sum_{m=1}^{\infty} s_{-m}\right)$$

where G is the Barnes G-function, and where

$$C_2 = \exp(s_0).$$

In our case, $\gamma = k + 1$ and $\delta = 1$ and $b(\vartheta) = \exp(kF_X(-\vartheta))$. To complete the proof, we simply need to calculate the Fourier coefficients of $kF_X(-\vartheta)$ (those coefficients being the desired s_m).

We will, in fact, delay the exact calculation of s_m to Lemma 5.13 to quickly complete the proof of the theorem. All we need from the lemma is that $s_m = 0$ if $m \leq 0$. From (5.21) and using Ehrhardt and Silbermann's formula above, with the values $\gamma = k + 1$ and $\delta = 1$, we have

$$\mathbb{E}_{N}\left[Z_{N,X}'(\theta_{N})^{k}\right] \sim e^{ik\pi/2} e^{kF_{X}(0)} \frac{1}{N} \frac{G(k+2)G(2)}{G(k+3)} \exp\left(-\sum_{m=1}^{\infty} s_{m}\right) N^{k+1}.$$

Finally, note that G(2) = 1, $G(k + 3) = \Gamma(k + 2)G(k + 2)$ and

$$e^{kF_X(0)} = \exp\left(\sum_{m=-\infty}^{\infty} s_m\right) = \exp\left(\sum_{m=1}^{\infty} s_m\right)$$

(the first equality following from setting $\vartheta = 0$ in the Fourier series expansion; the second equality comes from knowing $s_m = 0$ for $m \leq 0$). Therefore,

$$\mathbb{E}_N\left[Z'_{N,X}(\theta_N)^k\right] \sim \frac{e^{ik\pi/2}}{\Gamma(k+2)}N^k$$

as $N \to \infty$, as required.

5.3. The hybrid model approach

Lemma 5.13. Let $F_X(\vartheta)$ be defined in (5.20), then

$$kF_X(-\vartheta) = \sum_{m=-\infty}^{\infty} s_m e^{im\vartheta}$$

where

$$s_m = \begin{cases} 0 & \text{if } m \le 0\\ \frac{k}{m} \left(1 - \int_1^{\exp(m/\log X)} u(y) \, dy \right) & \text{if } 1 \le m < \log X\\ 0 & \text{if } m \ge \log X. \end{cases}$$

Remark. Since u has total mass 1, for m > 0 we can write

$$1 - \int_{1}^{\exp(m/\log X)} u(y) \, dy = \int_{\exp(m/\log X)}^{\infty} u(y) \, dy$$

although due to the support condition on u, if $m \ge \log X$ then $\exp(m/\log X) > e$ and so the integral vanishes.

Proof of Lemma 5.13. Note that

$$s_m = \frac{1}{2\pi} \int_{-\pi}^{\pi} k F_X(-\vartheta) e^{-im\vartheta} \, d\vartheta.$$

From (5.20) we have

$$kF_X(-\vartheta) = -k\log\left(1-e^{i\vartheta}\right) - k\sum_{j=-\infty}^{\infty} U(i(-\vartheta+2\pi j)\log X).$$

For m an integer, we have

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} -k \log\left(1 - e^{i\vartheta}\right) e^{-im\vartheta} \, d\vartheta = \begin{cases} 0 & \text{if } m \le 0\\ \frac{k}{m} & \text{if } m \ge 1. \end{cases}$$
(5.22)

(To non-rigorously see why, note that $\log(1 - e^{i\vartheta}) = -\sum_{\ell=1}^{\infty} e^{i\ell\vartheta}/\ell$ and the only term in the sum that survives the integration is when $\ell = m$ for positive integers m.)

Furthermore, we have

$$\frac{1}{2\pi} \int_{-\pi}^{\pi} -k \sum_{j=-\infty}^{\infty} U\left(i(-\vartheta+2\pi j)\log X\right) e^{-im\vartheta} \, d\vartheta = \frac{-k}{2\pi} \int_{-\infty}^{\infty} U(-i\vartheta\log X) e^{-im\vartheta} \, d\vartheta$$
$$= \frac{-k}{2\pi} \int_{-\infty}^{\infty} \int_{1}^{e} u(y) E_1(-i\vartheta\log X\log y) e^{-im\vartheta} \, dy \, d\vartheta$$

where the final equality comes from inserting the definition of U from (5.12) and the support of u. Swapping the order of integration, this is

$$\frac{-k}{2\pi} \int_1^e u(y) \int_{-\infty}^\infty E_1(-i\vartheta \log X \log y) e^{-im\vartheta} \, d\vartheta \, dy.$$

The Fourier transform of the exponential integral is

$$\frac{1}{2\pi} \int_{-\infty}^{\infty} E_1(-i\vartheta \log X \log y) e^{-im\vartheta} d\vartheta = \begin{cases} 0 & \text{if } m < \log X \log y \\ \frac{1}{2m} & \text{if } m = \log X \log y \\ \frac{1}{m} & \text{if } m > \log X \log y \end{cases}$$

Therefore, making use of the fact that u is supported on [1, e] and has total mass 1, we have

$$\begin{aligned} \frac{-k}{2\pi} \int_{1}^{e} u(y) \int_{-\infty}^{\infty} E_{1}(-i\vartheta \log X \log y) e^{-im\vartheta} \ d\vartheta \ dy &= \frac{-k}{m} \int_{1}^{\max(1,\exp(m/\log X))} u(y) \ dy \\ &= \begin{cases} 0 & \text{if } m \leq 0 \\ -\frac{k}{m} \int_{1}^{\exp(m/\log X)} u(y) \ dy & \text{if } 1 \leq m < \log X \\ -\frac{k}{m} & \text{if } m \geq \log X. \end{cases} \end{aligned}$$

The Fourier coefficient, s_m is the sum of this and (5.22). Note that both terms are zero for $m \leq 0$ and the two terms perfectly cancel each other if $m \geq \log X$.

5.4 NEGATIVE MOMENTS OF THE DERIVATIVES OF THE CHARACTERIS-TIC POLYNOMIAL

As we have seen in Section 1.3.6, Section 1.4.1 and Section 2.4, negative moments are of interest. Recall that Conjecture 5.5 states that we can take any $\Re(k) > -3$. We consider some small negative moments, specifically the cases when k = -1, -2, -3 (pushing our model to the point of breaking). We consider

$$\mathbb{E}_N\left[\frac{1}{N}\sum_{m=1}^N Z'_N(\theta_m)^k\right],\,$$

where we remind the reader that $Z_N(\theta)$ is the characteristic polynomial of a random unitary matrix, given by

$$Z_N(\theta) = \det(I - Ue^{-i\theta}) = \prod_{m=1}^N (1 - e^{i(\theta_m - \theta)}).$$

For the analogous symbol to (5.5) for the negative moments problem, we need to consider for k = -1, -2, -3,

$$f(z) = |1 - e^{iz}|^2 (1 - e^{iz})^k.$$

For k = -1, we have

$$f(z) = |1 - e^{iz}|^2 (1 - e^{iz})^{-1} = (1 - e^{-it})$$

so the Fourier coefficients are $c_{-1} = -1$, $c_0 = 1$, and $c_j = 0$ for $j \ge 1$. Then the Toeplitz determinant is given by

$$\mathbb{E}_{N}\left[\frac{1}{N}\sum_{m=1}^{N}Z_{N}'(\theta_{m})^{-1}\right] = \frac{1}{N}\det\begin{bmatrix}1 & 0 & 0 & \dots\\-1 & 1 & 0 & \dots\\0 & -1 & 1 & \dots\\\vdots & \vdots & \vdots & \vdots\end{bmatrix} = \frac{1}{N}.$$

This suggests as $T \to \infty$,

$$\frac{1}{N(T)}\sum_{0<\gamma\leq T}\zeta'\left(\frac{1}{2}+i\gamma\right)^{-1}\sim\frac{1}{\log T}.$$

Multiplying through by N(T) gives the result from Theorem 2.3.

For k = -2, our symbol is of the form

$$f(z) = |1 - e^{iz}|^2 (1 - e^{iz})^{-2} = (1 - e^{-it})(1 - e^{it})^{-1} = (1 - e^{-it}) \left(\sum_{k=0}^{\infty} e^{ikt}\right).$$

This sum telescopes to give

$$f(z) = -e^{-it}$$

so the Fourier coefficients are $c_{-1} = -1$, and $c_j = 0$ for $j \ge 0$. Then the Toeplitz determinant is given by

$$\mathbb{E}_{N}\left[\frac{1}{N}\sum_{m=1}^{N}Z_{N}'(\theta_{m})^{-2}\right] = \frac{1}{N}\det\begin{bmatrix}0 & 0 & 0 & \dots\\-1 & 0 & 0 & \dots\\0 & -1 & 0 & \dots\\\vdots & \vdots & \vdots & \vdots\end{bmatrix} = 0.$$

This suggests as $T \to \infty$,

$$\frac{1}{N(T)}\sum_{0<\gamma\leq T}\zeta'\left(\frac{1}{2}+i\gamma\right)^{-2}=O\left(T^{-1/2+\varepsilon}\right),$$

with some numerical evidence given in Figure 5.1, where we suggest an error term of 'typical' size for the zeta function.

Finally for k = -3, our symbol is of the form

$$f(z) = |1 - e^{iz}|^2 (1 - e^{iz})^{-3} = (1 - e^{-it})(1 - e^{it})^{-2} = (1 - e^{-it})(1 - e^{it})^{-1}(1 - e^{it})^{-1}.$$

Then by the k = -2 case, the first two terms in the last equality cancel to give $-e^{-it}$ so we have

$$f(z) = -e^{-it} \left(\sum_{k=0}^{\infty} e^{ikt} \right) = -e^{-it} - 1 - e^{it} - e^{2it} - \dots$$



Figure 5.1: The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^{-2}$ plotted over the first million zeros.

so the Fourier coefficients are $c_j = -1$ for $j \ge -1$. Hence,

$$\mathbb{E}_{N}\left[\frac{1}{N}\sum_{m=1}^{N}Z_{N}'(\theta_{m})^{-3}\right] = \frac{1}{N}\det\begin{bmatrix}-1 & -1 & -1 & \dots\\-1 & -1 & -1 & \dots\\0 & -1 & -1 & \dots\\\vdots & \vdots & \vdots & \vdots\end{bmatrix}.$$

Then expanding along the first column gives two copies of the same matrix, with a minus sign from expanding a determinant, meaning that this expectation is also zero.

Therefore, we could conjecture the follow result for the Riemann zeta function. As $T \to \infty$,

$$\frac{1}{N(T)} \sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^{-3} = O\left(T^{-1/2 + \varepsilon}\right),$$

where again we suggest an error term of 'typical' size for the zeta function.

However, we do emphasise that our model breaks down exactly at k = -3 so we can't give this conjecture with a huge degree of confidence, unlike the above which we have given good supporting evidence for. It would be of great interest to explore how to model the Riemann zeta function at this breaking point, and for higher negative moments.

5.5 A RECURRENCE RELATION HIDDEN IN THE RMT CALCULATION

Consider the left-hand side of (5.4), given by

$$\frac{1}{N}\prod_{r=1}^{k}(1-e^{-i\alpha_{r}})\times\mathbb{E}_{N-1}\left[\prod_{m=1}^{N-1}(1-e^{i\phi_{m}})(1-e^{-i\phi_{m}})\prod_{r=1}^{k}(1-e^{i\phi_{m}}e^{-i\alpha_{r}})\right].$$

By following the method in Section 5.2, that is, following the method from Basor and Forrester [21], we have shown in (5.9) that this equals (after a trivial simplification)

$$\frac{1}{N}\prod_{r=1}^{k} \left(1 - A_r^{-1}\right) \sum_{j_1=0}^{N-1} \sum_{j_2=0}^{N-1-j_1} \cdots \sum_{j_k=0}^{N-1-j_1-\dots-j_{k-1}} A_1^{-j_1} \dots A_k^{-j_k} (N - j_1 - \dots - j_k),$$

where $A_r = e^{i\alpha_r}$.

Hidden behind this method is a recurrence relation, which we now derive and give a solution for. Our previous method in Section 5.2 proves that the solution we give here is the correct one.

Rather than using Heine's Lemma [295] to write the expectation of the right-hand side of (5.4) as a Toeplitz determinant, and then applying the method of Basor and Forrester [21], we instead consider the left-hand side of (5.4) and rearrange it in a naïve way. That is, we first use Heine's Lemma to write

$$\mathbb{E}_{N-1}\left[\prod_{m=1}^{N-1} (1-e^{i\phi_m})(1-e^{-i\phi_m}) \prod_{r=1}^k (1-e^{i\phi_m}e^{-i\alpha_r})\right] = D_{N-1}[f]$$
(5.23)

for the symbol

$$f(z) = (1-z)(1-1/z) \prod_{r=1}^{k} (1-zA_r^{-1}).$$

For ease of notation, we relabel $A_r^{-1} = \mathcal{A}_r$. Expanding f(z) gives

$$f(z) = (1-z)(1-1/z)(1-\mathcal{A}_1 z)\dots(1-\mathcal{A}_k z)$$

= $(-1/z+2-z)(1-\mathcal{A}_1 z)\dots(1-\mathcal{A}_k z)$
= $(-1/z+2-z)(B_0-B_1 z+B_2 z^2+\dots+(-1)^k B_k z^k)$
= $-\frac{1}{z} + (2B_0+B_1) - z(B_0+2B_1+B_2) + z^2(B_1+2B_2+B_3) + \dots +$
+ $(-1)^{k-1} z^{k-1}(B_{k-2}+2B_{k-1}+B_k) + (-1)^k z^k(B_{k-1}+2B_k) + (-1)^{k+1} z^{k+1} B_k$

where

$$B_0 = 1$$

$$B_1 = \sum_{1 \le r \le k} \mathcal{A}_r$$

$$B_2 = \sum_{1 \le r_1 < r_2 \le k} \mathcal{A}_{r_1} \mathcal{A}_{r_2}$$

$$B_3 = \sum_{1 \le r_1 < r_2 < r_3 \le k} \mathcal{A}_{r_1} \mathcal{A}_{r_2} \mathcal{A}_{r_3}$$

$$\vdots$$

$$B_k = \mathcal{A}_1 \dots \mathcal{A}_k$$

are the elementary symmetric polynomials in k variables.

The Fourier coefficients we need to write this expectation as the Toeplitz determinant $D_{N-1}[f]$ are then given by

$$f_{j} = \begin{cases} 0 & \text{for } j \leq -2 \\ -1 & \text{for } j = -1 \\ 2B_{0} + B_{1} & \text{for } j = 0 \\ (-1)^{j}(B_{j-1} + 2B_{j} + B_{j+1}) & \text{for } 1 \leq j \leq k - 1 \\ (-1)^{k}(B_{k-1} + 2B_{k}) & \text{for } j = k \\ (-1)^{k+1}B_{k} & \text{for } j = k + 1 \\ 0 & \text{for } j \geq k + 2. \end{cases}$$

The expectation in (5.23) (with $N \ge k+4$) may be written as the $(N-1) \times (N-1)$ Toeplitz determinant

$$D_{N-1}[f] = \det \begin{bmatrix} f_0 & f_1 & f_2 & f_3 & \dots & f_{k+1} & 0 & 0 & \dots \\ -1 & f_0 & f_1 & f_2 & \dots & f_k & f_{k+1} & 0 & \dots \\ 0 & -1 & f_0 & f_1 & \dots & f_{k-1} & f_k & f_{k+1} & \dots \\ 0 & 0 & -1 & f_0 & \dots & f_{k-2} & f_{k-1} & f_k & \dots \\ \vdots & \vdots & \ddots & \ddots & \ddots & \ddots & \ddots & \ddots & \cdots \end{bmatrix}$$

with the f_j for $j \ge 0$ as given above. Expanding this determinant along the leading column repeatedly, we are lead to the following recurrence relation (where we have not substituted the f_j given above for ease of notation)

$$D_{N-1}[f] = f_0 D_{N-2}[f] + f_1 D_{N-3}[f] + \ldots + f_{k+1} D_{N-k-3}[f].$$
(5.24)

We then know by (5.9) that the solution to this recurrence relation is

$$\sum_{j_1=0}^{N-1} \sum_{j_2=0}^{N-1-j_1} \cdots \sum_{j_k=0}^{N-1-j_1-\cdots-j_{k-1}} A_1^{-j_1} \dots A_k^{-j_k} (N-j_1-\cdots-j_k),$$

which can be written as

$$D_{N-1}[f] = N + (N-1)C_1 + (N-2)C_2 + \ldots + C_{N-1}$$

where

$$C_j = \sum_{1 \le r_1 \le r_2 \le r_3 \cdots \le r_j \le k} \mathcal{A}_{r_1} \mathcal{A}_{r_2} \mathcal{A}_{r_3} \dots \mathcal{A}_{r_j}$$

are defined similarly to the B_j other than none of the inequalities are strict. Note that this gives a solution to the hidden recurrence relation (5.24) as claimed.

Higher Moments of the Riemann Zeta Function via the Ratios Conjecture

In this chapter we assume the Riemann Hypothesis, unless otherwise stated.

In Section 6.1 we begin this chapter with a summary of some results that one can obtain from the Ratios Conjecture, many of which are rigorously known. Its strength is in deriving these results in a generally easier way, and often giving more information, such as the lower order terms for asymptotics. The Ratios Conjecture is a heuristic argument based on certain RMT results. For an integral containing quotients of products of zeta functions (or more generally *L*-functions), we perform the follow steps (ignoring all error terms at all steps):

- 1. Replace all zeta functions in the denominator by its Dirichlet series (which we know conditionally converges up to the critical line under the Riemann Hypothesis).
- 2. Replace all zeta functions in the numerator by the approximate functional equation.
- 3. Expand the new terms in the integral, keeping only those terms with an equal number of $\chi(s)$ and $\chi(1-s)$ terms. These are the terms that are not rapidly oscillating.
- 4. In these remaining terms, keep only the diagonal terms.
- 5. Complete the resulting sums, factoring out various zeta factors to make the resulting terms converge.
- 6. A power-saving error term is then conjectured for the final result.

A full example will be seen when we use the Ratios Conjecture to calculate our conjectures in Section 6.2.

In Section 6.2 we will use the Ratios Conjecture in several different ways. First we give a general closed integral form to find a full asymptotic expansion for any moment for any number of derivatives. Then we re-derive Conjectures 5.3 and 5.4 following the Ratios Conjecture methodology. We have discussed that the Ratios Conjecture can be used to find lower order terms for discrete moments of the Riemann zeta function. Applying similar ideas, we can use the Ratios Conjecture to find lower order terms for the conjectures formulated in Chapter 5.

Finally we give two examples of our result, the first of which is to find an integral form of the full asymptotic for the Generalised Shanks' Conjecture, first given in Theorem 3.1. This integral form is simpler than our previous result, and as has been seen in recent times, the 'correct' way of writing moments are as integrals.

Theorem 6.1. For *n* a positive integer and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function,

$$\sum_{0<\gamma\leq T} \zeta^{(n)} \left(\frac{1}{2} + i\gamma\right) = \frac{n!}{2\pi} \int_{1}^{T} \left(A_n + \frac{(-1)^{n+1}L^{n+1}}{(n+1)!} + \sum_{m=0}^{n} \frac{(-1)^{m+1}L^m \gamma_{n-m}}{m!(n-m)!}\right) dt + O\left(T^{1/2+\varepsilon}\right),$$

where $L = \log \frac{t}{2\pi}$, and the γ_m are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1, and the A_n are the coefficients in the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1.

Note that while we assume the Riemann Hypothesis in this chapter, but as previously seen, this result is actually unconditional with the only change in the error term.

The second example is a conjecture for the second moment of the first derivative of the Riemann zeta function, a previously unknown result. Specifically, we conjecture the following result.

Conjecture 6.2. For $L = \log \frac{t}{2\pi}$ and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function,

$$\begin{split} \sum_{0<\gamma\leq T} \zeta' \left(\frac{1}{2} + i\gamma\right)^2 &= \\ \frac{1}{2\pi} \int_1^T \left(\frac{1}{6}L^3 + \frac{1}{2}L^2 \left(2\gamma_0 + A^{(0,0,1)}\right) + \frac{1}{2}L \left(-8\gamma_1 + 4\gamma_0 A^{(0,0,1)} + A^{(0,0,2)} + 2A^{(0,1,1)}\right) \\ &+ \frac{1}{6} \left(-12\gamma_0^3 - 36\gamma_0\gamma_1 + 6\gamma_2 - 24\gamma_1 A^{(0,0,1)} + 6\gamma_0 A^{(0,0,2)} + A^{(0,0,3)} \right. \\ &\left. + 12\gamma_0 A^{(0,1,1)} + 3A^{(0,1,2)} - 3A^{(0,2,1)} + 6A^{(1,1,1)}\right) dt + O\left(T^{1/2+\varepsilon}\right) \end{split}$$

where the γ_m are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1 and the $A^{(i,j,k)}$ are arithmetic terms that are various products and sums over primes.

Much of this chapter is based on an upcoming paper, and will soon appear in [185]. Notation and cross-references have been updated for this thesis.

6.1 THE RATIOS CONJECTURE

The recipe of Conrey, Farmer, Keating, Rubinstein and Snaith [60] is used to calculate products of L-functions averaged over a family with full asymptotic formula with a power-saving error term. The Ratios Conjecture generalises the recipe to quotients of products of L-functions. The recipe was created through comparison with the analogous quantities for the characteristic polynomials of matrices averaged over classical compact groups [60], and various analogous RMT results for the Ratios Conjecture can be found in [63, 62, 36, 64, 180].

A conjecture of Farmer [112] states that for $s = \sigma + it$ and for complex shifts $\alpha, \beta, \gamma, \delta$ with small positive real parts,

$$\frac{1}{T} \int_0^T \frac{\zeta(s+\alpha)\zeta(1-s+\beta)}{\zeta(s+\gamma)\zeta(1-s+\delta)} \, dt \sim \frac{(\alpha+\delta)(\beta+\gamma)}{(\alpha+\beta)(\gamma+\delta)} - T^{-(\alpha+\beta)} \frac{(\alpha-\gamma)(\beta-\delta)}{(\alpha+\beta)(\gamma+\delta)}$$

The Ratios Conjecture generalises Farmer's conjecture to quotients of an arbitrary number of L-functions averaged over a family, with a full asymptotic expansion and a power-saving error term in each case. It does so through a similar process to the recipe, with one extra step. In the first step of the Ratios Conjecture, for any L-function in the denominator, we replace it by its Dirichlet series, and carry on the recipe as usual. This is acceptable as under the (Generalised) Riemann Hypothesis, the Dirichlet series converges up to the critical line.

The reason we are so interested in the Ratios Conjecture is that it has numerous applications to a number of questions that arise from number theory. We give a brief overview of some of these applications below. Conrey and Snaith [87] go into a great deal of detail in their paper but we give a short summary of some of their results here. The versatility of the Ratios Conjecture allows us to recover many previously known or conjectured results, and often pushes these further.

Throughout the following subsections we assume the truth of both the Ratios Conjecture, and of the Riemann Hypothesis.

6.1.1 PAIR CORRELATION AND *n*-LEVEL CORRELATIONS

Farmer noted that his original conjecture implies the leading order terms of Montgomery's Pair Correlation Conjecture. The Ratios Conjecture can be used to calculate full asymptotics of all n-level correlations of zeros, as predicted heuristically by Bogomolny and Keating [30, 31], and stated by Rudnick and Sarnark [274] and later proved by them in [275], albeit in a restricted range.

As an example, Conrey and Snaith [87] explicitly calculate the full asymptotic for Pair Correlation, and show how one can recover the leading order behaviour, exactly as in Conjecture 1.3.

Additionally, Conrey and Snaith have found full asymptotics for other n-level correlations via the Ratios Conjecture, for example in [89, 88, 86].

6.1.2 Moments of Logarithmic Derivatives of the Riemann zeta function

Goldston, Gonek and Montgomery [141] considered a second continuous moment of the logartihmic derivative of the Riemann zeta function, and found an equivalence with it and Montgomery's Pair Correlation Conjecture. Using the Ratios Conjecture, lower order terms for such a problem were found. Higher moments are considered in [12, 139], and a weighted moment version is studied in [114]. Similarly, Farmer [112] considered discrete moments of the ratio $\zeta(\rho + \alpha)/\zeta(\rho + \beta)$ for small shifts α, β and obtained a leading order asymptotic. Using the Ratios Conjecture, Conrey, Farmer and Zirnbauer [63] found a full asymptotic for moments of this form.

6.1.3 Mollified Moments of L-functions

As we discussed in detail in Section 1.3.7.2, mollifying moments is used for computing information about zeros, for example for obtaining lower bounds for the proportion of zeros on the critical line as in 1.3.7.3, and as have seen in Section 1.4.3.1, for obtaining lower bounds for the proportion of simple zeros as in [75, 39].

Recall that when we want to perform a mollified moment calculation, we take our L-function and multiply it by a Dirichlet polynomial which acts as an approximation to the inverse of the L-function. As discussed in Section 1.3.7.3, mollifier calculations are generally very complicated. The Ratios Conjecture gives us a relatively simple way to obtain the relevant asymptotic formulae, and serves as a guide and goal to aim for. Recall that we described how Young [316] shortened Levinson's result that one-third of the zeros of $\zeta(s)$ lie on the critical line from 50 pages to 8 pages, using this idea.

These mollified calculations via the Ratios Conjecture also provide evidence that meanvalue formulae, that can be proved for short mollifiers, remain true for long mollifiers. This gives more evidence for Farmer's $\theta = \infty$ conjecture. While we have assumed the Riemann Hypothesis in this section, these calculations retain their value.

6.1.4 Non-vanishing results for various families of L-functions

While we have not discussed other L-functions in much detail in this thesis, we note that the Ratios Conjecture can be used to show that the proportion of non-vanishing for 6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 175

 $L^{(k)}(1/2, \chi)$ approaches 100% as $k \to \infty$ for this family. The Ratios Conjecture can also be used to deduce results about the non-vanishing of automorphic *L*-functions, and with non-vanishing of Dirichlet *L*-functions for real quadratic characters, at the central point and on the real axis. For further reading see [201, 232, 289].

6.1.5 Discrete Moments of $\zeta(s)$ and Lower Order Terms

The Ratios Conjecture can be used to recover discrete moment results, with full asymptotic expansions. Conrey and Snaith [87] show how to obtain this asymptotic for the second moment, and in particular, preempted Milinovich's result [233], referred to in (1.33) (albeit in integral form), for the full asymptotic for the second discrete moment. In particular, they showed that assuming the Ratios Conjecture, we have

$$\begin{split} \sum_{0<\gamma\leq T} |\zeta'(\rho)|^2 &= \int_0^T \left(\frac{1}{24\pi} \left(\log \frac{t}{2\pi} \right)^4 + \frac{\gamma_0}{3\pi} \left(\log \frac{t}{2\pi} \right)^3 + \left(\frac{\gamma_0^2}{2\pi} - \frac{\gamma_1}{\pi} \right) \left(\log \frac{t}{2\pi} \right)^2 \\ &- \left(\frac{\gamma_0^3}{\pi} + \frac{5\gamma_0\gamma_1}{\pi} + \frac{\gamma_2}{2\pi} \right) \log \frac{t}{2\pi} + \left(\frac{\gamma_0^4}{\pi} + \frac{6\gamma_0^2\gamma_1}{\pi} + \frac{7\gamma_1^2}{\pi} + \frac{4\gamma_0\gamma_2}{\pi} + \frac{5\gamma_3}{3\pi} \right) \right) \\ &\times \left(1 + O\left(t^{-1/2 + \varepsilon} \right) \right) \, dt. \end{split}$$

Note that performing the integral recovers Gonek's [143] result to leading order. Milinovich's theorem [233] is proved in a different way but he obtains the same asymptotic which can be seen by computing the integral. Milinovich's asymptotic also follows from a result of Fujii's [127]. Note that none of these proofs require the Ratios Conjecture.

6.2 LOWER ORDER TERMS FOR MOMENTS OF THE DERIVATIVES OF THE RIEMANN ZETA FUNCTION

The aim of this chapter is to extend the Conjectures 5.3 and 5.4 to include the lower order terms with a power-saving error term. We do this by evaluating discrete moments of zeta with small shifts α_j . Throughout this chapter we assume that $|\Re(\alpha_j)| < \frac{1}{4}$ and $|\Im(\alpha_j)| \ll_{\varepsilon} T^{1-\varepsilon}$ for every $\varepsilon > 0$. We will also assume the Riemann Hypothesis.

Specifically, from the ratios methodology we will obtain the following conjecture.

Conjecture 6.3. Assume the Riemann Hypothesis. For $|\delta| < 1/4$, let

$$A_{\{\alpha_1,\dots,\alpha_k\}}(\delta) := \prod_p \frac{1 + F_1(p,\delta) + F_2(p,\delta) + \dots + F_k(p,\delta)}{(1 - p^{-(1 + \alpha_1)}) \dots (1 - p^{-(1 + \alpha_k)})}$$
(6.1)

where for $1 \leq m \leq k$

$$F_m(p,\delta) = (-1)^m \sum_{\substack{J \subset \{1,...,k\} \\ |J|=m}} p^{-\left(m + (m-1)\delta + \sum_{j \in J} \alpha_j\right)}.$$

The discrete moments of shifts of $\zeta(s)$, evaluated at the non-trivial zeros $\rho = 1/2 + i\gamma$ of $\zeta(s)$, are given by

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_1\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_k\right)$$
$$=\frac{1}{2\pi}\int_1^T \left(A'_{\{\alpha_1,\dots,\alpha_k\}}(0)+\sum_{j=1}^k W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t)\right) dt+O\left(T^{1/2+\varepsilon}\right).$$

where

$$W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t) = \frac{\zeta'(1+\alpha_j)}{\zeta(1+\alpha_j)} - \left(\frac{t}{2\pi}\right)^{-\alpha_j} \zeta(1-\alpha_j) A_{\{\alpha_1,\dots,\alpha_{j-1},0,\alpha_{j+1},\dots,\alpha_k\}}(-\alpha_j) \prod_{\substack{\ell=1\\\ell\neq j}}^k \frac{\zeta(1+\alpha_\ell-\alpha_j)}{\zeta(1+\alpha_\ell)}.$$

Remark. Note that if $\delta = 0$ we write $F_m(p,0) = F_m(p)$, and in this case have

$$1 + F_1(p) + F_2(p) + \dots + F_k(p) = (1 - p^{-(1 + \alpha_1)}) \dots (1 - p^{-(1 + \alpha_k)}),$$

so $A_{\{\alpha_1,...,\alpha_k\}}(0) = 1.$

Remark. To obtain the mixed derivative that we are interested in, Taylor expand each $\zeta(1/2 + i\gamma + a_j)$ about $a_j = 0$, take derivatives with respect to a_j and then set each α_j equal to zero.

Remark. As we previously discussed, the error term here can be contentious. While some authors prefer an error term of $O(T^{1-\delta})$ for $0 < \delta < 1/2$, in this conjecture and our later examples, we suggest an error term of $O(T^{1/2+\varepsilon})$, in keeping with the original statements of the recipe/Ratios Conjecture.

6.2.1 Preliminary Lemmas on Vandermonde Determinants

As part of our later calculations, a certain identity between Vandemonde determinants will prove useful.

Let

$$\Delta_k (\alpha_1, \dots, \alpha_k) = \det \begin{pmatrix} 1 & \alpha_1 & \alpha_1^2 & \dots & \alpha_1^{k-1} \\ 1 & \alpha_2 & \alpha_2^2 & \dots & \alpha_2^{k-1} \\ \vdots & \vdots & \vdots & \dots & \vdots \\ 1 & \alpha_k & \alpha_k^2 & \dots & \alpha_k^{k-1} \end{pmatrix}$$
$$= \prod_{1 \le i < j \le k} (\alpha_j - \alpha_i)$$

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta FUNCTION

denote the Vandermonde determinant with k parameters, and also let

$$\Delta_{k-1}(\alpha_1,\ldots,\alpha_k;\alpha_\ell) = \prod_{\substack{1 \le i < j \le k \\ i \ne \ell \\ j \ne \ell}} (\alpha_j - \alpha_i)$$

denote the Vandermonde determinant with the ℓ^{th} element removed.

Lemma 6.4. For $k \in \mathbb{N}$ and $n \in \mathbb{Z}$, if $\alpha_1, \ldots, \alpha_k \neq 0$ then

$$\sum_{\ell=1}^{k} (-1)^{k+\ell} \Delta_{k-1} (\alpha_1, \dots, \alpha_k; \alpha_\ell) \alpha_\ell^{n-1}$$

$$= \begin{cases} \Delta_k (\alpha_1, \dots, \alpha_k) \sum_{\substack{n_1 \ge 0, \dots, n_k \ge 0 \\ n_1 + \dots + n_k = n-k}} \alpha_1^{n_1} \alpha_2^{n_2} \dots \alpha_k^{n_k} & \text{if } n \ge k \\ 0 & \text{if } 1 \le n \le k-1 \\ (-1)^{k+1} \Delta_k (\alpha_1, \dots, \alpha_k) \sum_{\substack{n_1 \ge 0, \dots, n_k \ge 0 \\ n_1 + \dots + n_k = |n|}} \alpha_1^{-n_1 - 1} \alpha_2^{-n_2 - 1} \dots \alpha_k^{-n_k - 1} & \text{if } n \le 0. \end{cases}$$

Proof. Let

$$X = \sum_{\ell=1}^{k} (-1)^{k+\ell} \Delta_{k-1} (\alpha_1, \dots, \alpha_k; \alpha_\ell) \alpha_\ell^{n-1}.$$

Observe that X can be expressed as a determinant,

$$X = \det \begin{pmatrix} 1 & \alpha_1 & \alpha_1^2 & \dots & \alpha_1^{k-2} & \alpha_1^{n-1} \\ 1 & \alpha_2 & \alpha_2^2 & \dots & \alpha_2^{k-2} & \alpha_2^{n-1} \\ \vdots & \vdots & \vdots & \dots & \vdots & \vdots \\ 1 & \alpha_k & \alpha_k^2 & \dots & \alpha_k^{k-2} & \alpha_k^{n-1} \end{pmatrix}$$

where the terms in the last column are replaced by α_j^{n-1} (in the standard Vandermonde these terms would be α_j^{k-1}). Expanding the determinant down its last column, it is clear that it equals X.

Note that if $n \leq k-1$ then the last column will be equal to one of the other columns in the matrix, and thus the determinant will be zero.

To complete the proof of the lemma, we now evaluate X in a different manner, via row manipulations.

Subtract the first row from all subsequent rows. Pull out a factor of $(\alpha_i - \alpha_1)$ from the j^{th} row (for $j \ge 2$).

Then subtract the new second row from all subsequent rows, and pull out a factor of $(\alpha_j - \alpha_2)$ from the j^{th} row (for $j \ge 3$ this time).

Repeat this process all the way down to subtracting what remains in the the $k - 1^{\text{st}}$ row from what remains in the k^{th} row, and pulling out a factor of $(\alpha_k - \alpha_{k-1})$.

The terms that are factored out during this process are

$$\prod_{\ell=1}^{k-1} \prod_{j=\ell+1}^{k} (\alpha_j - \alpha_\ell) = \Delta_k (\alpha_1, \dots, \alpha_k)$$

and the remaining matrix determinant is

$$\det \begin{pmatrix} A_{1,1} & A_{1,2} & \dots & A_{1,k-1} & B_1 \\ A_{2,1} & A_{2,2} & \dots & A_{2,k-1} & B_2 \\ \vdots & \vdots & \vdots & \vdots & \vdots \\ A_{k,1} & A_{k,2} & \dots & A_{k,k-1} & B_k \end{pmatrix}$$

where

$$A_{\ell,m} = \sum_{\substack{n_1,\dots,n_\ell \ge 0\\n_1 + \dots + n_\ell = m - \ell}} \alpha_1^{n_1} \dots \alpha_\ell^{n_\ell}$$

and where $B_{\ell} = A_{\ell,n}$ in the case $n \ge 1$, and in the case $n \le 0$,

$$B_{\ell} = (-1)^{\ell+1} \sum_{\substack{n_1, \dots, n_{\ell} \ge 0\\ n_1 + \dots + n_{\ell} = |n|}} \alpha_1^{-n_1 - 1} \dots \alpha_{\ell}^{-n_{\ell} - 1}.$$

Crucially note that $A_{\ell,m} = 0$ for $m < \ell$ and $A_{\ell,\ell} = 1$. This makes the determinant trivial to evaluate - the matrix is upper triangular, and so the determinant will equal B_k .

That is, we have shown that

$$X = \Delta_k \left(\alpha_1, \ldots, \alpha_k \right) B_k$$

and this is the right-hand of the statement of the lemma.

From this we can derive two useful combinatorial sums.

Lemma 6.5. For $k \in \mathbb{N}$ and distinct nonzero $\alpha_1, \ldots, \alpha_k \in \mathbb{C}$, we have

$$\sum_{\ell=1}^k \frac{1}{\alpha_\ell} \prod_{\substack{j=1\\j \neq \ell}}^k \frac{\alpha_j}{(\alpha_j - \alpha_\ell)} = \sum_{\ell=1}^k \frac{1}{\alpha_\ell}.$$

Proof. From Lemma 6.4 with n = -1 we have

$$\sum_{\ell=1}^{k} (-1)^{k+\ell} \Delta_{k-1} (\alpha_1, \dots, \alpha_k; \alpha_\ell) \alpha_\ell^{-2} = (-1)^{k+1} \Delta_k (\alpha_1, \dots, \alpha_k) \frac{1}{\alpha_1 \dots \alpha_k} \sum_{\ell=1}^{k} \frac{1}{\alpha_\ell} \alpha_\ell^{-2}$$

Since the α_i are all distinct, Δ_k is nonzero, so dividing both sides by

$$(-1)^{k+1}\frac{\Delta_k(\alpha_1,\ldots,\alpha_k)}{\alpha_1\ldots\alpha_k}$$

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 179

yields

$$\sum_{\ell=1}^{k} \frac{(-1)^{\ell-1} \Delta_{k-1} (\alpha_1, \dots, \alpha_k; \alpha_\ell)}{\Delta_k(\alpha_1, \dots, \alpha_k)} \frac{\alpha_1 \dots \alpha_k}{\alpha_\ell^2} = \sum_{\ell=1}^{k} \frac{1}{\alpha_\ell}.$$

Note that

$$\frac{(-1)^{\ell-1}\Delta_{k-1}(\alpha_1,\ldots,\alpha_k;\alpha_\ell)}{\Delta_k(\alpha_1,\ldots,\alpha_k)} = \prod_{\substack{j=1\\j\neq\ell}}^k \frac{1}{\alpha_j - \alpha_\ell}$$

where we observe that for $j = 1, ..., \ell - 1$ there are terms $\alpha_{\ell} - \alpha_{j}$ in the Vandermonde, whereas in the product on the RHS they are $\alpha_{j} - \alpha_{\ell}$, thus accounting for the $(-1)^{\ell-1}$ term). Therefore we have shown

$$\frac{(-1)^{\ell-1}\Delta_{k-1}(\alpha_1,\ldots,\alpha_k;\alpha_\ell)}{\Delta_k(\alpha_1,\ldots,\alpha_k)}\frac{\alpha_1\ldots\alpha_k}{\alpha_\ell^2} = \frac{1}{\alpha_\ell}\prod_{\substack{j=1\\j\neq\ell}}^k \frac{\alpha_j}{\alpha_j-\alpha_\ell}$$

as required.

Lemma 6.6. For distinct $\alpha_j \in \mathbb{C}$, for $n \in \mathbb{N}$ and $k \in \mathbb{N}$ we have

$$\sum_{\ell=1}^{k} \alpha_{\ell}^{n} \prod_{\substack{j=1\\ j \neq \ell}}^{k} \frac{\alpha_{j}}{(\alpha_{j} - \alpha_{\ell})} = \begin{cases} (-1)^{k+1} \sum_{\substack{n_{1} \ge 1, \dots, n_{k} \ge 1\\ n_{1} + \dots + n_{k} = n}} \alpha_{1}^{n_{1}} \alpha_{2}^{n_{2}} \dots \alpha_{k}^{n_{k}} & \text{if } n \ge k \\ 0 & \text{if } n < k. \end{cases}$$

Proof. As in the previous lemma, we multiply both sides of the statement of Lemma 6.4 for the case $n \ge 1$ by $\prod_{j=1}^{k} \alpha_j$, and divide by the Vandermonde determinant $\Delta_k(\alpha_1, \ldots, \alpha_k)$. Again, note that the assumption that all the variables α_j are distinct means their Vandermonde determinant is non-zero, so this is a valid division.

On the left-hand side of Lemma 6.4 this yields terms such as

$$(-1)^{k+\ell} \frac{\Delta_{k-1}(\alpha_1, \dots, \alpha_k; \alpha_\ell)}{\Delta_k(\alpha_1, \dots, \alpha_k)} \alpha_\ell^{n-1} \prod_{j=1}^k \alpha_j = (-1)^{k-1} \alpha_\ell^n \prod_{\substack{j=1\\j\neq\ell}}^k \frac{\alpha_j}{\alpha_j - \alpha_\ell}$$

and the effect on the right-hand side of Lemma 6.4 is that

$$\prod_{j=1}^{k} \alpha_{j} \sum_{\substack{n_{1} \ge 0, \dots, n_{k} \ge 0\\ n_{1} + \dots + n_{k} = n - k}} \alpha_{1}^{n_{1}} \alpha_{2}^{n_{2}} \dots \alpha_{k}^{n_{k}} = \sum_{\substack{n_{1} \ge 1, \dots, n_{k} \ge 1\\ n_{1} + \dots + n_{k} = n}} \alpha_{1}^{n_{1}} \alpha_{2}^{n_{2}} \dots \alpha_{k}^{n_{k}}$$

which yields the statement of this lemma, when we note that if $1 \le n \le k$ then the sum is empty, so equals 0.

6.2.2 DERIVING CONJECTURE 6.3

We are required to find

$$\sum_{0 < \gamma \le T} \prod_{j=1}^{k} \zeta \left(\frac{1}{2} + i\gamma + \alpha_j \right)$$

where the α_j satisfy certain conditions as specified at the start of Section 6.2.

We may use Cauchy's theorem to write

$$\sum_{0<\gamma\leq T}\prod_{j=1}^{k}\zeta\left(\frac{1}{2}+i\gamma+\alpha_{j}\right)=\frac{1}{2\pi i}\int_{R}\frac{\zeta'}{\zeta}(s)\zeta\left(s+\alpha_{1}\right)\ldots\zeta\left(s+\alpha_{k}\right)\ ds,$$

where R is a positively oriented rectangular contour with vertices c + i, c + iT, 1 - c + iT, and 1 - c + i, where 1/2 < c < 3/4. For simplicity we assume that $|T - \gamma| \gg 1/\log T$ for any zero γ and T sufficiently large, although this constraint has no effect on the final answer.

One can prove that the bottom, right, and top sides of the contour do not contribute anything except to the error term, with the largest contribution coming from the top piece of the contour at $O(T^{1/2+\varepsilon})$.

We use the functional equation of $\zeta(s)$ given in (1.1) to write

$$\frac{\zeta'}{\zeta}(s) = \frac{\chi'}{\chi}(s) - \frac{\zeta'}{\zeta}(1-s)$$

and rewrite the left-hand side of the contour. Again the part that comes from the $\chi'(s)/\chi(s)$ term is also small and only contributes to the error term.

Therefore we have shown that

$$\sum_{0<\gamma\leq T}\prod_{j=1}^{k}\zeta\left(\frac{1}{2}+i\gamma+\alpha_{j}\right)=\frac{1}{2\pi i}\int_{c+i}^{c+iT}\frac{\zeta'}{\zeta}(s)\zeta(1-s+\alpha_{1})\ldots\zeta(1-s+\alpha_{k})\,ds+O\left(T^{1/2+\varepsilon}\right)$$

This integral can be rewritten as

$$\frac{d}{d\delta} \frac{1}{2\pi} \int_{1}^{T} \frac{\zeta(c+it+\delta)}{\zeta(c+it)} \zeta(1-c-it+\alpha_{1}) \dots \zeta(1-c-it+\alpha_{k}) ds \bigg|_{\delta=0}$$
(6.2)

for some small shift δ .

We now apply the recipe from the Ratios Conjecture. Step 1 tells us that for the $\zeta(s)$ term in the denominator, we replace it with its Dirichlet series

$$\frac{1}{\zeta(s)} = \sum_{n=1}^{\infty} \frac{\mu(n)}{n^s}$$

noting that, assuming the Riemann Hypothesis, this is conditionally convergent on the line of integration. Step 2 tells us that for any $\zeta(s)$ terms in the numerator, we replace
6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 181

them using the approximate functional equation

$$\zeta(s) = \sum_{n \le \sqrt{t/2\pi}} \frac{1}{n^s} + \chi(s) \sum_{n \le \sqrt{t/2\pi}} \frac{1}{n^{1-s}}$$

ignoring the error terms.

Step 3 of the Ratio's Conjecture tells us to now expand the numerator, yielding 2^{k+1} terms coming from picking either the sum without a χ factor or the sum with a χ factor in each zeta function.

For example, we can keep every term from each of the approximate functional equations without a $\chi(s)$ factor, which gives

$$\sum_{m} \frac{1}{n^{c+it+\delta}} \sum_{n_1} \frac{1}{n_1^{1-c-it+\alpha_1}} \cdots \sum_{n_k} \frac{1}{n_k^{1-c-it+\alpha_1}} \sum_{h} \frac{\mu(h)}{h^{c+it}}.$$

Step 4 tells us that when we integrate over t we want to keep only the "diagonal terms", which are the terms in the sum where there are no oscillations in t, namely where

$$mh = n_1 \dots n_k.$$

The ratios methodology assumes all the non-diagonal terms oscillate away, and so we are left with

$$\sum_{mh=n_1\dots n_k} \frac{\mu(h)}{m^{1/2+\delta} h^{1/2} n_1^{1/2+\alpha_1} \dots n_k^{1/2+\alpha_k}}.$$

Since this sum is multiplicative, we may write this as an Euler product. Writing

$$n_j = p^{a_j}, m = p^c, h = p^d$$

we have

$$\prod_{p} \sum_{c+d=a_1+\dots+a_k} \frac{\mu(p^d)}{p^{c(1/2+\delta)} p^{d(1/2)} p^{a_1(1/2+\alpha_1)} \dots p^{a_k(1/2+\alpha_k)}}.$$

We may only have d = 0 (in which case $\mu(1) = 1$), or d = 1 (in which case $\mu(p) = -1$).

Summing over both possibilities for d gives

$$\prod_{p} \sum_{a_1=0}^{\infty} \cdots \sum_{a_k=0}^{\infty} \frac{1}{p^{a_1(1+\alpha_1+\delta)}} \cdots \frac{1}{p^{a_k(1+\alpha_k+\delta)}}$$
$$-\prod_{p} p^{\delta} \left(\sum_{a_1=0}^{\infty} \cdots \sum_{a_k=0}^{\infty} \frac{1}{p^{a_1(1+\alpha_1+\delta)}} \cdots \frac{1}{p^{a_k(1+\alpha_k+\delta)}} - 1 \right).$$

Factorising and using the sum of the geometric formula gives

$$\prod_{p} \frac{1}{1 - p^{-(1 + \alpha_1 + \delta)}} \cdots \frac{1}{1 - p^{-(1 + \alpha_k + \delta)}} \left(1 - p^{\delta} (1 - (1 - p^{-(1 + \alpha_1 + \delta)}) \cdots (1 - p^{-(1 + \alpha_k + \delta)}) \right)$$

The Euler product for $\zeta(s)$ allows us to rewrite this as

$$\zeta(1+\alpha_1+\delta)\dots\zeta(1+\alpha_k+\delta) = \prod_p \left(1-p^{-(1+\alpha_1)}-\dots-p^{-(1+\alpha_k)}+p^{-(2+\alpha_1+\alpha_2+\delta)}+\dots+p^{-(2+\alpha_{k-1}+\alpha_k+\delta)} + \dots + (-1)^k p^{-(k+\alpha_1+\dots+\alpha_k+(k-1)\delta)}\right).$$

Finally, in Step 5, to make this product over primes converge we take out factors of $\zeta(1 + \alpha_j)$ for each $j = 1, \ldots, k$. In doing so, we have

$$\frac{\zeta(1+\alpha_1+\delta)\dots\zeta(1+\alpha_k+\delta)}{\zeta(1+\alpha_1)\dots\zeta(1+\alpha_k)}\prod_p\frac{1-p^{-(1+\alpha_1)}+\dots+(-1)^kp^{-(k+\alpha_1+\dots+\alpha_k+(k-1)\delta)}}{(1-p^{-(1+\alpha_1)})\dots(1-p^{-(1+\alpha_k)})} =: Z_{\alpha_1,\dots,\alpha_k,\delta}.$$

For later simplicity, noticing the symmetry in the parameters $\alpha_1, \ldots, \alpha_k$, we will denote the product over primes in the previous line by

$$A_{\{\alpha_1,\dots,\alpha_k\}}(\delta) := \prod_p \frac{1 - p^{-(1+\alpha_1)} + \dots + (-1)^k p^{-(k+\alpha_1+\dots+\alpha_k+(k-1)\delta)}}{(1 - p^{-(1+\alpha_1)}) \dots (1 - p^{-(1+\alpha_k)})}$$

We now return to the numerator in (6.2) when the zeta terms have been replaced by the appropriate approximate functional equation. The ratios methodology tells us to keep only the terms with an equal number of $\chi(s)$ and $\chi(1-s)$ terms, since all the other terms will be highly oscillatory, and hence are assumed to not contribute to the final answer.

Other than the case when there are no χ terms (dealt with above), it is clear from (6.2) that the only such terms will come from $\chi(c+it+\delta)$ and $\chi(1-c-it+\alpha_j)$ for each j.

Note that

$$\chi(c+it+\delta)\chi(1-c-it+\alpha_j) = \left(\frac{t}{2\pi}\right)^{-\delta-\alpha_j} \left(1+O\left(\frac{1}{|t|}\right)\right)$$

and the rest of the calculation (namely, the summing over the diagonal terms) proceeds as in the "no swap" case, given above. Thus the term arising from the swap of α_j with δ contributes

$$\left(\frac{t}{2\pi}\right)^{-\alpha_j-\delta} Z_{\alpha_1,\dots,\alpha_{j-1},-\delta,\alpha_{j+1},\dots,\alpha_k,-\alpha_j}.$$

Overall, we have

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_{1}\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_{k}\right) = \frac{d}{d\delta}\frac{1}{2\pi}\int_{1}^{T}Z_{\alpha_{1},\dots,\alpha_{k},\delta}+\left(\frac{t}{2\pi}\right)^{-\alpha_{1}-\delta}Z_{-\delta,\alpha_{2},\dots,\alpha_{k},-\alpha_{1}}+\dots+\left(\frac{t}{2\pi}\right)^{-\alpha_{k}-\delta}Z_{\alpha_{1},\dots,-\delta,-\alpha_{k}}dt\Big|_{\delta=0} +O\left(T^{1/2+\varepsilon}\right), \quad (6.3)$$

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 183

where

$$Z_{\alpha_1,\dots,\alpha_k,\delta} = \frac{\zeta(1+\alpha_1+\delta)\dots\zeta(1+\alpha_k+\delta)}{\zeta(1+\alpha_1)\dots\zeta(1+\alpha_k)} A_{\{\alpha_1,\dots,\alpha_k\}}(\delta).$$

Remark. It is not obvious from inspection that the main term of (6.3) is holomorphic in terms of the shift parameters. However the symmetries of the expression imply that the poles cancel to form a holomorphic function. Lemma 6.7 of [63] exhibits an integral representation for the permutation sum that proves the holomorphy.

We proceed to pull the $\frac{d}{d\delta}$ inside the integral. Note that

$$\frac{d}{d\delta}Z_{\alpha_1,\dots,\alpha_k,\delta}\Big|_{\delta=0} = A'_{\{\alpha_1,\dots,\alpha_k\}}(0) + \sum_{j=1}^k \frac{\zeta'(1+\alpha_j)}{\zeta(1+\alpha_j)}$$

and due to the fact that $\frac{1}{\zeta(1-\delta)}\Big|_{\delta=0} = 0$ and $\lim_{\delta\to 0} \frac{\zeta'}{\zeta^2}(1-\delta) = -1$ we have

$$\begin{aligned} \frac{d}{d\delta} \left(\frac{t}{2\pi}\right)^{-\alpha_1 - \delta} Z_{-\delta,\alpha_2,\dots,\alpha_k,-\alpha_1} \bigg|_{\delta=0} \\ &= -\left(\frac{t}{2\pi}\right)^{-\alpha_1} Z_{0,\alpha_2,\dots,\alpha_k,-\alpha_1} - \left(\frac{t}{2\pi}\right)^{-\alpha_1} \zeta(1-\alpha_1) A_{\{0,\alpha_2,\dots,\alpha_k\}}(-\alpha_1) \prod_{j=2}^k \frac{\zeta(1+\alpha_j - \alpha_1)}{\zeta(1+\alpha_j)} \\ &= -\left(\frac{t}{2\pi}\right)^{-\alpha_1} \zeta(1-\alpha_1) A_{\{0,\alpha_2,\dots,\alpha_k\}}(-\alpha_1) \prod_{j=2}^k \frac{\zeta(1+\alpha_j - \alpha_1)}{\zeta(1+\alpha_j)} \end{aligned}$$

where the first term in the second line vanishes (again due to the fact that $\frac{1}{\zeta(1-\delta)}\Big|_{\delta=0} = 0$). Therefore, if we let

$$W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t) = \frac{\zeta'(1+\alpha_j)}{\zeta(1+\alpha_j)} - \left(\frac{t}{2\pi}\right)^{-\alpha_j} \zeta(1-\alpha_j) A_{\{\alpha_1,\dots,\alpha_{j-1},0,\alpha_{j+1},\dots,\alpha_k\}}(-\alpha_j) \prod_{\substack{\ell=1\\\ell\neq j}}^k \frac{\zeta(1+\alpha_\ell-\alpha_j)}{\zeta(1+\alpha_\ell)}$$

then we see that the Ratios Conjecture yields

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2} + i\gamma + \alpha_1\right) \dots \zeta\left(\frac{1}{2} + i\gamma + \alpha_k\right) =$$
$$= \frac{1}{2\pi} \int_1^T \left(A'_{\{\alpha_1,\dots,\alpha_k\}}(0) + \sum_{j=1}^k W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_j\}}(\alpha_j,t)\right) dt + O\left(T^{1/2+\varepsilon}\right), \quad (6.4)$$

which is Conjecture 6.3.

6.2.3 Recovering Conjectures 5.3 and 5.4

We aim to find the coefficient of $\alpha_1^{n_1} \dots \alpha_k^{n_k}$, or rather its leading order in term of the largest power of $L = \log \frac{t}{2\pi}$. We will follow the Ratios methodology, which is to replace each arithetmic piece by its leading order, which in our case means

$$A_{\{\alpha_1,...,\alpha_k\}}(0) \sim 1.$$

We also replace each zeta function by its expansion about its pole, that is,

$$\zeta(1+x) = \frac{1}{x} + O(1).$$

Then we can write

$$\frac{\zeta'}{\zeta}(1+x) = -\frac{1}{x} + O(1).$$

Writing $L = \log \frac{t}{2\pi}$ we have

$$\left(\frac{t}{2\pi}\right)^{-\alpha_1} = 1 - L\alpha_1 + \frac{1}{2!}L^2\alpha_1^2 + \dots$$

Combing all this, we have shown that

$$W_{\{\alpha_1,\dots,\alpha_k\}\setminus\{\alpha_\ell\}}(\alpha_\ell,t)\sim -\frac{1}{\alpha_\ell}+\left(1-L\alpha_1+\frac{1}{2!}L^2\alpha_1^2+\dots\right)\frac{1}{\alpha_\ell}\prod_{\substack{1\leq j\leq k\\j\neq\ell}}\frac{\alpha_j}{\alpha_j-\alpha_\ell}.$$

Returning to (6.4), we see that

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_{1}\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_{k}\right)$$
$$=\frac{1}{2\pi}\int_{1}^{T} \left(A'_{\{\alpha_{1},\dots,\alpha_{k}\}}(0)+\sum_{\ell=1}^{k}W_{\{\alpha_{1},\dots,\alpha_{k}\}\setminus\{\alpha_{\ell}\}}(\alpha_{\ell},t)\right) dt+O\left(T^{1/2+\varepsilon}\right)$$
$$\sim\frac{1}{2\pi}\int_{1}^{T}-\sum_{\ell=1}^{k}\frac{1}{\alpha_{\ell}}+\sum_{\ell=1}^{k}\left(1-L\alpha_{1}+\frac{1}{2!}L^{2}\alpha_{1}^{2}+\dots\right)\frac{1}{\alpha_{\ell}}\prod_{\substack{1\leq j\leq k\\j\neq\ell}}\frac{\alpha_{j}}{\alpha_{j}-\alpha_{\ell}} dt$$

We can now see by Lemma 6.5 that the poles from the sums over ℓ , where we take the 1-term in the bracket containing the logarithms, cancels perfectly. This means that we have to evaluate

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_1\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_k\right)$$
$$\sim \frac{1}{2\pi} \int_1^T \sum_{\ell=1}^k \left(-L\alpha_1+\frac{1}{2!}L^2\alpha_1^2+\dots\right) \frac{1}{\alpha_\ell} \prod_{\substack{1\leq j\leq k\\ j\neq\ell}} \frac{\alpha_j}{\alpha_j-\alpha_\ell} dt.$$

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 185

To find the coefficient of the α_{ℓ}^n term, say, we need to take the $(-1)^{n+1}\alpha_{\ell}^{n+1}/(n+1)!$ term for each α_{ℓ} from the bracket containing the logarithms, as one α_{ℓ} will cancel with the one in the denominator for each ℓ . Summing over all such ℓ , we have

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_1\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_k\right)$$
$$\sim \frac{1}{2\pi} \int_1^T \left(\frac{(-1)^{n+1}}{(n+1)!}L^{n+1}\right) \sum_{\ell=1}^k \alpha_\ell^n \prod_{\substack{j=1\\j\neq\ell}}^k \frac{\alpha_j}{\alpha_j-\alpha_\ell} dt.$$

By Lemma 6.6, for $n \ge k$, this is

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_1\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_k\right)$$
$$\sim \frac{1}{2\pi} \int_1^T \left(\frac{(-1)^{n+1}}{(n+1)!} L^{n+1}\right) (-1)^{k+1} \sum_{\substack{n_1\geq 1,\dots,n_k\geq 1\\n_1+\dots+n_k=n}} \alpha_1^{n_1} \alpha_2^{n_2}\dots\alpha_k^{n_k} dt.$$

Since $n = n_1 + \ldots n_k$, we have

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha_1\right)\dots\zeta\left(\frac{1}{2}+i\gamma+\alpha_k\right)$$
$$\sim \frac{1}{2\pi} \int_1^T \left(\frac{(-1)^{n_1+\dots+n_k+k}}{(n_1+\dots+n_k+1)!} L^{n_1+\dots+n_k+1}\right) \sum_{\substack{n_1\geq 1,\dots,n_k\geq 1\\n_1+\dots+n_k=n}} \alpha_1^{n_1}\alpha_2^{n_2}\dots\alpha_k^{n_k} dt.$$

Finally, Taylor expand the left-hand side around each α_{ℓ} , differentiate with respect of each α_{ℓ} n_{ℓ} -times and set $\alpha_{\ell} = 0$. Multiplying through by n_{ℓ} for each ℓ gives

$$\sum_{0 < \gamma \le T} \zeta^{(n_1)} \left(\frac{1}{2} + i\gamma\right) \dots \zeta^{(n_k)} \left(\frac{1}{2} + i\gamma\right) \\ \sim \frac{1}{2\pi} \int_1^T \left((-1)^{n_1 + \dots + n_k + k} \frac{n_1! \dots n_k!}{(n_1 + \dots + n_k + 1)!} L^{n_1 + \dots + n_k + 1} \right) dt.$$

Integrating gives Conjecture 5.3 and additionally, setting all n_{ℓ} equal to n gives Conjecture 5.4.

6.2.4 Examples

6.2.4.1 Recovering the Shanks' conjecture asymptotic (k = 1)

For the case k = 1 we will recover the full Generalised Shanks' Conjecture in an integral form, including the lower order terms, which is equivalent to the asymptotic found in Chapter 3 and in [186] for all derivatives.

By Conjecture 6.3 we can write

$$\sum_{0<\gamma\leq T}\zeta\left(\frac{1}{2}+i\gamma+\alpha\right) = \frac{1}{2\pi}\int_{1}^{T}\left(A'_{\{\alpha\}}(0)+W(\alpha,t)\right) dt + O\left(T^{1/2+\varepsilon}\right)$$
(6.5)

where

$$W(\alpha, t) = \frac{\zeta'(1+\alpha)}{\zeta(1+\alpha)} - \left(\frac{t}{2\pi}\right)^{-\alpha} \zeta(1-\alpha) A_{\{0\}}(-\alpha)$$

Now by (6.1), we see that

$$A_{\{\alpha\}}(0) = \prod_{p} \frac{(1 - p^{-(1+\alpha)})}{(1 - p^{-(1+\alpha)})} = 1$$

and so $A'_{\{\alpha\}}(0) = 0$. Then (6.5) can be simplified to

$$\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha\right) = \frac{1}{2\pi} \int_{1}^{T} \left(\frac{\zeta'(1+\alpha)}{\zeta(1+\alpha)} - \left(\frac{t}{2\pi}\right)^{-\alpha} \zeta(1-\alpha)\right) dt + O\left(T^{1/2+\varepsilon}\right).$$
(6.6)

The Laurent expansion for $\frac{\zeta'(1+\alpha)}{\zeta(1+\alpha)}$ about $\alpha = 0$ is given by

$$\frac{\zeta'(1+\alpha)}{\zeta(1+\alpha)} = -\frac{1}{\alpha} + \sum_{j=0}^{\infty} A_j \alpha^j,$$

the Laurent expansion for $-\zeta(1-\alpha)$ about $\alpha = 0$ gives

$$-\zeta(1-\alpha) = \frac{1}{\alpha} - \sum_{j=0}^{\infty} \frac{\gamma_j}{j!} \alpha^j,$$

and the Taylor expansion of $(t/2\pi)^{-\alpha}$ about $\alpha = 0$ gives

$$\left(\frac{t}{2\pi}\right)^{-\alpha} = \sum_{j=0}^{\infty} \frac{(-1)^j \alpha^j L^j}{j!},$$

where $L = \log(t/2\pi)$, the γ_j are the Stieltjes coefficients from the expansion of $\zeta(s)$ about s = 1, and A_n is the n^{th} coefficient from the expansion of $\zeta'(s)/\zeta(s)$ about s = 1.

Substitute these series into (6.6) and simplify to give

$$\begin{split} &\sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2} + i\gamma + \alpha\right) \\ &= \frac{1}{2\pi} \int_{1}^{T} \left(-\frac{1}{\alpha} + \sum_{j=0}^{\infty} A_{j} \alpha^{j}\right) + \left(\sum_{j=0}^{\infty} \frac{(-1)^{j} \alpha^{j} L^{j}}{j!}\right) \left(\frac{1}{\alpha} - \sum_{j=0}^{\infty} \frac{\gamma_{j}}{j!} \alpha^{j}\right) \ dt + O\left(T^{1/2+\varepsilon}\right) \\ &= \frac{1}{2\pi} \int_{1}^{T} \sum_{j=0}^{\infty} A_{j} \alpha^{j} + \sum_{j=1}^{\infty} \frac{(-1)^{j} L^{j} \alpha^{j-1}}{j!} + \sum_{j=0}^{\infty} \sum_{m=0}^{j} \frac{(-1)^{m+1} L^{m} \gamma_{j-m} \alpha^{j}}{m! (j-m)!} \ dt + O\left(T^{1/2+\varepsilon}\right). \end{split}$$

186

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 187

For the general n^{th} derivative of $\zeta(s)$, where $n \in \mathbb{N}$, we have after Taylor expanding $\zeta(\rho + \alpha)$ about $\alpha = 0$, differentiating n times, setting $\alpha = 0$, and multiplying through by n!,

$$\sum_{0<\gamma\leq T}\zeta^{(n)}\left(\frac{1}{2}+i\gamma\right) = \frac{n!}{2\pi}\int_{1}^{T}A_{n} + \frac{(-1)^{n+1}L^{n+1}}{(n+1)!} + \sum_{m=0}^{n}\frac{(-1)^{m+1}L^{m}\gamma_{n-m}}{m!(n-m)!}\,dt + O\left(T^{1/2+\varepsilon}\right),$$

which is equivalent to Theorem 3.1 from Chapter 3 and from Hughes and Pearce-Crump [186].

6.2.4.2 The second moment of $\zeta'(1/2 + i\gamma)$ (k = 2)

To calculate the full asymptotic for the second moment of the first derivative of the Riemann zeta function, begin with Conjecture 6.3 with two shifts α, β to write

$$\begin{split} \sum_{0<\gamma\leq T} \zeta\left(\frac{1}{2}+i\gamma+\alpha\right)\zeta\left(\frac{1}{2}+i\gamma+\beta\right) \\ &= \frac{1}{2\pi}\int_{1}^{T} \left(A'_{\{\alpha,\beta\}}(0)+\frac{\zeta'(1+\alpha)}{\zeta(1+\alpha)}-\left(\frac{t}{2\pi}\right)^{-\alpha}\zeta(1-\alpha)A_{\{0,\beta\}}(-\alpha)\frac{\zeta(1+\beta-\alpha)}{\zeta(1+\beta)}\right. \\ &\quad +\frac{\zeta'(1+\beta)}{\zeta(1+\beta)}-\left(\frac{t}{2\pi}\right)^{-\beta}\zeta(1-\beta)A_{\{\alpha,0\}}(-\beta)\frac{\zeta(1+\alpha-\beta)}{\zeta(1+\alpha)}\right) dt \\ &\quad +O\left(T^{1/2+\varepsilon}\right). \end{split}$$

Note that by (6.1) we have that the arithmetic part can be written as

$$A_{\{\alpha,\beta\}}(\delta) = \prod_{p} \frac{1 - p^{-(1+\alpha)} - p^{-(1+\beta)} + p^{-(2+\alpha+\beta+\delta)}}{(1 - p^{-(1+\alpha)})(1 - p^{-(1+\beta)})}$$

By adding 'zero', we have

$$A_{\{\alpha,\beta\}}(\delta) = \prod_{p} \frac{1 - p^{-(1+\alpha)} - p^{-(1+\beta)} + p^{-(2+\alpha+\beta)}}{(1 - p^{-(1+\alpha)})(1 - p^{-(1+\beta)})} + \prod_{p} \frac{p^{-(2+\alpha+\beta+\delta)} - p^{-(2+\alpha+\beta)}}{(1 - p^{-(1+\alpha)})(1 - p^{-(1+\beta)})}$$
$$= 1 + \prod_{p} \frac{p^{-(2+\alpha+\beta+\delta)} - p^{-(2+\alpha+\beta)}}{(1 - p^{-(1+\alpha)})(1 - p^{-(1+\beta)})}$$
(6.7)

which is the in a nicer form to work with.

We now employ a computer package to perform the expansions. We substitute the Laurent expansion for $\zeta(s)$ about s = 1, which is where the various Stieltjes constants have come from.

Then for $L = \log \frac{t}{2\pi}$ and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function,

$$\sum_{0 < \gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^2 = \frac{1}{2\pi} \int_1^T \left(\frac{1}{6} L^3 A^{(0,0,0)} + \frac{1}{2} L^2 \left(2\gamma_0 A^{(0,0,0)} + A^{(0,0,1)} + A^{(0,1,0)}\right) + \frac{1}{2} L \left(-8\gamma_1 A^{(0,0,0)} + 4\gamma_0 A^{(0,0,1)} + A^{(0,0,2)} + 4\gamma_0 A^{(0,1,0)} + 2A^{(0,1,1)} - A^{(0,2,0)}\right) + \frac{1}{6} \left(-12\gamma_0^3 A^{(0,0,0)} - 36\gamma_0\gamma_1 A^{(0,0,0)} + 6\gamma_2 A^{(0,0,0)} - 24\gamma_1 A^{(0,0,1)} + 6\gamma_0 A^{(0,0,2)} + A^{(0,0,3)} + 12\gamma_0 A^{(0,1,1)} + 3A^{(0,1,2)} - 3A^{(0,2,1)} + 6A^{(1,1,1)} - 12\gamma_1 A^{(0,1,0)} - 12\gamma_1 A^{(1,0,0)} + 6\gamma_0^2 A^{(0,1,0)} - 6\gamma_0 A^{(0,2,0)} + A^{(0,3,0)} - 3A^{(2,1,0)} - 6\gamma_0^2 A^{(1,0,0)} + 12\gamma_0 A^{(1,1,0)} - 3A^{(1,2,0)} dt\right) + O\left(T^{1/2+\varepsilon}\right)$$

$$(6.8)$$

where the γ_m are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1 and the $A^{(i,j,k)}$ terms are various products over primes, where we are using the notation

$$A^{(i,j,k)} = \frac{\partial^i}{\partial \alpha^i} \frac{\partial^j}{\partial \beta^j} \frac{\partial^k}{\partial \delta^k} A_{\{\alpha,\beta\}}(\delta) \bigg|_{\alpha=\beta=\delta=0}$$

for complex numbers (shifts) α, β, δ satisfying the conditions of Conjecture 6.3.

We now simplify this result to state a final version of the conjecture for this moment. Clearly, say from (6.7), if we do not differentiate with respect to δ and set $\delta = 0$, then the arithmetic piece equals 1. If we then differentiate with respect to α or β , and then set these equal to 0, these terms disappear. That is, for i, j > 0,

$$A^{(i,j,0)} = 0,$$

while if we don't differentiate at all (that is, i = j = k = 0) and set $\alpha, \beta, \delta = 0$, we have

$$A^{(0,0,0)} = 1.$$

It is also clear that the arithmetic piece is symmetric in the first two derivatives, so

$$A^{(i,j,k)} = A^{(j,i,k)}.$$

This means that whenever we see these terms, we can combine them together which is why the conjecture does not appear symmetric in i, j, k.

After using these rules for $A^{(i,j,k)}$, we form the conjecture for the second moment of the first derivative of the Riemann zeta function.

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 189

Conjecture. Assume the Riemann Hypothesis. For $L = \log \frac{t}{2\pi}$ and $\rho = 1/2 + i\gamma$ a non-trivial zero of the Riemann zeta function,

$$\begin{split} \sum_{0<\gamma\leq T} \zeta' \left(\frac{1}{2} + i\gamma\right)^2 &= \\ \frac{1}{2\pi} \int_1^T \left(\frac{1}{6}L^3 + \frac{1}{2}L^2 \left(2\gamma_0 + A^{(0,0,1)}\right) + \frac{1}{2}L \left(-8\gamma_1 + 4\gamma_0 A^{(0,0,1)} + A^{(0,0,2)} + 2A^{(0,1,1)}\right) \\ &+ \frac{1}{6} \left(-12\gamma_0^3 - 36\gamma_0\gamma_1 + 6\gamma_2 - 24\gamma_1 A^{(0,0,1)} + 6\gamma_0 A^{(0,0,2)} + A^{(0,0,3)} \right. \\ &\left. + 12\gamma_0 A^{(0,1,1)} + 3A^{(0,1,2)} - 3A^{(0,2,1)} + 6A^{(1,1,1)}\right) dt + O\left(T^{1/2+\varepsilon}\right) \end{split}$$

where the γ_m are the coefficients in the Laurent expansion of $\zeta(s)$ about s = 1 and the $A^{(i,j,k)}$ are arithmetic terms that are various products and sums over primes.

We present some numerical evidence. To generate these numerics, we again use a computer package and describe how we have done so. Most of the calculations are standard so we just describe how to calculate the arithmetic pieces and so the full polynomial that appears in the integral version of this moment.

We begin by taking logarithms of the arithmetic pieces $A_{\{\alpha,\beta\}}(\delta)$ and expand it as a series about $\alpha, \beta, \delta = 0$, which we can do as they are all small by assumption. Note that since the highest derivative that we need to take is the third derivative for any of α, β, δ , we don't need to expand beyond the third powers. We then sum this expansion for the first 1000 prime numbers (note that for higher accuracy we could have done this calculation for more primes but at a higher computational cost - this would've generated better numerics and would be of future interest).

In doing these calculations, we have

$$\sum_{0<\gamma\leq T}\zeta'\left(\frac{1}{2}+i\gamma\right)^2 = \frac{1}{2\pi}\int_1^T \left(\frac{1}{6}L^3 - 0.03621L^2 + 2.12487L - 2.52789\right)dt + O\left(T^{1/2+\varepsilon}\right)dt + O\left(T^{1/$$

After integrating, we have

$$\begin{split} \sum_{0<\gamma \le T} \zeta' \left(\frac{1}{2} + i\gamma\right)^2 &= \frac{1}{6} \frac{T}{2\pi} \left(\log \frac{T}{2\pi}\right)^3 - 0.52037 \frac{T}{2\pi} \left(\log \frac{T}{2\pi}\right)^2 \\ &+ 2.95321 \frac{T}{2\pi} \left(\log \frac{T}{2\pi}\right) - 4.65238 \frac{T}{2\pi} + O\left(T^{1/2 + \varepsilon}\right). \end{split}$$

We are now in a place to compare our theoretical conjecture with the true numerics of the $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$, where we ignore the small imaginary part.

We begin by plotting the real part of the cumulative total of the sum $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$ for the first 1,000,000 zeros, given in Figure 6.1. Numerically we can show that the imaginary part is small and so don't plot this in the following figures (which agrees with our conjecture only containing real terms in the asymptotic).



Figure 6.1: The real part of the true value of $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$, for T up to the height of the 1,000,000th zero.

In Figure 6.2 we have subtracted the main term, which shows a decrease in size of an order of magnitude, and also shows a sign change suggesting that we have the correct leading order behaviour.



Figure 6.2: Difference in the real part of the actual value of $\sum_{0 < \gamma \leq T} \zeta' (1/2 + i\gamma)^2$ and the leading asymptotic result of the equation, for T up to the height of the 1,000,000th zero.

Finally, in Figure 6.3 we plot the sum over zeros minus all the remaining terms in the asymptotic expansion, leaving the error term. The arithmetic pieces are calculated for the first 1000 primes. Clearly there is excellent agreement, with a very small error compared with the original cumulative sum.

6.2. Lower Order Terms for Moments of the Derivatives of the Riemann Zeta Function 191



Figure 6.3: Difference in the real part of the actual value of $\sum_{0 < \gamma \le T} \zeta' (1/2 + i\gamma)^2$ and the whole asymptotic result of the equation, for T up to the height of the 1,000,000th zero, showing the real error at each point.

Remark. As we have already remarked in this chapter, the error term here could plausibly be of size $O(T^{1-\delta})$ for $0 < \delta < 1/2$, rather than of size $O(T^{1/2+\varepsilon})$. While we do not rule out the possibility of the first option, as discussed earlier, we have settled on this second option, in keeping with the original statements of the recipe/Ratios Conjecture.

— 7 —

Moments of the Riemann zeta function at its local extrema

In this chapter we assume the Riemann Hypothesis, unless otherwise stated.

Recall that under the Riemann Hypothesis, Conrey and Ghosh [66] showed that

$$\sum_{0 < \gamma \le T} \max_{\gamma \le t \le \gamma^+} \left| \zeta \left(\frac{1}{2} + it \right) \right|^2 \sim \frac{e^2 - 5}{4\pi} T(\log T)^2 \tag{7.1}$$

as $T \to \infty$ where for each non-trivial zero $\rho = 1/2 + i\gamma$, let γ^+ denote the successive ordinate of a non-trivial zero of $\zeta(s)$.

In [183], Hughes, Lugmayer and Pearce-Crump calculated the full asymptotic expansion of (7.1), showing that under the Riemann Hypothesis, for K a positive integer, that

$$\sum_{0 < \gamma \le T} \max_{\gamma \le t \le \gamma^+} \left| \zeta \left(\frac{1}{2} + it \right) \right|^2 = \frac{e^2 - 5}{2} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^2 \\ + \left(5 - e^2 - 10\gamma_0 + 2e^2\gamma_0 \right) \frac{T}{2\pi} \log \frac{T}{2\pi} + \frac{T}{2\pi} \sum_{k=0}^K \frac{c_k}{(\log T/2\pi)^k} \\ + O_K \left(\frac{T}{(\log T)^{K+1}} \right)$$
(7.2)

where γ_0 is Euler's constant and the other constants c_k are effectively computable constants.

Note that we can rewrite (7.1) and (7.2) in terms of the Hardy Z-function. Specifically, under the Riemann Hypothesis, we can rewrite the result of Conrey and Ghosh [66] as

$$\sum_{0<\lambda\leq T} Z(\lambda)^2 \sim \frac{e^2 - 5}{4\pi} T(\log T)^2$$
(7.3)

as $T \to \infty$, where we recall that the points λ that are being summed over are the zeros of the derivative of the Hardy Z-function, described in Section 1.3.8.1, up to a height T.

In this section we combine some of the results presented in this thesis with these asymptotics, specifically equations (7.1) and (2.4), or rather the full asymptotic expansions

given in (7.2) and Theorem 3.1. That is, under the Riemann Hypothesis, we calculate full asymptotics for the moments

$$\sum_{0<\lambda\leq T}\zeta^{(n)}\left(\frac{1}{2}+i\lambda\right)$$

where the sum is over the maxima of $|\zeta(s)|$ up to a height T, or equivalently, over those points λ such that $Z'(\lambda) = 0$, where Z(t) is the Hardy Z-function, up to a height T. This sum is considered for all non-negative integers n (specifically including the no derivative case which is non-trivial, compared with our previous sums over non-trivial zeros).

In addition, we also consider sums of the factor $\chi(s)$ from the non-symmetric functional equation for $\zeta(s)$ given in (1.1). These sums are considered both over the maxima of $|\zeta(s)|$ and over the non-trivial zeros of the Riemann zeta function.

We will also compare the results in this chapter with those from Chapter 3, specifically Theorem 3.1. In these comparisons we don't give many reasons beyond the obvious (that we are summing over different points/summing different functions), but the differences are worth pointing out explicitly in the remarks that follow the results. The actual, full reasons for these differences are of further interest and worth exploring.

The work presented in Section 7.1 has already appeared in print in [263]. Notation and cross-references have been updated for this thesis.

7.1 MOMENTS OF $\zeta(s)$, ITS DERIVATIVES, AND $\chi(s)$ AT THE LOCAL EX-TREMA OF THE RIEMANN ZETA FUNCTION

We now state the results that are to be proved in this chapter. There are four main results that we prove (with one simple corollary).

The first two results (along with the corollary) involve moments of the derivatives of the Riemann zeta function, evaluated at the local extrema of the Riemann zeta function (recall, equivalently, at the zeros of the derivative of the Hardy Z-function). We split the cases of derivatives and no derivatives into two separate results, both for clarity of the results and as the proofs are slightly different. The no derivative case is non-trivial, compared with typical moments where the sums are over non-trivial zeros of the Riemann zeta function.

For the second pair of results, we evaluate the sum of the factor $\chi(s)$ from the functional equation (1.1) for $\zeta(s)$ over the local extrema of the Riemann zeta function and over the non-trivial zeros of the Riemann zeta function.

Theorem 7.1. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K, n \ge 1$ positive integers, we have

$$\sum_{0<\lambda\leq T} \zeta^{(n)}\left(\frac{1}{2}+i\lambda\right) = a_{n+1}\frac{T}{2\pi}L^{n+1} + \frac{T}{2\pi}\sum_{\ell=0}^{n}a_{n-\ell}L^{n-\ell} + \frac{T}{2\pi}\sum_{m=1}^{K}\frac{b_m}{L^m} + O\left(\frac{T}{L^{K+1}}\right),$$

as $T \to \infty$ where:

1. The leading order coefficient is given by

$$a_{n+1} = (-1)^n \left(\frac{e^2 - 2}{n+1} + (-1)^{n+1} \frac{n!}{2^{n+1}} \left(1 - e^2 \sum_{k=0}^{n+1} \frac{(-2)^k}{k!} \right) \right).$$

2. The subleading, non-negative logarithm power coefficients are given by

$$\begin{split} a_{n-\ell} &= (-1)^n \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \left[\frac{c_{\ell+1}^{k,j}}{(k+j-\ell)!} + (k-n+j) \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \right. \\ &+ (k-n+j) \sum_{m=0}^{\ell-1} (-1)^{\ell-m} (n-m) \dots (n-\ell+1) \frac{c_m^{k,j}}{(k+j+1-m)!} \right] \\ &+ (-1)^{n+1} \sum_{\ell=0}^n \binom{n}{\ell} (-1)^{\ell} \ell! \left(-1 + \sum_{m=0}^{\ell} \frac{1}{m!} \gamma_m \right), \end{split}$$

where if $\ell = 0$ the summation over *m* in the square brackets is empty, and where if $\ell = n$, the coefficient of L^0 , that is to say a_0 , has an extra contribution of $n!A_n$, where the A_n are the coefficients in the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1.

3. The subleading, negative logarithm power coefficients are given by

$$b_m = (-1)^n \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \beta_m^{k,j},$$

where for $1 \le m \le k + j - n$, we have

$$\beta_m^{k,j} = \frac{c_{m+n+1}^{k,j}}{(k+j-n-m)!} + \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{m-1} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j}$$

and where for $m \leq k + j - n + 1$, we have

$$\beta_m^{k,j} = \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{k+j-n} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j}$$

In these coefficients a_m, b_m , the $c_{\ell}^{k,j}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \zeta^{(j)}(s) \frac{1}{s} = \sum_{\ell=0}^{\infty} c_{\ell}^{k,j} (s-1)^{-k-j-2+\ell}.$$

An immediate corollary of Theorem 7.1 follows by setting n = 1. We state this result explicitly to compare it with (2.1) and Shanks' Conjecture.

Corollary 7.2. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K \ge 1$ a positive integer, we have

$$\begin{split} \sum_{0<\lambda\leq T} \zeta'\left(\frac{1}{2}+i\lambda\right) &= -\frac{(e^2-3)}{4}\frac{T}{2\pi}L^2 + \frac{e^2-3+2\gamma_0}{2}\frac{T}{2\pi}L + \frac{3-e^2(1+2\gamma_0+2\gamma_1)}{2}\frac{T}{2\pi} \\ &+ \frac{T}{2\pi}\sum_{m=1}^{K}\frac{c_m}{L^k} + O_K\left(\frac{T}{L^{K+1}}\right) \end{split}$$

as $T \to \infty$, where the other constants c_m are special cases of the coefficients b_m from Theorem 7.1 with n = 1.

We may also prove the analogous result of Conrey and Ghosh in (7.1) from [66], with the modulus squared removed.

Theorem 7.3. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K \ge 1$ a positive integer, we have

$$\sum_{0<\lambda\leq T} \zeta\left(\frac{1}{2}+i\lambda\right) = \frac{e^2-3}{2}\frac{T}{2\pi}L + \frac{3-e^2-4\gamma_0}{2}\frac{T}{2\pi} + \frac{T}{2\pi}\sum_{m=1}^K \frac{d_m}{L^m} + O_K\left(\frac{T}{L^{K+1}}\right)$$

as $T \to \infty$, where

$$d_m = \sum_{k=m}^{\infty} 2^k \frac{c_{k,m+1}}{(k-m)!} + (m-1)! \sum_{k=1}^{\infty} \frac{2^k k}{(k-1)!} \sum_{\ell=0}^{\min\{m-1,k\}} \binom{k}{\ell} c_{k,\ell+1}$$

and where the $c_{k,\ell}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)'\left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1}\zeta(s)\frac{1}{s} = \sum_{\ell=0}^{\infty} c_{k,\ell}(s-1)^{-k-2+\ell}.$$

We evaluate the sum of the factor $\chi(s)$ from the functional equation (1.1) for $\zeta(s)$ over the local extrema of $\zeta(s)$.

Theorem 7.4. Assume the Riemann Hypothesis. Write $L = \log T/2\pi$. For λ defined by $Z'(\lambda) = 0$ and $K \ge 1$ a positive integer, we have

$$\sum_{0<\lambda\leq T} \chi\left(\frac{1}{2}+i\lambda\right) = (e^2-2)\frac{T}{2\pi} - 4e^2\gamma_0\frac{T}{2\pi}\frac{1}{L} + \frac{T}{2\pi}\sum_{m=2}^K \frac{e_m}{L^m} + O_K\left(\frac{T}{L^{K+1}}\right)$$

as $T \to \infty$, where

$$e_m = \sum_{k=m}^{\infty} 2^k \frac{c_{k,m}}{(k-m)!} + (m-1)! \sum_{k=1}^{\infty} \frac{2^k}{(k-1)!} \sum_{\ell=0}^{\min\{m-1,k\}} \binom{k}{\ell} c_{k,\ell},$$

and where the $c_{k,\ell}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \frac{1}{s} = \sum_{\ell=0}^{\infty} c_{k,\ell}(s-1)^{-k-1+\ell}.$$

Remark. The infinite descending chain of powers of $\log T$ in the previous results where each function is summed over the relative extrema of zeta is due to a pole in the function that we consider near 1, given by

$$\beta_Z = 1 + 2/L + O\left(L^{-2}\right),$$

where $L = \log T/2\pi$.

We expect that we would be able to give these results with a power-saving error term of $O\left(T^{1/2+\varepsilon}\right)$ for all $\varepsilon > 0$ if we were able to find a closed form for the asymptotic as a function of β_Z .

As in [183], we have set aside the search for such a form for the time being.

With only minor adjustments to our proof of Theorem 7.4 (which we will specify in Section 7.1.7), we are also able to prove the following result. It is a special case of a result from [208], but here we give it with a power-saving error term.

Theorem 7.5. Assume the Riemann Hypothesis. For $\rho = 1/2 + i\gamma$ a non-trivial zero of $\zeta(s)$, we have

$$\sum_{0 < \gamma \le T} \chi\left(\frac{1}{2} + i\gamma\right) = -\frac{T}{2\pi} + O\left(T^{1/2 + \varepsilon}\right)$$

as $T \to \infty$, where $\chi(s)$ is the factor from the functional equation for $\zeta(s)$ given in (1.1).

7.1.1 Remarks on the previous results

1. Theorem 7.1

We compare the result from this theorem with that of Kaptan, Karabulut and Yıldırım [207] and Hughes and Pearce-Crump [186] on the Generalised Shank's conjecture. In that result, the average of $\zeta^{(n)}(s)$, summed over the non-trivial zeros of $\zeta(s)$, is clearly real and alternates between positive and negative as *n* increases. In the case of Theorem 7.1, the behaviour is reversed; the average of $\zeta^{(n)}(s)$, summed over the maxima of $|\zeta(s)|$, is real and alternates between negative and positive as *n* increases.

It isn't immediately obvious from the leading order coefficient that this is the case, that is, that

$$(-1)^n \left(\frac{e^2 - 2}{n+1} + (-1)^{n+1} \frac{n!}{2^{n+1}} \left(1 - e^2 \sum_{k=0}^{n+1} \frac{(-2)^k}{k!} \right) \right)$$

alternates. Begin by noting by Taylor's theorem with remainder that

$$e^{-2} = \sum_{k=0}^{n+1} \frac{(-2)^k}{k!} + O\left(\frac{1}{(n+2)!}\right)$$

so the coefficient of the leading order can be rewritten as

$$(-1)^n \left(\frac{e^2 - 2}{n+1} + (-1)^{n+1} \frac{n!}{2^{n+1}} \left(1 - e^2 \left\{ e^{-2} + O\left(\frac{1}{(n+2)!}\right) \right\} \right) \right)$$

and simplifying gives

$$(-1)^n \left(\frac{e^2 - 2}{n+1} + O\left(\frac{1}{n^2}\right)\right).$$

Since $e^2 - 2 > 0$, the alternating behaviour is then clear, but in reverse to the Generalised Shanks' Conjecture as n increases.

2. Corollary 7.2

As noted in the general n^{th} derivative case, while Shanks' conjecture is that $\zeta'(\rho)$ is positive and real on average, the same question with regards to $\zeta'(\lambda)$ is negative and real on average.

We plot the first few terms of the asymptotic against the real part of the true value of the sum in Figure 7.1. If we were to plot the imaginary part of the true value, we would find that it remains small (at most $\approx \pm 10$ in absolute value).



Figure 7.1: The real part of the sum $\sum_{0 < \lambda \leq T} \zeta'(1/2 + i\lambda)$ with true value (blue), the true - leading (red), true - leading - subleading (green) and true - leading - subleading - subleading (green) and true - leading - subleading - subleading (purple) terms subtracted from the true value, for T up to the height of the 10,000th zero λ of Z'(t).

3. Theorem 7.3

The result of Conrey and Ghosh in (7.1) has coefficient $(e^2 - 5)/2$ to leading order,

197

with an extra power of a logarithm, compared with the $(e^2 - 3)/2$ here. The power of the logarithm can be explained by noting that we are taking a first moment in Theorem 7.3.

We plot the first few terms of the asymptotic against the real part of the true value of the sum in Figure 7.2. If we were to plot the imaginary part of the true value, we would find that it remains small (at most $\approx \pm 20$ in absolute value).



Figure 7.2: The real part of the sum $\sum_{0 < \lambda \leq T} \zeta(1/2 + i\lambda)$ with true value (blue), the true leading (red) and true - leading - subleading (green) terms subtracted from the true value, for T up to the height of the 1,000,000th zero λ of Z'(t).

4. Theorem 7.4

We plot the first few terms of the asymptotic against the real part of the true value of the sum in Figure 7.3. If we were to plot the imaginary part of the true value, we would find that it remains small (at most $\approx \pm 10$ in absolute value).

5. Theorem 7.5

We assume the Riemann Hypothesis in proving this result but can make it unconditional by taking better care when calculating the error term.

The behaviour is somewhat strange, however. Recall that the Gram points occur when $\theta(g_n) = n\pi$, and by the definition of the Riemann–Siegel function given in (1.28), we have

$$\chi\left(\frac{1}{2} + ig_n\right) = e^{-2i\theta(g_n)} = e^{-2in\pi} = 1$$

 \mathbf{SO}

$$\sum_{0 < \gamma \le T} \chi\left(\frac{1}{2} + ig_n\right) = \frac{T}{2\pi} \log T + O(T).$$

Given that a zero is expected to lie roughly in the middle of a Gram interval, it isn't overly surprising that the sum in Theorem 7.5 is negative, but this doesn't account



Figure 7.3: The real part of the sum $\sum_{0 < \lambda \leq T} \chi(1/2 + i\lambda)$ with true value (blue), the true leading (red) and true - leading - subleading (green) terms subtracted from the true value, for T up to the height of the 1,000,000th zero λ of Z'(t).

for the loss of the logarithm. Likewise, it then isn't overly surprising that the sum in Theorem 7.3 over λ is positive as we should expect these points to lie closer to the Gram points, but again doesn't account for the loss of the logarithm.

We plot the true value of the sum minus the main term of the asymptotic in Figure 7.4. As we can see, all that remains is clearly an error term. We have plotted both the real and imaginary parts as both are small and can be easily seen against each other, unlike in the previous figures where the imaginary parts are too small to be seen.



Figure 7.4: The real (blue) and imaginary (orange) part of $\sum_{0 < \gamma \leq T} \chi(1/2 + i\gamma) + \frac{T}{2\pi}$ for T up to the height of the 1,000,000th zero $1/2 + i\gamma$ of $\zeta(s)$.

7.1.2 Overview of the proofs

In Section 7.1.3, we consider properties of the auxiliary function

$$Z_1(s) = \zeta'(s) - \frac{1}{2}\frac{\chi'}{\chi}(s)\zeta(s)$$

used in the proof, where $\chi(s)$ is the factor from the non-symmetrical form of the functional equation of the Riemann zeta function, given in (1.1). This function is zero at the extrema of the Riemann zeta function. We also list other useful lemmas in this section, including functional equations, stationary phase methods, and a moments result from Chapter 3 (and from [186]) on the average of derivatives of $\zeta(s)$, evaluated over the non-trivial zeros of $\zeta(s)$.

In Section 7.1.4, we prove Theorem 7.1. This involves using Cauchy's Theorem to write the sum over the extrema as an integral

$$\sum_{0<\lambda\leq T} \zeta^{(n)}\left(\frac{1}{2}+i\lambda\right) = \frac{1}{2\pi i} \int_{\mathcal{C}} \frac{Z_1'}{Z_1}(s) \zeta^{(n)}(s) \ ds$$

(up to a small error), where C is a contour we specify later that encloses the critical line up to a height T.

We begin by showing that only the left-hand side of the contour contributes to the asymptotic, with the rest going into a power-saving error term. The bulk of the argument then comes down to evaluating

$$\frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{Z_1'}{Z_1}(s) \zeta^{(n)}(1-s) \ ds$$

for some specific c > 1 defined in the proof.

We are able to write the logarithmic derivative of $Z_1(s)$ as a series of the form

$$\frac{Z_1'}{Z_1}(s) = \sum_{m=1}^{\infty} \frac{a(m,s)}{m^s}$$

which converges for some c > 1 with coefficients a(n, s) given in Lemma 7.7. Using this and the Dirichlet series for the derivatives of $\zeta(s)$ allows us to apply the method of stationary phase to change this integral into a sum, given by

$$(-1)^n \left(\sum_{m_1 m_2 \le T/2\pi} \Lambda(m_1) (\log m_1)^n - \sum_{k=1}^\infty 2^k \sum_{j=0}^n \binom{n}{j} \sum_{m_1 m_2 \le T/2\pi} \frac{(-1)^j (\log m_1)^j a_k(m_2)}{(\log m_1 m_2)^{k-n+j}} \right)$$

up to a power-saving error term, where the coefficients $a_k(m)$ come from a(m, s).

An application of Perron's formula turns the first sum back into an integral, which we can evaluate the residue of at the pole. Another application of Perron's formula turns

the numerator of the inner sum of the second term into an integral that we evaluate the residue of at the pole, and complete the argument with partial summation and summing over j = 0, ..., n and $k \ge 1$.

We then outline the proofs of Theorem 7.3 in Section 7.1.5, of Theorem 7.4 in Section 7.1.6, and of Theorem 7.5 in 7.1.7, leaving the full details of the proofs to the interested reader as they are similar to that of Theorem 7.1.

7.1.3 Preliminary Lemmas

Define the function $Z_1(s)$ by

$$Z_1(s) = \zeta'(s) - \frac{1}{2} \frac{\chi'}{\chi}(s)\zeta(s).$$
(7.4)

201

Note that by taking the logarithmic derivative of Hardy's Z function, we see that $Z_1(s)$ is zero exactly when Z'(t) = 0, that is, at an extrema of $|\zeta(s)|$ on the critical line.

The function $Z_1(s)$ satisfies various properties which were proved by Conrey and Ghosh [66] in their lemma. Note that only the properties we use in our proof are listed below. For a proof of this lemma, we refer the reader to their paper.

Lemma 7.6. We have the following properties for $Z_1(s)$:

- 1. $|Z_1(1/2 + it)| = |Z'(t)|.$
- 2. $Z_1(s)$ satisfies the functional equation

$$Z_1(s) = -\chi(s)Z_1(1-s)$$

for all s, where $\chi(s)$ is the term in the functional equation for the Riemann zeta function, $\zeta(s) = \chi(s)\zeta(1-s)$.

- 3. The number of zeros of $Z_1(t)$ up to a height T which lie off the critical line is bounded by $\log T$.
- 4. If $Z_1(\beta_1 + i\gamma_1) = 0$ then

$$\left|\beta_1 - \frac{1}{2}\right| \le \frac{1}{9}$$

for γ_1 sufficiently large.

The logarithmic derivative of $Z_1(s)$ is given by

$$\frac{Z_1'}{Z_1}(s) = \frac{\chi'}{\chi}(s) - \frac{Z_1'}{Z_1}(1-s)$$
(7.5)

which follows directly from part (2) in Lemma 7.6.

For $s = \sigma + it$, where $|\sigma| \leq 2$, we have the following estimate for the logarithmic derivative for $\chi(s)$ given by

$$\frac{\chi'}{\chi}(s) = -\log\frac{t}{2\pi} + O\left(\frac{1}{t}\right),\tag{7.6}$$

(say for t > 1) which can be found in [66], for example. This enables us to rewrite the functional equation above as

$$\frac{Z_1'}{Z_1}(s) = -\log\frac{t}{2\pi} - \frac{Z_1'}{Z_1}(1-s) + O\left(\frac{1}{t}\right).$$
(7.7)

Lemma 7.7. For $\Re(s) > 1$ and $t \ge 100$, we have

$$\frac{Z_1'}{Z_1}(s) = \sum_{m=1}^{\infty} \frac{a(m,s)}{m^s} + O\left(\frac{1}{t\log t}\right)$$

where

$$a(m,s) = -\Lambda(m) + \sum_{k=1}^{\infty} \frac{1}{f(s)^k} a_k(m)$$
(7.8)

with

$$f(s) = -\frac{1}{2}\frac{\chi'}{\chi}(s) \tag{7.9}$$

and $a_k(m)$ is of the form

$$a_k(m) = \left(\left(\Lambda \log \right) * \Lambda_{k-1} \right)(m), \tag{7.10}$$

where we use the notation Λ_k to denote

$$\underbrace{\underbrace{\Lambda*\Lambda*\Lambda*\dots*\Lambda}_{k-1\ convolutions}}$$

with the convention that $\Lambda_0(m)$ takes the value 1 if m = 1 and 0 otherwise, and where $\Lambda_1(m) = \Lambda(m)$ is the usual von Mangoldt function.

We give a variation of the proof given in [183].

Proof. Recall that we have defined the function $Z_1(s)$ in (7.4) by

$$Z_1(s) = \zeta'(s) + f(s)\zeta(s)$$

where f(s) is given in (7.9). Differentiating this gives

$$Z'_{1}(s) = \zeta''(s) + f(s)\zeta'(s) + f'(s)\zeta(s).$$

Then

$$\frac{Z_1'}{Z_1}(s) = \frac{\zeta''(s) + f(s)\zeta'(s)}{\zeta'(s) + f(s)\zeta(s)} + f'(s)\frac{1}{\frac{\zeta'}{\zeta}(s) + f(s)}.$$
(7.11)

For $\Re(s) > 1$ with $t \ge 1$, the last term can be bounded by $O(1/t \log t)$ by noting that f'(s) = O(1/t) (which follows from (7.6)), $f(s) = O(\log t)$ and $\zeta'(s)/\zeta(s) = O(\log \log t)$ (due to Littlewood [228], under the Riemann Hypothesis) in the region under consideration. Therefore we can write

$$\frac{Z_1'}{Z_1}(s) = \frac{\frac{\zeta''}{\zeta}(s) + f(s)\frac{\zeta'}{\zeta}(s)}{\frac{\zeta'}{\zeta}(s) + f(s)} + O\left(\frac{1}{t\log t}\right).$$

For s with sufficiently large real part, we can expand the denominator as a geometric series. This gives

$$\frac{1}{\frac{\zeta'_{\zeta}(s) + f(s)}{\zeta}} = \frac{1}{f(s)} \left(1 + \frac{1}{f(s)} \frac{\zeta'}{\zeta}(s)\right)^{-1}$$
$$= \frac{1}{f(s)} \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \left(-\frac{\zeta'}{\zeta}(s)\right)^k.$$

This expansion will be valid so long as $\Re(s)$ is large enough so that $\left|\frac{1}{f(s)}\frac{\zeta'}{\zeta}(s)\right| \leq 1$. For example, we can take $\Re(s) > 1$ and $t \geq 100$.

Then we have

$$\frac{Z_1'}{Z_1}(s) = \frac{1}{f(s)} \left(\frac{\zeta''}{\zeta}(s) + f(s)\frac{\zeta'}{\zeta}(s)\right) \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \left(-\frac{\zeta'}{\zeta}(s)\right)^k + O\left(\frac{1}{t\log t}\right)$$
(7.12)
$$= \sum_{n=1}^{\infty} \frac{a(n,s)}{n^s} + O\left(\frac{1}{t\log t}\right)$$

where we calculate the coefficients a(n, s) below.

By repeated use of the quotient rule, we may write

$$\frac{d}{ds}\left(\frac{\zeta^{(k)}}{\zeta}(s)\right) = \frac{\zeta^{(k+1)}}{\zeta}(s) - \frac{\zeta'}{\zeta}(s)\frac{\zeta^{(k)}}{\zeta}(s).$$

Rearranging this we have

$$\frac{\zeta^{(k+1)}}{\zeta}(s) = \frac{d}{ds}\left(\frac{\zeta^{(k)}}{\zeta}(s)\right) + \frac{\zeta'}{\zeta}(s)\frac{\zeta^{(k)}}{\zeta}(s).$$

The Dirichlet coefficients of these series then give us the recurrence relation for any non-negative integer k,

$$\Lambda^{(k+1)}(n) = \Lambda^{(k)}(n)\log(n) + (\Lambda^{(k)}(n) * \Lambda)(n),$$
(7.13)

for each $n \in \mathbb{N}$, where $\Lambda^{(1)}(n) = \Lambda(n)$ is the usual von Mangoldt function, that is, the coefficient from the Dirichlet series for $-\zeta'(s)/\zeta(s)$. (Note that this notation is not standard - see the remark at the end of this proof.)

Consider now the coefficient from each of the Dirichlet series in (7.12). For the terms on the right-hand side of that equation (ignoring the error term for now), we have

$$a(n,s) = \left(\frac{1}{f(s)}\Lambda^{(2)}(n) - \Lambda(n)\right)\sum_{k=0}^{\infty} \frac{1}{f(s)^k}\Lambda_k(n).$$

Now use the recurrence relation (7.13) with k = 1 to write

$$\Lambda^{(2)}(n) = (\Lambda \log + \Lambda_2)(n).$$

Substitue this into the previous expression to obtain for each $n \in \mathbb{N}$,

$$\begin{aligned} a(n,s) &= \left(\frac{1}{f(s)}(\Lambda \log + \Lambda_2)(n) - \Lambda(n)\right) \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \Lambda_k(n) \\ &= \sum_{k=0}^{\infty} \frac{1}{f(s)^{k+1}} ((\Lambda \log) * \Lambda_k)(n) + \sum_{k=0}^{\infty} \frac{1}{f(s)^{k+1}} (\Lambda_2 * \Lambda_k)(n) - \sum_{k=0}^{\infty} \frac{1}{f(s)^k} (\Lambda * \Lambda_k)(n) \\ &= \sum_{k=1}^{\infty} \frac{1}{f(s)^{k+1}} ((\Lambda \log) * \Lambda_k)(n) + \sum_{k=1}^{\infty} \frac{1}{f(s)^k} \Lambda_{k+1}(n) - \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \Lambda_{k+1}(n) \\ &= \sum_{k=1}^{\infty} \frac{1}{f(s)^{k+1}} ((\Lambda \log) * \Lambda_k)(n) + \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \Lambda_{k+1}(n) - \Lambda(n) - \sum_{k=0}^{\infty} \frac{1}{f(s)^k} \Lambda_{k+1}(n) \\ &= \sum_{k=0}^{\infty} \frac{1}{f(s)^k} ((\Lambda \log) * \Lambda_{k-1})(n) - \Lambda(n), \end{aligned}$$

completing the proof of the Lemma.

Remark. Note that while we can only use the expansion in this Lemma for $t \ge 100$, in the proofs of our results in later sections we will ignore this slight technicality as using the result for $t \ge 1$ in our applications will only introduce an error of O(1), which is well within our other error terms.

Remark. As our notation Λ_k is used for (k-1)-convolutions, we use non-standard notation

$$\frac{\zeta^{(k)}}{\zeta}(s) = \sum_{n=1}^{\infty} \frac{\Lambda^{(k)}(n)}{n^s}.$$

Typically in the literature, $\Lambda_k(n)$ is used here but we have already used it in a different meaning in this section, and want to stay consistent with the notation in [183]. Since this proof is the only place in this section that we use this convention, we don't worry about any confusion or conflicts and remind the reader that we are using Λ_k to denote (k-1)-convolutions.

We will be applying the method of stationary phase at various points throughout our argument. The following version of this result can be found in [208], and is based on a similar result due to Gonek [143] where he took k in the following lemma to be a non-negative integer, which we have previously stated in this thesis as Lemma 3.10.

Lemma 7.8. Let $\{b_m\}_{m=1}^{\infty}$ be a sequence of complex numbers such that for any $\varepsilon > 0$, $b_m \ll m^{\varepsilon}$. Let c > 1 and suppose $|k| = o(\log T)$ as $T \to \infty$. Then for T sufficiently large,

$$\frac{1}{2\pi} \int_0^T \chi(1-c-it) \left(\log \frac{t}{2\pi}\right)^k \left(\sum_{m=1}^\infty b_m m^{-c-it}\right) dt = \sum_{m \le \frac{T}{2\pi}} b_m (\log m)^k + O\left(T^{c-1/2} (\log T)^k\right).$$

The last result we need is from the proof of the main result from Chapter 3, and from Hughes and Pearce-Crump [186]. This equation is stated in this thesis as (3.10), and involves calculating the sum

$$\sum_{m_1m_2 \le T/2\pi} \Lambda(m_1) (\log m_1)^n.$$

This result is unconditional in that section, but as we are assuming the Riemann Hypothesis here, we will also assume it here (note that this only affects the error term coming from this sum).

7.1.4 Proof Theorem 7.1

7.1.4.1 Initial Manipulations

We note that the beginning of the proofs of Theorems 7.1, 7.3, and 7.4 start in a similar way to that of Conrey and Ghosh [66], and of Hughes, Lugmayer and Pearce-Crump [183]. We repeat it here for completeness sake but skip over the details in the later cases.

Under the Riemann Hypothesis, we know that the zeros λ of Z'(t) interlace those of Z(t), and so the set of points where $|\zeta(1/2 + it)|$ achieves its local maximum are exactly the points where Z'(t) = 0.

Let $\rho_1 = \beta_1 + i\gamma_1$ be the zeros of $Z_1(s)$. By part (3) of Lemma 7.6, $Z_1(s)$ has $O(\log T)$ zeros off the critical line (that is, $\beta_1 \neq 1/2$). At the zeros off the critical line, we may use part (4) of Lemma 7.6 together with the Lindelöf bounds on $\zeta(s)$ and its derivatives to obtain

$$\sum_{\substack{0 < \gamma_1 \le T\\ \beta_1 \neq 1/2}} \zeta^{(n)}(\rho_1) \ll T^{1/9 + \varepsilon} \log T \ll T^{1/9 + \varepsilon}$$

for any integer $n \ge 0$.

Therefore, we have

$$\sum_{0<\lambda\leq T}\zeta^{(n)}\left(\frac{1}{2}+i\lambda\right) = \sum_{\substack{0<\gamma_1\leq T\\\beta_1=1/2}}\zeta^{(n)}\left(\frac{1}{2}+i\gamma_1\right).$$

We can write this as an integral using Cauchy's theorem,

$$\frac{1}{2\pi i} \int_{\mathcal{C}} \frac{Z'_1}{Z_1}(s) \zeta^{(n)}(s) \, ds = \sum_{\substack{0 < \gamma_1 \le T \\ \beta_1 = 1/2}} \zeta^{(n)}(\rho_1) \\ = \sum_{\substack{0 < \gamma_1 \le T \\ \beta_1 = 1/2}} \zeta^{(n)}\left(\frac{1}{2} + i\gamma_1\right) + O\left(T^{1/9 + \varepsilon}\right)$$
(7.14)

where C is a positively oriented contour with vertices c+i, c+iT, 1-c+iT, and 1-c+i, where $c = 1 + 1/\log T$. We may assume, without loss of generality, that the distance from the contour to any zero ρ_1 of $Z_1(s)$ is uniformly $\gg 1/\log T$.

We split the integral as

$$\begin{aligned} &\frac{1}{2\pi i} \left(\int_{c+i}^{c+iT} + \int_{c+iT}^{1-c+iT} + \int_{1-c+iT}^{1-c+i} + \int_{1-c+i}^{c+i} \right) \frac{Z_1'}{Z_1}(s) \zeta^{(n)}(s) \ ds \\ &= S^R + S^T + S^L + S^B, \end{aligned}$$

say. We will first bound S^B, S^T, S^R trivially within error term, and show that the leading contributions to the asymptotic will then come from S^L . Consider each of these integrals in turn.

First note that the integral S^B , which is the integral along the bottom of the contour, is O(1).

For the integral S^T , which is the integral along the top of the contour, we may use the bound proved in [183] that states if $|T - \rho_1| \gg 1/\log T$ for all zeros ρ_1 of Z_1 (which we are in this proof), we have

$$\frac{Z_1'}{Z_1}(\sigma + iT) \ll (\log T)^2$$

uniformly for $-1 < \sigma \leq 2$. Combining the Lindelöf bounds on $\zeta(s)$ and its derivatives with this shows that the integral along the top of the contour is $O(T^{1/2+\varepsilon})$.

Next note that for the integral S^R , which is the integral along the right-hand vertical side of the contour, we have c > 1 which is past the abscissa of convergence for $\zeta^{(n)}(s)$ and $Z_1(s)$ as given in (7.4). Then

$$S^{R} = \frac{1}{2\pi} \int_{0}^{T} \frac{Z'_{1}}{Z_{1}} (c+it) \zeta^{(n)}(c+it) \ dt \ll T^{\varepsilon}.$$

Note that S^B, S^T, S^R are all within an error term of $O\left(T^{1/2+\varepsilon}\right)$. All that remains is to evaluate S^L . Observe that

$$S^{L} = \frac{1}{2\pi i} \int_{1-c+iT}^{1-c+iT} \frac{Z'_{1}}{Z_{1}}(s) \zeta^{(n)}(s) \ ds = -\frac{1}{2\pi i} \int_{c-iT}^{c-iT} \frac{Z'_{1}}{Z_{1}}(1-s) \zeta^{(n)}(1-s) \ ds = -\overline{I},$$

where

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1} (1-s) \zeta^{(n)} (1-s) \, ds.$$
(7.15)

Overall, we have

$$\sum_{0<\lambda\leq T}\zeta^{(n)}\left(\frac{1}{2}+i\lambda\right)=-\overline{I}+O\left(T^{1/2+\varepsilon}\right).$$

7.1.4.2 Deriving the main terms

To begin manipulating I into a form that we can evaluate, we use the logarithmic derivative of the functional equation for $Z_1(s)$ given in (7.5). We will also use for functional equation for the derivatives of $\zeta(1-s)$ given in Lemma 3.5. Substituting these two expressions into (7.15) gives

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(-\log\frac{t}{2\pi} \right) (-1)^n \chi (1-s) \sum_{j=0}^n \binom{n}{j} \left(\log\frac{t}{2\pi} \right)^{n-j} \zeta^{(j)}(s) \, ds \\ + \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(-\frac{Z_1'}{Z_1}(s) \right) (-1)^n \chi (1-s) \sum_{j=0}^n \binom{n}{j} \left(\log\frac{t}{2\pi} \right)^{n-j} \zeta^{(j)}(s) \, ds + O\left(T^{1/2+\varepsilon}\right) ds$$

Simplifying slightly gives

$$\begin{split} I &= \frac{(-1)^{n+1}}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \sum_{j=0}^n \binom{n}{j} \left(\log \frac{t}{2\pi}\right)^{n-j+1} \zeta^{(j)}(s) \ ds \\ &+ \frac{(-1)^{n+1}}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \sum_{j=0}^n \binom{n}{j} \left(\log \frac{t}{2\pi}\right)^{n-j} \zeta^{(j)}(s) \frac{Z'_1}{Z_1}(s) \ ds + O\left(T^{1/2+\varepsilon}\right) \\ &= I_1 + I_2 + O\left(T^{1/2+\varepsilon}\right), \end{split}$$

say.

7.1.4.3 The integral I_1

First consider I_1 . We want to apply Lemma 7.8 here to extract terms for the asymptotic. To do this we begin by switching the order of integration and summation and substitute the Dirichlet series for $\zeta^{(j)}(s)$ to obtain

$$I_1 = (-1)^{n+1} \sum_{j=0}^n \binom{n}{j} \left[\frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \left(\log \frac{t}{2\pi} \right)^{n-j+1} \sum_{m=1}^\infty \frac{(-1)^j (\log m)^j}{m^s} \, ds \right]$$

Now apply Lemma 7.8 to the integral, obtaining

$$I_1 = (-1)^{n+1} \sum_{j=0}^n \binom{n}{j} \left[\sum_{m \le T/2\pi} (\log m)^{n-j+1} (-1)^j \log(m)^j \right] + O\left(T^{1/2+\varepsilon}\right).$$

Simplifying this gives

$$I_1 = (-1)^{n+1} \sum_{j=0}^n \binom{n}{j} (-1)^j \left[\sum_{m \le T/2\pi} (\log m)^{n+1} \right] + O\left(T^{1/2+\varepsilon}\right).$$

207

Now by the Binomial Theorem, this summation over j equals zero for n > 0 and equals 1 if n = 0. Hence in the case n = 0, we have a contribution of

$$I_1 = -\sum_{m \le T/2\pi} \log m + O\left(T^{1/2+\varepsilon}\right).$$

By standard methods, we have

$$I_1 = -\frac{T}{2\pi} \log \frac{T}{2\pi} + \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right)$$

and so we have proved the following result.

Lemma 7.9. For n a non-negative integer, the integral I_1 satisfies

$$I_1 = \begin{cases} -\frac{T}{2\pi} \log \frac{T}{2\pi} + \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right) & \text{if } n=0\\ O\left(T^{1/2+\varepsilon}\right) & \text{if } n>0. \end{cases}$$

7.1.4.4 The integral I_2

Similarly to when we evaluated the integral I_1 , we want to apply stationary phase calculations to extract terms for the asymptotic from I_2 . To do this we begin by switching the order of integration and summation and substitute the Dirichlet series for $\zeta^{(j)}(s)$ and the series for the logarithmic derivative of $Z_1(s)$ to obtain

$$\begin{split} I_{2} = (-1)^{n+1} \sum_{j=0}^{n} \binom{n}{j} \\ \times \left[\frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \left(\log \frac{t}{2\pi} \right)^{n-j} \sum_{m_{1}=1}^{\infty} \frac{(-1)^{j} (\log m_{1})^{j}}{m_{1}^{s}} \sum_{m_{2}=1}^{\infty} \frac{(-\Lambda(m_{2}))}{m_{2}^{s}} ds \right] \\ + \sum_{k=1}^{\infty} (-1)^{n+1} \sum_{j=0}^{n} \binom{n}{j} \\ \times \left[\frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{1}{f(s)^{k}} \left(\log \frac{t}{2\pi} \right)^{n-j} \sum_{m_{1}=1}^{\infty} \frac{(-1)^{j} (\log m_{1})^{j}}{m_{1}^{s}} \sum_{m_{2}=1}^{\infty} \frac{a_{k}(m_{2})}{m_{2}^{s}} ds \right] \\ = I_{2,1} + I_{2,2}, \end{split}$$

say, where we split the initial term of $Z'_1(s)/Z_1(s)$ from the rest as $I_{2,1}$ follows directly from stationary phase methods, while $I_{2,2}$ has a more involved argument.

For $I_{2,1}$ we may use Lemma 7.8 to write

$$I_{2,1} = (-1)^{n+1} \sum_{j=0}^{n} \binom{n}{j} \left[\sum_{m_1 m_2 \le T/2\pi} (-1)^j (\log m_1)^j (-\Lambda(m_2)) (\log m_1 m_2)^{n-j} \right] + O\left(T^{1/2+\varepsilon}\right)$$

Switching the order of summation, expanding the logarithms, and performing some basic algebraic manipulations allows us to rewrite this as

$$I_{2,1} = (-1)^n \sum_{m_1 m_2 \le T/2\pi} \Lambda(m_1) (\log m_1)^n + O\left(T^{1/2+\varepsilon}\right).$$

We note that this calculation was preformed in Hughes and Pearce-Crump [186], and is effectively given as the negative of the result of Theorem 3.1. That is,

$$I_{2,1} = (-1)^n \frac{1}{n+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n+1} + (-1)^n \sum_{k=0}^n \binom{n}{k} (-1)^k k! \left(-1 + \sum_{j=0}^k \frac{1}{j!} \gamma_j \right) \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n-k} - n! A_n \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right)$$
(7.16)

where we remind the reader that the γ_j are the coefficients from the Laurent expansion of $\zeta(s)$ about s = 1 and the A_n are the coefficients from the Laurent expansion of $\zeta'(s)/\zeta(s)$ about s = 1.

Finally we need to evaluate $I_{2,2}$. By (7.6) and (7.9), we can write

$$f(s) = -\frac{1}{2}\frac{\chi'}{\chi}(s) = \frac{1}{2}\log\frac{t}{2\pi} + O\left(\frac{1}{t}\right)$$

and so

$$\frac{1}{f(s)^k} = \frac{2^k}{(\log t/2\pi)^k} \left(1 + O\left(\frac{1}{t\log t}\right)\right)^{-k} = \frac{2^k}{(\log t/2\pi)^k} \left(1 + O\left(\frac{k}{t\log t}\right)\right).$$

Then

$$\begin{split} I_{2,2} &= \sum_{k=1}^{\infty} (-1)^{n+1} \sum_{j=0}^{n} \binom{n}{j} \\ &\times \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{2^{k}}{(\log t/2\pi)^{k}} \left(1 + O\left(\frac{k}{t\log t}\right)\right) \\ &\quad \times \left(\log \frac{t}{2\pi}\right)^{n-j} \sum_{m_{1}=1}^{\infty} \frac{(-1)^{j} (\log m_{1})^{j}}{m_{1}^{s}} \sum_{m_{2}=1}^{\infty} \frac{a_{k}(m_{2})}{m_{2}^{s}} ds \\ &= \sum_{k=1}^{\infty} (-1)^{n+1} \sum_{j=0}^{n} \binom{n}{j} \\ &\quad \times \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{2^{k}}{(\log t/2\pi)^{k}} \left(\log \frac{t}{2\pi}\right)^{n-j} \\ &\quad \times \sum_{m_{1}=1}^{\infty} \frac{(-1)^{j} (\log m_{1})^{j}}{m_{1}^{s}} \sum_{m_{2}=1}^{\infty} \frac{a_{k}(m_{2})}{m_{2}^{s}} ds + O\left(T^{1/2+\varepsilon}\right). \end{split}$$

After these initial manipulations performed above, we may apply Lemma 7.8. This gives

$$I_{2,2} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^{n} \binom{n}{j} \left[\sum_{m_1 m_2 \le T/2\pi} \frac{(-1)^j (\log m_1)^j a_k (m_2) (\log m_1 m_2)^{n-j}}{(\log m_1 m_2)^k} \right] + O\left(T^{1/2+\varepsilon}\right).$$
(7.17)

Remark. Now we note that we could repeat our trick that we used when switching the order of summation and expanding the logarithm when we evaluated $I_{2,1}$. However it is then unclear what the correct Dirichlet series is to use in the upcoming Perron argument. Instead we take the approach of cancelling the appropriate powers of the logarithm, and work with $I_{2,2}$ in this form.

Switching the order of summation in $I_{2,2}$ gives

$$I_{2,2} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \left[\sum_{m_1 m_2 \le T/2\pi} \frac{(-1)^j (\log m_1)^j a_k(m_2)}{(\log m_1 m_2)^{k-n+j}} \right] + O\left(T^{1/2+\varepsilon}\right)$$
(7.18)

To evaluate $I_{2,2}$, we begin with evaluating the numerator of the inner sum. After calculating the sum of the numerator of the inner sum, we can preform partial summation to reinsert the logarithm in the denominator of the inner sum, and then sum over j and k.

Lemma 7.10. Let

$$A_{k,j}(x) = \sum_{m \le x} \alpha_{k,j}(m),$$

where

$$\alpha_{k,j}(m) = \sum_{m_1 m_2 = m} (-1)^j (\log m_1)^j a_k(m_2)$$

and where $a_k(m_2)$ is given in (7.10). This sum can also be written as

$$A_{k,j}(x) = \sum_{1 \le mn_1n_2\dots n_k \le x} (-1)^j (\log m)^j \log(n_1) \Lambda(n_1) \Lambda(n_2) \dots \Lambda(n_k).$$

Then for large x,

$$A_{k,j}(x) = x \sum_{\ell=0}^{k+j+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} (\log x)^{k+j+1-\ell} + O\left(x^{1/2+\varepsilon}\right)$$

where the $c_{\ell}^{k,j}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \zeta^{(j)}(s) \frac{1}{s} = \sum_{\ell=0}^{\infty} c_{\ell}^{k,j} (s-1)^{-k-j-2+\ell}.$$

Remark. A long but straightforward calculation tells us what the coefficients $c_{\ell}^{k,j}$ equal. The leading coefficient is when $\ell = 0$ and is given by

211

$$c_0^{k,j} = (-1)^j j!$$

for all $j \ge 0$. When $\ell = 1$, the subleading coefficients are given by

$$c_1^{k,j} = \begin{cases} -1 + \gamma_0 + (1-k)\gamma_0 & \text{if } j = 0\\ (-1)^j j! (-1 + (1-k)\gamma_0) & \text{for all } j \ge 1. \end{cases}$$

Remark. The superscripts on the coefficients $c_{\ell}^{k,j}$ is to indicate that the terms depend on both k and j.

Proof. We begin by applying Perron's formula to $A_{k,j}(x)$, that is, we have

$$A_{k,j}(x) = \frac{1}{2\pi i} \int_{3/2 - iR}^{3/2 + iR} \left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \zeta^{(j)}(s) \frac{x^s}{s} \, ds + O\left(\frac{x^{3/2 + \varepsilon}}{R}\right)$$

Note that the integrand has a pole of order k + j + 2 at s = 1. We shift past the pole at s = 1 to the line $\Re(s) = 1/2 + \varepsilon$, and evaluate the residue at s = 1. Therefore, up to a power-saving error, $A_{k,j}(x)$ will equal the residue of the integrand around that point. That is,

$$A_{k,j}(x) = x \sum_{\ell=0}^{k+j+1} b_{\ell}^{k,j} (\log x)^{k+j+1-\ell} + O\left(R^{\varepsilon} x^{1/2+\varepsilon}\right) + O\left(\frac{x^{3/2+\varepsilon}}{R}\right)$$

for some constants $b_{\ell}^{k,j}$. We have started the sum at the largest power of $\log x$ as this simplifies future calculations. Choosing R = x optimises the error terms, giving an error of $O(x^{1/2+\varepsilon})$.

To evaluate the residue, we use the various Laurent expansions about s = 1 in the various terms in the integrand. The expansions we need are

$$\begin{pmatrix} \zeta'_{\zeta}(s) \end{pmatrix}' = \frac{1}{(s-1)^2} + \left(-2\gamma_1 - \gamma_0^2\right) + \left(6\gamma_0\gamma_1 + 3\gamma_2 + 2\gamma_0^3\right)(s-1) + \dots \\ \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} = \frac{1}{(s-1)^{k-1}} + \frac{(1-k)\gamma_0}{(s-1)^{k-2}} + \frac{\frac{1}{2}k(k-1)\gamma_0^2 + 2(k-1)\gamma_1}{(s-1)^{k-3}} + \dots \\ \zeta^{(j)}(s) = \frac{(-1)^j j!}{(s-1)^{j+1}} + (-1)^j \gamma_j + (-1)^{j+1} \gamma_{j+1}(s-1) + \dots \\ \frac{1}{s} = 1 - (s-1) + (s-1)^2 + \dots$$

where the Stieltjes constants are from the Laurent expansion of $\zeta(s)$ about s = 1.

212Chapter 7. Moments of the Riemann zeta function at its local extrema

Combining these Laurent expansions allows us to write

$$\left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \zeta^{(j)}(s) \frac{1}{s} = \sum_{\ell=0}^{\infty} c_{\ell}^{k,j} (s-1)^{-k-j-2+\ell}$$

where we have explicitly given the first few values of $c_{\ell}^{k,j}$ in the remark following the statement of the lemma.

The residue is the coefficient of 1/(s-1) which will come from combining these terms with the x^s in the integrand (which is the only term that contains an x). Its expansion about s = 1 is

$$x^{s} = x \left(1 + (s-1)\log x + \frac{(s-1)^{2}}{2!} (\log x)^{2} + \dots + \frac{(s-1)^{k}}{k!} (\log x)^{k} + \dots \right)$$

so the coefficient of $(\log x)^{k+j+1-\ell}$ in the $(s-1)^{-1}$ term equals

$$b_{\ell}^{k,j} = \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!}$$

and such a combination is possible for ℓ between 0 and k + j + 1.

To complete the proof of Theorem 7.1, we want to use the asymptotic from Lemma 7.10 in (7.18) and use partial summation to reinsert the logarithm in the denominator of the inner sum in (7.17).

This means that we need to calculate

$$I_{2,2} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \left[A_k \left(\frac{T}{2\pi} \right) f \left(\frac{T}{2\pi} \right) - \int_2^{T/2\pi} A_k(x) f'(x) \, dx \right] + O\left(T^{1/2+\varepsilon} \right)$$
with $f(x) = 1/(\log x)^{k-n+j}$ so that $f'(x) = -(k-n+i)/(x(\log x)^{k-n+j+1})$ and with

with $f(x) = 1/(\log x)^k$ $f'(x) = -(k - n + j)/(x(\log x))$), and with $A_{k,j}(x)$ as in Lemma 7.10.

Note that while we want to sum the inner sum in $I_{2,2}$ over all positive integers m_1, m_2 such that $1 \leq m_1 m_2 \leq T/2\pi$, this introduces a problem with the logarithms appearing in the denominator. Instead we will sum from 2 to avoid this problem, at the cost of a O(1)error which we ignore as it is smaller than our largest error.

Writing $L = \log T/2\pi$, by partial summation we have

$$I_{2,2} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \\ \left[\left(\frac{T}{2\pi} \right)^k \sum_{\ell=0}^{k+j+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} L^{n+1-\ell} + (k-n+j) \sum_{\ell=0}^{k+j+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \int_2^{T/2\pi} (\log x)^{n-\ell} dx \right] \\ + O\left(T^{1/2+\varepsilon} \right).$$
(7.19)

Lemma 7.11. Writing $L = \log T/2\pi$, for any integer $K \ge 1$, we have

$$\begin{split} I_{2,2} &= (-1)^{n+1} \frac{T}{2\pi} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \left(L^{n+1} \frac{c_0^{k,j}}{(k+j+1)!} \right. \\ &+ \sum_{\ell=0}^n L^{n-\ell} \left[\frac{c_{\ell+1}^{k,j}}{(k+j-\ell)!} + (k-n+j) \frac{c_\ell^{k,j}}{(k+j+1-\ell)!} \right. \\ &+ (k-n+j) \sum_{m=0}^{\ell-1} (-1)^{\ell-m} (n-m) \dots (n-\ell+1) \frac{c_m^{k,j}}{(k+j+1-m)!} \right] \right) \\ &+ (-1)^{n+1} \frac{T}{2\pi} \sum_{m=1}^K \frac{1}{L^m} \sum_{k=1}^\infty 2^k \sum_{j=0}^n \binom{n}{j} \beta_m^{k,j} + O\left(\frac{T}{L^{K+1}}\right), \end{split}$$

where for $1 \le m \le k+j-n$, we have

$$\beta_m^{k,j} = \frac{c_{m+n+1}^{k,j}}{(k+j-n-m)!} + \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{m-1} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j}$$

and where for $m \leq k + j - n + 1$, we have

$$\beta_m^{k,j} = \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{k+j-n} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j},$$

where the constants $c_{\ell}^{k,j}$ are given in Lemma 7.10.

Proof. We begin by noting from (7.19) that we can obtain the leading order, that is, the L^{n+1} , from the first sum within the square brackets by setting $\ell = 0$. There are then positive and negative powers of L^n which come from both sums in the square brackets.

We begin by splitting the sums into those that contribute to non-negative powers of L, and those that contribute to negative powers of L. This is because we can evaluate the positive powers exactly, while the negative powers are slightly more complicated. The positive powers occur for $\ell \leq n + 1$ in the first sum, and $\ell \leq n$ in the second sum from the

square brackets. This splitting gives

$$\begin{split} I_{2,2} &= (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \\ & \left[\left(\frac{T}{2\pi} \sum_{\ell=0}^{n+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} L^{n+1-\ell} \right. \\ & \left. + (k-n+j) \sum_{\ell=0}^n \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \int_2^{T/2\pi} (\log x)^{n-\ell} dx \right) \right. \\ & \left. + \left(\frac{T}{2\pi} \sum_{\ell=n+2}^{k+j+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} L^{n+1-\ell} \right. \\ & \left. + (k-n+j) \sum_{\ell=n+1}^{k+j+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \int_2^{T/2\pi} (\log x)^{n-\ell} dx \right) \right] \\ & \left. + O\left(T^{1/2+\varepsilon} \right) \end{split}$$

For ease of notation, we write the sum $I_{2,2}$ as

$$I_{2,2} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^{n} \binom{n}{j} [J_1 + J_2] + O\left(T^{1/2+\varepsilon}\right),$$

say, where both J_1 and J_2 depend on k and j, and the J_1 sum is for the non-negative powers of the logarithm, and the J_2 sum is for the negative powers of the logarithm.

We now consider the two sums J_1 and J_2 in turn.

Consider J_1 first. Recall that for $m \ge 0$, we have

$$\int_{2}^{T/2\pi} (\log x)^m \, dx = \frac{T}{2\pi} \sum_{r=0}^{m} (-1)^r \frac{m!}{(m-r)!} L^{m-r} + O(1).$$

Then using this (and ignoring the error term as it is consumed by larger error terms that we have already seen in this proof), we have

$$J_{1} = \frac{T}{2\pi} \left[\sum_{\ell=0}^{n+1} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} L^{n+1-\ell} + (k-n+j) \sum_{\ell=0}^{n} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \left(\sum_{r=0}^{n-\ell} (-1)^{r} \frac{(n-\ell)!}{(n-\ell-r)!} L^{n-\ell-r} \right) \right].$$

We first note that we can extract the leading order behaviour from the first sum, and

reindex the remaining terms in the first sum to obtain

$$J_{1} = \frac{T}{2\pi} L^{n+1} \frac{c_{0}^{k,j}}{(k+j+1)!} + \frac{T}{2\pi} \left[\sum_{\ell=0}^{n} \frac{c_{\ell+1}^{k,j}}{(k+j-\ell)!} L^{n-\ell} + (k-n+j) \sum_{\ell=0}^{n} \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \left(\sum_{r=0}^{n-\ell} (-1)^{r} \frac{(n-\ell)!}{(n-\ell-r)!} L^{n-\ell-r} \right) \right].$$

We now reorder the second summation to write

$$J_{1} = \frac{T}{2\pi} L^{n+1} \frac{c_{0}^{k,j}}{(k+j+1)!} + \frac{T}{2\pi} \sum_{\ell=0}^{n} \left[\frac{c_{\ell+1}^{k,j}}{(k+j-\ell)!} + (k-n+j) \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} + (k-n+j) \sum_{m=0}^{\ell-1} (-1)^{\ell-m} (n-m) \dots (n-\ell+1) \frac{c_{m}^{k,j}}{(k+j+1-m)!} \right] L^{n-\ell}$$

where if $\ell = 0$ the last sum is empty.

We now calculate the contribution from J_2 . By a simple relabelling, we have

$$J_{2} = \frac{T}{2\pi} \sum_{\ell=1}^{k+j+1} \frac{c_{\ell+n+1}^{k,j}}{(k+j-n-\ell)!} L^{-\ell} + (k-n+j) \sum_{\ell=1}^{k+j+1-n} \frac{c_{\ell+n}^{k,j}}{(k+j+1-n-\ell)!} \int_{2}^{T/2\pi} (\log x)^{-\ell} dx,$$

so both of our sums in J_2 start from $\ell = 1$.

Next we note that in the remaining integrals, all of the powers of $\log x$ are negative, so evaluate to an infinite chain of descending powers of L. For example, for $1 \le m \le M$, we have

$$\int_{2}^{T/2\pi} \frac{1}{(\log x)^{m}} dx = \frac{T}{2\pi} \frac{1}{L^{m}} + m \int_{2}^{T/2\pi} \frac{1}{(\log x)^{m+1}} dt + O(1)$$
$$= \frac{T}{2\pi} \sum_{r=m}^{M} \frac{(r-1)!}{(m-1)!} \frac{1}{L^{r}} + O\left(\frac{T}{L^{M+1}}\right).$$

We write $\beta_m^{k,j}$ for the coefficient of L^{-m} and split into two cases.

If $m \leq k+j-n$, then there will be a contribution from the first sum in J_2 when $m = \ell$ and from all integrals in the second sum for $\ell = 1, \ldots, m$. We have

$$\beta_m^{k,j} = \frac{c_{m+n+1}^{k,j}}{(k+j-n-m)!} + (k-n+j)\sum_{\ell=1}^m \frac{c_{\ell+n}^{k,j}}{(k+j+1-n-\ell)!} \frac{(m-1)!}{(\ell-1)!}.$$

Using the fact

$$\frac{(k-n+j)}{(k+j+1-n-\ell)!(\ell-1)!} = \binom{k-n+j}{\ell-1} \frac{1}{k-n+j-1},$$

this simplifies to

$$\beta_m^{k,j} = \frac{c_{m+n+1}^{k,j}}{(k+j-n-m)!} + \frac{(m-1)!}{k-n+j-1} \sum_{\ell=1}^m \binom{k-n+j}{\ell-1} c_{\ell+n}^{k,j}$$

If $m \ge k + j - n + 1$, then there will only be a contribution coming from the second sum of J_2 , but every term in that sum contributes. We have

$$\beta_m^{k,j} = \frac{(m-1)!}{k-n+j-1} \sum_{\ell=1}^{k+j+1-n} \binom{k-n+j}{\ell-1} c_{\ell+n}^{k,j}.$$

After a trivial relabelling for both versions of $\beta_m^{k,j}$, we have the statement of the lemma. \Box

We can see that we have

$$I_{2,2} = \mathfrak{a}_{n+1} \frac{T}{2\pi} L^{n+1} + \frac{T}{2\pi} \sum_{\ell=0}^{n} \mathfrak{a}_{n-\ell} L^{n-\ell} + \frac{T}{2\pi} \sum_{m=1}^{K} \frac{\mathfrak{b}_m}{L^m} + O\left(\frac{T}{L^{K+1}}\right),$$

where the leading order coefficient is given by

$$\mathfrak{a}_{n+1} = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \frac{c_0^{k,j}}{(k+j+1)!},$$

the subleading, non-negative logarithm power coefficients are given by

$$\begin{aligned} \mathfrak{a}_{n-\ell} &= (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \left[\frac{c_{\ell+1}^{k,j}}{(k+j-\ell)!} + (k-n+j) \frac{c_{\ell}^{k,j}}{(k+j+1-\ell)!} \right. \\ &+ (k-n+j) \sum_{m=0}^{\ell-1} (-1)^{\ell-m} (n-m) \dots (n-\ell+1) \frac{c_m^{k,j}}{(k+j+1-m)!} \right], \end{aligned}$$

and the subleading, negative logarithm power coefficients are given by

$$\mathfrak{b}_m = (-1)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \beta_m^{k,j},$$

where for $1 \le m \le k + j - n$, we have

$$\beta_m^{k,j} = \frac{c_{m+n+1}^{k,j}}{(k+j-n-m)!} + \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{m-1} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j}$$

and where for $m \leq k + j - n + 1$, we have

$$\beta_m^{k,j} = \frac{(m-1)!}{k-n+j-1} \sum_{\ell=0}^{k+j-n} \binom{k-n+j}{\ell} c_{\ell+n+1}^{k,j},$$
where the constants $c_{\ell}^{k,j}$ are given in Lemma 7.10.

By Lemma 7.11 we can see that to complete calculating $I_{2,2}$, all we need to do is sum over $j = 0, \ldots, n$ and over $k \ge 1$.

In the remark following Lemma 7.10 we calculated $c_0^{k,j}$ and $c_1^{k,j}$ explicitly, and so we could calculate the leading and subleading terms in $I_{2,2}$, using our result for J_1 .

In this case, to leading order we would have

$$(-1)^{n+1}\frac{T}{2\pi}L^{n+1}\sum_{k=1}^{\infty}2^k\sum_{j=0}^n\binom{n}{j}\frac{c_0^{k,j}}{(k+j+1)!}.$$

Substitute the value of $c_0^{k,j}$ from Lemma 7.10 to obtain

$$(-1)^{n+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n+1} \sum_{k=1}^{\infty} 2^k \sum_{j=0}^n \binom{n}{j} \frac{(-1)^j j!}{(k+j+1)!}$$
$$= (-1)^{n+1} \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n+1} \sum_{k=1}^{\infty} \left(\frac{2^k}{k!(k+n+1)} \right),$$

where the second line follows after summing over j. Now we need to sum over k. For this sum we obtain

$$\sum_{k=1}^{\infty} \frac{2^k}{k!(k+n+1)} = \frac{e^2 - 1}{n+1} + (-1)^{n+1} \left(\frac{\Gamma(n+2) - \Gamma(n+2,-2)}{2^{n+1}(n+1)} \right)$$
$$= \frac{e^2 - 1}{n+1} + (-1)^{n+1} \frac{n!}{2^{n+1}} \left(1 - e^2 \sum_{k=0}^{n+1} \frac{(-2)^k}{k!} \right)$$

where we have used the incomplete gamma function in simplifying the expressions above, which for integer n is given by

$$\Gamma(n,x) = (n-1)! e^{-x} \sum_{k=0}^{n-1} \frac{x^k}{k!}.$$

This means that to leading order, we have

$$I_{2,2} \sim (-1)^{n+1} \left(\frac{e^2 - 1}{n+1} + (-1)^{n+1} \frac{n!}{2^{n+1}} \left(1 - e^2 \sum_{k=0}^{n+1} \frac{(-2)^k}{k!} \right) \right) \frac{T}{2\pi} \left(\log \frac{T}{2\pi} \right)^{n+1}$$

In a similar way, we can calculate the subleading behaviour, and show that this contributes

$$(-1)^{n+1}\frac{T}{2\pi}L^n\left(1-(1+e^2)\gamma_0+(-1)^{n+1}\frac{(n+1)!}{2^{n+1}}(-1+2\gamma_0)\left(1-e^2\sum_{k=0}^n\frac{(-2)^k}{k!}\right)\right).$$

to the asymptotic.

Further calculations from Lemma 7.10 would give rise to more lower order terms in the obvious way.

217

We then recombine $I_{2,2}$ here with I_1 from Lemma 7.9 and with $I_{2,1}$ from (7.16), which gives I.

Finally, recall that

$$\sum_{0<\lambda\leq T}\zeta^{(n)}\left(\frac{1}{2}+i\lambda\right) = -\overline{I} + O\left(T^{1/2+\varepsilon}\right)$$

to obtain the asymptotic, proving Theorem 7.1.

7.1.5 Outline of the proof of Theorem 7.3

We begin by noting that the proof of Theorem 7.3 follows in a similar way to that of Theorem 7.1. We sketch the differences below but leave the full details to the interested reader.

The initial steps remain the same in the case n = 0 as in the proof of Theorem 7.1. We are able to write

$$\sum_{0 < \lambda \le T} \zeta \left(\frac{1}{2} + i\lambda \right) = -\overline{I} + O\left(T^{1/2 + \varepsilon} \right)$$

where

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1} (1-s)\zeta(1-s) \, ds,$$

where $c = 1 + 1 / \log T$.

To begin manipulating I into a form that we can evaluate, we use the logarithmic derivative of the functional equation for $Z_1(s)$, given in (7.5), and the functional equation for $\zeta(s)$, given in (1.1). Then

$$\begin{split} I &= \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1} (1-s)\zeta(1-s) \, ds \\ &= -\frac{1}{2\pi i} \int_{c+i}^{c+iT} \log \frac{t}{2\pi} \chi(1-s)\zeta(s) \, ds - \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1} (s)\chi(1-s)\zeta(s) \, ds + O\left(T^{1/2+\varepsilon}\right) \\ &= I_1 + I_2 + O\left(T^{1/2+\varepsilon}\right), \end{split}$$

say.

Then by Lemma 7.9, we have

$$I_1 = -\frac{T}{2\pi} \log \frac{T}{2\pi} + \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right).$$
 (7.20)

For I_2 , we have

$$\begin{split} I_2 &= -\frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1}(s) \chi(1-s) \zeta(s) \ ds \\ &= \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \sum_{n=1}^{\infty} \frac{\Lambda(n)}{n^s} \sum_{m=1}^{\infty} \frac{1}{m^s} \ ds \\ &\quad -\sum_{k=1}^{\infty} \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{1}{f(s)^k} \sum_{n=1}^{\infty} \frac{a_k(n)}{n^s} \sum_{m=1}^{\infty} \frac{1}{m^s} \ ds \\ &= I_{2,1} + I_{2,2}, \end{split}$$

say.

Then $I_{2,1}$ can be evaluated through a simple application of Lemma 7.8 and Perron's theorem, giving

$$I_{2,1} = \frac{T}{2\pi} \log \frac{T}{2\pi} - \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right).$$
(7.21)

Combining I_1 from (7.20) and $I_{2,1}$ from (7.21) gives a contribution of $O(T^{1/2+\varepsilon})$ to the asymptotic.

Finally, to evaluate $I_{2,2}$, we have after using Lemma 7.8

$$I_{2,2} = -\sum_{k=1}^{\infty} \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{1}{f(s)^k} \sum_{n=1}^{\infty} \frac{a_k(n)}{n^s} \sum_{m=1}^{\infty} \frac{1}{m^s} ds$$
$$= -\sum_{k=1}^{\infty} 2^k \sum_{nm \le T/2\pi} \frac{a_k(n)}{(\log nm)^k} + O\left(T^{1/2+\varepsilon}\right) + O\left(T^{1/2+\varepsilon}\right)$$

As in the proof of Theorem 7.1, we evaluate the numerator of the inner sum in the previous line via Perron. This is actually a special case of Lemma 7.10 in the case j = 0, so we have the following result.

Lemma 7.12. Let

$$A_k(x) = \sum_{mn \le x} a_k(n),$$

where $a_k(n)$ is given in (7.10). This sum can also be written as

$$A_k(x) = \sum_{1 \le mn_1n_2...n_k \le x} \log(n_1)\Lambda(n_1)\Lambda(n_2)\ldots\Lambda(n_k).$$

Then for large x,

$$A_k(x) = x \sum_{\ell=0}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} (\log x)^{k+1-\ell} + O\left(x^{1/2+\varepsilon}\right)$$

where the $c_{k,\ell}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)'\left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1}\zeta(s)\frac{1}{s} = \sum_{\ell=0}^{\infty} c_{k,\ell}(s-1)^{-k-2+\ell}.$$

219

Remark. A long but straightforward calculation tells us what the coefficients $c_{k,\ell}$ equal. The leading coefficient is when $\ell = 0$ and is given by

$$c_{k,0} = 1.$$

When $\ell = 1$, the subleading coefficient is given by

$$c_{k,1} = -1 + \gamma_0 + (1-k)\gamma_0,$$

Next we use partial summation to calculate

$$\sum_{nm \le T/2\pi} \frac{a_k(n)}{(\log nm)^k} = A_k\left(\frac{T}{2\pi}\right) f\left(\frac{T}{2\pi}\right) - \int_2^{T/2\pi} A_k(x) f'(x) \ dx,$$

in the case $f(x) = 1/(\log x)^k$ (so $f'(x) = -k/(x(\log x)^{k+1}))$). Then using $A_k(x)$ from Lemma 7.12, we have with $L = \log T/2\pi$,

$$\sum_{nm \le T/2\pi} \frac{a_k(n)}{(\log nm)^k} = \frac{T}{2\pi} \sum_{\ell=0}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} L^{1-\ell} + k \sum_{\ell=0}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} \int_2^{T/2\pi} (\log x)^{-\ell} dx + O\left(T^{1/2+\varepsilon}\right).$$

As before, we extract the non-negative powers of the logarithm and calculate the coefficients of the negative powers as we clearly have an infinite chain of decreasing powers of the logarithm. This gives

$$\sum_{nm \le T/2\pi} \frac{a_k(n)}{(\log nm)^k} = \frac{T}{2\pi} \frac{c_{k,0}}{(k+1)!} L + \frac{T}{2\pi} \frac{c_{k,1}}{k!} + \frac{T}{2\pi} k \frac{c_{k,0}}{(k+1)!} + \frac{T}{2\pi} \sum_{\ell=2}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} L^{1-\ell} + k \sum_{\ell=1}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} \int_2^{T/2\pi} (\log x)^{-\ell} dx + O\left(T^{1/2+\varepsilon}\right).$$

For the negative powers of the logarithm, we reindex so both sums start at $\ell = 1$ to give

$$\frac{T}{2\pi} \sum_{\ell=1}^{k} \frac{c_{k,\ell+1}}{(k-\ell)!} L^{-\ell} + k \sum_{\ell=1}^{k+1} \frac{c_{k,\ell}}{(k+1-\ell)!} \int_{2}^{T/2\pi} (\log x)^{-\ell} dx.$$

Recall that for $1 \leq m \leq M$, we have

$$\int_{2}^{T/2\pi} \frac{1}{(\log x)^{m}} dx = \frac{T}{2\pi} \frac{1}{L^{m}} + m \int_{2}^{T/2\pi} \frac{1}{(\log x)^{m+1}} dt + O(1)$$
$$= \frac{T}{2\pi} \sum_{r=m}^{M} \frac{(r-1)!}{(m-1)!} \frac{1}{L^{r}} + O\left(\frac{T}{L^{M+1}}\right).$$

We now write $\beta_m^{k,\ell}$ for the coefficient of L^{-m} . In an analogous way to to the proof of Theorem 7.1, we have for $m \leq k$,

$$\beta_m^{k,j} = \frac{c_{k,m+1}}{(k-m)!} + k\frac{(m-1)!}{(k-1)!} \sum_{\ell=0}^{m-1} \binom{k}{\ell} c_{k,\ell+1}$$

and where for $m \leq k + j - n + 1$, we have

$$\beta_m^{k,j} = k \frac{(m-1)!}{(k-1)!} \sum_{\ell=0}^k \binom{k}{\ell} c_{k,\ell+1},$$

where the constants $c_{k,\ell}$ are given in Lemma 7.12.

In order to complete the proof, we multiply through by -2^k , and sum over all $k \ge 1$ to obtain $I_{2,2}$. Since the combination from the rest of the calculation only contributes to the error term, we have evaluated I and so the sum in question.

Note that since we have a simple form for $c_{k,0}$ and $c_{k,1}$, we can easily calculate the leading and subleading behaviour explicitly, giving for K a positive integer

$$\sum_{0<\lambda\leq T} \zeta\left(\frac{1}{2}+i\lambda\right) = \frac{e^2-3}{2}\frac{T}{2\pi}L + \frac{3-e^2-4\gamma_0}{2}\frac{T}{2\pi} + \frac{T}{2\pi}\sum_{m=1}^K \frac{d_m}{L^m} + O_K\left(\frac{T}{L^{K+1}}\right)$$

as $T \to \infty$, where

$$d_m = \sum_{k=1}^{\infty} 2^k \beta_m^{k,\ell},$$

with the $\beta_m^{k,\ell}$ as above.

7.1.6 Proof Theorem 7.4

The main steps in the proof of Theorem 7.4 follow in a similar way to Theorem 7.1. We sketch the steps here for completeness sake.

Under the Riemann Hypothesis we can show that all the zeros off the critical line only contribute to the error term in our asymptotic with a similar justification to the proof of Theorem 7.1. Recall $\chi(s) \ll T^{1/2-\sigma}$ for $s = \sigma + it$ with $|t| \ll T$. Write $\rho_1 = \beta_1 + i\gamma_1$ for a zero of $Z_1(s)$. By part (3) of Lemma 7.6, $Z_1(s)$ has $O(\log T)$ zeros off the critical line (that is, $\beta_1 \neq 1/2$). At the zeros off the critical line, we may use part (4) of Lemma 7.6 to obtain

$$\sum_{\substack{0 < \gamma_1 \le T \\ \beta_1 \neq 1/2}} \chi(\rho_1) \ll T^{1/9} \log T \ll T^{1/9 + \varepsilon}.$$

Therefore, we have

$$\sum_{0<\lambda\leq T}\chi\left(\frac{1}{2}+i\lambda\right)=\sum_{\substack{0<\gamma_1\leq T\\\beta_1=1/2}}\chi\left(\frac{1}{2}+i\gamma_1\right).$$

We can write this as an integral using Cauchy's theorem,

$$\frac{1}{2\pi i} \int_{\mathcal{C}} \frac{Z_1'}{Z_1}(s)\chi(s) \ ds = \sum_{\substack{0 < \gamma_1 \le T \\ \beta_1 = 1/2}} \chi(\rho_1)$$
$$= \sum_{\substack{0 < \gamma_1 \le T \\ \beta_1 = 1/2}} \chi\left(\frac{1}{2} + i\gamma_1\right) + O\left(T^{1/9+\varepsilon}\right)$$
(7.22)

where C is a positively oriented contour with vertices c+i, c+iT, 1-c+iT, and 1-c+i, where $c = 1 + 1/\log T$. We may assume, without loss of generality, that the distance from the contour to any zero ρ_1 of $Z_1(s)$ is uniformly $\gg 1/\log T$.

We split the integral as

$$\begin{split} &\frac{1}{2\pi i} \left(\int_{c+i}^{c+iT} + \int_{c+iT}^{1-c+iT} + \int_{1-c+iT}^{1-c+i} + \int_{1-c+i}^{c+i} \right) \frac{Z_1'}{Z_1}(s) \chi(s) \ ds \\ &= S^R + S^T + S^L + S^B, \end{split}$$

say. Note that S^B, S^T, S^R are all trivially bounded within error term in a similar way to the proof of Theorem 7.1, using the bound $\chi(s) \ll T^{1/2-\sigma}$ where appropriate. This shows that the leading contributions to the asymptotic will then come from S^L .

Since S^B, S^T, S^R are all within an error term of $O(T^{1/2+\varepsilon})$, all that remains is to evaluate S^L . Note that

$$S^{L} = \frac{1}{2\pi i} \int_{1-c+iT}^{1-c+iT} \frac{Z'_{1}}{Z_{1}}(s)\chi(s) \ ds = -\frac{1}{2\pi i} \int_{c-iT}^{c-iT} \frac{Z'_{1}}{Z_{1}}(1-s)\chi(1-s) \ ds = -\overline{I},$$

where

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{Z_1'}{Z_1} (1-s)\chi(1-s) \, ds.$$
(7.23)

Overall, we have

$$\sum_{0<\lambda\leq T}\chi\left(\frac{1}{2}+i\lambda\right) = -\overline{I} + O\left(T^{1/2+\varepsilon}\right).$$

Using the version of the functional equation for $Z'_1(s)/Z_1(s)$ given in (7.7), we have

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(-\log \frac{t}{2\pi} \right) \chi(1-s) \, ds + \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(-\frac{Z_1'}{Z_1}(s) \right) \chi(1-s) \, ds + O\left(T^{1/2+\varepsilon}\right) = I_1 + I_2 + O\left(T^{1/2+\varepsilon}\right),$$

say.

Then I_1 doesn't contribute to anything beyond an error term, so we split I_2 as before to write

$$I_{2} = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \left(\sum_{n=1}^{\infty} \frac{\Lambda(n)}{n^{s}} \right) \chi(1-s) \, ds$$
$$- \sum_{k=1}^{\infty} \frac{1}{2\pi i} \int_{c+i}^{c+iT} \chi(1-s) \frac{1}{f(s)^{k}} \sum_{n=1}^{\infty} \frac{a_{k}(n)}{n^{s}} \, ds + O\left(T^{1/2+\varepsilon}\right)$$
$$= I_{2,1} + I_{2,2} + O\left(T^{1/2+\varepsilon}\right).$$

Then applying Lemma 7.8 to $I_{2,1}$ we have

$$I_{2,1} = \sum_{n \le T/2\pi} \Lambda(n) + O\left(T^{1/2+\varepsilon}\right).$$

By Perron/the Prime Number Theorem, we have

$$I_{2,1} = \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right).$$
 (7.24)

223

Finally, applying Lemma 7.8 to $I_{2,2}$ gives

$$I_{2,2} = -\sum_{k=1}^{\infty} 2^k \sum_{n \le T/2\pi} \frac{a_k(n)}{(\log n)^k} + O\left(T^{1/2+\varepsilon}\right).$$

As in the proof of Theorem 7.1, we evaluate the numerator of the inner sum in the previous line via Perron.

Lemma 7.13. Let

$$A_k(x) = \sum_{n \le x} a_k(n),$$

where $a_k(n)$ is given in (7.10). This sum can also be written as

$$A_k(x) = \sum_{1 \le n_1 n_2 \dots n_k \le x} \log(n_1) \Lambda(n_1) \Lambda(n_2) \dots \Lambda(n_k).$$

Then for large x,

$$A_k(x) = x \sum_{\ell=0}^k \frac{c_{k,\ell}}{(k-\ell)!} (\log x)^{k-\ell} + O\left(x^{1/2+\varepsilon}\right)$$

where the $c_{k,\ell}$ are the Laurent series coefficients around s = 1 of

$$\left(\frac{\zeta'}{\zeta}(s)\right)' \left(-\frac{\zeta'}{\zeta}(s)\right)^{k-1} \frac{1}{s} = \sum_{\ell=0}^{\infty} c_{k,\ell}(s-1)^{-k-1+\ell}.$$

Remark. A long but straightforward calculation tells us what the coefficients $c_{k,\ell}$ equal. The leading coefficient is when $\ell = 0$ and is given by

$$c_{k,0} = 1.$$

When $\ell = 1$, the subleading coefficient is given by

$$c_{k,1} = -1 + \gamma_0 - k\gamma_0.$$

Next we use partial summation to calculate

$$\sum_{n \le T/2\pi} \frac{a_k(n)}{(\log n)^k} = A_k\left(\frac{T}{2\pi}\right) f\left(\frac{T}{2\pi}\right) - \int_2^{T/2\pi} A_k(x) f'(x) \ dx,$$

in the case $f(x) = 1/(\log x)^k$ (so $f'(x) = -k/(x(\log x)^{k+1}))$). Then using $A_k(x)$ from Lemma 7.13, we have with $L = \log T/2\pi$,

$$\sum_{n \le T/2\pi} \frac{a_k(n)}{(\log n)^k} = \frac{T}{2\pi} \sum_{\ell=0}^k \frac{c_{k,\ell}}{(k-\ell)!} L^{-\ell} + k \sum_{\ell=0}^k \frac{c_{k,\ell}}{(k-\ell)!} \int_2^{T/2\pi} (\log x)^{-1-\ell} dx + O\left(T^{1/2+\varepsilon}\right).$$

As before, we extract the non-negative powers of the logarithm and calculate the coefficients of the negative powers as we clearly have an infinite chain of decreasing powers of the logarithm. This gives

$$\sum_{n \le T/2\pi} \frac{a_k(n)}{(\log n)^k} = \frac{T}{2\pi} \frac{c_{k,0}}{k!} L + \frac{T}{2\pi} \sum_{\ell=1}^k \frac{c_{k,\ell}}{(k-\ell)!} L^{-\ell} + k \sum_{\ell=0}^k \frac{c_{k,\ell}}{(k-\ell)!} \int_2^{T/2\pi} (\log x)^{-1-\ell} dx + O\left(T^{1/2+\varepsilon}\right).$$

For the negative powers of the logarithm, we reindex so both sums start at $\ell = 1$ to give

$$\frac{T}{2\pi} \sum_{\ell=1}^{k} \frac{c_{k,\ell}}{(k-\ell)!} L^{-\ell} + k \sum_{\ell=1}^{k+1} \frac{c_{k,\ell-1}}{(k+1-\ell)!} \int_{2}^{T/2\pi} (\log x)^{-\ell} dx.$$

Recall that for $1 \leq m \leq M$, we have

$$\int_{2}^{T/2\pi} \frac{1}{(\log x)^{m}} dx = \frac{T}{2\pi} \frac{1}{L^{m}} + m \int_{2}^{T/2\pi} \frac{1}{(\log x)^{m+1}} dt + O(1)$$
$$= \frac{T}{2\pi} \sum_{r=m}^{M} \frac{(r-1)!}{(m-1)!} \frac{1}{L^{r}} + O\left(\frac{T}{L^{M+1}}\right).$$

We now write $\beta_m^{k,\ell}$ for the coefficient of L^{-m} . In an analogous way to to the proof of Theorem 7.1, we have for $m \leq k$,

$$\beta_m^{k,j} = \frac{c_{k,m}}{(k-m)!} + \frac{(m-1)!}{(k-1)!} \sum_{\ell=0}^{m-1} \binom{k}{\ell} c_{k,\ell}$$

and where for $m \leq k + j - n + 1$, we have

$$\beta_m^{k,j} = \frac{(m-1)!}{(k-1)!} \sum_{\ell=0}^k \binom{k}{\ell} c_{k,\ell},$$

where the constants $c_{k,\ell}$ are given in Lemma 7.13.

In order to complete the proof, we multiply through by -2^k , and sum over all $k \ge 1$ to obtain $I_{2,2}$. Once we add in the contribution from $I_{2,1}$ given in (7.24), we have evaluated I and so the sum in question.

Note that since we have a simple form for $c_{k,0}$ and $c_{k,1}$, we can easily calculate the leading and subleading behaviour explicitly, giving for K a positive integer

$$\sum_{0<\lambda\leq T} \chi\left(\frac{1}{2}+i\lambda\right) = (e^2-2)\frac{T}{2\pi} - 4e^2\gamma_0\frac{T}{2\pi}\frac{1}{L} + \frac{T}{2\pi}\sum_{m=2}^{K}\frac{e_m}{L^m} + O_K\left(\frac{T}{L^{K+1}}\right)$$

as $T \to \infty$, where

$$e_m = \sum_{k=1}^{\infty} 2^k \beta_m^{k,\ell},$$

with the $\beta_m^{k,\ell}$ as above.

7.1.7 Outline of the proof of Theorem 7.5

The proof of Theorem 7.5 is very similar to that of Theorem 7.4. We just sketch out the main steps here. Note that this result can be made unconditional but we assume the Riemann Hypothesis to make the error terms trivial to deal with.

We begin by using Cauchy's theorem to write

$$\sum_{0 < \gamma \le T} \chi\left(\frac{1}{2} + i\gamma\right) = \frac{1}{2\pi i} \int_{\mathcal{C}} \frac{\zeta'}{\zeta}(s)\chi(s) \ ds,$$

where C is a positively oriented contour with vertices c+i, c+iT, 1-c+iT, and 1-c+i, where $c = 1 + 1/\log T$. We may assume, without loss of generality, that the distance from the contour to any zero of $\zeta(s)$ is uniformly $\gg 1/\log T$.

Note that the integrals along the bottom, top and right-hand side of the contour are bounded by $O(T^{1/2+\varepsilon})$, using the bound $\chi(s) \ll T^{1/2-\sigma}$ for $s = \sigma + it$ with $|t| \ll T$. As

before, the only contribution comes from the left-hand side of the contour, and as before, we have

$$\frac{1}{2\pi i} \int_{1-c+iT}^{1-c+i} \frac{\zeta}{\zeta}(s)\chi(s) \, ds = -\frac{1}{2\pi i} \int_{c-iT}^{c-i} \frac{\zeta}{\zeta}(1-s)\chi(1-s) \, ds = -\overline{I},$$

where

$$I = \frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{\zeta}{\zeta} (1-s)\chi(1-s) \, ds.$$

Overall, we have

$$\sum_{0 < \gamma \le T} \chi\left(\frac{1}{2} + i\gamma\right) = -\overline{I} + O\left(T^{1/2+\varepsilon}\right).$$

Next we need the functional equation for $\zeta'(s)/\zeta(s)$, which can be derived easily from either the functional equation for $\zeta(s)$ in (1.1), or Lemma 3.5. This gives

$$\frac{\zeta'}{\zeta}(1-s) = -\log\frac{t}{2\pi} - \frac{\zeta'}{\zeta}(s) + O\left(\frac{1}{|t|}\right).$$

Substituting this into I gives

$$I = -\frac{1}{2\pi i} \int_{c+i}^{c+iT} \log \frac{t}{2\pi} \chi(1-s) \, ds$$
$$-\frac{1}{2\pi i} \int_{c+i}^{c+iT} \frac{\zeta'}{\zeta}(s) \chi(1-s) \, ds + O\left(T^{1/2+\varepsilon}\right)$$
$$= I_1 + I_2 + O\left(T^{1/2+\varepsilon}\right).$$

Clearly I_1 doesn't contribute to the asymptotic, so consider I_2 . Then by Lemma 7.8

$$I_2 = \sum_{n \le T/2\pi} \Lambda(n) + O\left(T^{1/2+\varepsilon}\right).$$

Then by Perron/the Prime Number Theorem,

$$I_2 = \frac{T}{2\pi} + O\left(T^{1/2+\varepsilon}\right).$$

The result then follows by working back up the proof.

Remark. We could see this result directly from the proof of Theorem 7.4, noticing that the logarithmic derivative of $\zeta(s)$ is included in that of $Z_1(s)$, and making the obvious changes to the proof. We have included the basic steps above for clarity.

References

- E. Amzallag, L.-P. Arguin, E. Bailey, K. Hui, and R. Rao. Evidence of random matrix corrections for the large deviations of Selberg's central limit theorem. *Exp. Math.*, 33(1):123–135, 2024.
- J. Andrade. Random Matrix Theory and L-functions in function fields. PhD thesis, University of Bristol, 2012.
- [3] G. Andrews, R. Askey, and R. Roy. Special Functions, volume 71 of Encyclopedia of Mathematics and its Applications. Cambridge University Press, 1999.
- [4] R. Apéry. Irrationalité de $\zeta(2)$ et $\zeta(3)$. Astérisque, 61(11-13):1, 1979.
- [5] T. M. Apostol. Introduction to Analytic Number Theory. Springer Science & Business Media, 2013.
- [6] L.-P. Arguin, P. Bourgade, and M. Radziwiłł. The Fyodorov-Hiary-Keating conjecture. I. arXiv preprint arXiv:2007.00988, 2020.
- [7] L.-P. Arguin, P. Bourgade, and M. Radziwiłł. The Fyodorov-Hiary-Keating conjecture. II. arXiv preprint arXiv:2307.00982, 2023.
- [8] T. Assiotis, M. A. Gunes, J. P. Keating, and F. Wei. Exchangeable arrays and integrable systems for characteristic polynomials of random matrices. arXiv preprint arXiv:2407.19233, 2024.
- [9] T. Assiotis, J. P. Keating, and J. Warren. On the joint moments of the characteristic polynomials of random unitary matrices. *Int. Math. Res. Not. IMRN*, 18:14564–14603, 2022.
- [10] F. V. Atkinson. The mean value of the zeta-function on the critical line. Proc. London Math. Soc. (2), 2(1):174–200, 1941.
- [11] F. V. Atkinson. A mean value property of the Riemann zeta-function. J. London Math. Soc., 1(2):128–135, 1948.

- [12] E. C. Bailey, S. Bettin, G. Blower, J. B. Conrey, A. Prokhorov, M. O. Rubinstein, and N. C. Snaith. Mixed moments of characteristic polynomials of random unitary matrices. J. Math. Phys., 60, 2019.
- [13] E. C. Bailey and J. P. Keating. Maxima of log-correlated fields: some recent developments. J. Phys. A, 55(5):053001, 2022.
- [14] A. Baker. Linear forms in the logarithms of algebraic numbers. Mathematika, 13(1):204–216, 1966.
- [15] R. Balasubramanian, J. B. Conrey, and D. R. Heath-Brown. Asymptotic mean square of the product of the Riemann zeta-function and a Dirichlet polynomial. J. *Reine Angew. Math.*, 357:161–181, 1985.
- [16] R. Balasubramanian and K. Ramachandra. Proof of some conjectures on the meanvalue of Titchmarsh series I. *Hardy-Ramanujan J.*, 13, 1990.
- [17] S. A. C. Baluyot. On the zeros of Riemann's zeta-function on the critical line. J. Number Theory, 165:203–269, 2016.
- [18] S. A. C. Baluyot, D. A. Goldston, A. I. Suriajaya, and C. L. Turnage-Butterbaugh. An unconditional Montgomery theorem for pair correlation of zeros of the Riemann zeta-function. Acta Arith., 214:357–376, 2024.
- [19] E. W. Barnes. The theory of the *G*-function. *Quart. J. Math.*, 31:264–314, 1899.
- [20] E. L. Basor, P. Bleher, R. Buckingham, T. Grava, A. Its, E. Its, and J. P. Keating. A representation of joint moments of CUE characteristic polynomials in terms of Painlevé functions. *Nonlinearity.*, 32:4033–4078, 2019.
- [21] E. L. Basor and P. J. Forrester. Formulas for the evaluation of Toeplitz determinants with rational generating functions. *Math. Nachr.*, 170:5–18, 1994.
- [22] K. Benli, E. Elma, and N. Ng. A discrete mean value of the Riemann zeta function. Mathematika, 70, 2024.
- [23] M. V. Berry and J. P. Keating. Clusters of near-degenerate levels dominate negative moments of spectral determinants. J. Phys. A, 35(1):L1, 2001.
- [24] J. Bertrand. Mémoire sur le nombre de valeurs que peut prendre une fonction: quand on y permute les lettres qu'elle renferme. Bachelier, 1845.
- [25] S. Bettin. The second moment of the Riemann zeta function with unbounded shifts. Int. J. Number Theory, 6(08):1933–1944, 2010.
- [26] S. Bettin, H. M. Bui, X. Li, and M. Radziwiłł. A quadratic divisor problem and moments of the Riemann zeta-function. J. Eur. Math. Soc., 22(12):3953–3980, 2020.

- [27] S. Bettin, V. Chandee, and M. Radziwiłł. The mean square of the product of the Riemann zeta-function with Dirichlet polynomials. J. Reine Angew. Math., 2017(729):51–79, 2017.
- [28] S. Bettin, V. Chandee, and M. Radziwiłł. The mean square of the product of the Riemann zeta-function with Dirichlet polynomials. *Journal für die reine und* angewandte Mathematik (Crelles Journal), 2017(729):51–79, 2017.
- [29] S. Bettin and S. M. Gonek. The $\theta = \infty$ conjecture implies the Riemann Hypothesis. Mathematika, 63(1):29–33, 2017.
- [30] E. B. Bogomolny and J. P. Keating. Random matrix theory and the Riemann zeros.I. Three-and four-point correlations. *Nonlinearity*, 8(6):1115, 1995.
- [31] E. B. Bogomolny and J. P. Keating. Random matrix theory and the Riemann zeros II: n-point correlations. Nonlinearity, 9(4):911, 1996.
- [32] E. B. Bogomolny and P. Leboeuf. Statistical properties of the zeros of zeta functionsbeyond the Riemann case. *Nonlinearity*, 7(4):1155, 1994.
- [33] H. Bohr and E. Landau. Sur les zéros de la fonction $\zeta(s)$ de Riemann. C. R. Math. Acad. Sci. Paris, 158:106–110, 1914.
- [34] E. Bombieri. Problems of the Millennium: The Riemann Hypothesis. *Clay Mathematics Institute*, 2000.
- [35] E. Bombieri and J. C. Lagarias. Complements to Li's criterion for the Riemann Hypothesis. J. Number Theory, 77(2):274–287, 1999.
- [36] A. Borodin and E. Strahov. Averages of characteristic polynomials in random matrix theory. Comm. Pure Appl. Math., 59(2):161–253, 2006.
- [37] J. Bourgain. Decoupling, exponential sums and the Riemann zeta function. J. Amer. Math. Soc., 30(1):205-224, 2017.
- [38] N. Bruijn. The roots of trigonometric integrals. Duke Math. J, 17(3):197–226, 1950.
- [39] H. Bui and D. Heath-Brown. On simple zeros of the Riemann zeta-function. Bull. Lond. Math. Soc., 45(5):953–961, 2013.
- [40] H. M. Bui. Mean values of L-functions. PhD thesis, University of Bristol, 2007.
- [41] H. M. Bui. Critical zeros of the Riemann zeta-function. arXiv preprint arXiv:1410.2433, 2014.
- [42] H. M. Bui, J. B. Conrey, and M. P. Young. More than 41% of the zeros of the zeta function are on the critical line. Acta Arith., 150:35–64, 2011.
- [43] H. M. Bui and A. Florea. Negative moments of the Riemann zeta-function. J. Reine Angew. Math., 2024(806):247–288, 2024.

- [44] H. M. Bui, A. Florea, and J. P. Keating. The Ratios Conjecture and upper bounds for negative moments of *L*-functions over function fields. *Trans. Amer. Math. Soc.*, 376(06):4453–4510, 2023.
- [45] H. M. Bui, A. Florea, and M. B. Milinovich. Negative discrete moments of the derivative of the Riemann zeta-function. *Bull. Lond. Math. Soc.*, 2024.
- [46] H. M. Bui, D. A. Goldston, M. B. Milinovich, and H. L. Montgomery. Small gaps and small spacings between zeta zeros. Acta Arith., 210, 2023.
- [47] H. M. Bui, S. M. Gonek, and M. B. Milinovich. A hybrid Euler-Hadamard product and moments of $\zeta'(\rho)$. Forum Math., 27(3):1799–1828, 2015.
- [48] F. Çiçek. On the logarithm of the Riemann zeta-function near the nontrivial zeros. Trans. Amer. Math. Soc., 374:5995–6037, 2021.
- [49] V. Chandee and K. Soundararajan. Bounding $|\zeta(1/2+it)|$ on the Riemann Hypothesis. Bull. Lond. Math. Soc., 43(2):243–250, 2011.
- [50] P. L. Chebyshev. Lettre de M. le Professeur Tchébychev à M. Fuss sur un nouveau théorème relatif aux nombres premiers contenus dans les formes 4n + 1 et 4n + 3. Bull. Classe Phys. Acad. Imp. Sci. St. Petersburg, 11:208, 1853.
- [51] P. L. Chebyshev. *Oeuvres de PL Tchebychef*, volume 1. Commissionaires de l'Académie impériale des sciences, 1899.
- [52] A. Y. Cheer and D. A. Goldston. Simple zeros of the Riemann zeta-function. Proc. Amer. Math. Soc., 118(2):365–372, 1993.
- [53] A. Chirre, F. Gonçalves, and D. De Laat. Pair correlation estimates for the zeros of the zeta function via semidefinite programming. Adv. Math., 361:106926, 2020.
- [54] N. G. Chudakov. On the functions $\zeta(s)$ and $\pi(x)$ CR Acad. Sci. USSR, NS, 21:421–422, 1938.
- [55] J. B. Conrey. The fourth moment of derivatives of the Riemann zeta-function. Quart. J. Math. Oxford Ser. (2), 39.1:21–36, 1988.
- [56] J. B. Conrey. More than two fifths of the zeros of the Riemann zeta function are on the critical line. J. Reine Angew. Math., 399:1–26, 1989.
- [57] J. B. Conrey. A note on the fourth power moment of the Riemann zeta-function. In Analytic Number Theory: Proceedings of a Conference In Honor of Heini Halberstam Volume 1, pages 225–230. Springer, 1996.
- [58] J. B. Conrey. The Riemann Hypothesis. Notices Amer. Math. Soc., 50(3):341–353, 2003.

- [59] J. B. Conrey and D. W. Farmer. Mean values of L-functions and symmetry. Int. Math. Res. Not., 2000(17):883–908, 2000.
- [60] J. B. Conrey, D. W. Farmer, J. P. Keating, M. O. Rubinstein, and N. C. Snaith. Autocorrelation of random matrix polynomials. *Comm. Math. Phys.*, 237:365–395, 2003.
- [61] J. B. Conrey, D. W. Farmer, J. P. Keating, M. O. Rubinstein, and N. C. Snaith. Integral moments of *L*-functions. *Proc. London Math. Soc.* (3), 91(1):33–104, 2005.
- [62] J. B. Conrey, D. W. Farmer, and M. R. Zirnbauer. Howe pairs, supersymmetry, and ratios of random characteristic polynomials for the unitary groups U(N). arXiv preprint arXiv:math-ph/0511024, 2005.
- [63] J. B. Conrey, D. W. Farmer, and M. R. Zirnbauer. Autocorrelation of ratios of L-functions. Commun. Number Theory Phys., 2(3):593-636, 2008.
- [64] J. B. Conrey, P. J. Forrester, and N. C. Snaith. Averages of ratios of characteristic polynomials for the compact classical groups. *Int. Math. Res. Not.*, 2005(7):397–431, 2005.
- [65] J. B. Conrey and A. Ghosh. On mean values of the zeta-function. Mathematika, 31(1):159–161, 1984.
- [66] J. B. Conrey and A. Ghosh. A mean value theorem for the Riemann zeta-function at its relative extrema on the critical line. J. Lond. Math. Soc. (2), 32:193–202, 1985.
- [67] J. B. Conrey and A. Ghosh. On mean values of the zeta-function, II. Acta Arith., 52:367–371, 1989.
- [68] J. B. Conrey and A. Ghosh. A conjecture for the sixth power moment of the Riemann zeta-function. *Internat. Math. Res. Notices*, pages 775–780, 1998.
- [69] J. B. Conrey, A. Ghosh, and S. M. Gonek. A note on gaps between zeros of the zeta function. Bull. Lond. Math. Soc., 16(4):421–424, 1984.
- [70] J. B. Conrey, A. Ghosh, and S. M. Gonek. Large gaps between zeros of the zetafunction. *Mathematika*, 33(2):212–238, 1986.
- [71] J. B. Conrey, A. Ghosh, and S. M. Gonek. Simple zeros of the zeta function of a quadratic number field. I. *Invent. Math.*, 86(3):563–576, 1986.
- [72] J. B. Conrey, A. Ghosh, and S. M. Gonek. Simple zeros of the zeta-function of a quadratic number field, II. In Analytic Number Theory and Diophantine Problems: Proceedings of a Conference at Oklahoma State University, 1984, pages 87–114. Springer, 1987.

- [73] J. B. Conrey, A. Ghosh, and S. M. Gonek. Simple zeros of zeta functions. In Colloque de Théorie Analytique des Nombres "Jean Coquet" (Marseille, 1985), volume 88-02 of Publ. Math. Orsay, pages 77–83. Univ. Paris XI, Orsay, 1988.
- [74] J. B. Conrey, A. Ghosh, and S. M. Gonek. Mean values of the Riemann zeta-function with application to the distribution of zeros. In *Number Theory, Trace Formulas and Discrete Groups (Oslo, 1987)*, pages 185–199. Academic Press, Boston, MA, 1989.
- [75] J. B. Conrey, A. Ghosh, and S. M. Gonek. Simple zeros of the Riemann zeta-function. Proc. London Math. Soc. (3), 76(3):497–522, 1998.
- [76] J. B. Conrey and S. M. Gonek. High moments of the Riemann zeta-function. Duke Math. J., 107(3):577–604, 2001.
- [77] J. B. Conrey and H. Iwaniec. Spacing of zeros of Hecke L-functions and the class number problem. Acta Arith., 103(3):259–312, 2002.
- [78] J. B. Conrey and J. P. Keating. Moments of zeta and correlations of divisor-sums: I. Philos. Trans. Roy. Soc. A, 373(2040):20140313, 2015.
- [79] J. B. Conrey and J. P. Keating. Moments of zeta and correlations of divisor-sums:
 II. In Advances in the Theory of Numbers: Proceedings of the Thirteenth Conference of the Canadian Number Theory Association, pages 75–85. Springer, 2015.
- [80] J. B. Conrey and J. P. Keating. Moments of zeta and correlations of divisor-sums: III. Indag. Math, 26(5):736–747, 2015.
- [81] J. B. Conrey and J. P. Keating. Moments of zeta and correlations of divisor-sums: IV. Res. Number Theory, 2(1):24, 2016.
- [82] J. B. Conrey and J. P. Keating. Pair correlation and twin primes revisited. R. Soc. Lond. Proc. Ser. A Math. Phys. Eng. Sci., 472(2194):20160548, 2016.
- [83] J. B. Conrey and J. P. Keating. Averages of ratios of the Riemann zeta-function and correlations of divisor sums. *Nonlinearity*, 30(10):R67, 2017.
- [84] J. B. Conrey and J. P. Keating. Moments of zeta and correlations of divisor-sums:
 V. Proc. Lond. Math. Soc. (3), 118(4):729–752, 2019.
- [85] J. B. Conrey, M. O. Rubinstein, and N. C. Snaith. Moments of the derivative of characteristic polynomials with an application to the Riemann Zeta function. *Comm. Math. Phys.*, 267:611–629, 2006.
- [86] J. B. Conrey and N. Snaith. In support of n-correlation. Comm. Math. Phys., 330(2):639–653, 2014.
- [87] J. B. Conrey and N. C. Snaith. Applications of the L-functions ratios conjectures. Proc. London Math. Soc. (3), 94(3):594–646, 2007.

- [88] J. B. Conrey and N. C. Snaith. Correlations of eigenvalues and Riemann zeros. Commun. Number Theory Phys., 2(3):477–536, 2008.
- [89] J. B. Conrey and N. C. Snaith. Triple correlation of the Riemann zeros. J. Théor. Nombres Bordeaux, 20(1):61–106, 2008.
- [90] J. B. Conrey and C. L. Turnage-Butterbaugh. On r-gaps between zeros of the Riemann zeta-function. Bull. Lond. Math. Soc., 50(2):349–356, 2018.
- [91] G. Csordas, W. Smith, and R. S. Varga. Lehmer pairs of zeros, the de Bruijn-Newman constant Λ, and the Riemann hypothesis. *Constr. Approx.*, 10(1):107–129, 1994.
- [92] M. J. Curran. Upper bounds for fractional joint moments of the Riemann zeta function. Acta Arith., 204(1):83–96, 2022.
- [93] M. J. Curran and A. Heycock. Sharp bounds for joint moments of the Riemann zeta function. arXiv preprint arXiv:2403.00902, 2024.
- [94] R. de la Bretèche and G. Tenenbaum. Sommes de Gál et applications. Proc. Lond. Math. Soc. (3), 119(1):104–134, 2019.
- [95] C. J. de La Vallée Poussin. Recherches analytiques sur la théorie des nombres premiers. Ann. Soc. Sc. Bruxelles, 1896.
- [96] C. J. de La Vallée Poussin. Sur la fonction $\zeta(s)$ de Riemann et le nombre des nombres premiers inferieurs à une limite donnée, volume 51. Hayez, 1899.
- [97] P.-O. Dehaye. Joint moments of derivatives of characteristic polynomials. Algebra Number Theory, 2:31–68, 2008.
- [98] P.-O. Dehaye. A note on moments of derivatives of characteristic polynomials. 22nd International Conference on Formal Power Series and Algebraic Combinatorics (FPSAC 2010), 681-692, Discrete Math. Theor. Comput. Sci. Proc. AN, Assoc. Discrete Math. Theor. Comput. Sci., pages 681–692, 2010.
- [99] P. Deligne. La conjecture de Weil. I. Inst. Hautes Études Sci. Publ. Math., 43:273–307, 1974.
- [100] P. Deligne. La conjecture de Weil: II. Inst. Hautes Études Sci. Publ. Math., 52:137– 252, 1980.
- [101] J.-M. Deshouillers and H. Iwaniec. Power mean values of the Riemann zeta-function. Mathematika, 29(2):202–212, 1982.
- [102] A. Diaconu. On the third moment of $L(\frac{1}{2}, \chi_d)$ I: The rational function field case. J. Number Theory, 198:1–42, 2019.
- [103] A. Diaconu, D. Goldfeld, and J. Hoffstein. Multiple Dirichlet series and moments of zeta and L-functions. *Compos. Math.*, 139(3):297–360, 2003.

- [104] A. Diaconu and I. Whitehead. On the third moment of $L(\frac{1}{2}, \chi_d)$ II: the number field case. J. Eur. Math. Soc. (JEMS), 23(6):2051–2070, 2021.
- [105] P. L. Dirichlet. Beweis eines Satzes über die arithmetische Progression. Bericht über die Verhandlungen der königlich Presussischen Akademie der Wissenschaften Berlin, pages 309–312, 1837.
- [106] W. Duke, J. B. Friedlander, and H. Iwaniec. A quadratic divisor problem. Invent. Math., 115(1):209–217, 1994.
- [107] F. J. Dyson. Statistical theory of the energy levels of complex systems. I. J. Mathematical Phys., 3:140–156, 1962.
- [108] F. J. Dyson. Statistical theory of the energy levels of complex systems. II. J. Mathematical Phys., 3:157–165, 1962.
- [109] F. J. Dyson. Statistical theory of the energy levels of complex systems. III. J. Mathematical Phys., 3:166–175, 1962.
- [110] T. Ehrhardt and B. Silbermann. Toeplitz determinants with one Fisher–Hartwig singularity. J. Funct. Anal., 148:229–256, 1997.
- [111] Euclid. The thirteen books of Euclid's Elements translated from the text of Heiberg. Vol. I: Introduction and Books I, II. Vol. II: Books III–IX. Vol. III: Books X–XIII and Appendix. Dover Publications, Inc., New York, 1956. Translated with introduction and commentary by Thomas L. Heath, 2nd ed.
- [112] D. W. Farmer. Long mollifiers of the Riemann zeta-function. Mathematika, 40(1):71– 87, 1993.
- [113] D. W. Farmer, S. M. Gonek, and C. P. Hughes. The maximum size of L-functions. J. Reine Angew. Math., 609:215–236, 2007.
- [114] A. Fazzari. On the joint second moment of zeta and its logarithmic derivative. arXiv preprint arXiv:2310.15918, 2023.
- [115] S. Feng. Zeros of the Riemann zeta function on the critical line. J. Number Theory, 132(4):511–542, 2012.
- [116] A. Florea. Negative moments of L-functions with small shifts over function fields. Int. Math. Res. Not., 2024(3):2298–2337, 2024.
- [117] K. Ford. Zero-free regions for the Riemann zeta function. In Number theory for the millennium, II (Urbana, IL, 2000), pages 25–56. AK Peters, Natick, MA, 2002.
- [118] K. Ford, K. Soundararajan, and A. Zaharescu. On the distribution of imaginary parts of zeros of the Riemann zeta function, II. Math. Ann., 343(3):487–505, 2009.

- [119] K. Ford and A. Zaharescu. On the distribution of imaginary parts of zeros of the Riemann zeta function. J. Reine Angew. Math., pages 145–158, 2005.
- [120] P. J. Forrester and J. P. Keating. Singularity dominated strong fluctuations for some random matrix averages. *Comm. Math. Phys.*, 250:119–131, 2004.
- [121] P. J. Forrester and N. S. Witte. Boundary conditions associated with the Painlevé III and V evaluations of some random matrix averages. J. Phys., 28:8983–8995, 2006.
- [122] A. Fujii. On the distribution of the zeros of the Riemann zeta function in short intervals. Bull. Amer. Math. Soc., 81(1):139–142, 1975.
- [123] A. Fujii. On the zeros of Dirichlet L-functions. II. Trans. Amer. Math. Soc., 267(1):33–40, 1981.
- [124] A. Fujii. On a theorem of Landau. Proc. Japan Acad. Ser. A Math. Sci., 65(10):51–54, 1989.
- [125] A. Fujii. On a theorem of Landau, II. Proc. Japan Acad. Ser. A Math. Sci., 66(10):291–296, 1990.
- [126] A. Fujii. On a conjecture of Shanks. Proc. Japan Acad. Ser. A Math. Sci., 70(10):109– 114, 1994.
- [127] A. Fujii. On a mean value theorem in the theory of the Riemann zeta function. Comment. Math. Univ. St. Pauli, 44(1):59–67, 1995.
- [128] A. Fujii. On the distribution of values of the derivative of the Riemann zeta function at its zeros. I. Proc. Steklov Inst. Math., 276:51–76, 2012.
- [129] Y. V. Fyodorov, S. Gnutzmann, and J. P. Keating. Extreme values of CUE characteristic polynomials: a numerical study. J. Phys. A, 51(46):464001, 2018.
- [130] Y. V. Fyodorov, G. A. Hiary, and J. P. Keating. Freezing transition, characteristic polynomials of random matrices, and the Riemann zeta function. *Phys. Rev. Lett.*, 108(17):170601, 2012.
- [131] Y. V. Fyodorov and J. P. Keating. Freezing transitions and extreme values: random matrix theory, and disordered landscapes. *Philos. Trans. Roy. Soc. A*, 372(2007):20120503, 2014.
- [132] P. Gao. Sharp lower bounds for moments of $\zeta'(\rho)$. arXiv preprint arXiv:2106.03057, 2021.
- [133] P. Gao and L. Zhao. Lower bounds for negative moments of $\zeta'(\rho)$. Mathematika, 69(4):1081–1103, 2023.
- [134] M. Garaev and A. Sankaranarayanan. The sum involving derivative of $\zeta(s)$ over simple zeros. J. Number Theory, 117(1):122–130, 2006.

- [135] M. Z. Garaev. One inequality involving simple zeros of $\zeta(s)$. Hardy-Ramanujan J., pages 18–22, 2003.
- [136] R. Garunkštis and J. Steuding. Simple zeros and discrete moments of the derivative of the Riemann zeta-function. J. Number Theory, 115(2):310–321, 2005.
- [137] C. F. Gauss. Carl Friedrich Gauss Werke II, volume 6. Gedruckt in der Dieterichschen universitätsdruckerei (WF Kaestner), 1874.
- [138] C. F. Gauss. *Disquisitiones arithmeticae*. Yale University Press, 1966.
- [139] F. Ge. Mean values of the logarithmic derivative of the Riemann zeta-function near the critical line. *Mathematika*, 69(2):562–572, 2023.
- [140] D. M. Goldfeld. The class number of quadratic fields and the conjectures of Birch and Swinnerton-Dyer. Ann. Sc. Norm. Super. Pisa Cl. Sci. (5), 3(4):623–663, 1976.
- [141] D. A. Goldston, S. M. Gonek, and H. L. Montgomery. Mean values of the logarithmic derivative of the Riemann zeta-function with applications to primes in short intervals. *J. Reine Angew. Math.*, 2001(537), 2001.
- [142] D. A. Goldston and H. L. Montgomery. Pair correlation of zeros and primes in short intervals. In Analytic Number Theory and Diophantine Problems: Proceedings of a Conference at Oklahoma State University, 1984, pages 183–203. Springer, 1987.
- [143] S. M. Gonek. Mean values of the Riemann zeta-function and its derivatives. Invent. Math., 75:123–141, 1984.
- [144] S. M. Gonek. A formula of Landau and mean values of ζ(s). In Topics in analytic number theory (Austin, Tex., 1982), pages 92–97. Univ. Texas Press, Austin, TX, 1985.
- [145] S. M. Gonek. On negative moments of the Riemann zeta-function. Mathematika, 36(1):71–88, 1989.
- [146] S. M. Gonek. An explicit formula of Landau and its applications to the theory of the zeta-function. *Contemp. Math.*, 143:395–395, 1993.
- [147] S. M. Gonek. The second moment of the reciprocal of the Riemann zeta-function and its derivative. *Talk at Mathematical Sciences Research Institute, Berkeley*, 1999.
- [148] S. M. Gonek. Three lectures on the Riemann zeta-function. arXiv preprint arXiv:math/0401126, 2004.
- [149] S. M. Gonek, C. P. Hughes, and J. P. Keating. A hybrid Euler-Hadamard product for the Riemann zeta function. *Duke Math. J.*, 136(3):507–549, 2007.
- [150] J. P. Gram. Note sur les zéros de la fonction $\zeta(s)$ de Riemann. Acta Math., 27:289–304, 1903.

- [151] A. Granville and G. Martin. Prime number races. Amer. Math. Monthly, 113(1):1–33, 2006.
- [152] B. H. Gross and D. B. Zagier. Heegner points and derivatives of L-series. Invent. Math., 84:225–320, 1986.
- [153] L. Guth and J. Maynard. New large value estimates for Dirichlet polynomials. arXiv preprint arXiv:2405.20552, 2024.
- [154] J. Hadamard. Étude sur les propriétés des fonctions entières et en particulier d'une fonction considérée par Riemann. J. Math. Pures Appl., 9:171–215, 1893.
- [155] J. Hadamard. Sur la distribution des zéros de la fonction $\zeta(s)$ et ses conséquences arithmétiques. Bulletin de la Societé mathematique de France, 24:199–220, 1896.
- [156] G. Halász. Über die Mittelwerte multiplikativer zahlentheoretischer Funktionen. Acta Math. Hungar., 19(3-4):365–403, 1968.
- [157] R. R. Hall. The behaviour of the Riemann zeta function on the critical line. Mathematika, 46:281–314, 1999.
- [158] R. R. Hall. A Wirtinger type inequality and the spacing of the zeros of the Riemann zeta-function. J. Number Theory, 93(2):235–245, 2002.
- [159] R. R. Hall. On the extreme values of the Riemann zeta-function between its zeros on the critical line. J. Reine Angew. Math., 560:29–41, 2003.
- [160] R. R. Hall. Extreme values of the Riemann zeta-function on short zero intervals. Acta Arith., 121:259–273, 2006.
- [161] A. Hamieh and N. Ng. Mean values of long Dirichlet polynomials with higher divisor coefficients. Adv. Math., 410:108759, 2022.
- [162] C. Hanga and C. P. Hughes. Probabilistic models for Gram's Law. arXiv preprint arXiv:1911.03190, 2019.
- [163] G. H. Hardy. Sur les zéros de la fonction $\zeta(s)$ de Riemann. CR Acad. Sci. Paris, 158:1012–1014, 1914.
- [164] G. H. Hardy and J. E. Littlewood. Contributions to the theory of the Riemann zeta-function and the theory of the distribution of primes. Acta Math., 41:119–196, 1918.
- [165] G. H. Hardy and J. E. Littlewood. The zeros of Riemann's zeta-function on the critical line. *Math. Z.*, 10(3):283–317, 1921.
- [166] A. J. Harper. Sharp conditional bounds for moments of the Riemann zeta function. arXiv preprint arXiv:1305.4618, 2013.

- [167] C. B. Haselgrove and J. C. P. Miller. Tables of the Riemann Zeta Function. Published for the Royal Society at the University Press, 1960.
- [168] W. Heap, J. Li, and J. Zhao. Lower bounds for discrete negative moments of the Riemann zeta function. Algebra Number Theory, 16(7):1589–1625, 2022.
- [169] W. Heap, M. Radziwiłł, and K. Soundararajan. Sharp upper bounds for fractional moments of the Riemann zeta function. Q. J. Math., 70(4):1387–1396, 2019.
- [170] W. Heap and K. Soundararajan. Lower bounds for moments of zeta and L-functions revisited. Mathematika, 68(1):1–14, 2022.
- [171] D. R. Heath-Brown. The twelfth power moment of the Riemann-function. Q. J. Math., 29(4):443–462, 1978.
- [172] D. R. Heath-Brown. The fourth power moment of the Riemann zeta function. Proc. London Math. Soc. (3), 38(3):385–422, 1979.
- [173] D. R. Heath-Brown. Simple zeros of the Riemann zeta-function on the critical line. Bull. Lond. Math. Soc., 11(1):17–18, 1979.
- [174] D. R. Heath-Brown. Fractional Moments of the Riemann Zeta-Function. J. London Math. Soc., 2(1):65–78, 1981.
- [175] D. R. Heath-Brown and M. Huxley. Exponential sums with a difference. In Proc. London Math. Soc. (3), volume 61, pages 227–250, 1990.
- [176] K. Heegner. Diophantische analysis und modulfunktionen. Math. Z., 56(3):227–253, 1952.
- [177] D. A. Hejhal. On the distribution of $\log |\zeta'(\frac{1}{2}+it)|$. In Number theory, trace formulas and discrete groups (Oslo, 1987), pages 343–370. Academic Press, Boston, MA, 1989.
- [178] D. A. Hejhal. On the triple correlation of zeros of the zeta function. Int. Math. Res. Not., 1994(7):293–302, 1994.
- [179] G. A. Hiary and A. M. Odlyzko. Numerical study of the derivative of the Riemann zeta function at zeros. *Comment. Math. Univ. St. Pauli*, 60(1-2):47–60, 2011.
- [180] A. Huckleberry, A. Püttmann, and M. Zirnbauer. Haar expectations of ratios of random characteristic polynomials. *Complex Anal. Synerg.*, 2(1):1, 2016.
- [181] C. P. Hughes. On the characteristic polynomial of a random unitary matrix and the Riemann zeta function. PhD thesis, University of Bristol, 2001.
- [182] C. P. Hughes, J. P. Keating, and N. O'Connell. Random matrix theory and the derivative of the Riemann zeta function. R. Soc. Lond. Proc. Ser. A Math. Phys. Eng. Sci., 456(2003):2611–2627, 2000.

- [183] C. P. Hughes, S. Lugmayer, and A. Pearce-Crump. The second moment of the Riemann zeta function at its local extrema. arXiv preprint arXiv:2411.05573, 2024.
- [184] C. P. Hughes, G. Martin, and A. Pearce-Crump. A heuristic for discrete mean values of the derivative of the Riemann zeta function. *Integers*, 24, 2024.
- [185] C. P. Hughes and A. Pearce-Crump. Moments of derivatives of the Riemann zeta function: The Ratios Conjecture. (In preparation).
- [186] C. P. Hughes and A. Pearce-Crump. A discrete mean-value theorem for the higher derivatives of the Riemann zeta function. J. Number Theory, 241:142–164, 2022.
- [187] C. P. Hughes and A. Pearce-Crump. Moments of derivatives of the Riemann zeta function: Characteristic polynomials and the hybrid formula. arXiv preprint arXiv:2406.08121, 2024.
- [188] C. P. Hughes and M. P. Young. The twisted fourth moment of the Riemann zeta function. J. Reine Angew. Math., 2010(641), 2010.
- [189] M. N. Huxley. On the difference between consecutive primes. Invent. Math., 15:164– 170, 1971.
- [190] M. N. Huxley. Exponential sums and lattice points III. Proc. London Math. Soc. (3), 87(3):591–609, 2003.
- [191] M. N. Huxley. Exponential sums and the Riemann zeta function V. Proc. London Math. Soc. (3), 90(1):1–41, 2005.
- [192] A. E. Ingham. Mean-value theorems in the theory of the Riemann zeta-function. Proc. Lond. Math. Soc., 2:273–300, 1926.
- [193] A. E. Ingham. On the difference between consecutive primes. Q. J. Math., 8(1):255– 266, 1937.
- [194] A. E. Ingham. On the estimation of $N(\sigma, T)$. The Quarterly Journal of Mathematics, 11(1):201–202, 1940.
- [195] A. E. Ingham. On two conjectures in the theory of numbers. Amer. J. Math., 64(1):313–319, 1942.
- [196] K. Ireland and M. Rosen. A Classical Introduction to Modern Number Theory, volume 84. Springer Science & Business Media, 2013.
- [197] M. Israilov. The Laurent expansion of the Riemann zeta function. Tr. Mat. Inst. Steklova, 158:98–104, 1981.
- [198] A. Ivic. The Riemann zeta-function: theory and applications. Courier Corporation, 2012.

- [199] A. Ivić. The Theory of Hardy's Z-function (No. 196). Cambridge University Press, 2013.
- [200] A. Ivić and Y. Motohashi. On the fourth power moment of the Riemann zeta-function. J. Number Theory, 51:16–45, 1995.
- [201] H. Iwaniec and P. Sarnak. Dirichlet L-functions at the central point. In Number theory in progress, Vol. 2 (Zakopane-Kościelisko, 1997), pages 941–952. de Gruyter, Berlin, 1999.
- [202] M. T. Jakhlouti and K. Mazhouda. Distribution of the values of the derivative of the Dirichlet L-functions at its a-points. Bull. Korean Math. Soc., 54(4):1141–1158, 2017.
- [203] J. A. G. Jara. Asymptotic mean square of the product of the second power of the Riemann zeta function and a Dirichlet polynomial. PhD thesis, University of Rochester, 1997.
- [204] M. Jutila. Zeros of the zeta-function near the critical line. In Studies in Pure Mathematics: To the Memory of Paul Turán, pages 385–394. Springer, 1983.
- [205] J. Kaczorowski, A. Languasco, and A. Perelli. A note on Landau's formula. Funct. Approx. Comment. Math, 28:173–186, 2000.
- [206] H. Kadiri. Une région explicite sans zéro pour la fonction zeta de Riemann. arXiv preprint arXiv:math/0401238, 2004.
- [207] D. A. Kaptan, Y. Karabulut, and C. Y. Yildirim. Some mean value theorems for the Riemann zeta-function and Dirichlet L-functions. Comment. Math. Univ. St. Pauli, 60(1-2):83–87, 2011.
- [208] C. Karabulut, Y. and Yıldırım. On some averages at the zeros of the derivatives of the Riemann zeta-function. J. Number Theory, 131:1939–1961, 2011.
- [209] J. P. Keating, H. L. Montgomery, and P. Sarnak. Opening Remarks and History of the math talks. 50 Years of Number Theory and Random Matrix Theory Conference, I.A.S, 2022.
- [210] J. P. Keating and N. C. Snaith. Random matrix theory and L-functions at s = 1/2. Comm. Math. Phys., 214:91–100, 2000.
- [211] J. P. Keating and N. C. Snaith. Random matrix theory and $\zeta(1/2 + it)$. Comm. Math. Phys., 214:57–89, 2000.
- [212] J. P. Keating and F. Wei. Joint moments of higher order derivatives of CUE characteristic polynomials I: asymptotic formulae. arXiv preprint arXiv:2307.01625, 2023.

- [213] J. P. Keating and F. Wei. Joint moments of higher order derivatives of CUE characteristic polynomials II: Structures, recursive relations, and applications. arXiv preprint arXiv:2307.0283, 2023.
- [214] S. Kirila. An upper bound for discrete moments of the derivative of the Riemann zeta-function. *Mathematika*, 66(2):475–497, 2020.
- [215] H. Kobayashi. On a certain sum of the derivative of Dirichlet L-functions. Tsukuba J. Math., 46(1):121–143, 2022.
- [216] H. Kober. Eine Mittelwertformel der Riemannschen Zetafunktion. Compos. Math., 3:174–189, 1936.
- [217] N. M. Korobov. Estimates of trigonometric sums and their applications (Russian). Uspekhi Mat. Nauk, 13(4):185–192, 1958.
- [218] Y. Lamzouri. An effective Linear Independence conjecture for the zeros of the Riemann zeta function and applications. arXiv preprint arXiv:2311.04860, 2023.
- [219] E. Landau. Über die nullstellen der zetafunktion. Math. Ann., 71(4):548–564, 1912.
- [220] A. Laurincikas and J. Steuding. A note on the moments of $\zeta(1/2 + i\gamma)$. Publ. Inst. Math. (Beograd)(NS), 76:57–63, 2004.
- [221] A. M. Legendre. Essai sur la théorie des nombres. Cambridge Library Collection. Cambridge University Press, Cambridge, 2009. Reprint of the second (1808) edition.
- [222] N. Levinson. More than one third of zeros of Riemann's zeta-function are on $\sigma = 1/2$. Adv. Math., 13(4):383–436, 1974.
- [223] N. Levinson and H. L. Montgomery. Zeros of the derivatives of the Riemann zetafunction. Acta Math., 133(1):49–65, 1974.
- [224] J. E. Littlewood. Sur la distribution des nombres premiers. CR Acad. Sci. Paris, 158(1869):1872, 1914.
- [225] J. E. Littlewood. Sur la distribution des nombres premiers. CR Acad. Sci. Paris, 158(1869):1872, 1914.
- [226] J. E. Littlewood. Researches in the theory of the Riemann ζ -function. In *Proc.* London Math. Soc.(2), volume 20, pages 22–28, 1922.
- [227] J. E. Littlewood. On the zeros of the Riemann zeta-function. Math. Proc. Cambridge Philos. Soc., 22:295–318, 1924.
- [228] J. E. Littlewood. On the Riemann zeta-function. Proc. London Math. Soc., 24:175– 201, 1925.
- [229] G. Martin, P. J. S. Yang, A. Bahrini, P. Bajpai, K. Benli, J. Downey, Y. Y. Li, X. Liang, A. Parvardi, R. Simpson, et al. An annotated bibliography for comparative prime number theory. arXiv preprint arXiv:2309.08729, 2023.

- [230] K. Maslanka. Effective method of computing Li's coefficients and their properties. arXiv preprint arXiv:math/0402168, 2004.
- [231] M. L. Mehta. Random Matrices. Elsevier, 2004.
- [232] P. Michel and J. VanderKam. Non-vanishing of high derivatives of Dirichlet Lfunctions at the central point. J. Number Theory, 81(1):130–148, 2000.
- [233] M. B. Milinovich. Mean-Value Estimates for the Derivative of the Riemann Zeta-Function. PhD thesis, University of Rochester, 2008.
- [234] M. B. Milinovich. Upper bounds for moments of $\zeta'(\rho)$. Bull. Lond. Math. Soc., 42(1):28–44, 2010.
- [235] M. B. Milinovich. Moments of the Riemann zeta-function at its relative extrema on the critical line. Bull. Lond. Math. Soc., 43.6:1119–1129, 2011.
- [236] M. B. Milinovich and N. Ng. A note on a conjecture of Gonek. Funct. Approx. Comment. Math, 46(2):177–187, 2012.
- [237] M. B. Milinovich and N. Ng. Lower bounds for moments of $\zeta'(\rho)$. Int. Math. Res. Not., 2014(12):3190–3216, 2014.
- [238] S.-H. Min. On the zeros and the order of the Riemann zeta-function on the critical line. PhD thesis, University of Oxford, 1947.
- [239] H. L. Montgomery. The pair correlation of zeros of the zeta function. In Proc. Symp. Pure Math, volume 24, pages 181–193, 1973.
- [240] H. L. Montgomery. Distribution of the zeros of the Riemann zeta function. Proc. Internat. Congr. Mathematicians, pages 379–381, 1974.
- [241] H. L. Montgomery. The zeta function and prime numbers. In Proceedings of the Queen's Number Theory Conference, 1979, pages 1–31, 1980.
- [242] H. L. Montgomery. Topics in Multiplicative Number Theory, volume 227. Springer, 2006.
- [243] H. L. Montgomery and R. C. Vaughan. Hilbert's inequality. J. London Math. Soc. (2), 8(73-82):4–8, 1974.
- [244] H. L. Montgomery and R. C. Vaughan. Multiplicative number theory. I. Classical theory, volume 97 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2007.
- [245] Y. Motohashi. An explicit formula for the fourth power mean of the Riemann zeta-function. Acta Math., 170(2):181–220, 1993.
- [246] Y. Motohashi. A relation between the Riemann zeta-function and the hyperbolic Laplacian. Ann. Scuola Norm. Sup. Pisa Cl. Sci. (4), 22:299–313, 1995.

- [247] Y. Motohashi. Spectral theory of the Riemann zeta-function. Cambridge University Press, 1997.
- [248] Y. Motohashi. The Riemann zeta-function and Hecke congruence subgroups. II. J. Res. Inst. Sci. Technol. Nihon Univ., 2009(119):119_29-119_64, 2009.
- [249] J. Mueller. Gaps between consecutive zeta zeros. In Analytic Number Theory: Proceedings of a Conference Held at Temple University, Philadelphia, May 12–15, 1980, pages 141–147. Springer, 1981.
- [250] J. Mueller. On the difference between consecutive zeros of the Riemann zeta function. J. Number Theory, 14(3):327–331, 1982.
- [251] N. Ng. Limiting distributions and zeros of Artin L-functions. PhD thesis, University of British Columbia, 2000.
- [252] N. Ng. The distribution of the summatory function of the Möbius function. Proc. London Math. Soc. (3), 89(2):361–389, 2004.
- [253] N. Ng. The fourth moment of $\zeta'(\rho)$. Duke Math. J., 125(2), 2004.
- [254] N. Ng. Extreme values of $\zeta'(\rho)$. J. London Math. Soc., 78(2):273–289, 2008.
- [255] N. Ng. The sixth moment of the Riemann zeta function and ternary additive divisor sums. Discrete Anal., 2021.
- [256] N. Ng, Q. Shen, and P.-J. Wong. The eighth moment of the Riemann zeta function. arXiv preprint arXiv:2204.13891, 2022.
- [257] A. M. Odlyzko. On the distribution of spacings between zeros of the zeta function. Math. Comp., 48(177):273–308, 1987.
- [258] A. M. Odlyzko. The 10²⁰ zero of the Riemann zeta function and 70 million of its neighbors. AT&T Bell Lab preprint, 1989.
- [259] A. M. Odlyzko. The 10²²-nd zero of the Riemann zeta function. Dynamical, spectral, and arithmetic zeta functions (San Antonio, TX, 1999). *Contemp. Math*, 290:139–144, 2001.
- [260] A. M. Odlyzko and H. J. J. te Riele. Disproof of the Mertens conjecture. J. Reine Angew. Math., 357:138–160, 1985.
- [261] E. Paquette and O. Zeitouni. The extremal landscape for the CβE ensemble. arXiv preprint arXiv:2209.06743, 2022.
- [262] A. Pearce-Crump. A further generalisation of sums of higher derivatives of the Riemann Zeta Function. arXiv preprint arXiv:2111.02756, 2021. To appear in Int. J. Number Theory.

- [263] A. Pearce-Crump. Moments of the Riemann zeta function at its local extrema. arXiv preprint arXiv:2411.05568, 2024.
- [264] D. Platt and T. S. Trudgian. The Riemann Hypothesis is true up to 3.10¹². Bull. Lond. Math. Soc., 53(3):792–797, 2021.
- [265] K. Pratt, N. Robles, A. Zaharescu, and D. Zeindler. More than five-twelfths of the zeros of ζ are on the critical line. *Res. Math. Sci.*, 7(1):2, 2020.
- [266] M. Radziwill. Limitations to mollifying $\zeta(s)$. arXiv preprint arXiv:1207.6583, 2012.
- [267] M. Radziwiłł and K. Soundararajan. Continuous lower bounds for moments of zeta and L-functions. *Mathematika*, 59(1):119–128, 2013.
- [268] K. Ramachandra. Some remarks on the mean value of the Riemann zeta-function and other Dirichlet series-II. *Hardy-Ramanujan J.*, 3, 1980.
- [269] K. Ramachandra. Some remarks on the mean-value of the Riemann zeta-function and other Dirichlet series, III. Annales Acad. Sci. Fenn., 5:145–158, 1980.
- [270] K. Ramachandra. On the mean-value and omega-theorems for the Riemann zetafunction. Tata Institute of Fundamental Research, 1995.
- [271] G. Riemann. Über die Anzahl der Primzahlen unter einer gegebenen Grösse (On the number of primes less than a given quantity). Monatsber. Knigl. Preuss. Akad. Wiss. Berlin, pages 671–680, 1859.
- [272] B. Rodgers and T. Tao. The de Bruijn–Newman constant is non-negative. Forum Math. Pi, 8, 2020.
- [273] M. O. Rubinstein and P. Sarnak. Chebyshev's bias. Exp. Math., 3(3):173–197, 1994.
- [274] Z. Rudnick and P. Sarnak. The n-level correlations of zeros of the zeta function. C. R. Acad. Sci. Paris Sér. I Math., 319(10):1027–1032, 1994.
- [275] Z. Rudnick and P. Sarnak. Zeros of principal L-functions and random matrix theory. Duke Math. J., 81(2):269–322, 1996.
- [276] A. Selberg. On the zeros of Riemann's zeta-function. Skr. Norske Vid. Akad. Oslo, 1942.
- [277] A. Selberg. Remarks on a multiple integral. Norsk Mat. Tidsskr., 26:71–78, 1944.
- [278] A. Selberg. Contributions to the theory of the Riemann zeta-function. Arch. for Math. og Naturv. B, 1946.
- [279] A. Selberg. The zeta-function and the Riemann Hypothesis, CR Dixieme Congres Math. Scandinaves, Copenhague, page 187, 1946.
- [280] D. Shanks. Review of 'Tables of the Riemann Zeta Function' by C.B. Haselgrove in collaboration with J.C.P. Miller. *Math. Comp.*, 15:84–86, 1961.

- [281] C. Siegel. Über die classenzahl quadratischer zahlkörper. Acta Arith., 1(1):83–86, 1935.
- [282] C. L. Siegel. Über Riemanns Nachlas zur analytischen Zahlentheorie. Quellen und Studien zur Geschichte der Math. Astr. Phys. Abt. B, 2:45–80, 1932.
- [283] S. Skewes. On the Difference $\pi(x) li(x)$ (I). J. London Math. Soc., 1(4):277–283, 1933.
- [284] S. Skewes. On the Difference $\pi(x) li(x)$ (II). Proc. London Math. Soc. (3), 5(1):48–70, 1955.
- [285] R. Šleževičien and J. Steuding. Short series over simple zeros of the Riemann zeta-function. *Indag. Math*, 15(1):129–132, 2004.
- [286] C. R. Smith. The hunt for Skewes' number. Master's thesis, University of York, 2016.
- [287] N. C. Snaith. Random Matrix Theory and Zeta Functions. PhD thesis, University of Bristol, 2000.
- [288] K. Soundararajan. Mean-values of the Riemann zeta-function. Mathematika, 42(1):158–174, 1995.
- [289] K. Soundararajan. Nonvanishing of quadratic Dirichlet L-functions at s = 1/2. Ann. Math., 152:447–488, 2000.
- [290] K. Soundararajan. Extreme values of zeta and L-functions. Math. Ann., 2(342):467– 486, 2008.
- [291] K. Soundararajan. Moments of the Riemann zeta function. Ann. Math., 170:981–993, 2009.
- [292] A. Speiser. Geometrisches zur Riemannschen zetafunktion. Math. Ann., 110(1):514– 521, 1935.
- [293] H. M. Stark. A complete determination of the complex quadratic fields of class-number one. Michigan Math. J., 14(1):1–27, 1967.
- [294] R. Steuding and J. Steuding. Large gaps between zeros of the zeta-function on the critical line and moment conjectures from random matrix theory. *Comput. Methods Funct. Theory*, 8(1):121–132, 2008.
- [295] G. Szegő. Orthogonal Polynomials. AMS Colloquium Publications XXII, 1939.
- [296] E. C. Titchmarsh. The mean-value of the zeta-function on the critical line. Proc. London Math. Soc. (2), 2(1):137–150, 1928.
- [297] E. C. Titchmarsh. The zeros of the Riemann zeta-function. Proc. Roy. Soc. London Ser. A, 151(873):234–255, 1935.

- [298] E. C. Titchmarsh and D. R. Heath-Brown. The Theory of the Riemann Zeta-function. Oxford University Press, 1986.
- [299] T. S. Trudgian. On a conjecture of Shanks. J. Number Theory, 130(12):2635–2638, 2010.
- [300] T. S. Trudgian and A. Yang. On optimal exponent pairs. arXiv preprint arXiv:2306.05599, 2023.
- [301] A. M. Turing. Some calculations of the Riemann zeta-function. Proc. London Math. Soc. (3), 3(1):99–117, 1953.
- [302] C. L. Turnage-Butterbaugh. Moments Of Products Of L-Functions. PhD thesis, University of Mississippi, 2014.
- [303] I. M. Vinogradov. A new estimate for $\zeta(1+it)$ (Russian). Izv. Akad. Nauk SSSR, Ser. Mat., 22:161–164, 1958.
- [304] H. von Koch. Sur la distribution des nombres premiers. Acta Math., 24(1):159, 1901.
- [305] H. von Mangoldt. Zur Verteilung der Nullstellen der Riemannschen Funktion $\xi(t)$. Math. Ann., 61:1–19, 1905.
- [306] M. Watkins. Class numbers of imaginary quadratic fields. Math. Comp., 73(246):907– 938, 2004.
- [307] N. Watt. Kloosterman sums and a mean value for Dirichlet polynomials. J. Number Theory, 53(1):179–210, 1995.
- [308] N. Watt. A note on the mean square of $|\zeta(1/2 + it)|$. J. London Math. Soc., 82(2):279–294, 2010.
- [309] A. Weil. Courbes algébriques et variétés abéliennes. Hermann, Paris, 1971.
- [310] H. Weyl. Zur Abschätzung von $\zeta(1+ti)$. Math. Z., 10:88–101, 1921.
- [311] H. Weyl. The Classical Groups: Their Invariants and Representations. Princeton University Press, 1946.
- [312] B. Winn. Derivative moments for characteristic polynomials from the CUE. Commun. Math. Phys., 315:531–563, 2012.
- [313] A. Wintner. On the asymptotic distribution of the remainder term of the primenumber theorem. *Amer. J. Math.*, 57(3):534–538, 1935.
- [314] A. Wintner. On the distribution function of the remainder term of the prime number theorem. Amer. J. Math., 63(2):233–248, 1941.
- [315] X. Wu. The twisted mean square and critical zeros of Dirichlet L-functions. Math. Z., 293(1):825–865, 2019.

- [316] M. P. Young. A short proof of Levinson's theorem. Arch. Math. (Basel), 95(6), 2010.
- [317] N. I. Zavorotnyi. On the fourth moment of the Riemann zeta-function. In Automorphic functions and number theory I (Russian), pages 69–125. Collected Scientific Works, Vladivostok, 1989.
- [318] Q. Zhang. On the cubic moment of quadratic Dirichlet *L*-functions. *Math. Res. Lett.*, 12(3):413–424, 2005.